#### IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re

SCOTTISH HOLDINGS, INC., et al.,

Debtors.<sup>1</sup>

Chapter 11

Case No. 18-10160 (LSS)

Jointly Administered

### NOTICE OF FILING DEBTORS' THIRD AMENDED JOINT CHAPTER 11 PLAN

PLEASE TAKE NOTICE that on April 18, 2018, the Debtors filed the *Joint* Chapter 11 Plan of Reorganization of Scottish Holdings, Inc., and Scottish Annuity & Life Insurance Company (Cayman) Ltd. (D.I. 213) (the "Plan").

PLEASE TAKE FURTHER NOTICE that on June 15, 2018, the Debtors filed the Notice of Filing of (I) Amended [Proposed] Disclosure Statement for Debtors' First Amended Joint Chapter 11 Plan; (II) Debtors' First Amended Joint Chapter 11 Plan; and (III) Revised Proposed Disclosure Statement Order (D.I. 357) (the "First Amended Plan").

PLEASE TAKE FURTHER NOTICE that on June 27, 2018, the Debtors filed the solicitation version of the Second Amended Joint Chapter 11 Plan of Reorganization of Scottish Holdings, Inc., and Scottish Annuity & Life Insurance Company (Cayman) Ltd. (D.I. 381) (the "Second Amended Plan").

PLEASE TAKE FURTHER NOTICE that attached hereto as **Exhibit A** is the Third Amended Joint Chapter 11 Plan of Reorganization of Scottish Holdings, Inc., and Scottish Annuity & Life Insurance Company (Cayman) Ltd. (the "<u>Third Amended Plan</u>").

PLEASE TAKE FURTHER NOTICE that attached hereto as **Exhibit B** is a redline comparing the Third Amended Plan to the Second Amended Plan.

1

The Debtors, along with the last four digits of their federal tax identification numbers, are as follows: Scottish Holdings, Inc. (4408) and Scottish Annuity & Life Insurance Company (Cayman) Ltd. (3285). The Debtors' mailing address for purposes of these chapter 11 cases is 14120 Ballantyne Corporate Place, Suite 300, Charlotte, NC 28277.

Dated: August 9, 2018 Wilmington, Delaware

#### **MORRIS, NICHOLS, ARSHT & TUNNELL LLP**

/s/ Paige N. Topper Eric D. Schwartz (No. 3134) Gregory W. Werkheiser (No. 3553) Matthew B. Harvey (No. 5186) Paige N. Topper (No. 6470) 1201 N. Market St., 16th Floor PO Box 1347 Wilmington, DE 19899-1347 Telephone: (302) 658-9200 Facsimile: (302) 658-3989 eschwartz@mnat.com gwerkheiser@mnat.com mharvey@mnat.com

- and –

Peter Ivanick Lynn W. Holbert John D. Beck HOGAN LOVELLS US LLP 875 Third Avenue New York, NY 10022 Telephone: (212) 918-3000 Facsimile: (212) 918-3100 peter.ivanick@hoganlovells.com lynn.holbert@hoganlovells.com john.beck@hoganlovells.com

Counsel for Debtors and Debtors in Possession

# EXHIBIT A

#### IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re

SCOTTISH HOLDINGS, INC., et al.,

Debtors.<sup>1</sup>

Chapter 11

Case No. 18-10160 (LSS)

Jointly Administered

## THIRD AMENDED JOINT CHAPTER 11 PLAN OF REORGANIZATION OF SCOTTISH HOLDINGS, INC., AND SCOTTISH ANNUITY & LIFE <u>INSURANCE COMPANY (CAYMAN) LTD.</u>

#### **HOGAN LOVELLS US LLP**

Peter A. Ivanick Lynn W. Holbert John D. Beck 875 Third Avenue New York, NY 10022

Telephone: (212) 918-3000 Facsimile: (212) 918-3100 peter.ivanick@hoganlovells.com lynn.holbert@hoganlovells.com john.beck@hoganlovells.com

#### **MORRIS, NICHOLS, ARSHT & TUNNELL LLP**

Eric D. Schwartz (No. 3134) Gregory W. Werkheiser (No. 3553) Matthew B. Harvey (No. 5186) Paige N. Topper (No. 6470) 1201 N. Market St., 16th Floor P.O. Box 1347 Wilmington, DE 19899-1347 Telephone: (302) 658-9200 Facsimile: (302) 658-3989 eschwartz@mnat.com gwerkheiser@mnat.com mharvey@mnat.com

#### Counsel for Debtors and Debtors in Possession

Dated: August 9, 2018 Wilmington, Delaware

<sup>&</sup>lt;sup>1</sup>The Debtors, along with the last four digits of their federal tax identification numbers, are as follows: Scottish Holdings, Inc. (4408) and Scottish Annuity & Life Insurance Company (Cayman) Ltd. (3285). The Debtors' mailing address for purposes of these Chapter 11 Cases is 14120 Ballantyne Corporate Place, Suite 300, Charlotte, NC 28277.

# TABLE OF CONTENTS

I.	SUMM	UMMARY OF THE PLAN1				
II.	DEFIN	NITIONS AND INTERPRETATION				
III.	CLASS	SIFICA	TION OF CLAIMS AND INTERESTS			
	3.1.	Introdu	action2			
	3.2.	Unclas	sified Claims			
		(a)	Administrative Claims			
		(b)	Priority Tax Claims			
	3.3.	Classif	fication of Claims and Interests			
IV.	TREA	TMENT	Γ OF CLAIMS AND INTERESTS			
	4.1.	Unclas	sified Claims			
		(a)	Administrative Claims			
		(b)	Priority Tax Claims			
		(c)	Other Provisions Governing Administrative Claims4			
		(d)	Indenture Trustee Fees			
	4.2.	Unimp	aired Classes of Claims and Interests7			
		(a)	Class 1 – Secured Claims			
		(b)	Class 2 – Priority Non-Tax Claims			
		(c)	Class 3 – Intercompany Claims			
		(d)	Class 10 – SALIC Existing Equity Interests			
	4.3.	Impair	ed Voting Classes of Claims9			
		(a)	Class 4 – SHI TruPS Claims			
		(b)	Class 5 – SHI General Unsecured Claims			
		(c)	Class 6 – SALIC TruPS Claims & SFL Claims11			

# Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 4 of 241

# TABLE OF CONTENTS (Continued)

		(d)	Class 7 – SALIC General Unsecured Claims	16
	4.4.	Impai	red Non-Voting Classes of Claims and Interests	17
		(a)	Class 8 – Subordinated Claims	17
		(b)	Class 9 – SHI Existing Equity Interests	17
V.	ACCE	EPTAN	CE REQUIREMENTS	17
	5.1.	Impai	red Classes of Claims Entitled to Vote	17
	5.2.	Accep	ptance by an Impaired Class	17
	5.3.	Presu	med Acceptance by Unimpaired Classes	
	5.4.		med Rejection by Impaired Classes Not Receiving Any bution under the Plan	
	5.5.	Confi	rmation Pursuant to Section 1129(b) of the Bankruptcy Code	
	5.6.	Elimi	nation of Vacant Classes	18
	5.7.		med Acceptance by Voting Classes in Which No Votes Are	19
	5.8.	Conse	olidation of Classes	19
	5.9.	Separ	ate Classes of Secured Claims	19
VI.	MEA	NS FOI	R IMPLEMENTATION OF THE PLAN	19
	6.1.	Plan 7	Fransactions	19
		(a)	Stock Purchase Agreement Closing	19
		(b)	Funding, Allocation and Use of Plan Funding Payment; Allocation and Use of Debtors' Unrestricted Cash	19
		(c)	Funding of Recapitalization Funding Payment	20
		(d)	Cancellation of SHI Existing Equity Interests; Issuance of New SHI Equity	20

# TABLE OF CONTENTS (Continued)

	(e)	Final Share Surrender	21
	(f)	New Equity Issuance and Distribution	21
	(g)	Distribution of Offered New Equity	22
6.2.	Vesti	ng of Estate Property	22
6.3.	The I	Distribution Trust	23
	(a)	Execution of Distribution Trust Agreement	23
	(b)	Purpose of the Distribution Trust	23
	(c)	Distribution Trust Assets and Other Property	23
	(d)	The Administration of the Distribution Trust and Authority and Powers of the Distribution Trustee	23
	(e)	Mutual Cooperation	24
	(f)	Establishment and Funding of Distribution Trust Reserves	24
	(g)	Intentionally Omitted	25
	(h)	Cash Investments	25
	(i)	Distributions to Holders of Claims and Distribution Trust Beneficiaries	25
	(j)	Federal Income Tax Treatment of Distribution Trust	26
	(k)	Tax Reporting	27
	(1)	Dissolution	
6.4.	The F	Reorganized Debtors	29
	(a)	Continued Corporate Existence	29
	(b)	Directors and Officers of the Reorganized Debtors	29
	(c)	Corporate Action	

# TABLE OF CONTENTS (Continued)

		(d)	Effectuating Documents; Further Transactions	
	6.5.	Retai	ned Causes of Action	
	6.6.	The (	Closing	
	6.7.		ellation of Agreements, Securities and Other Documents ting to TruPS Transactions & SFL Note	31
	6.8.	Com	prehensive Settlement of Claims and Controversies	
VII.	CLAI	MS RE	ESOLUTION & DISTRIBUTIONS	
	7.1.	Right	t to Object to Claims	
	7.2.	Dead	lline for Objecting to Claims	
	7.3.	Dead	lline for Responding to Claim Objections	
	7.4.	Right	t to Request Estimation of Claims	
	7.5.	Distr	ibution Procedures Regarding Allowed Claims	
		(a)	In General	
		(b)	Distributions on Allowed Claims Only	
		(c)	Method of Distributions	
		(d)	Undeliverable Distributions	
		(e)	Unclaimed Distributions	
		(f)	Taxes; Withholding	
		(g)	Special Distribution Provisions for TruPS Claims	
		(h)	Additional Procedures Regarding Distributions from the Distribution Trust	
		(i)	Allocation of Distributions between Principal and Interest	
	7.6.	Imple	ementation of New Equity and Cash Elections	

# Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 7 of 241

# TABLE OF CONTENTS (Continued)

# <u>Page</u>

		(a)	Making of New Equity or Cash Election by New Equity Eligible Beneficial Holders	
		(b)	Implementation of New Equity or Cash Election for SFL Note Claim	
	7.7.	Claim	s Paid or Payable by Third Parties	39
		(a)	Claims Paid by Third Parties	39
		(b)	Claims Payable by Third Parties	40
	7.8.	Foreig	gn Currency Exchange Rate	40
	7.9.	Setoff	、 ······	40
	7.10.	De Mi	nimis Distributions	40
	7.11.	Fraction	onal Shares	41
	7.12.	No Int	terest	41
VIII.	EXEC	UTOR	Y CONTRACTS AND UNEXPIRED LEASES	41
	8.1.	Assun	nption of Executory Contracts	41
	8.2.	Reject	tion of Executory Contracts	42
	8.3.	Detern	nination of Assumption Disputes and Deemed Consent	42
	8.4.	Effect	of Assumption of Contracts and Leases	43
	8.5.	Payme	ent of Cure Amounts	43
		(a)	No Admission of Liability	44
		(b)	Reservation of Rights	44
		(c)	Rejection Claim Bar Date	44
		(d)	Continuing Obligations Owed to the Debtors	44
		(e)	Postpetition Contracts	45

# TABLE OF CONTENTS (Continued)

IX. CONDITIONS PRECEDENT TO CONFIRMATION OF THE PLAN AND OCCURRENCE OF THE EFFECTIVE DATE					
	9.1.	Conditions to Confirmation	5		
	9.2.	Conditions to Effectiveness	5		
X.		LEMENT, DISCHARGE, RELEASE, INJUNCTION AND TED PROVISIONS40	5		
	10.1.	Compromise and Settlement of Claims, Interests, and Controversies	5		
	10.2.	Releases by the Debtors in Favor of Third Parties46	5		
	10.3.	Releases by Holders of Claims and Interests47	7		
	10.4.	Discharge and Discharge Injunction	)		
		(a) Discharge of Claims	)		
		(b) Discharge Injunction	)		
	10.5.	Exculpation	)		
	10.6.	Post-Effective Date Indemnification	)		
XI.	RETE	NTION OF JURISDICTION	l		
	11.1.	Retention of Jurisdiction	l		
	11.2.	Limitation on Jurisdiction	3		
XII.	MISC	ELLANEOUS PROVISIONS	1		
	12.1.	Legally Binding Effect	1		
	12.2.	Exemption from Transfer Taxes	1		
	12.3.	Securities Exemption	5		
	12.4.	Defects, Omissions and Amendments of the Plan	5		

# Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 9 of 241

# TABLE OF CONTENTS (Continued)

12.5.	Due Authorization by Creditors	.55
12.6.	Filing of Additional Documentation	.56
12.7.	Dissolution of the Official Committee	.56
12.8.	Governing Law	.56
12.9.	Successors and Assigns	.56
12.10.	Transfer of Claims	.56
12.11.	Notices	.57
12.12.	U.S. Trustee Fees and Reports	.59
12.13.	Implementation	.60
12.14.	No Admissions	.60
12.15.	SRGL Consent Rights Reserved	.60
12.16.	Substantial Consummation	.60
12.17.	Final Decree	.60

Scottish Holdings, Inc. ("<u>SHI</u>") and Scottish Annuity & Life Insurance Company (Cayman) Ltd. ("<u>SALIC</u>"), debtors and debtors in possession (the "<u>Debtors</u>") in these Chapter 11 Cases, jointly propose this Plan. Reference is made to the Disclosure Statement in support of the Plan for a discussion of the Debtors' history, business, property and results of operations, and for a summary of the Plan and certain related matters.

All Creditors and other parties in interest are encouraged to read the Plan and the Disclosure Statement in their entirety, as well as the Stock Purchase Agreement, the Restructuring Implementation Agreement and the RIA Order, the terms of which are incorporated into the Plan by reference and form integral parts of the Plan. No materials, other than the Disclosure Statement and any exhibits and schedules thereto or referenced therein, have been approved by the Debtors or the Bankruptcy Court for use in soliciting acceptances or rejections of the Plan.

#### I. SUMMARY OF THE PLAN

An overview of the Plan is set forth in the Disclosure Statement. Generally, the Plan provides for, among other things, the following: (1) the reorganization and recapitalization of the Debtors and certain of their non-debtor Affiliates through a new money contribution of \$12,500,000 by the Purchaser in the form of the Recapitalization Funding Payment; (2) the funding of distributions to the Debtors' Creditors through an additional new money contribution of \$21,500,000 by the Purchaser in the form of the Plan Funding Payment subject to reduction by the amount of the TruPS Returned Cash; (3) in exchange for the foregoing payments and other consideration, the issuance or assignment to the Purchaser of one hundred percent (100%) of the New Equity, subject to downward adjustment to no less than seventy percent (70%), to the extent that eligible unsecured creditors elect to receive their pro rata share of up to thirty percent (30%) of the New Equity, in lieu of a cash distribution under the Plan; (4) the assumption by the Reorganized Debtors of the SALIC Net Worth Maintenance Agreements and all or substantially all reinsurance treaties for which SALIC acts as reinsurer or retrocessionaire; (5) the creation of the Distribution Trust (a) for payment of all Secured Claims, Administrative Claims, and Priority Claims to the extent Allowed and not paid or otherwise satisfied prior to the Effective Date, and (b) for the benefit of Holders of SHI TruPS Claims, SHI General Unsecured Claims, SALIC TruPS Claims and SALIC General Unsecured Claims, all to the extent Allowed; and (6) the funding of the Distribution Trust with the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves.

Pursuant to sections 105(a), 363 and 1123 of the Bankruptcy Code and Bankruptcy Rule 9019, and in consideration for the Distributions and other benefits provided pursuant to the Plan, the provisions of the Plan shall constitute a good faith compromise of all Claims, Interests and controversies relating to the contractual, legal, and subordination rights that a Holder of a Claim or Interest may have with respect to any Allowed Claim or Interest, or any Distribution to be made on account of such Allowed Claim or Interest, as well as the allocation of the Plan Funding Payment among the Debtors' Estates and Creditors. In addition, the Plan contains certain, release, injunction and exculpation provisions all as set forth in Article X of the Plan.

## II. DEFINITIONS AND INTERPRETATION

2.1 All capitalized terms used but not defined elsewhere in the Plan have the meanings assigned to them in the Glossary of Defined Terms attached as <u>Exhibit A</u> to the Plan. Any capitalized term used and not otherwise defined by the Plan has the meaning ascribed to that term in the Bankruptcy Code and/or Bankruptcy Rules.

2.2 For purposes of the Plan, any reference in the Plan to an existing document or exhibit Filed or to be Filed means that document or exhibit as it may have been or may be amended, supplemented, or otherwise modified.

2.3 The words "herein," "hereof" and "hereunder" and other words of similar import refer to the Plan as a whole and not to any particular section, subsection or clause contained in the Plan, unless the context requires otherwise. Whenever from the context it appears appropriate, each term stated in either the singular or the plural includes the singular and the plural, and pronouns stated in the masculine, feminine or neuter gender include the masculine, feminine and the neuter.

2.4 Captions and headings to articles, sections and exhibits are inserted for convenience of reference only and are not intended to be part of or to affect the interpretation of the Plan.

2.5 The rules of construction set forth in section 102 of the Bankruptcy Code shall apply.

2.6 In computing any period of time prescribed or allowed by the Plan, the provisions of Bankruptcy Rule 9006(a) shall apply.

# III. CLASSIFICATION OF CLAIMS AND INTERESTS

#### 3.1. Introduction

All Claims and Interests, except Administrative Claims and Priority Tax Claims, are placed in the Classes set forth below. In accordance with section 1123(a)(1) of the Bankruptcy Code, Administrative Claims and Priority Tax Claims have not been classified, and the treatment of such unclassified Claims is set forth below in <u>Section 4.1</u> of the Plan.

A Claim or Interest is placed in a particular Class only to the extent that the Claim or Interest falls within the description of that Class, and is classified in other Classes to the extent that any portion of the Claim or Interest falls within the description of such other Classes. A Claim is also placed in a particular Class for the purpose of receiving Distributions pursuant to the Plan only to the extent that such Claim is an Allowed Claim in that Class and such Claim has not been paid, released, or otherwise settled prior to the Effective Date.

## **3.2.** Unclassified Claims

- (a) Administrative Claims
- (b) **Priority Tax Claims**

## **3.3.** Classification of Claims and Interests

The following table designates the Classes of Claims against and Interests in the Debtors and specifies which Classes are: (a) Impaired and Unimpaired under the Plan; (b) entitled to vote to accept or reject the Plan in accordance with section 1126 of the Bankruptcy Code; and (c) deemed to accept or reject the Plan:

Class	Type of Claim or Interest	Impairment	<b>Entitled to Vote</b>
Class 1	Secured Claims	Unimpaired	No (deemed to accept)
Class 2	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 3	Intercompany Claims	Unimpaired	No (deemed to accept)
Class 4	SHI TruPS Claims	Impaired	Yes
Class 5	SHI General Unsecured Claims	Impaired	Yes
Class 6	SALIC TruPS Claims & SFL	Impaired	Yes
	Claims		
Class 7	SALIC General Unsecured Claims	Impaired	Yes
Class 8	Subordinated Claims	Impaired	No (deemed to reject)
Class 9	SHI Existing Equity Interests	Impaired	No (deemed to reject)
Class 10	SALIC Existing Equity Interests	Unimpaired	No (deemed to accept)

# IV. TREATMENT OF CLAIMS AND INTERESTS

# 4.1. Unclassified Claims

#### (a) Administrative Claims

Except to the extent that an Allowed Administrative Claim has been satisfied prior to the Effective Date, and except as otherwise provided for herein (including <u>Section</u> <u>4.1(c)(ii)</u> with respect to Professional Fee Claims), each Holder of an Allowed Administrative Claim shall be entitled to receive in full, final and complete settlement, release, and discharge of such Claim, either (i) to the extent such Administrative Claim is Allowed as of the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Administrative Claim as a Closing Date Plan Distribution on the Effective Date, or as soon as reasonably practicable thereafter, or (ii) to the extent such Administrative Claim is Allowed after the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Administrative Claim from the Distribution Trust, in an amount equal to the unpaid portion of such claim, at such time as such Administrative Claim is Allowed, or as soon as reasonably practicable thereafter.

#### (b) **Priority Tax Claims**

Except to the extent that an Allowed Priority Tax Claim has been satisfied prior to the Effective Date, each Holder of an Allowed Priority Tax Claim shall be entitled to receive, in full, final and complete settlement, release, and discharge of such Claim, at the election of the Debtors or the Distribution Trustee, one of the following treatments: (i) to the extent such Priority Tax Claim is Allowed as of the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Priority Tax Claim as a Closing Date Plan Distribution on the Effective Date, or as soon as reasonably practicable thereafter; (ii) to the extent such Priority Tax Claim is Allowed after the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Priority Tax Claim is Allowed, or as soon as reasonably practicable thereafter; or (iii) or such other treatment or payment from the Distribution Trust as permitted under section 1129(a)(9)(C) of the Bankruptcy Code.

#### (c) Other Provisions Governing Administrative Claims

General Provisions. Except as otherwise provided in this Article i. IV, requests for payment of Administrative Claims must be included within an application (setting forth the amount of, and basis for, such Administrative Claims, together with documentary evidence) and Filed and served on respective counsel for the Debtors, the Reorganized Debtors, the Purchaser, and the Distribution Trustee no later than the applicable Administrative Claims Bar Date. Holders of Administrative Claims (including, without limitation, Holders of any Claims for federal, state or local taxes) that are required to File a request for payment of such Claims and that do not File such requests by the applicable Administrative Claims Bar Date shall be forever barred from asserting such Claims against the Debtors, Reorganized Debtors, the Purchaser, the Distribution Trust or any of their respective property. Requests for payments of Administrative Claims included within a Proof of Claim are of no force and effect, and are deemed disallowed in their entirety as of the Effective Date, and shall be satisfied only to the extent such Administrative Claim is subsequently Filed in a timely fashion as provided by this subsection and subsequently becomes an Allowed Claim.

ii. *Professionals.* All Professionals or other Entities requesting compensation or reimbursement of expenses pursuant to sections 327, 328, 330, 331, 503(b) and 1103 of the Bankruptcy Code for services rendered before the Effective Date (including, without limitation, any compensation requested by any Professional or any other Entity for making a substantial contribution in the Chapter 11 Cases) shall File an application for final allowance of compensation and reimbursement of expenses no later than thirty (30) days after the Effective Date and simultaneously serve such application on counsel for the following entities: the Reorganized Debtors, the Purchaser, the Official Committee, the Distribution Trustee, and the U.S. Trustee.

Objections, if any, to a Professional's application for compensation or reimbursement of expenses must be Filed no later than twenty-one (21) days after the date the application is Filed, and simultaneously served on the applicant (and its counsel, if any) and

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 14 of 241

counsel for the following Entities: the Reorganized Debtors, the Purchaser, the Official Committee, the Distribution Trustee, and the U.S. Trustee. If no objections are received, the Bankruptcy Court may enter a final order approving the applications and authorizing final Allowance and payment of compensation and reimbursement of expenses without a hearing. If any objection cannot be resolved consensually, the Bankruptcy Court will hold a hearing on the affected application(s).

The amount of compensation and reimbursement of expenses Allowed by the Bankruptcy Court (less (i) amounts previously received by the Professional in respect of interim compensation and (ii) any unapplied retainer or advance held by the Professional) shall be paid by the Distribution Trustee from the Professional Fee Reserve.

Any professional fees and reimbursements or expenses incurred by the Distribution Trust subsequent to the Effective Date may be paid in accordance with the Distribution Trust Agreement. Any professional fees and reimbursements or expenses incurred by the Reorganized Debtors subsequent to the Effective Date may be paid without further order of, or application to, the Bankruptcy Court.

#### (d) Indenture Trustee Fees

i. Indenture Trustee Fees Incurred through the Effective Date. On the DT Initial Distribution Date, prior to making the Distributions under Section 4.3(c) of Cash or Offered New Equity (as applicable) on account of each Allowed TruPS Claim, subject to the review of the IT Fee Notice Parties as set forth in Section 4.1(d)(iii), the Distribution Trustee shall, in connection with such TruPS Indenture Trustee's Charging Lien, first (a) pay to each TruPS Indenture Trustee in Cash in full the undisputed Indenture Trustee Fees incurred by such TruPS Indenture Trustee in connection with the Chapter 11 Cases through the Effective Date, and then (b) reserve for each TruPS Indenture Trustee in Cash in full any disputed Indenture Trustee Fees incurred by such TruPS Indenture Trustee through the Effective Date. The payment to each TruPS Indenture Trustee shall, as set forth more fully in Section 4.1(d)(ii), come from the DT Initial Distribution made to Holders of Allowed TruPS Claims related to such trustee, such that:

A. With respect to the SHST I TruPS Indenture Trustee, the payment shall come from the DT Initial Distribution made on account of the Allowed SHST I TruPS Claims;

B. With respect to the SHST II TruPS Indenture Trustee, the payment shall come from the DT Initial Distribution made on account of the Allowed SHST II TruPS Claims;

C. With respect to the GPIC TruPS Indenture Trustee, the payment shall come from the DT Initial Distribution made on account of the Allowed GPIC TruPS Claims;

D. With respect to the SHST III TruPS Indenture

Trustee, the payment shall come from the DT Initial Distribution made on account of the Allowed SHST III TruPS Claims; and

E. With respect to the SFLST I TruPS Indenture Trustee, the payment shall come from the DT Initial Distribution made on account of the Allowed SFLST I TruPS Claims.

The DT Initial Distribution for each Allowed TruPS Claim shall ii. each be used: (1) first, to satisfy the undisputed Indenture Trustee Fees for the respective TruPS Debenture in Cash; (2) second, to fund in Cash the Post-Effective Date Indenture Trustee Fee Reserve on account of disputed Indenture Trustee Fees for the respective TruPS Debenture incurred before the Effective Date; (3) third, to fund \$50,000 to thePost-Effective Date Indenture Trustee Fee Reserve for the TruPS Indenture Trustee Fees that may be incurred post-Effective Date with respect to the SHST I TruPS, the SHST III TruPS and the SFLST I TruPS, which amount shall be an aggregate amount for all such post-Effective Date Indenture Trustee Fees, but shall be in addition to any amounts reserved pursuant to clause (2) with respect to Indenture Trustee Fees incurred before the Effective Date, and (4) fourth, to distribute the remaining DT Initial Distribution to be made to Allowed TruPS Claims in Cash or Offered New Equity based upon the election of the respective Holders of Allowed TruPS Claims; provided, that, the amount of Offered New Equity to be distributed to Holders of Allowed TruPS Claims that make the New Equity Election will be adjusted downward to account for such Allowed TruPS Claims' respective share of Indenture Trustee Fees payable and reserved.

iii. Notwithstanding the foregoing, the IT Fee Notice Parties shall have the right to review such Indenture Trustee Fees as set forth in <u>Section 4.1(d)(iv)</u>. The Charging Lien of each Indenture Trustee will be satisfied to the extent that such trustee receives payment of its Indenture Trustee Fees; provided, however, that nothing herein will prevent the a TruPS Indenture Trustee from asserting its respective Charging Lien for unpaid Indenture Trustee Fees owed to it, including the SFL I TruPS Indenture Trustee asserting its Charging Lien against any distributions it receives from SFL. Other than as expressly set forth herein, no TruPS Indenture Trustee shall be eligible to receive payment from the Distribution Trust or maintain a Charging Lien for Indenture Trustee Fees incurred after the Effective Date for services related to Distributions on account of Allowed TruPS Claims. For the avoidance of doubt, the Distribution Trustee for the SHST II Debenture Trustee Fees that may be incurred by the TruPS Indenture Trustees for the SHST II Debentures or the GPIC Debentures.

iv. Review of Indenture Trustee Fees and Disputes Regarding Indenture Trustee Fees.

A. (x) Five (5) days after the conclusion of the Confirmation Hearing and (y) five (5) days after the Effective Date, each TruPS Indenture Trustee shall deliver by email to the IT Fee Notice Parties and their counsel invoices setting forth in reasonable detail (but subject to appropriate redactions to preserve confidentiality or any applicable privileges or protections) the Indenture Trustee Fees incurred by such TruPS Indenture Trustee and its counsel and agents. If an IT Fee Notice Party has not served the applicable TruPS Indenture Trustee with a written objection to the TruPS Indenture Trustee Fees set forth in such invoices within ten (10) Business Days after the date of service of such invoices, then the subject Indenture Trustee Fees shall be Allowed and paid in accordance with section 4.1(d)(i) without the need for application to, or approval by, any court.

B. If any IT Fee Notice Party disputes any requested Indenture Trustee Fees, such party shall notify the applicable TruPS Indenture Trustee and the Distribution Trustee, and the other IT Fee Notice Parties, and, upon such notification, the applicable TruPS Indenture Trustee may submit such dispute for resolution to the Bankruptcy Court on an expedited basis if the IT Fee Notice Party and the applicable TruPS Indenture Trustee are unable to resolve such dispute consensually within five (5) days of receipt of a written objection. The Distribution Trustee will reserve the disputed Indenture Trustee Fees in the Post-Effective Date Indenture Trustee Fee Reserve pending resolution of the dispute. If there are excess funds in the Post-Effective Date Indenture Trustee Fee Reserve after resolution of the dispute, such funds will be distributed to the applicable holders of Allowed TruPS Claim.

# 4.2. Unimpaired Classes of Claims and Interests

# (a) Class 1 – Secured Claims

i. *Classification*. Class 1 consists of all Secured Claims, to the extent such Claims have not already been satisfied during the Chapter 11 Cases.

ii. Treatment. Unless a Holder of an Allowed Secured Claim agrees to lesser treatment, on the Effective Date, or as soon as reasonably practicable thereafter, each Holder of an Allowed Secured Claim shall receive one of the following treatments on account of such Allowed Secured Claim, at the option of the Debtors or the Distribution Trustee, as applicable: (a) reinstatement of the Allowed Secured Claim as against any collateral or proceeds thereof held by the Distribution Trust; (b) with the consent of the Purchaser, reinstatement of the Allowed Secured Claim as against any collateral or proceeds thereof held by the Reorganized Debtors; (c) in full and final satisfaction, compromise, settlement, release, and discharge of and in exchange for such Allowed Secured Claim, Cash equal to the full Allowed amount of such Secured Claim, with such Cash to be paid (i) as a Closing Date Plan Distribution to the extent that such Secured Claim is Allowed as of the Effective Date or (ii) from the assets of the Distribution Trust to the extent that such Secured Claim is Allowed after the Effective Date; or (d) with the consent of the Purchaser as to any asset that is not a Distribution Trust Asset, delivery of the collateral securing any such Claim and payment of any interest required under section 506(b) of the Bankruptcy Code.

iii. *Voting.* Class 1 is Unimpaired and the Holders of Claims in Class 1 will be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code. Therefore, the Holders of Claims in Class 1 will not be entitled to vote to accept or reject the Plan.

### (b) Class 2 – Priority Non-Tax Claims

i. *Classification*. Class 2 consists of all Priority Non-Tax Claims, to the extent such Claims have not already been satisfied during the Chapter 11 Cases.

ii. *Treatment.* Unless a Holder of an Allowed Priority Non-Tax Claim agrees to lesser treatment, on the Effective Date, or as soon as reasonably practicable thereafter, each Holder of an Allowed Priority Non-Tax Claim shall receive in full, final and complete satisfaction, settlement, release, and discharge of such Allowed Priority Non-Tax Claim, either: (i) to the extent such Priority Non-Tax Claim is Allowed as of the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Priority Non-Tax Claim as a Closing Date Plan Distribution on the Effective Date, or as soon as reasonably practicable thereafter, or (ii) to the extent such Priority Non-Tax Claim is Allowed after the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Priority Non-Tax Claim from the Distribution Trust at such time as such Priority Non-Tax Claim is Allowed, or as soon as reasonably practicable thereafter.

iii. *Voting.* Class 2 is Unimpaired and the Holders of Claims in Class 2 will be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code. Therefore, the Holders of Claims in Class 2 will not be entitled to vote to accept or reject the Plan.

## (c) Class 3 – Intercompany Claims

i. *Classification*. Class 3 consists of all Intercompany Claims, to the extent such Claims have not already been satisfied during the Chapter 11 Cases.

ii. *Treatment.* Intercompany Claims shall be paid, adjusted, continued, settled, reinstated, discharged, eliminated, or otherwise managed, in each case to the extent determined to be appropriate by the applicable Debtor(s) or Reorganized Debtor(s) and certain of their non-debtor Affiliates with the consent of the Purchaser. For the avoidance of doubt, Intercompany Claims shall not receive a Distribution and shall not otherwise be entitled to any of the assets of the Distribution Trust.

iii. *Voting.* Class 3 is Unimpaired and the Holders of Claims in Class 3 will be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code. Therefore, the Holders of Claims in Class 3 will not be entitled to vote to accept or reject the Plan.

# (d) Class 10 – SALIC Existing Equity Interests

Interests.

i. Classification. Class 10 consists of all SALIC Existing Equity

ii. *Treatment.* SALIC Existing Equity Interests are Unimpaired by the Plan and will be treated in accordance with the Stock Purchase Agreement, the New SALIC

Shares Issuance Documents, the Share Surrender Documents, and the Restructuring Implementation Agreement, as provided in <u>Section 6.1</u> of the Plan.

iii. *Voting.* Class 10 is Unimpaired and the Holders of Interests in Class 10 will be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code. Therefore, the Holders of Interests in Class 10 will not be entitled to vote to accept or reject the Plan.

## 4.3. Impaired Voting Classes of Claims

# (a) Class 4 – SHI TruPS Claims

i. *Classification*. Class 4 consists of all SHI TruPS Claims.

ii. *Allowance*. The SHI TruPS Claims shall be allowed as follows (inclusive of all principal amount issued pursuant to the applicable TruPS Indentures and related documents and accrued but unpaid interest as of the Petition Date at the applicable rates specified in the applicable TruPS Indentures and related documents, other than Indenture Trustee Fees), and shall not be subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual or otherwise) except as provided herein, counterclaim, cross-claim, defense or disallowance under Applicable Law:

TruPS Debenture Issuance	TruPS Indenture Trustee	Holder of Allowed SHI TruPS Claim	Principal	Interest through Petition Date	Total Allowed SHI TruPS Claim
SHST I TruPS Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, on behalf of Beneficial Holders	\$18,042,000.00	\$4,805,863.87	\$22,847,863.87
SHST II TruPS Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, exclusively for SRGL	\$20,619,000.00	\$5,528,239.88	\$26,147,239.88
GPIC TruPS Debentures	BNYM, as Indenture Trustee	BNYM, as Indenture Trustee, exclusively for SRGL	\$10,310,000.00	\$2,561,006.29	\$12,873,506.29
SHST III TruPS Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, on	\$32,990,000.00	\$8,310,979.84	\$41,300,979.84

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 19 of 241

TruPS Debenture Issuance	TruPS Indenture Trustee	Holder of Allowed SHI TruPS Claim	Principal	Interest through Petition Date	Total Allowed SHI TruPS Claim
		behalf of Beneficial Holders			
TOTAL			\$81,961,000.00	\$21,206,089.88	\$103,169,589.88

iii. Treatment.

(1)

A. With respect to Eligible SHI TruPS Claims, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all Eligible SHI TruPS Claims, each Beneficial Holder of TruPS on account of its Allocated Portion of SHI TruPS Claim will receive (to be distributed in accordance with the procedures set forth in Sections 6.1(f) & (g), 6.3(i), 7.5 and 7.6 of this Plan) the following Distributions:

Beneficial Holder, the following:

(a) Either (x) if the Beneficial Holder makes the New Equity Election, then the Beneficial Holder's TruPS Claims Equity Distribution Amount; or (y) if the Beneficial Holder makes the Cash Election, the Beneficial Holder's TruPS/GUC Claims Cash Distribution Amount; and

(b) The Beneficial Holder's applicable percentage of the Distribution Trust Asset Proceeds, calculated as the amount of such New Equity Eligible Beneficial Holder's Allocated Portion of the Allowed Eligible SHI TruPS Claim divided by the TruPS/GUC Claims Aggregate Amount.

(2) If such Beneficial Holder is SRGL, the following:

(a) SRGL's TruPS/GUC Claims Cash

Distribution Amount; and

(b) SRGL's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of SRGL's Allocated Portion of the Allowed Eligible SHI TruPS Claim divided by the TruPS/GUC Claims Aggregate Amount.

If such Beneficial Holder is a New Equity Eligible

B. With respect to SRGL Exclusively Held SHI TruPS Claims, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all SRGL Exclusively Held SHI TruPS Claims, SRGL shall receive (to be distributed in accordance with the procedures set forth in <u>Sections 6.1(f) & (g), 6.3(i), 7.5 and 7.6</u> of this Plan):

(1) SRGL's TruPS/GUC Claims Cash Distribution

(2) SRGL's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SRGL Exclusively Held SHI TruPS Claims divided by TruPS/GUC Claims Aggregate Amount.

iv. *Voting*. Class 4 is Impaired, and the Holders of Allowed Class 4 Claims will be entitled to vote to accept or reject the Plan.

# (b) Class 5 – SHI General Unsecured Claims

(1)

Claims.

Amount; and

Classification. Class 5 consists of all SHI General Unsecured

#### ii. *Treatment*.

i.

A. On or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all SHI General Unsecured Claims, each Holder of an Allowed SHI General Unsecured Claim shall receive the following Distributions:

Amount; and

(2) The Holder's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed Amount of the Holder's SHI General Unsecured Claims divided by the TruPS/GUC Claims Aggregate Amount.

iii. *Voting.* Class 5 is Impaired, and the Holders of Allowed Class 5 Claims will be entitled to vote to accept or reject the Plan.

# (c) Class 6 – SALIC TruPS Claims & SFL Claims

i. *Classification*. Class 6 consists of all SALIC TruPS Claims and the SFL Note Claim.

ii. Allowance.

A. <u>SALIC TruPS Claims</u>: The SALIC TruPS Claims

The Holder's TruPS/GUC Claims Cash Distribution

shall be allowed as follows (inclusive of all principal amount issued pursuant to the applicable TruPS Indentures and related documents and accrued but unpaid interest as of the Petition Date at the applicable rates specified in the applicable TruPS Indentures and related documents, other than Indenture Trustee Fees), and shall not be subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual or otherwise) except as provided herein, counterclaim, cross-claim, defense or disallowance under Applicable Law:

TruPS Debenture Issuance	TruPS Indenture Trustee	Holder of Allowed SALIC TruPS Claim	Principal	Interest through Petition Date	Total Allowed SALIC TruPS Claim
SHST I Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, on behalf of Beneficial Holders	\$18,042,000.00	\$4,805,863.87	\$22,847,863.87
SHST II Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, exclusively for SRGL	\$20,619,000.00	\$5,528,239.88	\$26,147,239.88
GPIC Debentures	BNYM, as Indenture Trustee	BNYM, as Indenture Trustee, exclusively for SRGL	\$10,310,000.00	\$2,561,006.29	\$12,873,506.29
SHST III Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, on behalf of Beneficial Holders	\$32,990,000.00	\$8,310,979.84	\$41,300,979.84
SFLST I Debentures	WTC, as Indenture Trustee	WTC, as Indenture Trustee, on behalf of Beneficial Holders	\$51,547,000.00	\$11,989,041.32	\$63,536,041.32
TOTAL			\$133,508,000.00	\$33,195,131.20	\$166,705,631.20

#### B. <u>SFL Note Claim</u>:

(1) If the SFLST I TruPS CDO Facility Holders each have voted their respective Allocated Portions of SALIC TruPS Claims to accept the Plan, then (a) the SFL Note Claim Allowance Conditions (as defined below) will be deemed satisfied, such that the Holder of the SFL Note Claim will be deemed (i) to have voted the entire SFL Note Claim to accept the Plan and (ii) to not have opted out of the "Releases by Holders of Claims and Interests" set forth in <u>Section 10.3</u> of the Plan, and (b) the SFL Note Claim shall be deemed Allowed as a Class 6 Claim in the amount of \$63,536,014.32, and shall not be subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual or otherwise) except as provided herein, counterclaim, cross-claim, defense or disallowance under Applicable Law.

(2) If one or more of the SFLST I TruPS CDO Facility Holders has not voted its Allocated Portion of SALIC TruPS Claims to accept the Plan, then whether or not the SFL Note Claim will be Allowed pursuant to this Plan as a Class 6 Claim in the amount of \$63,536,014.32 will depend on whether the SFL Note Claim Allowance Conditions are satisfied or waived as follows:

(a) If the Holder of the SFL Note Claim (1) votes the SFL Note Claim to accept the Plan, (2) does not object to confirmation of the Plan, and (3) does not opt out of the "Releases by Holders of Claims and Interests" set forth in <u>Section 10.3</u> of the Plan (together, the "<u>SFL Note Claim Allowance Conditions</u>"), then upon the occurrence of the Effective Date, the SFL Note Claim shall be deemed Allowed as a Class 6 Claim in the amount of \$63,536,014.32, and shall not be subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual or otherwise) except as provided herein, counterclaim, cross-claim, defense or disallowance under Applicable Law, and shall be entitled, at such Holder's option to make (i) the New Equity Election, (ii) the Cash Election, or (iii) elect to allocate its Claim between the New Equity Election or the Cash Election, as set forth in <u>Section 4.3(c)(iii)</u> below.

(b) If the Holder of the SFL Note Claim fails to satisfy one or more of the SFL Note Claim Allowance Conditions, then, unless otherwise agreed in a writing signed by an authorized representative of the Debtors (or, if on or after the Effective Date, the Distribution Trustee) and consented to by the Purchaser (which consent shall not be unreasonably withheld) or adjudicated by a Final Order of the Bankruptcy Court, the SFL Note Claim shall (a) remain fully subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual or otherwise), counterclaim, cross-claim, defense or disallowance under Applicable Law and treated as a Disputed Claim and (b) be deemed to have made the Cash Election and will be reserved for in Cash as a Disputed Class 6 Claim. For the avoidance of doubt, after the Effective Date, the Distribution Trustee shall have the sole right and authority, but not the obligation, to object to, litigate, and settle the amount, priority or extent of the SFL Note Claim and to make a Cash Distribution thereon to the extent Allowed. iii. Treatment:

A. With respect to Eligible SALIC TruPS Claims, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all Eligible SALIC TruPS Claims, each Beneficial Holder of TruPS on account of its Allocated Portion of SALIC TruPS Claim arising from or relating to the TruPS issuance for which it is a Beneficial Holder shall receive (to be distributed in accordance with the procedures set forth in Sections <u>6.1(f) & (g), 6.3(i), 7.5 and 7.6</u> of this Plan) the following Distributions:

Beneficial Holder, the following:

(1) If such Beneficial Holder is a New Equity Eligible

(a) Either (x) if the Beneficial Holder makes the New Equity Election, then the Beneficial Holder's TruPS Claims Equity Distribution Amount; or (y) if the Beneficial Holder makes the Cash Election, the Beneficial Holder's TruPS/GUC Claims Cash Distribution Amount; and

(b) The Beneficial Holder's applicable percentage of the Distribution Trust Asset Proceeds, calculated as the amount of such New Equity Eligible Beneficial Holder's Allocated Portion of the Allowed Eligible SALIC TruPS Claim divided by the TruPS/GUC Claims Aggregate Amount.

(2) If such Beneficial Holder is SRGL, the following:

(a) SRGL's TruPS/GUC Claims Cash

Distribution Amount; and

(b) SRGL's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of SRGL's Allocated Portion of the Allowed Eligible SALIC TruPS Claim divided by the TruPS/GUC Claims Aggregate Amount.

B. With respect to SRGL Exclusively Held SALIC TruPS Claims, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Date(s) (as applicable), in full and final satisfaction of and in exchange for all SRGL Exclusively Held SALIC TruPS Claims, SRGL shall receive (to be distributed in accordance with the procedures set forth in Sections 6.1(f) & (g), 6.3(i), 7.5 and 7.6 of this Plan) the following Distributions:

Amount; and

(1) SRGL's TruPS/GUC Claims Cash Distribution

(2) SRGL's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SRGL Exclusively Held SALIC TruPS Claims divided by the TruPS/GUC Claims Aggregate Amount. C. With respect to the SFL Note Claim, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all SFL Claims, the Holder of the SFL Note Claim shall receive (to be distributed in accordance with the procedures set forth in Sections 6.1(f) & (g), 6.3(i), 7.5 and 7.6 of this Plan) the following:

(1) If the SFL Note Claim Allowance Conditions are deemed satisfied in accordance with Section 4.3(c)(ii)(B)(1) of this Plan because the SFLST I TruPS CDO Facility Holders each have voted their respective Allocated Portions of SALIC TruPS Claims to accept the Plan, then the Distributions to be received on account of the Allowed SFL Note Claim shall consist of the following:

(a) Subject to the SFL Receiver's right to make the SFL Bankruptcy Administration Senior Cash Election, the Holder of the SFL Note Claim will be deemed to have made the New Equity Election for the remaining portion of the Allowed amount of the SFL Note Claim (*i.e.*, the Allowed SFL Note Claim net of the portion for which the SFL Receiver has made the SFL Bankruptcy Administration Senior Cash Election) that corresponds to the percentage of SFLST I TruPS to have made the New Equity Election (as determined by the Ballots received from the SFLST I TruPS CDO Facility Holders). Accordingly, the Holder of the Allowed SFL Note Claim will be entitled to receive the TruPS Claims Equity Distribution Amount calculated based on the portion of the Allowed SFL Note Claim for which it has deemed to have made the New Equity Election as described immediately above. Any remaining portion of the Holder's Allowed SFL Note Claim shall be treated as if the Cash Election had been made and entitled to receive the TruPS/GUC Claims Cash Distribution Amount calculated based on such remaining portion of the Allowed SFL Note Claim.

(b) The Holder of the Allowed SFL Note Claim's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SFL Note Claim divided by the TruPS/GUC Claims Aggregate Amount.

(2) If one or more of the SFLST I TruPS CDO Facility Holders has not voted its Allocated Portion of SALIC TruPS Claims to accept the Plan, but the Holder of the SFL Note Claim has otherwise satisfied the SFL Note Claim Allowance Conditions in accordance with Section 4.3(c)(ii)(B)(2)(a) of this Plan, then the Distributions to be received on account of the Allowed SFL Note Claim shall consist of the following:

(a) Either (x) if the Holder of the Allowed SFL Note Claim makes the New Equity Election, such Holder's TruPS Claims Equity Distribution Amount; or (y) if the Holder of the Allowed SFL Note Claim makes the Cash Election, such Holder's TruPS/GUC Claims Cash Distribution Amount and

(b) The Holder of the Allowed SFL Note Claim's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SFL Note Claim divided by the TruPS/GUC Claims Aggregate Amount. (3) If the SFL Note Claim Allowance Conditions are not satisfied or deemed satisfied in accordance with this Plan, then subject to and upon the Allowance of the SFL Note Claim post-Effective Date as a Class 6 Claim, then the Distributions to be received on account of the Allowed SFL Note Claim shall consist of the following:

(a) The Holder of the Allowed SFL Note Claim's TruPS/GUC Claims Cash Distribution Amount; and

(b) The Holder of the Allowed SFL Note Claim's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SFL Note Claim divided by the TruPS/GUC Claims Aggregate Amount.

D. <u>Voting</u>. Class 6 is Impaired, and the Holders of Allowed Class 6 Claims will be entitled to vote to accept or reject the Plan; *provided, however*, that with respect to the Allowed SFL Note Claim, if the SFLST I TruPS CDO Facility Holders each have voted their respective Allocated Portions of SALIC TruPS Claims to accept the Plan as contemplated by Section 4.3(c)(ii)(B)(1) of this Plan, then the Holder of the SFL Note Claim shall be deemed to have voted the SFL Note Claim to accept the Plan.

# (d) Class 7 – SALIC General Unsecured Claims

Classification. Class 7 consists of all SALIC General Unsecured

Claims.

ii. *Treatment*.

i.

A. On or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all Allowed SALIC General Unsecured Claims, each Holder of an Allowed SALIC General Unsecured Claim shall receive the following Distributions:

(1) The Holder's TruPS/GUC Claims Cash Distribution

Amount; and

(2) The Holder's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed Amount of the Holder's SALIC General Unsecured Claims divided by the TruPS/GUC Claims Aggregate Amount.

iii. *Voting.* Class 7 is Impaired, and the Holders of Allowed Class 7 Claims will be entitled to vote to accept or reject the Plan.

## 4.4. Impaired Non-Voting Classes of Claims and Interests

## (a) Class 8 – Subordinated Claims

i. *Classification*. Class 8 consists of all Subordinated Claims.

ii. *Treatment.* Holders of Subordinated Claims will not receive or retain any property on account of such Claims. On the Effective Date, Subordinated Claims shall be deemed automatically cancelled, released, and extinguished without further action by any Debtor, any Reorganized Debtor or the Distribution Trustee, and the obligations of the Debtors thereunder shall be forever discharged.

iii. *Voting.* Class 8 is Impaired, and each Holder of a Subordinated Claim will be conclusively deemed to have rejected the Plan pursuant to section 1126(g) of the Bankruptcy Code. Therefore, Holders of Subordinated Claims shall not be entitled to vote on the Plan.

#### (b) Class 9 – SHI Existing Equity Interests

Interests.

i. Classification. Class 9 consists of all SHI Existing Equity

ii. *Treatment*. All SHI Existing Equity Interests will be cancelled and reissued at the direction of the Purchaser as described in <u>Section 6.1(d)</u> of the Plan.

iii. *Voting.* Class 9 is Impaired, and each Holder of an SHI Existing Equity Interest will be conclusively deemed to have rejected the Plan pursuant to section 1126(g) of the Bankruptcy Code. Therefore, Holders of SHI Existing Equity Interests shall not be entitled to vote on the Plan.

# V. ACCEPTANCE REQUIREMENTS

# 5.1. Impaired Classes of Claims Entitled to Vote

Holders of Allowed Claims in each Impaired Class that will receive a Distribution are entitled to vote as a Class to accept or reject the Plan. Accordingly, only the votes of Holders of Claims in Classes 4, 5, 6 and 7 shall be solicited with respect to the Plan. A Holder of a Disputed Claim which has not been temporarily allowed for purposes of voting on the Plan may vote only such Disputed Claim in an amount equal to the portion, if any, of such Claim shown as fixed, liquidated, and undisputed in the Schedules.

# 5.2. Acceptance by an Impaired Class

In accordance with section 1126(c) of the Bankruptcy Code, and except as provided in section 1126(e) of the Bankruptcy Code, an Impaired Class shall have accepted the Plan if the Plan is accepted by the Holders of at least two-thirds (2/3) in dollar amount and more

than one-half (1/2) in number of the Allowed Claims in such Class that have timely and properly voted to accept or reject the Plan.

# 5.3. Presumed Acceptance by Unimpaired Classes

Classes 1, 2, 3, and 10 are Unimpaired under the Plan. Under section 1126(f) of the Bankruptcy Code, Holders of such Unimpaired Claims and Interests are conclusively presumed to have accepted the Plan, and the votes of Holders of such Claims and Interests shall not be solicited.

# 5.4. Presumed Rejection by Impaired Classes Not Receiving Any Distribution under the Plan

Classes 8 and 9 are Impaired under the Plan, and Holders of Claims and Interests in such Classes will not receive or retain any property under the Plan on account of such Claims or Interests. Under section 1126(g) of the Bankruptcy Code, Holders of such Claims and Interests are conclusively presumed to have rejected the Plan, and the votes of Holders of such Claims and Interests shall not be solicited.

# 5.5. Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code

To the extent that any Impaired Class rejects the Plan or is deemed to have rejected the Plan, the Debtors shall request Confirmation of the Plan under section 1129(b) of the Bankruptcy Code. The Debtors, subject to the terms of the Plan, the Stock Purchase Agreement, and the Restructuring Implementation Agreement, reserve the right to alter, amend, modify, revoke, or withdraw the Plan, the Plan Supplement, or any schedule or exhibit, including to amend or modify it to satisfy the requirements of section 1129(b) of the Bankruptcy Code, if necessary.

In connection with requesting Confirmation of the Plan under section 1129(b) of the Bankruptcy Code, the Debtors, with the consent of the Purchaser, reserve the right to seek Confirmation of the Plan on a "per plan" basis (as opposed to a "per debtor" basis) consistent with *In re Matter of Transwest Resort Properties, Inc.*, 881 F.3d 724 (9th Cir. 2018).

# 5.6. Elimination of Vacant Classes

Any Class that, as of the date of commencement of the Confirmation Hearing, does not contain any Allowed Claim or Interest, or any Claim temporarily allowed under Bankruptcy Rule 3018, shall be deemed eliminated from the Plan for purposes of voting to accept or reject the Plan, and disregarded for purposes of determining whether the Plan satisfies section 1129(a)(8) of the Bankruptcy Code with respect to such Class.

# 5.7. Presumed Acceptance by Voting Classes in Which No Votes Are Cast

If a Class contains Claims eligible to vote and no Holder of a Claim eligible to vote in such Class votes to accept or reject the Plan, the Plan shall be presumed accepted by the Holders of such Claims in such Class.

# 5.8. Consolidation of Classes

To the extent permitted under section 1122 of the Bankruptcy Code, and subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement, the Debtors reserve the right to consolidate one or more Classes of Claims, including for purposes of sections 1126, 1129(a)(8), 1129(a)(10) or 1129(b) of the Bankruptcy Code.

# 5.9. Separate Classes of Secured Claims

Although all Secured Claims have been placed in one Class for purposes of nomenclature within the Plan, each Secured Claim, to the extent secured by a Lien on collateral different from the collateral securing another Secured Claim, shall be treated as being in a separate sub-Class for the purposes of receiving Distributions.

# VI. MEANS FOR IMPLEMENTATION OF THE PLAN

# 6.1. Plan Transactions

# (a) Stock Purchase Agreement Closing

On or prior to the Effective Date, and as a condition to the Effective Date, each of the actions, transactions, and deliveries described in the Stock Purchase Agreement shall occur and shall have occurred, including, without limitation, the actions, transactions, and deliveries described in section 2.4 of the Stock Purchase Agreement.

# (b) Funding, Allocation and Use of Plan Funding Payment; Allocation and Use of Debtors' Unrestricted Cash

On the Effective Date, the Plan Funding Payment shall be funded by the Purchaser in accordance with the terms of the Stock Purchase Agreement and this Plan and allocated and used as follows (such waterfall, the "<u>Allocation/Use Priorities</u>"):

- (1) <u>First</u>, to fund all Closing Date Plan Distributions; <u>provided</u>, <u>however</u>, that such amounts shall be funded on or before the Effective Date from the unrestricted Cash of SALIC and SHI to the extent of such unrestricted Cash;
- (2) <u>Second</u>, to fund the Professional Fee Reserve;

- (3) <u>Third</u>, to fund the Trust Administration Reserve up to the Trust Administration Reserve Cap;
- (4) <u>Fourth</u>, to fund the Disputed Claims Reserve to be maintained by the Distribution Trust;
- (5) <u>Fifth</u>, to fund Distributions to the Holders of Allowed SHI General Unsecured Claims, Allowed SALIC General Unsecured Claims, Allowed SHI TruPS Claims (that make the Cash Election), Allowed SALIC TruPS Claims (that make the Cash Election), and Allowed SFL Note Claim (if the Cash Election is made or deemed to have been made) with such Distributions to be made on the DT Initial Distribution Date or DT Subsequent Distribution Dates, as applicable.

The Plan Funding Payment shall be adjusted downward by the TruPS Returned Cash on account of any amount of the Available Plan Funding Distribution Amount allocable to Allowed TruPS Claims that elect to receive the New Equity. For the avoidance of doubt, the TruPS Returned Cash shall be returned by the Distribution Trust to the Hildene Master Funds or their designee in such proportion as they direct on the DT Initial Distribution Date. Any portion of the Plan Funding Payment that is subsequently released from the Disputed Claims Reserve, the Professional Fee Reserve, or the Trustee Administration Reserve, shall be released by the Distribution Trustee pro rata to (x) the Hildene Master Funds or their designee on account of the TruPS Returned Cash and (y) Holders of Allowed SHI General Unsecured Claims, Allowed SALIC General Unsecured Claims, Allowed SHI TruPS Claims (that make the Cash Election), and Allowed SFL Note Claim (if the Cash Election is made or deemed to have been made). Except as stated in Section 6.1(b)(1) of the Plan, the unrestricted Cash of SALIC and SHI shall remain with the Reorganized Debtors.

# (c) Funding of Recapitalization Funding Payment

On the Effective Date, the Recapitalization Funding Payment shall be funded to Reorganized SALIC by the Purchaser in accordance with the terms of the Stock Purchase Agreement and this Plan. The Recapitalization Funding Payment shall not be used to make Distributions.

# (d) Cancellation of SHI Existing Equity Interests; Issuance of New SHI Equity

On the Effective Date, all SHI Existing Equity Interests shall be cancelled and New SHI Equity shall be issued to the Purchaser or to another entity at the direction of the Purchaser in its sole discretion. Unless the Purchaser determines otherwise in its sole discretion, the New SHI Equity shall be deemed immediately contributed by the Purchaser to Reorganized SALIC.

#### (e) Final Share Surrender

On the Effective Date and immediately following the New SALIC Equity issuance to Purchaser, in accordance with the terms and conditions of the Restructuring Implementation Agreement, the Stock Purchase Agreement and this Plan, SRGL shall complete the Final Share Surrender (as defined in the Restructuring Implementation Agreement). For the avoidance of doubt, as a result of the Plan, SRGL as the holder of the SALIC Existing Equity Interests shall not receive or retain any property under the Plan on account of such SALIC Existing Equity Interests.

#### (f) New Equity Issuance and Distribution

On the Effective Date, without further act or action under Applicable Law (other than as required by Applicable Law of the Cayman Islands with respect to SRGL and SALIC and provided for in the Restructuring Implementation Agreement and the RIA Order), in accordance with the terms and conditions of the Stock Purchase Agreement, the Restructuring Implementation Agreement, the RIA Order and this Plan, the New Equity shall be issued and distributed by New Holdco as reflected in the organizational chart included in the Plan Supplement. Such New Equity shall be issued and distributed free and clear of all Liens, Claims and other Interests, except as expressly provided in this Plan.

On or before the deadline established by the Disclosure Statement Order for the filing of the Plan Supplement, the Debtors shall File a notice stating whether the New Equity will be issued by Reorganized SALIC or New Holdco, which notice may be Filed as part of the Plan Supplement in the form of an organizational chart. Any recipient or subsequent holder of shares of New Equity shall be required to enter into the Stockholders Agreement, whether such recipient or holder acquires such shares as of the Effective Date or subsequent thereto. The New Corporate Governance Documents (including the Stockholders Agreement) will include certain restrictions on transfers of the New Equity, which shall be reasonably acceptable to the Purchaser in consultation with the Official Committee and the Debtors, and disclosed in the Plan Supplement.

The New Equity when issued or distributed as provided in the Plan, will be duly authorized, validly issued and, if applicable, fully paid and nonassessable. Each Distribution and issuance of such New Equity shall be governed by the terms and conditions set forth in the Plan applicable to such Distribution or issuance and by the terms and conditions of the instruments evidencing or relating to such Distribution or issuance, which terms and conditions shall bind each Person receiving such Distribution or issuance.

The Debtors, the Purchaser, the Indenture Trustees, the Official Committee, SRGL, the Voting Agent, and each of their respective Representatives have, and upon Confirmation of the Plan shall be deemed to have, participated in good faith and in compliance with the applicable provisions of the Bankruptcy Code and Applicable Law with regard to the Distribution of the New Equity under the Plan, and therefore are not, and on account of such Distributions will not be, liable at any time for the violation of any Applicable Law, rule, or regulation governing the solicitation of acceptances or rejections of the Plan or such Distributions made pursuant to the Plan. Upon entry of the Confirmation Order, all provisions of the Plan addressing Distribution of the New Equity shall be deemed necessary and proper.

## (g) Distribution of Offered New Equity

On the DT Initial Distribution Date, the Offered New Equity shall be distributed to all New Equity Eligible Holders that make the New Equity Election.

On the Effective Date, all New Equity, other than the Offered New Equity that is distributed to New Equity Eligible Holders that make the New Equity Election, shall be distributed to the Hildene Master Funds or their designee in such proportion as they direct.

Any shares of Offered New Equity that, as of the DT Initial Distribution Date, have not already been distributed to (or earmarked for distribution to) New Equity Eligible Holders or the Holder of the Allowed SFL Note Claim, shall be distributed to the Hildene Master Funds or their designee in such proportion as they direct.

## 6.2. Vesting of Estate Property

On the Effective Date, all property of the Debtors and their Estates shall vest automatically in the Reorganized Debtors or the Distribution Trust as described in this section of the Plan.

On the Effective Date, except as otherwise expressly provided in the Confirmation Order, the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves shall automatically vest in the Distribution Trust free and clear of all Claims, Liens and Interests (other than the Purchaser and Reorganized Debtors' respective reversionary interests in the Distribution Trust Reserves).

Except for the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves, or as otherwise expressly provided in the Confirmation Order, pursuant to sections 1123(b)(3) and 1141(b)–(c) of the Bankruptcy Code, on the Effective Date, all of the property and assets of each Debtor shall automatically vest in the respective Reorganized Debtor, free and clear of all Claims, Liens and Interests. For the avoidance of doubt, such property and assets of each Debtor that will automatically vest in the Reorganized Debtors includes, but is not limited to, any SALIC Group Services Claim incurred on or after the Petition Date. The Reorganized Debtors may operate their business and may use, acquire, and dispose of such property free of any restrictions of the Bankruptcy Code, the Bankruptcy Rules, and the Bankruptcy Court. As of the Effective Date, all such property of the Reorganized Debtors shall be free and clear of all Claims, Liens and Interests, except as specifically provided in the Confirmation Order, and the Reorganized Debtors shall receive the benefit of any and all discharges and injunctions under the Plan.

## 6.3. The Distribution Trust

### (a) Execution of Distribution Trust Agreement

On or prior to the Effective Date, the Debtors shall execute the Distribution Trust Agreement, and shall take all other necessary steps to establish the Distribution Trust, which shall be for the payment of Allowed Administrative Claims, Allowed Priority Claims, and Allowed Secured Claims not satisfied by Closing Date Plan Distributions, and for the benefit of the Distribution Trust Beneficiaries. In the event of any conflict between the terms of this Section 6.3(a) and the terms of the Distribution Trust Agreement as such conflict relates to the establishment of the Distribution Trust, the terms of this Section 6.3(a) shall govern. The Distribution Trust Agreement may provide powers, duties and authorities in addition to those explicitly stated herein, but only to the extent that such powers, duties, and authorities do not affect the status of the Distribution Trust Agreement shall be Filed with the Plan Supplement.

## (b) Purpose of the Distribution Trust

The Distribution Trust shall be established for the sole purpose of liquidating and distributing the assets of the Debtors contributed to such Distribution Trust in accordance with Treas. Reg. § 301.7701-4(d), with no objective to continue or engage in the conduct of a trade or business.

## (c) Distribution Trust Assets and Other Property

The Distribution Trust shall consist of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves. On the Effective Date, all of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves shall transfer to and be vested in the Distribution Trust. All attorney-client privilege associated with the Retained Causes of Action remains with and vests in the Reorganized Debtors.

# (d) The Administration of the Distribution Trust and Authority and Powers of the Distribution Trustee

The Distribution Trust shall be administered by the Distribution Trustee pursuant to the Distribution Trust Agreement. The initial Distribution Trustee shall be a Person selected by the Official Committee and reasonably acceptable to the Debtors and the Purchaser. The identity of the initial Distribution Trustee will be disclosed in the Plan Supplement, and any successor Distribution Trustee shall be appointed in the manner set forth in the Distribution Trust Agreement. In the event of any inconsistency between the Plan and the Distribution Trust Agreement as such conflict relates to anything other than the establishment of the Distribution Trust, the Plan shall control. All compensation for the Distribution Trust e and other costs of administration for the Distribution Trust shall be paid from the Trust Administration Reserve or proceeds of the Distribution Trust Assets in accordance with this Plan and the Distribution Trust Agreement. The Distribution Trust e shall be a representative of each Debtor's Estate in accordance with section 1123(b)(3)(B) of the Bankruptcy Code for the purposes of the DT Post-Closing Rights.

#### (e) Mutual Cooperation

As the Reorganized Debtors or the Distribution Trustee may reasonably request, each shall use commercially reasonable efforts to cooperate with the other with respect to the implementation of the Plan (including, without limitation, the resolution of Disputed Claims, the determination of taxes and the preparation and filing of tax returns), with all reasonable out-of-pocket expenses incurred by the (i) Reorganized Debtors in connection with a request by the Distribution Trustee made pursuant to this paragraph being borne by the Distribution Trustee and (ii) Distribution Trustee in connection with a request by the Reorganized Debtors made pursuant to this paragraph being borne by the Reorganized Debtors; *provided, however*, that neither party shall be required to (i) provide information, records or employees or other personnel under circumstances which the providing party believes in its sole reasonable determination may waive privilege, confidentiality or a similar protection or expose it to material liability to any person or may prejudice any legal interest of the providing party, or (ii) take any action that in the providing party's reasonable determination unreasonably interferes with its business. For the avoidance of doubt, nothing herein is intended to limit the DT Post-Closing Rights of the Distribution Trust and the Distribution Trustee.

## (f) Establishment and Funding of Distribution Trust Reserves

On the Effective Date, the following Distribution Trust Reserves shall be established and funded from the Plan Funding Payment, each in accordance with the Allocation/Use Priorities:

i. Disputed Claims Reserve. The Disputed Claims Reserve shall be established and funded with Cash (including an amount for the SFL Note Claim if the SFL Note Claim Allowance Conditions are not met as of the Effective Date) in an amount sufficient to cover pro rata distributions to each Disputed Claim that, as of the Effective Date, is neither an Allowed Claim nor a Disallowed Claim, and includes, without limitation, a Claim that is the subject of a timely objection or request for estimation with the Bankruptcy Court, which has not been withdrawn, settled or overruled by a Final Order; *provided, however*, that if the Disputed Claim is an Administrative Claim (other than a Professional Fee Claim), Priority Claim or Secured Claim, an amount sufficient to cover payment in full of the Face Amount of such Disputed Claim shall be funded to the Disputed Claims Reserve; provided further, that if any dispute arises regarding any increase or reduction of the Disputed Claims Reserve, the Distribution Trustee shall consult with the Purchaser and shall obtain approval of the Bankruptcy Court, which shall have jurisdiction and power to set the amount of the reserve applying the principals of section 502(c) of the Bankruptcy Code to estimate any Claim.

ii. *Professional Fee Reserve*. The Professional Fee Reserve shall be established and funded in an amount that the Debtors estimate in good faith, after consultation

with the relevant Professionals, the Purchaser, and the Official Committee, to be necessary to pay in full all amounts then owing or that may later become owing to such Professionals for professional fees and expenses incurred through the Effective Date. For the avoidance of doubt, the estimated amount initially funded to the Professional Fee Reserve is not intended as and shall not be deemed to be a cap on the funds available to pay or satisfy Allowed Administrative Claims of Professionals for compensation or reimbursement of expenses. Following the Effective Date, the Distribution Trustee shall have the discretion to increase the Professional Fee Reserve as the Distribution Trustee deems necessary or appropriate to pay or satisfy Allowed Administrative Claims of Professionals for compensation or reimbursement of expenses.

Trust Administration Reserve. The Trust Administration Reserve iii. shall be established and funded in an amount, mutually agreed by the Debtors, the Official Committee, and the Purchaser, estimated in good faith to be necessary to cover the costs of administration of the Distribution Trust and subject in all respects to the Trust Administration Reserve Cap. The Trust Administration Reserve shall be used to administer the Distribution Trust, including to (a) fund the reasonable fees and expenses of the Distribution Trustee and any employees, attorneys, accountants, financial advisors, consultants, other professional persons or independent contractors that the Distribution Trustee may engage to assist him, her or it in the discharge of the Distribution Trustee's duties under the Plan and the Distribution Trust Agreement, including, without limitation, fees and expenses related to prosecution and resolution of Causes of Action and objections to Claims; (b) fund premium payments for an errors and omissions insurance policy for the benefit of the Distribution Trust, the Distribution Trustee and the Distribution Trustee's agents and representatives, (c) meet contingent liabilities and to maintain the value of the Distribution Trust Assets during liquidation, (d) pay other reasonably incurred or anticipated expenses (including, without limitation, any taxes imposed on or payable by the Distribution Trust or in respect of the Distribution Trust Assets, including with respect to such assets as are allocable to Disputed Claims), and (e) satisfy other liabilities incurred or anticipated by such Distribution Trust in accordance with the Plan or Distribution Trust Agreement.

#### (g) Intentionally Omitted.

#### (h) Cash Investments

The Distribution Trustee may invest Cash (including any earnings thereon or proceeds therefrom); *provided, however*, that such investments are investments permitted to be made by a "liquidating trust" within the meaning of Treas. Reg. § 301.7701-4(d), as reflected therein, or under applicable IRS guidelines, rulings or other controlling authorities.

#### (i) Distributions to Holders of Claims and Distribution Trust Beneficiaries

The Distribution Trustee shall be responsible for making all Distributions to Holders of Allowed Claims required to be made on or after the Effective Date pursuant to the Plan; *provided*, that the Reorganized Debtors or the Disbursing Agent (as applicable) shall make
the Distributions to Holders of Allowed Claims on the Effective Date on behalf of the Distribution Trustee. The Distribution Trustee will make all Distributions to Holders of Allowed Claims, or in the case of Holders of Allowed Claims in Classes 4 and 6, the Beneficial Holders thereof, as required by this Plan at: (i) the address of any such Holder on the books and records of the Debtors, the Distribution Trustee or their agents; or (ii) at the address in any written notice of address change delivered to the Debtors, the Distribution Trustee or the Distribution Trust Agreement. The procedures for Distributions to be made to the holder of the SFL Note Claim shall be set forth in the SFL Distribution Agreement.

#### i. DT Initial Distribution

As soon as reasonably practicable after (i) funding of all Distribution Trust Reserves and (ii) payment in full (or reserving for payment in full) of all Allowed Administrative Claims, Allowed Priority Claims, and Allowed Secured Claims, the Distribution Trustee shall distribute to the Holders of Allowed Claims in Classes 4, 5, 6 and 7, or in the case of Holders of Allowed Claims in Classes 4 and 6, to the Beneficial Holders thereof, on account of their Allowed Claims their pro rata share of the Available Plan Distribution Funding Amount and Distribution Trust Asset Proceeds, as applicable to Allowed Claims.

#### ii. DT Subsequent Distribution

After completion of the DT Initial Distribution, the Distribution Trustee shall make the DT Subsequent Distribution(s) in a reasonably timely manner after Distribution Trust Assets Proceeds become available, and in accordance with such procedures as may be provided in the Distribution Trust Agreement. Such DT Subsequent Distributions shall be made no less frequently than every twelve (12) months; *provided, however*, that the Distribution Trustee shall not be required to make a Distribution pursuant to this <u>Section 6.3(h)(ii)</u> of the Plan if the Distribution Trustee determines that the expense associated with making the Distribution would likely utilize a substantial portion of the amount to be distributed, thus making the Distribution impracticable.

#### (j) Federal Income Tax Treatment of Distribution Trust

Subject to definitive guidance from the IRS or a court of competent jurisdiction to the contrary (including the receipt of an adverse determination by the IRS upon audit if not contested by such Distribution Trustee), for all U.S. federal income tax purposes, all parties (including, without limitation, the Debtors, the Distribution Trustee and Distribution Trust Beneficiaries) shall treat the transfer of Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves to the Distribution Trust as (1) a transfer of property (subject to any and all Allowed Administrative Claims, Allowed Priority Claims, and Allowed Secured Claims to the extent not satisfied by the Debtors on or prior to the Effective Date, that are payable by the Distribution Trust of Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves. Accordingly, except in the event of contrary definitive guidance, Distribution Trust Beneficiaries shall be treated for U.S. federal income tax purposes as the grantors and owners of their respective share of Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves (other than that which is allocable to Disputed Claims). The foregoing treatment shall also apply, to the extent permitted by Applicable Law, for state and local income tax purposes. For the avoidance of doubt, the term "party" as herein used shall not include the United States or any agency or department thereof, or any officer or employee thereof acting in such capacity. The Distribution Trust and the qualification of the Distribution Trust as a "liquidating trust" for U.S. federal income tax purposes.

## (k) Tax Reporting

i. The Distribution Trustee shall file tax returns for the Distribution Trust treating such Distribution Trust as a grantor trust pursuant to Treas. Reg. § 1.671-4(a) and in accordance with this <u>Section 6.3</u>. The Distribution Trustee also shall annually send or otherwise provide to each Holder of the Distribution Trust Interest a separate statement regarding the receipts and expenditures of the Distribution Trust as relevant for U.S. federal income tax purposes.

ii. Allocations of Distribution Trust taxable income among Distribution Trust Beneficiaries (other than taxable income allocable to any assets allocable to, or retained on account of, Disputed Claims, if such income is otherwise taxable at the Distribution Trust) shall be determined by reference to the manner in which an amount of Cash representing such taxable income would be distributed (were such Cash permitted to be distributed at such time) if, immediately prior to such deemed Distribution, the Distribution Trust had distributed all its assets (valued at their tax book value, other than, if applicable, assets allocable to Disputed Claims) to the Holders of Distribution Trust Interests, adjusted for prior taxable income and loss and taking into account all prior and concurrent Distributions from the Distribution Trust. Similarly, taxable loss of the Distribution Trust shall be allocated by reference to the manner in which an economic loss would be borne immediately after a hypothetical liquidating distribution of the remaining Distribution Trust Assets, Available Plan Distribution Funding Amount and Distribution Trust Reserves. The tax book value of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves for purposes of this paragraph shall equal their fair market value on the date such assets are transferred to the Distribution Trust, adjusted in accordance with tax accounting principles prescribed by the IRC, the applicable Treasury Regulations, and other applicable administrative and judicial authorities and pronouncements.

iii. As soon as reasonably practicable after the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves are transferred to the Distribution Trust, the Distribution Trustee shall make a good faith valuation of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves. Such valuation shall be made available from time to time to all

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 37 of 241

parties to the Distribution Trust (including, without limitation, the Debtors (or, as the case may be, the Reorganized Debtors), and the Distribution Trust Beneficiaries), to the extent relevant to such parties for tax purposes, and shall be used consistently by such parties for all U.S. federal income tax purposes.

iv. Subject to definitive guidance from the IRS or a court of competent jurisdiction to the contrary (including the receipt by the Distribution Trustee of a private letter ruling if the Distribution Trustee so requests one, or the receipt of an adverse determination by the IRS upon audit if not contested by such Distribution Trustee), the Distribution Trustee (i) shall treat any Distribution Trust Reserves allocable to Disputed Claims as a "disputed ownership fund" governed by Treas. Reg. § 1.468B-9 (and make any necessary elections with respect thereto) and (ii) to the extent permitted by Applicable Law, shall report consistently for state and local income tax purposes. All parties (including the Distribution Trustee, the Debtors and Distribution Trust Beneficiaries) shall report for U.S. federal, state and local income tax purposes consistently with the foregoing.

v. The Distribution Trustee shall be responsible for payment, out of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves, of any taxes imposed on the Distribution Trust or its assets (including with respect to assets allocable to Disputed Claims).

vi. The Distribution Trustee may request an expedited determination of taxes of the Distribution Trust, including any reserve for Disputed Claims, or of the Debtors as to whom the Distribution Trust was established, under section 505(b) of the Bankruptcy Code for all tax returns filed for, or on behalf of, such Distribution Trust or the Debtors for all taxable periods through the dissolution of such Distribution Trust.

#### (l) Dissolution

i. The Distribution Trustee and Distribution Trust shall be discharged or dissolved, as the case may be, at such time as (i) all of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves have been expended or distributed pursuant to the Plan and the Distribution Trust Agreement, (ii) the Distribution Trustee determines, in its sole discretion, that the administration of any remaining Distribution Trust Assets, Available Plan Distribution Funding Amount or Distribution Trust Reserves is not likely to yield sufficient additional Distribution Trust proceeds to justify further pursuit, or (iii) all Distributions required to be made by the Distribution Trustee under the Plan and the Distribution Trust Agreement have been made; provided, however, that in no event shall the Distribution Trust be dissolved later than three (3) years from the creation of such Distribution Trust pursuant to this Section 6.3 of the Plan, unless the Bankruptcy Court, upon motion within the six-month period prior to the third (3rd) anniversary (or within the six-month period prior to the end of an extension period), determines that a fixed period extension (not to exceed three (3) years, together with any prior extensions, without a favorable private letter ruling from the IRS or an opinion of counsel in form and substance satisfactory to the Distribution Trustee that any further extension would not adversely affect the status of the trust as the Distribution Trust for U.S. federal income tax purposes) is necessary to facilitate or complete the recovery and liquidation of the Distribution Trust Assets.

ii. If at any time the Distribution Trustee determines, in reliance upon such professionals as the Distribution Trustee may retain, that the expense of administering the Distribution Trust so as to make a final Distribution to Distribution Trust Beneficiaries is likely to exceed the value of the assets remaining in such Distribution Trust, such Distribution Trustee may apply to the Bankruptcy Court for authority to (i) reserve any amount necessary to dissolve such Distribution Trust, (ii) donate any balance to a charitable organization (A) described in section 501(c)(3) of the IRC, (B) exempt from U.S. federal income tax under section 501(a) of the IRC, (C) not a "private foundation", as defined in section 509(a) of the IRC, and (D) that is unrelated to the Debtors, such Distribution Trust, and any insider of such Distribution Trustee, and (iii) dissolve such Distribution Trust.

## 6.4. The Reorganized Debtors

## (a) Continued Corporate Existence

Except as otherwise provided in the Plan, each of the Debtors, as Reorganized Debtors, shall continue to exist after the Effective Date as a corporate entity, with all of the powers of a corporation or limited company, as the case may be, under Applicable Law in the jurisdiction in which such Debtor is incorporated or organized and pursuant to the New Corporate Governance Documents. After the Effective Date, the Reorganized Debtors may operate their business and use, acquire, and dispose of property without the supervision of the Bankruptcy Court, free of any restrictions of the Bankruptcy Code or Bankruptcy Rules.

The New Corporate Governance Documents shall be consistent with section 1123(a)(6) of the Bankruptcy Code, to the extent applicable, and in form and substance acceptable to the Purchaser.

### (b) Directors and Officers of the Reorganized Debtors

The officers and the members of each board of directors of each of the Reorganized Debtors shall be selected and appointed in the sole discretion of the Purchaser. To the extent required by section 1129(a)(5) of the Bankruptcy Code, the identity of such officers and members shall be disclosed prior to the Confirmation Hearing.

Except to the extent that a member of the board of directors of a Debtor continues to serve as a director of such Debtor following the Effective Date, the members of the board of directors of each Debtor prior to the Effective Date, in their capacities as such, shall have no continuing obligations to the Reorganized Debtors on or after the Effective Date and each such member will be deemed to have resigned or shall otherwise cease to be a director of the applicable Debtor on the Effective Date without any further action required on the part of any such Debtor or member. Commencing on the Effective Date, each of the directors of each of the Reorganized Debtors shall serve pursuant to the terms of the applicable organizational documents of such Reorganized Debtor and may be replaced or removed in accordance with such organizational documents.

Except as otherwise provided in the Stock Purchase Agreement with respect to the Employment Agreements (as defined therein), subject to the discretion of the Reorganized Debtors' boards of directors, the Reorganized Debtors shall enter into new employment agreements with key executives on a case-by-case basis.

## (c) Corporate Action

On the Effective Date, the adoption and filing of the New Corporate Governance Documents, the appointment of officers of the Reorganized Debtors, and all actions contemplated by the Plan will be authorized and approved in all respects pursuant to the Plan. On the Effective Date, pursuant to section 1142(b) of the Bankruptcy Code and section 303 of the Delaware General Corporation Law (to the extent applicable) and any comparable provision of other Applicable Law, the appropriate officers or directors of each Reorganized Debtor shall be authorized and directed to issue, execute, and deliver the agreements, documents, securities, and instruments contemplated by the Plan with like effect as if exercised and taken by unanimous action of the directors and stockholders of each Debtor.

## (d) Effectuating Documents; Further Transactions

On the Effective Date or as soon as reasonably practicable thereafter, the Reorganized Debtors and/or the Purchaser may take all actions as may be necessary or appropriate to effect any transaction described in, approved by, or necessary or appropriate to effectuate the Plan, including: (1) the execution and delivery of appropriate agreements or other documents of merger, consolidation, restructuring, conversion, disposition, transfer, dissolution or liquidation to the extent consistent with the terms of the Plan and the Plan Documents; (2) the execution and delivery of appropriate instruments of transfer, assignment, assumption or delegation of any asset, property, right, liability, debt or obligation on terms consistent with the terms of the Plan and Plan Documents or having other terms to which the Debtors, the Reorganized Debtors, the Purchaser, and other applicable parties may agree; (3) the filing of appropriate certificates or articles of incorporation, reincorporation, merger, consolidation, conversion or dissolution pursuant to applicable state law; and (4) all other actions that the Purchaser and any other applicable Entities determine to be necessary or appropriate, including making filings or recordings that may be required by Applicable Law.

## 6.5. Retained Causes of Action

Except to the extent any Claim against an Entity is expressly waived, relinquished, exculpated, released, compromised or settled in the Plan or by a Final Order or is a Distribution Trust Asset, all Causes of Action of the Debtors (the "<u>Retained Causes of Action</u>") shall, in accordance with section 1123(b) of the Bankruptcy Code, vest in and be retained by the Reorganized Debtors, including any Causes of Action against the Specified SALIC Officers and Directors. The applicable Reorganized Debtors (with respect to the Retained Causes of Action and any Causes of Action arising after the Petition Date), in accordance with section 1123(b) of

the Bankruptcy Code, shall retain and may enforce, sue on, settle, or compromise (or decline to do any of the foregoing) all Claims, rights, Causes of Action, suits, and proceedings, whether in law or in equity, whether known or unknown, that they each may respectively hold against any Person without the approval of the Bankruptcy Court and the Reorganized Debtors' rights to commence, prosecute, or settle such Causes of Action shall be preserved notwithstanding the occurrence of the Effective Date. For the avoidance of doubt, all Causes of Action against the SRGL Equity Holders shall not be Retained Causes of Action, and shall be transferred to and vest in the Distribution Trust. On the Effective Date, all Causes of Action that are Distribution Trust Assets shall, in accordance with Section 1123(b) of the Bankruptcy Code, vest in the Distribution Trust, and the Distribution Trust may enforce, sue on, settle, or compromise (or decline to do any of the foregoing) all such Claims, rights, Causes of Action, suits, and proceedings, whether in law or in equity, known or unknown, without approval of the Bankruptcy Court, and the Distribution Trust's rights to commence, prosecute, or settle such Causes of Action shall be preserved notwithstanding the Bankruptcy Court, and the Distribution Trust's rights to commence, prosecute, or settle such Causes of Action shall be preserved notwithstanding the occurrence of the Effective Date.

## 6.6. The Closing

The Closing as defined in the Stock Purchase Agreement shall be subject to the conditions in the Stock Purchase Agreement, including, without limitation, the conditions to closing set out in Article VII of the Stock Purchase Agreement and the actions and deliveries set out in section 2.4 of the Stock Purchase Agreement, unless waived in accordance with the Stock Purchase Agreement. The Closing shall occur simultaneously with the Effective Date of the Plan.

## 6.7. Cancellation of Agreements, Securities and Other Documents Relating to TruPS Transactions & SFL Note

Except for purposes of evidencing a right to a Distribution under the Plan or otherwise as provided in the Plan, the Confirmation Order or the Distribution Trust Agreement, on the Effective Date, the TruPS Indentures, the TruPS Debentures, the TruPS Declarations, the TruPS Sponsor Guarantees, the TruPS Parent Guarantees, all other TruPS Documents, the SFL Note and all corresponding documents issued in connection with such documents shall be deemed automatically cancelled, terminated and of no further force or effect, without further act or action under any applicable agreement, law, regulation, order, or rule and the obligations of the Debtors, the TruPS Indenture Trustees, and the TruPS Institutional Trustees, as applicable, thereunder shall be deemed terminated; provided, however, that TruPS Indentures and TruPS Declarations shall continue in effect only as follows: (1) for the TruPS Indenture Trustees and the TruPS Institutional Trustees to discharge any responsibilities they have under the Plan, the Confirmation Order, the Distribution Trust Agreement, the TruPS Indentures, the TruPS Parent Guarantees, the TruPS Sponsor Guarantees and the TruPS Declarations in connection with Distributions to be made to the Holders of the TruPS Debentures, the Beneficial Holders of TruPS and SRGL in accordance with the terms of the Plan (including Sections 4.3(a), and 4.3(c) thereof), the Confirmation Order, the Distribution Trust Agreement, the TruPS Indentures, the TruPS Declarations and, as to SRGL only, the Restructuring Implementation Agreement and RIA Order, it being understood that the TruPS Indentures, the TruPS Parent Guarantees, the

TruPS Sponsor Guarantees and the TruPS Declarations shall continue in effect only so long as is necessary to permit such Distributions to be made; (2) subject to <u>Section 4.1(d)(iii)</u> hereof, to allow each TruPS Indenture Trustee and any predecessor trustee under any of the TruPS Indentures to exercise its Charging Lien for the payment of its fees and expenses incurred post-Closing (if any) and for indemnification as provided in the applicable TruPS Indentures solely to the extent requested by the Debtors, the Reorganized Debtors or the Distribution Trustee; (3) to preserve any rights of the TruPS Indenture Trustees pursuant to any direction provided by Holders of the TruPS pursuant to the terms of the TruPS documents; and (4) solely with respect to the SFLST I TruPS Indenture, the SFLST I TruPS Debentures, the SFLST I TruPS Documents (except the SFLST I TruPS Parent Guarantee), the foregoing SFLST I TruPS Documents shall not be deemed cancelled, terminated or of no force or effect as against SFL. For the avoidance of doubt, nothing in this <u>Section 6.7</u> is intended to or shall extinguish or impair any liability or obligation of SFL under any SFLST I TruPS Document.

The TruPS Indenture Trustees, in their capacity as such, shall have no liability in connection with Distributions made in accordance with the terms of the Plan and shall have no obligation to take any action in connection with such Distributions or otherwise; provided, however, that for so long as the SFLST I TruPS Indenture, the SFLST I TruPS Debentures, the SFLST I Trust Declaration, the SFLST I TruPS Sponsor Guarantee, and any other SFLST I TruPS Documents (except the SFLST I TruPS Parent Guarantee) remain in effect, the SFLST I TruPS Indenture Trustee shall only have such liability and obligations as are expressly set forth therein. To the extent that, after the Effective Date, (x) the Debtors, the Reorganized Debtors or the Distribution Trustee requests services from the TruPS Indenture Trustees in connection with Distributions or otherwise, or (y) material action by the TruPS Indenture Trustee is reasonably necessitated in response to actions taken by a Holder of an Allowed TruPS Claim, a TruPS Indenture Trustee's reasonable and documented costs and out-of-pocket expenses, including professional fees, for such post-Effective Date services of the TruPS Indenture Trustees shall be paid by the Distribution Trust from the Post-Effective Date Indenture Trustee Fee Reserve of \$50,000 and such invoices for such Indenture Trustee Fees shall be subject to the review procedures set forth in section 4.1(d). Such payment shall be in lieu of the TruPS Indenture Trustee's ability to effectuate its Charging Liens under the TruPS Indentures. The \$50,000 in aggregate to be held in the Post-Effective Date Indenture Trustee Fee Reserve for payment of post-Effective Date Indenture Trustee Fees shall be held by the Distribution Trustee for a period of one (1) year after the Effective Date and any remainder of such funds will be released thereafter from the Post-Effective Date Indenture Trustee Fee Reserve ratably to the Holders of Allowed TruPS Claims. For the avoidance of doubt, the Distribution Trustee shall be under no obligation to satisfy or reserve any amount on account of post-Effective Date Indenture Trustee Fees that may be incurred by the TruPS Indenture Trustees for the SHST II Debentures or the **GPIC** Debentures.

#### 6.8. Comprehensive Settlement of Claims and Controversies

Pursuant to sections 363 and 1123 of the Bankruptcy Code and Bankruptcy Rule 9019, and in consideration for the distributions and other benefits provided pursuant to the Plan,

the provisions of the Plan shall constitute a good faith compromise of all Claims, Interests and controversies relating to the contractual, legal, and subordination rights that a Holder of a Claim may have with respect to any Allowed Claim, or any distribution to be made on account of such Allowed Claim, as well as the allocation of the Plan Funding Payment among the Debtors' Estates and Creditors. The entry of the Confirmation Order shall constitute the Bankruptcy Court's approval of the compromise or settlement of all such Claims, Interests, and controversies, as well as a finding by the Court that such compromise or settlement is in the best interests of the Debtors, their Estates, and Holders of Claims and Interests and is fair, equitable, and reasonable. In accordance with the provisions of the Plan, pursuant to Bankruptcy Rule 9019, without any further notice to or action, order, or approval of the Court, after the Effective Date, the Reorganized Debtors or the Distribution Trustee, as applicable, may compromise and settle Claims against, and Interests in, the Debtors and their Estates and Causes of Action against other entities.

## VII. CLAIMS RESOLUTION & DISTRIBUTIONS

### 7.1. Right to Object to Claims

The Distribution Trustee shall have the authority, but not the obligation, to object to, litigate, and settle, the amount, priority or the extent of any Administrative Claim, Secured Claim, Priority Claim, SHI General Unsecured Claim, SALIC General Unsecured Claim, any SFL Claims (if not settled on or before the Effective Date) or Subordinated Claim (including, with respect to any other of the foregoing, to argue that such Claim constitutes a Subordinated Claim). Notwithstanding anything to the contrary herein, subject to the terms and conditions set forth in the Distribution Trust Agreement, and notwithstanding any requirements that may be imposed pursuant to Bankruptcy Rule 9019, except insofar as a Claim is Allowed under the Plan on and after the Effective Date, the Distribution Trustee shall have the authority, but not the obligation, to: (1) file, withdraw or litigate to judgment objections to and requests for estimation of Claims; (2) settle or compromise any Disputed Claim without any further notice to or action, order or approval by the Bankruptcy Court; and (3) administer and adjust the Claims register to reflect any such settlements or compromises without any further notice to or action, order or approval by the Bankruptcy Court. The Distribution Trustee shall succeed to any pending objections to Claims filed by the Debtors prior to the Effective Date, and, at the Distribution Trustee's election, any other pending objections to Claims filed by any other party, and shall have and retain any and all rights and defenses the Debtors had immediately prior to the Effective Date with respect to any Disputed Claim, including pursuant to the DT Post-Closing The Reorganized Debtors shall provide commercially reasonable assistance and Rights. cooperation to the Distribution Trustee in connection with the Distribution Trustee's prosecution of objections to Claims, including, without limitation, access to the books and records of the Debtors or the Reorganized Debtors (as the case may be) and other information reasonably requested by the Distribution Trustee to enable the Distribution Trustee to perform its obligations under the Distribution Trust Agreement, including pursuant to the DT Post-Closing Rights.

## 7.2. Deadline for Objecting to Claims

Objections to Claims must be Filed with the Bankruptcy Court, and a copy of the objection must be served on the subject Creditor, before the expiration of the Claims Objection Deadline (unless such period is further extended by subsequent orders of the Bankruptcy Court); otherwise such Claims shall be deemed Allowed in accordance with section 502 of the Bankruptcy Code. The objection shall notify the Creditor of the deadline for responding to such objection.

## 7.3. Deadline for Responding to Claim Objections

Within twenty-one (21) days after service of an objection, or such other date as is indicated on such objection or the accompanying notice thereof, the Creditor whose Claim was objected to must File a written response to the objection with the Bankruptcy Court and serve a copy on the Distribution Trustee. Failure to File a written response within such time period shall constitute a waiver and release of that portion of the subject Claim that was subject to the objection, and shall constitute cause for the Bankruptcy Court to enter a default judgment against the non-responding Creditor or grant the relief requested in the Claim objection.

## 7.4. Right to Request Estimation of Claims

Pursuant to section 502(c) of the Bankruptcy Code, the Debtors, the Reorganized Debtors, and the Distribution Trustee may request estimation or liquidation of any Disputed Claim that is contingent or unliquidated or any Disputed Claim arising from a right to an equitable remedy or breach of performance.

## 7.5. Distribution Procedures Regarding Allowed Claims

## (a) In General

Other than Closing Date Plan Distributions and initial Distributions of New Equity to New Equity Eligible Holders that make the New Equity Election, the Distribution Trustee shall make all Distributions required to be made under the Plan, including Distributions from the Distribution Trust. Each Creditor or Holder receiving any Distribution from the Distribution Trust shall be deemed to have ratified and become bound by the terms and conditions of the Distribution Trust Agreement.

## (b) Distributions on Allowed Claims Only

Distributions from the Distribution Trust shall be made only to the Holders of Allowed Claims or in the case of Holders of Allowed Claims in Classes 4 and 6, the Beneficial Holders thereof. Until a Disputed Claim becomes an Allowed Claim, the Holder of that Disputed Claim shall not receive a Distribution. Other than as specifically set forth in section 6.1(b)(i), Allowed Claims shall not be entitled to distributions from any source other than the Plan Funding Payment or the Distribution Trust.

#### (c) Method of Distributions

Use of Distribution Agent. The Reorganized Debtors with respect i. to Closing Date Plan Distributions and initial Distributions of Offered New Equity and the Distribution Trustee with respect to Distributions from the Distribution Trust shall have the authority, in their respective sole discretion, to enter into agreements with a third-party Distribution Agent to facilitate the Distributions required hereunder. For the avoidance of doubt, Prime Clerk, LLC, which was previously engaged to serve as the Debtors' Voting Agent, is an acceptable choice to serve as the Distribution Agent and shall be engaged as the shared Distribution Agent in the event that the Reorganized Debtors and Distribution Trustee both want to use the services of a Distribution Agent and cannot agree to an alternate choice. The Distribution Trustee shall be authorized, but not directed, to pay to any third-party Distribution Agent all reasonable and documented fees and expenses of such Distribution Agent without the need for any approvals, authorizations, actions, or consents. The Distribution Agent shall be authorized, but not directed, to submit detailed invoices to the Debtors or the Reorganized Debtors, as applicable, for all fees and expenses for which the Distribution Agent seeks reimbursement and the Distribution Trustee shall pay those amounts from the Trust Administration Reserve that it, in its sole discretion, deems reasonable, and shall object in writing to those fees and expenses, if any, that the Distribution Trustee deems to be unreasonable. In the event that the Distribution Trustee objects to all or any portion of the amounts requested to be reimbursed in a Distribution Agent's invoice, the Distribution Trustee and such Distribution Agent shall endeavor, in good faith, to reach mutual agreement on the amount of the appropriate payment of such disputed fees or expenses. In the event that the Distribution Trustee and a Distribution Agent are unable to resolve any differences regarding disputed fees or expenses, either party shall be authorized to move to have such dispute heard by the Bankruptcy Court.

Cash Distributions. Except as otherwise specified in the Plan or ii. the Distribution Trust Agreement (with respect to Distributions other than Closing Date Plan Distributions), any Distribution of Cash made by the Reorganized Debtors as a Closing Date Plan Distribution or made by the Distribution Trustee from the Available Plan Funding Distribution Amount shall be made by mailing such Distribution to the Creditor or Holder, as applicable, at the address listed in any Proof of Claim Filed by such Entity or at such other address as such Entity shall have specified for payment purposes in a written notice received by the Reorganized Debtors or Distribution Trustee, as applicable, at least twenty-one (21) days before a Distribution Date. If a Creditor or Holder has not Filed a Proof of Claim or sent the Distribution Trustee a written notice of payment address, then the Distribution(s) for such Entity will be mailed to the address identified in the Schedules. Notwithstanding the foregoing, any Cash may be distributed by wire, check, or such other method as the Distribution Trustee or Reorganized Debtors, as applicable, deem appropriate under the circumstances. Any Cash Distribution to be made to SRGL under the Plan shall be made by wire transfer unless otherwise agreed by SRGL.

iii. Offered New Equity Distributions. On or as soon as practicable after the Effective Date, the Reorganized Debtors (or the New HoldCo (as applicable)) shall

distribute to the New Equity Eligible Holders that have made the New Equity Election a Distribution of each such holder's TruPS Claims Equity Distribution Amount. For the avoidance of doubt, such Distribution shall not be certificated, and will rather be in book-entry form.

iv. *Tax Information Required for Distributions*. Before receiving any Distributions, all Creditors and Holders, at the request of the Reorganized Debtors or Distribution Trustee, as applicable, must provide written notification of their respective Federal Tax Identification Numbers or Social Security Numbers to the requesting Entity; otherwise, the Reorganized Debtors or Distribution Trustee, as applicable, may suspend Distributions to any Creditors or Holders who have not provided their Federal Tax Identification Numbers or Social Security Numbers.

v. Notwithstanding the foregoing, the Debtors may cause there to be established a register on which there shall be recorded the Beneficial Holders of the Claims in Class 4 and Class 6, and the New Equity Election made by any of them. The Reorganized Debtors or the Distribution Trustee, as applicable, shall make Distributions of TruPS/GUC Claims Cash Distribution Amount, the TruPS Claims Equity Distribution Amounts and/or the Distribution Trust Asset Proceeds, as the case may be, to such Beneficial Holders, or their transferees, as recorded on such register in accordance with such transfer procedures as the Debtors may establish. The provisions of Section 7.5(c)(iv) shall similarly apply to Distributions to such Beneficial Holders or their transferees

## (d) Undeliverable Distributions

If a Distribution is returned as undeliverable, the Distribution Trustee shall use reasonable efforts to determine such Creditor's or Holder's then current address. If the Distribution Trustee cannot determine, or is not notified of, a Creditor's or Holder's then current address within six (6) months after the Effective Date, the Distribution reserved for such Creditor or Holder shall be deemed an unclaimed Distribution, and <u>Section 7.5(e)</u> of the Plan shall be applicable thereto.

### (e) Unclaimed Distributions

If the current address for a Creditor or Holder entitled to a Distribution (whether in the form of Cash or Offered New Equity) under the Plan has not been determined within six (6) months after the Effective Date or such Entity has otherwise not been located, or if such Entity has not submitted a valid Federal Tax Identification Number or Social Security Number to the Distribution Trustee within six (6) months after the Effective Date, then such Creditor or Holder, as applicable, (i) shall no longer be a Creditor or Holder and (ii) shall be deemed to have released such Claim and Interest, if any. If such unclaimed Distribution consists of Cash, then the Cash shall remain property of the Distribution Trust and be used or distributed in accordance with the terms of this Plan and the Distribution Trust Agreement. If such unclaimed Distribution consists of New Equity, then such New Equity shall not be issued to the forfeiting Claim Holder and shall instead be issued to the Hildene Master Funds.

#### (f) Taxes; Withholding

In connection with the Plan, any party issuing any instrument or making any Distribution described in the Plan shall comply with all applicable withholding and reporting requirements imposed by any federal, state, or local taxing authority, and all Distributions and all related agreements shall be subject to any such withholding or reporting requirements. Notwithstanding the foregoing, each Holder of an Allowed Claim or any other Person that receives a Distribution shall have responsibility for any taxes imposed by any Governmental Unit, including, without limitation, income, withholding, and other taxes, on account of such Distribution. Any party issuing any instrument or making any Distribution has the right, but not the obligation, to not make a Distribution until such Holder has made arrangements satisfactory to such issuing or disbursing party for payment of any such tax obligations. The Distribution Trustee, in the exercise of its sole discretion and judgment, may enter into agreements with taxing or other authorities for the payment of such amounts that may be withheld in accordance with the provisions of this section. Any party entitled to receive any property as an issuance or distribution under the Plan shall, upon request, by the Debtors or Distribution Trustee, as applicable, provide an appropriate Form W-9 or (if the payee is a foreign Person, as applicable) Form W-8. If such request is made and such party fails to comply before the date that is 180 days after the request is made, the amount of such Distribution shall irrevocably revert to the Distribution Trust, and any Claim in respect of such Distribution shall be discharged and forever barred from assertion against the Debtors, the Reorganized Debtors, the Distribution Trust and their respective property.

#### (g) Special Distribution Provisions for TruPS Claims

#### i. Service of TruPS Indenture Trustees; In General.

Except with respect to SRGL Exclusively Held TruPS Claims, Distributions on account of the TruPS Claims shall be made by the Distribution Trustee directly to each beneficial holder of Allowed TruPS Claims in accordance with the Distribution Trust Agreement, rather than to the applicable TruPS Indenture Trustee or Security Depository or Custodian. In the event that a beneficial holder of Allowed TruPS Claims is unknown to the Debtors, the Reorganized Debtors or the Distribution Trustee (as applicable), then any Distribution for such unknown beneficial holder shall be made to the applicable Security Depository or Custodian.

#### ii. Distributions Relating to SRGL Exclusively Held TruPS

Notwithstanding anything to the contrary in this Plan, Distributions on account of the SRGL Exclusively Held TruPS Claims shall be made by the Distribution Trustee directly to SRGL.

#### iii. Distributions Arising From SALIC Claims Against SRGL

Upon any distribution from the SRGL estate to the Distribution Trust in respect of the SALIC Claims, the Joint Liquidators shall apply such provisioning or withholding (if any) as they deem appropriate in accordance with their duties under Cayman Islands law.

All distributions from the SRGL estate to the Distribution Trust in respect of the SALIC Claims shall be distributed pro rata to all Holders of Allowed Claims in Classes 4, 5, 6 and 7, except for SRGL.

Notwithstanding anything contained herein, (i) any funds received by SRGL from the Distribution Trust and any distributions by SRGL on account of such funds shall be governed by, and construed in accordance with, the laws of the Cayman Islands and (ii) any funds received by the Debtors or the Distribution Trust from SRGL and any Distribution from the Distribution Trust or the Debtors on account of such funds shall be subject to the terms of the Plan and the jurisdiction of the Bankruptcy Court.

## (h) Additional Procedures Regarding Distributions from the Distribution Trust

Additional procedures regarding Distributions from the Distribution Trust to Holders of Allowed Claims shall be governed by the Distribution Trust Agreement.

# (i) Allocation of Distributions between Principal and Interest

Except as otherwise provided in the Plan, to the extent that any Claim entitled to a Distribution under the Plan is comprised of indebtedness and accrued but unpaid interest thereon, such Distribution shall be allocated first to the principal amount (as determined for U.S. federal income tax purposes) of the Claim, and then to accrued but unpaid interest.

## 7.6. Implementation of New Equity and Cash Elections

The Plan affords New Equity Eligible Holders the opportunity to make the New Equity Election or the Cash Election and with respect to the Holder of the SFL Note Claim, deems a New Equity Election and/or Cash Election to have been made under certain circumstances.

## (a) Making of New Equity or Cash Election by New Equity Eligible Beneficial Holders

Each New Equity Eligible Beneficial Holder must elect to take its entire Distribution (other than for its applicable percentage of the Distribution Trust Asset Proceeds) either exclusively in the form of Offered New Equity (by making the New Equity Election) or exclusively in the form of Cash (by making the Cash Election). For the avoidance of doubt, a New Equity Eligible Beneficial Holder may not split its Distribution between New Equity and Cash.

A New Equity Election will be recognized as valid only if the electing New Equity Eligible Holder checks the box for the "New Equity Election" on its Ballot and such Ballot is otherwise properly completed and timely received by the Voting Agent in accordance with the requirements of the Disclosure Statement Order. Any New Equity Election that is not

properly made as set forth herein shall be disregarded and such New Equity Eligible Holder shall be deemed to have made the Cash Election. Except as otherwise agreed in writing by (a) if prior to the Effective Date, by the Debtors, the Official Committee and the Purchaser or (b) if on or after the Effective Date, the Distribution Trustee and the Reorganized Debtors, all New Equity Elections and Cash Elections (including any deemed Cash Elections) will be final and irrevocable after the Voting Deadline.

## (b) Implementation of New Equity or Cash Election for SFL Note Claim

Notwithstanding the foregoing, with respect to Distributions to be made under the Plan on account of the Allowed SFL Note Claim:

i. If the SFL Note Claim Allowance Conditions are deemed satisfied in accordance with Section 4.3(c)(ii)(B)(1) of this Plan because the SFLST I TruPS CDO Facility Holders each have voted their respective Allocated Portions of SALIC TruPS Claims to accept the Plan, then the Holder of the Allowed SFL Note Claim shall be deemed to have made the New Equity Election and Cash Election in the manner described in Section 4.3(c)(iii)(C)(1)(a) of this Plan, subject to the SFL Receiver's right to make the SFL Bankruptcy Administration Senior Cash Election.

ii. If one or more of the SFLST I TruPS CDO Facility Holders has not voted its Allocated Portion of SALIC TruPS Claims to accept the Plan, but the Holder of the SFL Note Claim has otherwise satisfied the SFL Note Claim Allowance Conditions in accordance with Section 4.3(c)(ii)(B)(2)(a) of this Plan, then the Holder of the Allowed SFL Note Claim shall be entitled to: (a) apply its Allowed SFL Note Claim amount entirely to the New Equity Election; (b) apply its Allowed SFL Note Claim amount entirely to the Cash Election; or (c) allocate its Allowed SFL Note Claim Amount between the New Equity Election and the Cash Election.

iii. If the SFL Note Claim Allowance Conditions are not satisfied or deemed satisfied in accordance with this Plan, then the Holder of the SFL Note Claim shall be deemed to have made the Cash Election with respect to the entire SFL Note Claim, subject to the SFL Note Claim becoming an Allowed Class 6 Claim.

## 7.7. Claims Paid or Payable by Third Parties

## (a) Claims Paid by Third Parties

To the extent a Holder has received a Distribution on account of a Claim and also receives payment from a party that is not a Debtor or the Distribution Trustee on account of such Claim, such Holder shall, within 30 calendar days of receipt thereof, repay and/or return the Distribution to the Distribution Trustee to the extent the recipient-Holder's total recovery on account of such Claim from the third party and under this Plan exceeds the amount of the Holder's Allowed Claim as of the date of any such Distribution under this Plan.

Any such Claim shall be expunged from the official claims register without a claims objection having to be filed and without any further notice to or action, order, or approval of the Bankruptcy Court, to the extent that the Holder receives payment in full on account of such Claim; *provided, however*, that to the extent the non-Debtor party making the payment is subrogated to such Holder's Claim, the non-Debtor party shall have a 30-calendar-day grace period to notify the Distribution Trustee of such subrogation rights and, if they are valid and enforceable, the expungement will be reversed to the extent of such subrogation rights.

## (b) Claims Payable by Third Parties

To the extent that one or more of the Debtors' insurers agrees to satisfy a Claim (if and to the extent adjudicated by a court of competent jurisdiction), then immediately upon such insurers' agreement, such Claim may be expunged on the official claims register (in each case to the extent of any agreed-upon satisfaction) by the Clerk of Court or Distribution Trustee, as applicable, without a claims objection having to be filed and without any further notice to or action, order, or approval of the Bankruptcy Court.

## 7.8. Foreign Currency Exchange Rate

As of the Effective Date, any Claim asserted in a currency other than U.S. dollars shall be automatically deemed converted to the equivalent U.S. dollar value using the exchange rate on the Petition Date, as quoted at 4:00 p.m., mid-range spot rate of exchange for the applicable currency as published in The Wall Street Journal, Eastern Edition, on the day after the Petition Date.

### 7.9. Setoff

Except as otherwise provided in this Plan, the Restructuring Implementation Agreement, the RIA Order or another Final Order of the Bankruptcy Court, (a) nothing contained in this Plan shall constitute a waiver or release by the Debtors, the Reorganized Debtors, the Distribution Trustee or the Distribution Trust of any right of setoff or recoupment that any of the foregoing Entities may have against any Entity, and (b) to the extent permitted by Applicable Law, the Distribution Trustee or Reorganized Debtors, as applicable, may setoff or recoup (but shall not be required to do so) against any Claim (and any Interest) and the payments or other Distributions to be made under the Plan in respect of such Claim (or Interest), claims of any nature whatsoever that the Debtors may have against the Holder of such Claim or Interest.

### 7.10. De Minimis Distributions

If any interim Distribution under the Plan to the Holder of an Allowed Claim or Interest would be less than \$100.00 or a fractional number of Offered New Equity, the Distribution Trustee or Reorganized Debtors, as applicable, may withhold such Distribution until the next Subsequent Distribution Date or the date of a final Distribution, as applicable, is made to such Holder. If any final Distribution under the Plan to the Holder of an Allowed Claim or Interest would be less than \$25.00 or a fractional number of Offered New Equity, then such Distribution may be canceled in its entirety. Any unclaimed Distributions pursuant to this Section 7.9 shall be treated as an unclaimed Distribution under Section 7.5(e) of the Plan.

#### 7.11. Fractional Shares

No fractional shares or number of the Offered New Equity shall be issued or distributed under the Plan. The actual Distribution of shares or number of the Offered New Equity shall be rounded to the next higher or lower whole number as follows: (i) fractions less than one-half ( $\frac{1}{2}$ ) shall be rounded to the next lower whole number and (ii) fractions equal to or greater than one-half ( $\frac{1}{2}$ ) shall be rounded to the next higher whole number. The total amount of shares or number of Offered New Equity to be distributed hereunder shall be adjusted as necessary to account for such rounding. No consideration shall be provided in lieu of fractional shares or numbers that are rounded down.

#### 7.12. No Interest

Other than as provided by section 506(b) of the Bankruptcy Code or as specifically provided for in the Plan, the Confirmation Order, or another Final Order of the Bankruptcy Court, postpetition interest shall not accrue or be paid on Claims and no Holder of a Claim shall be entitled to interest accruing on or after the Petition Date on any Claim or right. Additionally, and without limiting the foregoing, interest shall not accrue or be paid on any Claim or Disputed Claim with respect to the period from and after the Effective Date; provided, however, that nothing in this Section 7.12 shall limit any rights of any governmental unit (as defined in section 101(27) of the Bankruptcy Code) to interest under sections 503, 506(b), 1129(a)(9)(A), or 1129(a)(9)(C) of the Bankruptcy Code or as otherwise provided for under Applicable Law.

## VIII. EXECUTORY CONTRACTS AND UNEXPIRED LEASES

#### 8.1. Assumption of Executory Contracts

Unless otherwise specified herein, as of, and subject to, the occurrence of the Effective Date and the payment of any applicable Cure Amounts, all Executory Contracts and Unexpired Leases to which either Debtor is a party, and which have not expired by their own terms on or prior to the Confirmation Date shall be deemed assumed except for any Executory Contract or Unexpired Lease that (a) previously has been assumed or rejected pursuant to a Final Order of the Bankruptcy Court, (b) is the subject of a separate motion for assumption or rejection under section 365 of the Bankruptcy Code filed by a Debtor before the Confirmation Date, (c) is designated as a contract or lease to be rejected on the Rejection Schedule, or (d) is the subject of a pending Assumption Dispute. Subject to the occurrence of the Effective Date, entry of the Confirmation Order by the Bankruptcy Court shall constitute approval of the assumptions and rejections provided for in the Plan pursuant to sections 365(a) and 1123 of the Bankruptcy Code. Each Executory Contract and Unexpired Lease assumed pursuant to the Plan shall vest in, and be fully enforceable by, the Reorganized Debtors in accordance with its terms, except as modified by the provisions of the Plan, and any order of the Bankruptcy Court authorizing and providing for its assumption or Applicable Law.

For the avoidance of doubt, unless otherwise expressly provided in the Plan, the Plan Supplement or the Confirmation Order, all SRUS Retrocession Agreements, Third-Party Reinsurance Agreements, Trust Agreements, SALIC Net Worth Maintenance Agreements, and any and all other reinsurance treaties and trust agreements shall be deemed assumed by SALIC.

## 8.2. Rejection of Executory Contracts

On the Effective Date, all Executory Contracts identified on the Rejection Schedule shall be deemed rejected. The initial Rejection Schedule shall be filed with, and as a part of, the Plan Supplement, and, except as otherwise provided in the Stock Purchase Agreement, may be amended by the Purchaser at any time prior to the Effective Date. Entry of the Confirmation Order shall constitute, subject to the occurrence of the Effective Date, approval of the rejection of such Executory Contracts under sections 365 and 1123 of the Bankruptcy Code.

## 8.3. Determination of Assumption Disputes and Deemed Consent

(a) On or before the date that is at least twenty-one (21) calendar days prior to the Confirmation Objection Deadline (as defined in the Disclosure Statement Order), the Debtors shall File and serve on parties to Executory Contracts and Unexpired Leases to be assumed a notice reflecting the Debtors' intention to assume the contract or lease in connection with the Plan and setting forth the proposed Cure Amount (if any). Such notice shall not prejudice any of the defenses and counterclaims the Debtors, Reorganized Debtors or Distribution Trust may have with respect to any such Executory Contract or Unexpired Lease. If the counterparty believes any different Cure Amount is due in connection with the assumption or otherwise objects to the assumption, it must assert such Cure Amount or any other Assumption Dispute by Filing an objection with the Bankruptcy Court and serving such objection on counsel for the notice parties identified in <u>Section 12.11</u> hereof so as to be received by such parties by no later than twenty-one (21) calendar days after the date of service of such notice.

(b) Cure Amount disputes shall be resolved by the Debtors or Reorganized Debtors and the applicable counterparty in the ordinary course, and the agreed-upon amounts shall be paid by the Debtors or Reorganized Debtors in the ordinary course, with any such payments of Cure Amounts to be funded in accordance with Section 2.4(d) of the Stock Purchase Agreement. If there is an Assumption Dispute pertaining to assumption of an Executory Contract or Unexpired Lease, such dispute shall be heard by the Bankruptcy Court prior to such assumption being effective; *provided, however*, the Debtors or the Reorganized Debtors, as applicable, may settle any dispute regarding the Cure Amount or the Assumption Dispute without any further notice to any party or any action, order, or approval of the Bankruptcy Court. To the extent the Assumption Dispute is resolved or determined by a Final Order unfavorable to the Debtors or Reorganized Debtors, as applicable, shall have fourteen (14) calendar days following entry of such Final Order to File a motion to reject such contract or lease.

Any counterparty to an Executory Contract or Unexpired Lease that fails (c) to object timely to the notice of the proposed assumption of such contract or lease within twentyone (21) calendar days after the date of service thereof shall be deemed to have assented to (i) the proposed Cure Amount, and (ii) the assumption of the applicable contract or lease notwithstanding any provision thereof that purports to (A) prohibit, restrict, or condition the transfer or assignment of such contract or lease, (B) terminate or modify, or permit the termination or modification of, a contract or lease as a result of any direct or indirect transfer or assignment of the rights of a Debtor under such contract or lease or a change, if any, in the ownership or control of a Debtor, (C) increase, accelerate, or otherwise alter any obligations or liabilities of a Debtor under such Executory Contract or Unexpired Lease, or (D) create or impose a Lien upon any property or asset of a Debtor or a Reorganized Debtor. Each such provision shall be deemed to not apply to the assumption of such Executory Contract or Unexpired Lease pursuant to the Plan and counterparties to assumed Executory Contracts or Unexpired Leases that fail to object to the proposed assumption in accordance with the terms set forth in this Section 8.3(c) shall forever be barred and enjoined from objecting to the validity of such assumption or to the Cure Amount, and from taking any other action prohibited by the foregoing on account of transactions contemplated by the Plan.

### 8.4. Effect of Assumption of Contracts and Leases

Subject to resolution of any Assumption Dispute, all Cure Amounts shall be satisfied by the Purchaser (subject to the terms of Section 2.4(d) of the Stock Purchase Agreement) or the Reorganized Debtors, as the case may be, upon assumption of the applicable Executory Contracts and Unexpired Leases in the ordinary course. Assumption of any Executory Contract or Unexpired Lease pursuant to the Plan, or otherwise, subject to satisfaction of the Cure Amount, shall result in the full release and satisfaction of any Claims or defaults, whether monetary or nonmonetary, including defaults of provisions restricting the change in control or ownership interest composition or other bankruptcy-related defaults, arising under any assumed Executory Contract or Unexpired Lease at any time before the effective date of the assumption. On the Effective Date (or with respect to any assumption of an Executory Contract or Unexpired Lease that becomes effective), any proofs of Claim Filed with respect to an Executory Contract or Unexpired Lease that has been assumed shall be deemed withdrawn without further notice to, action, order, or approval of the Bankruptcy Court or any other Entity.

## 8.5. Payment of Cure Amounts

Within thirty (30) days after the Effective Date, the Reorganized Debtors shall pay, in Cash (or as otherwise agreed or ordered by the Bankruptcy Court), all Cure Amounts related to Executory Contracts and Unexpired Leases that are assumed pursuant to this <u>Article VIII of the Plan</u>, other than Cure Amounts that are subject to an Assumption Dispute; *provided*, that the Reorganized Debtors shall pay all Cure Amounts that are subject to an Assumption Dispute on the Effective Date within twenty-one (21) days after entry of an order by the Bankruptcy Court resolving the Assumption Dispute or approving an agreement between the parties concerning the Cure Amount. For the avoidance of doubt, funding of Cure Amounts

shall be subject to sections 2.2(b), 2.3(f) and 2.4(e) of the Stock Purchase Agreement; in particular, the amount contributed by the Purchaser for payment of the Cure Amounts shall not exceed \$100,000 and the Recapitalization Funding Payment shall be used by the Reorganized Debtors to pay any amounts in respect of the Cure Amounts in excess of \$100,000.

## (a) No Admission of Liability

Neither the inclusion nor exclusion of any Executory Contract by the Debtors on the Assumption Schedule, nor anything contained in the Plan, shall constitute an admission by the Debtors that any such contract or lease is in fact an Executory Contract or that the Debtors, the Reorganized Debtors, the Purchaser or Distribution Trust has any liability thereunder.

## (b) Reservation of Rights

Nothing in the Plan shall waive, excuse, limit, diminish, or otherwise alter any of the defenses, claims, causes of action, or other rights of the Debtors, the Reorganized Debtors or Distribution Trust under any executory or non-executory contract or any unexpired or expired lease, nor shall any provision of the Plan increase, augment, or add to any of the duties, obligations, responsibilities, or liabilities of the Debtors or the Reorganized Debtors under any such contract or lease.

## (c) Rejection Claim Bar Date

Each Claim resulting from the rejection of an Executory Contract pursuant to <u>Section 8.2</u> of the Plan shall be Filed with the Bankruptcy Court no later than the Rejection Claim Bar Date. Any Claim resulting from the rejection of an Executory Contract not Filed by the applicable deadline shall be discharged and forever barred, and shall not be entitled to any Distributions under the Plan. The Distribution Trustee shall have the right to object to any rejection damage Claim. All rejection damage Claims shall be treated in Class 5 or 7, respectively, and shall be paid out of the Distribution Trust.

## (d) Continuing Obligations Owed to the Debtors

Any continuing obligations of third parties to the Debtors under insurance policies, contracts, or leases that have otherwise ceased to be executory or have otherwise expired on or prior to the Effective Date, including, without limitation, continuing obligations to pay insured claims, to defend against and process claims, to refund premiums or overpayments, to provide indemnification, contribution or reimbursement, to grant rights of first refusal, to maintain confidentiality, or to honor releases, will continue and will be binding on such third parties, notwithstanding any provision to the contrary herein, unless otherwise specifically terminated by the Debtors or the Reorganized Debtors, or by order of the Bankruptcy Court.

To the extent any insurance policy under which the insurer has a continuing obligation to pay the Debtors or a third party on behalf of the Debtors is held by the Bankruptcy Court to be an Executory Contract, such insurance policy will be treated as though it is an Executory Contract that is assumed by the Reorganized Debtors pursuant to section 365 of the Bankruptcy Code and <u>Sections 8.1</u> and <u>8.3</u> of the Plan.

## (e) **Postpetition Contracts**

The Debtors will not be required to assume or reject any contract or lease entered into by the Debtors after the Petition Date. Any such contract or lease will continue in effect in accordance with its terms after the Effective Date, unless the Reorganized Debtors have obtained a Final Order of the Bankruptcy Court approving rejection of such contract or lease. Contracts or leases entered into after the Petition Date will be performed by the Reorganized Debtors in the ordinary course of their business.

## IX. CONDITIONS PRECEDENT TO CONFIRMATION OF THE PLAN AND OCCURRENCE OF THE EFFECTIVE DATE

## 9.1. Conditions to Confirmation

The Confirmation Order will not be effective unless the final version of the Plan, Plan Supplement, and any other documents, or schedules thereto, including the Filed Confirmation Order, shall have been Filed in form and substance acceptable to the Purchaser in its reasonable discretion, and the Restructuring Implementation Agreement shall not have been terminated and shall be in full force and effect.

## 9.2. Conditions to Effectiveness

Unless the following conditions (except with respect to the Distribution Trust Agreement and the Restructuring Implementation Agreement) are waived by the Purchaser, the Plan will not be effective unless: (a) the conditions to Confirmation above have either been satisfied, or (except with respect to the Restructuring Implementation Agreement) waived by the Purchaser; (b) the Confirmation Order has been entered by the Bankruptcy Court, is not subject to appeal, and no stay or injunction is in effect with respect thereto; (c) the Closing shall have occurred or shall occur simultaneously with the Effective Date; (d) the Hildene Master Funds shall acquire the New Equity (subject to the New Equity Election), directly or indirectly, free and clear of all Liens, Claims, and Interests and in accordance with the Plan; (e) the Distribution Trust Agreement shall have been executed by all parties thereto; (f) the Restructuring Implementation Agreement shall not have been terminated and shall be in full force and effect; and (g) the Purchaser shall have demonstrated to the reasonable satisfaction of the Debtors and the Official Committee that all actions have occurred or will occur on or before the Effective Date necessary to fund the Plan Funding Payment to the Distribution Trust and the Recapitalization Funding Payment to Reorganized SALIC, each as provided in the Plan and the Stock Purchase Agreement; and (h) all governmental, judicial, and third party approvals and consents that are required in connection with the transactions contemplated by the Plan, including regulatory approvals of the transaction in Delaware by the Delaware DOI in accordance with Chapter 101 of Title 29 and Chapter 50 of Title 18 of the Delaware Code Annotated, the Cayman Islands, Bermuda and Ireland, shall have been obtained, not subject to unfulfilled conditions, and shall be in full force and effect.

## X. SETTLEMENT, DISCHARGE, RELEASE, INJUNCTION AND RELATED PROVISIONS

## 10.1. Compromise and Settlement of Claims, Interests, and Controversies

As discussed in detail in the Disclosure Statement and as otherwise provided herein, pursuant to sections 105(a), 363 and 1123 of the Bankruptcy Code and Bankruptcy Rule 9019 and in consideration for the distributions and other benefits provided pursuant to the Plan, which distributions and other benefits shall be irrevocable and not subject to challenge upon the Effective Date, the provisions of the Plan, and the distributions and other benefits provided hereunder, shall constitute a good-faith compromise and settlement of all Claims, Interests, and controversies relating to the contractual, legal, and subordination rights that a Holder of a Claim or Interest may have with respect to any Allowed Claim or Interest, or any distribution to be made on account of such Allowed Claim or Interest. Without limiting the foregoing, the Plan incorporates and is predicated upon the good-faith compromise and settlement of (i) any disputes regarding the appropriate allocation of general and administrative costs across the Debtors' assets, (ii) any disputes regarding the allocation of the Plan Funding Payment and any other value received by the Debtors under the Stock Purchase Agreement between the Debtors' Estates, and (iii) any disputes regarding whether and, if so, to what extent the Debtors' assets and liabilities should be pooled for voting, distribution and other purposes into a single, substantively consolidated estate.

The Plan shall be deemed a motion to approve the good-faith compromise and settlement of all such Claims, Interests, and controversies pursuant to Bankruptcy Rule 9019, and the entry of the Confirmation Order shall constitute the Bankruptcy Court's approval of the compromise or settlement of all such Claims, Interests, and controversies, as well as a finding by the Bankruptcy Court that such compromise or settlement is in the best interests of the Debtors, their Estates, and Holders of Claims and Interests and is fair, equitable, and reasonable. In accordance with the provisions of the Plan, pursuant to Bankruptcy Rule 9019, without any further notice to or action, order, or approval of the Bankruptcy Court, after the Effective Date, the Distribution Trustee may compromise and settle Claims against, and Interests in, the Debtors and their Estates and Causes of Action against other Entities.

#### 10.2. Releases by the Debtors in Favor of Third Parties

As of the Effective Date, for good and valuable consideration, the adequacy of which is hereby confirmed, the Debtors, the Reorganized Debtors, and any Person or Entity seeking to exercise the rights of the Estates, including, without limitation the Distribution Trust, the Distribution Trustee, any successor to the Debtors or any Estate representative appointed or selected pursuant to section 1123(b)(3) of the Bankruptcy Code, hereby forever release, waive and discharge, and shall be deemed to forever release, waive, and discharge each of the Released Parties from any and all Claims, obligations, suits, judgments, damages, demands, debts, rights, causes of action, and liabilities whatsoever, whether liquidated or unliquidated, fixed or contingent, matured or

unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, at equity, or otherwise, that are based in whole or in part on any act, omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the conduct of the Debtors' business, the Chapter 11 Cases, the Disclosure Statement, the Plan, or other documents implementing the Plan, provided, however, that nothing in this Section 10.2 of the Plan shall be deemed to release, or otherwise to prohibit the Reorganized Debtors or the Distribution Trustee from asserting and enforcing, any Claims, obligations, suits, judgments, demands, debts, rights, causes of action, or liabilities any of them may hold related to, or arising out of, the Plan, the DT Post-Closing Rights, the SALIC Claims, the Admitted SALIC/SRGL Revolver Claim, the Retained Causes of Action (solely as to the Reorganized Debtors), Causes of Action that are Distribution Trust Assets (solely as to the Distribution Trustee), the Stock Purchase Agreement, the Restructuring Implementation Agreement, the Distribution Trust Agreement, and the other documents implementing the Plan, provided, further, that nothing in this Section 10.2 of the Plan; (i) shall be deemed to release, or otherwise to prohibit the Reorganized Debtors or the Distribution Trustee, or anyone claiming through them from enforcing any confidentiality or non-disclosure agreement or any Claim, right or cause of action related thereto, (ii) shall be deemed to release, impair, or otherwise affect any parties' rights or interests under any Executory Contract or Unexpired Lease that is assumed by the Reorganized Debtors, and all such rights and interests shall be unaffected by the Plan and this Section 10.2 (subject, however, to the effects of Section 8.3(a), (c), and (h) of the Plan); (iii) shall be deemed to release any Intercompany Claims; (iv) shall be deemed to release any Causes of Action specifically identified in this Plan as Distribution Trust Assets; (v) shall be deemed to release any Claims or liabilities arising out of or relating to any act or omission of a Released Party that constitutes willful misconduct, actual fraud, or gross negligence as determined by a Final Order; or (vi) shall be deemed to release any Person's rights under the Plan.

Entry of the Confirmation Order shall constitute the Bankruptcy Court's approval, pursuant to Bankruptcy Rule 9019, of the Releases by the Debtors described in this Section 10.2 which includes by reference each of the related provisions and definitions contained herein, and, further, shall constitute the Bankruptcy Court's finding that such releases are: (1) in exchange for the good and valuable consideration provided by the Released Parties; (2) a good faith settlement and compromise of the Claims released by such releases; (3) in the best interests of the Debtors and all Holders of Claims and Interests; (4) fair, equitable and reasonable; (5) given and made after due notice and opportunity for hearing; and (6) a bar to any of the Debtors, the Estates or the Distribution Trust asserting any Claim or cause of action released pursuant to such releases.

#### 10.3. Releases by Holders of Claims and Interests

To the fullest extent permitted by law and except as otherwise specifically provided in the Plan, for good and valuable consideration, including the contributions of the Released Parties in facilitating the expeditious reorganization of the Debtors and the implementation of the restructuring contemplated by the Plan, effective as of the Effective Date, (i) Holders of Claims who vote to accept the Plan, (ii) Holders of Claims who are Unimpaired under the Plan, (iii) Holders of Claims entitled to vote on the Plan who do not submit a Ballot and do not timely object to or opt-out of the releases, and (iv) each of the Purchaser, the Joint Liquidators, SRGL, and for each of the foregoing, their respective Representatives (other than with respect to SRGL or the SRGL Equity Holders), hereby release, waive and discharge, and shall be deemed to forever release, waive, and discharge the Released Parties of any and all Claims, obligations, suits, judgments, damages, demands, debts, rights, causes of action, and liabilities whatsoever in connection with or related to the Debtors, whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, at equity, or otherwise, including those that any of the Debtors, the Reorganized Debtors, the Estates, or their Affiliates would have been legally entitled to assert in their own right (whether individually or collectively) or on behalf of the Holder of any Claim or Interest or other Entity based on or relating to, or in any manner arising from, in whole or in part, the Debtors, the conduct of the Debtors' business, the Chapter 11 Cases, the Disclosure Statement, the Plan or the documents implementing the Plan, and the negotiation, formulation, or preparation thereof, the solicitation of votes with respect to the Plan, the subject matter of, or the transactions or events giving rise to, any Claim or Interest that is treated in the Plan, the business or contractual arrangements between any of the Debtors and any Released Party, the restructuring of Claims and Interests prior to or in the Chapter 11 Cases, or upon any other act or omission, transaction, agreement, event or other occurrence taking place on or before the Effective Date, other than Claims or liabilities arising out of or relating to any act or omission of a Released Party that constitutes willful misconduct, actual fraud, or gross negligence as determined by a Final Order. Notwithstanding the foregoing, nothing in this Section 10.3, shall cause (i) a release, waiver or discharge of SFL from any Claim against it under the SFLST I TruPS Documents or (ii) SRGL to release, waive or discharge any Claim it may have against any of SRGL's Representatives (except for the Debtors and the Debtors' Representatives).

Each Person or Entity providing releases under the Plan, including the Debtors, the Estates, and the Holders of Claims (regardless of whether such Holder is a Released Party), shall be deemed to have granted the releases set forth above notwithstanding that such Person or Entity may hereafter discover facts in addition to, or different from, those which it now knows or believes to be true, and without regard to the subsequent discovery or existence of such different or additional facts, and such Person or Entity expressly waives any and all rights that it may have under any statute or common law principle which would limit the effect of such releases to those Claims or causes of action actually known or suspected to exist at the time of execution of such release.

Entry of the Confirmation Order shall constitute the Court's approval, pursuant to Bankruptcy Rule 9019, of the Releases by Holders of Claims and Interests described in this Section 10.3, which includes by reference each of the related provisions and definitions contained herein, and, further, shall constitute the Court's finding that such releases are: (1) in exchange for the good and valuable consideration provided by the Released Parties; (2) a good faith settlement and compromise of the Claims released by such releases; (3) in the best interests of the Debtors and all Holders of Claims and Interests; (4) fair, equitable and reasonable; (5) given and made after due notice and opportunity for hearing; and (6) a bar to any of the Releasing Parties asserting any Claim or cause of action released pursuant to such releases.

#### **10.4.** Discharge and Discharge Injunction

(a) Discharge of Claims

On and after the Effective Date: (a) the rights afforded in the Plan and the treatment of all Claims and Interests shall be in exchange for and in complete satisfaction, discharge, and release of all Claims and Interests of any nature whatsoever, including any interest accrued on such Claims from and after the Petition Date, against the Debtors, the Reorganized Debtors or any of their assets, property, or Estates; (b) the Plan shall bind all Holders of Claims and Interests, notwithstanding whether any such Holders failed to vote to accept or reject the Plan or voted to reject the Plan; (c) all Claims and Interests shall be satisfied, discharged, and released, and the Debtors' and Reorganized Debtors' liability with respect thereto shall be extinguished completely, including any liability of the kind specified under Bankruptcy Code section 502(g); and (d) all entities shall be precluded from asserting against the Debtors, the Reorganized Debtors, the Estates, the Distribution Trust, the Distribution Trustee their successors and assigns, and their assets and properties any other Claims or Interests based upon any documents, instruments, or any act or omission, transaction, or other activity of any kind or nature that occurred prior to the Effective Date, provided, however, that the foregoing discharge shall not apply to the ability of Holders of Allowed Claims to recover from the Distribution Trust on account of such Allowed Claims and/or Interests, all in accordance with the terms of the Plan and **Distribution Trust Agreement.** 

(b) Discharge Injunction

Except as provided in the Plan, to the fullest extent permitted by law, or the Confirmation Order, as of the Effective Date, all Persons that have held, currently hold, may hold, or allege that they hold, a Claim, Interest, or other debt or liability that is satisfied, released and discharged pursuant to the terms of the Plan are permanently enjoined from taking any of the following actions against the Debtors, the Reorganized Debtors, the Distribution Trust, the Distribution Trustee and their respective subsidiaries or their property on account of any such discharged Claims, debts, liabilities or rights: (i) commencing or continuing, in any manner or in any place, any action or other proceeding; (ii) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree, or order; (iii) creating, perfecting, or enforcing any Lien or encumbrance; (iv) asserting a setoff, right of subrogation, or recoupment of any kind against any debt, liability, or obligation due to the Debtors or the Reorganized Debtors; or (v) commencing or continuing any action or other proceeding of any kind, in each such case in any manner, in any place, or against any Person that does not comply with or is inconsistent with the provisions of the Plan.

Except as provided in the Plan or the Confirmation Order, as of the Effective Date, all Persons that have held, currently hold, or may hold, a Claim, Interest, obligation, suit, judgment, damage, demand, debt, right, cause of action, or liability that is released pursuant to Article X of the Plan are permanently enjoined from taking any of the following actions on account of such released Claims, Interests, obligations, suits, judgments, damages, demands, debts, rights, causes of action, or liabilities or rights: (i) commencing or continuing, in any manner or in any place, any action or other proceeding; (ii) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree, or order; (iii) creating, perfecting, or enforcing any Lien or encumbrance; (iv) asserting a setoff against any debt, liability, or obligation due to any Released Party; or (v) commencing or continuing any action, in any manner, in any place, or against any Person or Entity that does not comply with or is inconsistent with the provisions of the Plan.

Without limiting the effect of the foregoing provisions of this <u>Section 10.4</u> of the Plan upon any Person or Entity, by accepting Distributions pursuant to the Plan, each Holder of an Allowed Claim shall be deemed to have specifically consented to the injunctions set forth in this <u>Section 10.4</u> of the Plan.

#### 10.5. Exculpation

To the fullest extent permitted by law and except as otherwise specifically provided in the Plan, each of the Exculpated Parties will not have or incur any liability for any act or omission in connection with, or arising out of, the formulation, negotiation, preparation, dissemination, implementation or pursuit of approval of the Plan, the Disclosure Statement, the Restructuring Implementation Agreement, the Stock Purchase Agreement, the Plan Supplement or any documents, instruments or agreements implementing or related to the foregoing, or the solicitation of votes for or Confirmation of the Plan, or the consummation of the Plan, the Restructuring Implementation Agreement, the Stock Purchase Agreement, the Plan Supplement, or the transactions contemplated, implemented and effectuated thereby or the administration of the Plan or the property to be distributed under the Plan, or any other act or omission during the administration of the Debtors' Estates or in contemplation of the Chapter 11 Cases, except for willful misconduct, actual fraud or gross negligence as determined by a Final Order, and in all respects, will be entitled to rely upon the advice of counsel with respect to their duties and responsibilities under the Plan.

### **10.6.** Post-Effective Date Indemnification

Indemnification Obligations of the Debtors that are owed to Indemnified D&O Parties will be deemed to be, and will be treated as though they are, Executory Contracts that are assumed by the Reorganized Debtors pursuant to section 365 of the Bankruptcy Code, and such Indemnification Obligations shall not be discharged or otherwise impaired by Confirmation of the Plan.

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 60 of 241

From and after the Closing Date, to the extent permitted by Applicable Law, the certificate of incorporation, certificate of formation, bylaws or limited liability company operating agreement (or similar organizational documents) of each SALIC Group Company shall continue to contain provisions no less favorable with respect to indemnification, advancement of expenses and exculpation of each Indemnified D&O Party than are set forth in the organizational documents of the SALIC Group Companies as of the date hereof, which provisions shall not be amended, repealed or otherwise modified for a period of six (6) years after the Closing Date in any manner that would adversely affect the rights thereunder of any such individual.

For a period of six (6) years from and after the Closing Date, to the extent that the Indemnified D&O Parties are not otherwise covered as insureds under an existing policy of directors' and officers' liability insurance in accordance with the requirements set forth in Section 5.8(b) of the Stock Purchase Agreement, Purchaser shall cause the SALIC Group Companies to maintain in effect policies of directors' and officers' liability insurance comparable to those maintained by the SALIC Group Companies or SRGL with respect to matters existing or occurring at or prior to the Closing Date; provided, that Purchaser or the SALIC Group Companies may substitute therefor policies of at least the same coverage containing terms and conditions that are not less advantageous than the existing policies (including with respect to the period covered); provided, further, that in lieu of maintaining the current policies of directors' and officers' liability insurance, Purchaser may (or may cause the SALIC Group Companies to) purchase "tail" coverage or otherwise replace such policies with coverage with a scope, policy limits and retained coverage not less favorable than the scope, policy limits and retained coverage currently provided. Notwithstanding the foregoing, at Purchaser's direction and in satisfaction of Purchaser's obligations under Section 5.8(b) of the Stock Purchase Agreement, SALIC shall obtain such "tail" coverage in respect of SALIC's existing policy of directors' and officers' liability insurance identified in Section 3.15 of the SALIC Disclosure Schedules (Policy No. ELU154535-18) on terms acceptable to Purchaser, to be effective as of the Closing Date, provided that the cost of such coverage shall be funded from unrestricted Cash of SALIC and SHI.

#### XI. RETENTION OF JURISDICTION

#### **11.1. Retention of Jurisdiction**

Pursuant to sections 105(a) and 1142 of the Bankruptcy Code, and notwithstanding entry of the Confirmation Order and occurrence of the Effective Date, except as otherwise ordered by the Bankruptcy Court, or contemplated by the RIA Order or Restructuring Implementation Agreement, the Bankruptcy Court will retain jurisdiction over all matters arising out of, and related to, the Chapter 11 Cases and the Plan to the fullest extent permitted by law, including, among other things, jurisdiction to:

(a) allow, disallow, determine, liquidate, classify, estimate, or establish the priority, secured, unsecured, or subordinated status of any Claim or Interest not otherwise Allowed under the Plan (other than personal injury or wrongful death Claims, unless agreed by

### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 61 of 241

the Holder), including the resolution of any request for payment of any Administrative Claim and the resolution of any objections to the Allowance or priority of Claims;

(b) hear and determine all applications for compensation and reimbursement of expenses of Professionals under the Plan or under sections 327, 328, 330, 331, 503(b), 1103, and 1129(a)(4) of the Bankruptcy Code; provided, however, that from and after the Effective Date, the payment of the fees and expenses of the professionals of the Reorganized Debtors or the Distribution Trust shall be made in the ordinary course of business and shall not be subject to the approval of the Bankruptcy Court;

(c) hear and determine all matters with respect to the assumption or rejection of any Executory Contract or Unexpired Lease to which a Debtor is a party or with respect to which a Debtor may be liable, including, if necessary, the nature or amount of any required Cure or the liquidation or Allowance of any Claims arising therefrom;

(d) effectuate performance of and payments under the provisions of the Plan and enforce remedies upon any default under the Plan;

(e) hear and determine any and all adversary proceedings, motions, applications, and contested or litigated matters arising out of, under, or related to, the Chapter 11 Cases or the Plan;

(f) enter such orders as may be necessary or appropriate to execute, implement, or consummate the provisions of the Plan and all contracts, instruments, releases, and other agreements or documents created in connection with the Plan, the Disclosure Statement or the Confirmation Order;

(g) hear and determine disputes arising in connection with the interpretation, implementation, consummation, or enforcement of the Plan, including disputes arising under agreements, documents, or instruments executed in connection with the Plan;

(h) consider any modifications of the Plan, cure any defect or omission, or reconcile any inconsistency in any order of the Bankruptcy Court, including, without limitation, the Confirmation Order;

(i) issue injunctions, enter and implement other orders, or take such other actions as may be necessary or appropriate to restrain interference by any Person or Entity with the implementation, consummation, or enforcement of the Plan or the Confirmation Order;

(j) enter and implement such orders as may be necessary or appropriate if the Confirmation Order is for any reason reversed, stayed, revoked, modified, or vacated;

(k) hear and determine any matters arising in connection with or relating to the Plan, the Plan Supplement, the schedules to the Plan, the Disclosure Statement, the Confirmation Order, or any contract, instrument, release, or other agreement or document created

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 62 of 241

in connection with the Plan, the Plan Supplement, the Schedules to the Plan, the Disclosure Statement, or the Confirmation Order;

(1) hear and determine any matters arising in connection with or relating to the Distribution Trust, the interpretation, implementation or operation of the Distribution Trust Agreement or the consummation of the transactions contemplated thereby;

(m) enforce, interpret, and determine any disputes arising in connection with any stipulations, orders, judgments, injunctions, releases, exculpations, indemnifications, and rulings entered in connection with the Chapter 11 Cases (whether or not the Chapter 11 Cases have been closed);

(n) except as otherwise limited in the Plan, recover all assets of the Debtors and property of the Estates, wherever located;

(o) hear and determine matters concerning state, local, and federal taxes in accordance with sections 346, 505, and 1146 of the Bankruptcy Code;

(p) hear and determine all disputes involving the existence, nature, or scope of the Debtors' discharge;

(q) hear and determine such other matters as may be provided in the Confirmation Order or as may be authorized under, or not inconsistent with, provisions of the Bankruptcy Code;

(r) hear and determine matters relating to the Stock Purchase Agreement and the Restructuring Implementation Agreement, to the extent provided for in such documents; and

(s) enter a final decree closing the Chapter 11 Cases.

If the Bankruptcy Court abstains from exercising, or declines to exercise, jurisdiction or is otherwise without jurisdiction over any matter arising in, arising under, or related to the Chapter 11 Cases, including the matters set forth in <u>Section 11.1</u> of the Plan, the provisions of Article XI of the Plan will have no effect upon and will not control, prohibit, or limit the exercise of jurisdiction by any other court having jurisdiction with respect to such matter.

#### **11.2.** Limitation on Jurisdiction

In no event shall the provisions of the Plan be deemed to confer in the Bankruptcy Court jurisdiction greater than that established by the provisions of 28 U.S.C. §§ 157 and 1334, as well as the applicable circumstances that continue jurisdiction for defense and enforcement of the Plan and Plan Documents. For the avoidance of doubt, however, such jurisdiction shall be deemed, by the entry of the Confirmation Order, to: (a) Permit entry of a final judgment by the Bankruptcy Court in any core proceeding referenced in 28 U.S.C. § 157(b) and to hear and resolve such proceedings in accordance with 28 U.S.C. § 157(c) and any and all related proceedings, including, without limitation, (i) all proceedings concerning disputes with, or Causes of Action or Claims against, any Entity that the Distribution Trust, the Debtors or the Reorganized Debtors or their successors or assigns, may have, and (ii) any and all Causes of Action or other Claims against any Entity for harm to or with respect to (x) any property of the Estates, or (y) any property of the Estate liened or transferred by the Debtors to any other Entity;

(b) Include jurisdiction over the recovery of any property of the Estates (or property transferred by the Debtors with Bankruptcy Court approval) from any Entity wrongly asserting ownership, possession or control of the same, whether pursuant to sections 542, 543, 549, 550 of the Bankruptcy Code or otherwise, as well as to punish any violation of the automatic stay under section 362 of the Bankruptcy Code or any other legal rights of the Debtors under or related to the Bankruptcy Code; and

(c) Permit the taking of any default judgment against any Entity that has submitted itself to the jurisdiction of the Bankruptcy Court.

## XII. MISCELLANEOUS PROVISIONS

## **12.1.** Legally Binding Effect

The provisions of the Plan shall bind all Creditors and Interest Holders, whether or not they accept the Plan and wherever located. On and after the Effective Date, all Holders of Claims and Interests shall be precluded and enjoined from asserting any Claim against or Interest in the Debtors or their assets or properties based on any transaction or other activity of any kind that occurred prior to the Effective Date except as may be expressly provided for by the Plan.

### **12.2.** Exemption from Transfer Taxes

Pursuant to section 1146 of the Bankruptcy Code and the Plan, any of the following acts or any similar act otherwise contemplated in the Plan will not be subject to any stamp tax, transfer tax, filing or recording tax, or other similar tax: (a) the issuance, transfer or exchange of notes, debt instruments and equity securities under or in connection with the Plan; (b) the creation, assignment, recordation or perfection of any lien, pledge, other security interest or other instruments of transfer; (c) the making or assignment of any lease; (d) the creation, execution and delivery of any agreements or other documents creating or evidencing the formation of the Reorganized Debtors or the issuance or ownership of any interest in the Reorganized Debtors; or (e) the making or delivery of any deed or other instrument of transfer under the Plan in connection with the vesting of the Debtors' assets in the Reorganized Debtors or the Distribution Trust or Distribution Trustee pursuant to or in connection with the Plan, including, without limitation, merger agreements, stock purchase agreement, agreements of consolidation, restructuring, disposition, liquidation or dissolution, and transfers of tangible property.

#### **12.3.** Securities Exemption

Any rights issued under, pursuant to or in effecting the Plan, including, without limitation, the New Equity, the New SHI Equity and any beneficial interests in the Distribution Trust, and the offering and issuance thereof by any party, including without limitation the Debtors, the Estates, or New Holdco (if applicable), shall be exempt from Section 5 of the Securities Act of 1933, if applicable, and from any state or federal securities laws requiring registration for offer or sale of a security or registration or licensing of an issuer of, underwriter of, or broker or dealer in, a security, and shall otherwise enjoy all exemptions available for Distributions of securities under a plan of reorganization in accordance with all Applicable Law, including without limitation section 1145 of the Bankruptcy Code. If the issuance of the New Equity and the New SHI Equity does not qualify for an exemption under section 1145 of the Bankruptcy Code, the New Equity and the New SHI Equity shall be issued in a manner that qualifies for any other available exemption from registration, whether as a private placement under Section 4(a)(2) of the Securities Act and/or the safe harbor provisions promulgated thereunder, Regulation D of 1993, or otherwise.

### 12.4. Defects, Omissions and Amendments of the Plan

The Plan may be amended, modified, or supplemented by the Debtors, subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement, in the manner provided for by section 1127 of the Bankruptcy Code or as otherwise permitted by law, without additional disclosure pursuant to section 1125 of the Bankruptcy Code, except as otherwise ordered by the Bankruptcy Court. In addition, after the Confirmation Date, so long as such action does not materially and adversely affect the treatment of Holders of Allowed Claims pursuant to the Plan, the Debtors, subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement, may remedy any defect or omission or reconcile any inconsistencies in the Plan or the Confirmation Order with respect to such matters as may be necessary to carry out the purposes or effects of the Plan, and any Holder of a Claim that has accepted the Plan shall be deemed to have accepted the Plan as amended, modified, or supplemented. Subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement, prior to the Effective Date, the Debtors may make appropriate technical adjustments and modifications to the Plan without further order or approval of the Bankruptcy Court; provided, that such technical adjustments and modifications do not adversely affect in a material way the treatment of Holders of Claims under the Plan.

### 12.5. Due Authorization by Creditors

Each and every Creditor who elects to participate in the Distributions provided for herein warrants that the Creditor is authorized to accept in consideration of its Claim against the Debtors the Distributions provided for in the Plan, and that there are no outstanding commitments, agreements, or understandings, express or implied, that may or can in any way defeat or modify the rights conveyed or obligations undertaken by the Creditor under the Plan.

### 12.6. Filing of Additional Documentation

No later than seven (7) calendar days prior to the Voting Deadline, subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement and subject to approval in form and substance by the Purchaser, the Debtors may file with the Bankruptcy Court such Plan Supplement, agreements and other documents as may be reasonably necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan or any Plan Document, which shall also constitute "Plan Documents."

## 12.7. Dissolution of the Official Committee

On the Effective Date, the Official Committee shall dissolve and all members, employees, or agents thereof shall be released and discharged from all rights and duties arising from or related to the Chapter 11 Cases, provided, however, that (a) the Official Committee and its respective Professionals shall be retained with respect to applications Filed or to be Filed by Professionals pursuant to sections 330 and 331 of the Bankruptcy Code and (b) the Distribution Trust shall be deemed the successor of the Official Committee with respect to any motions seeking to enforce the Plan and the transactions contemplated hereunder or the Confirmation Order and any pending appeals and related proceedings.

## 12.8. Governing Law

Except to the extent the Bankruptcy Code or the Bankruptcy Rules are applicable, the rights and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with the laws of the State of Delaware, without giving effect to the principles of conflicts of law thereof.

## 12.9. Successors and Assigns

The rights, benefits and obligations of any Entity named or referred to in the Plan or any Plan Document shall be binding on, and shall inure to the benefit of, any heir, executor, administrator, successor or assign of such Entity.

### 12.10. Transfer of Claims

Any transfer of a Claim shall be in accordance with Bankruptcy Rule 3001(e) and the terms of this <u>Section 12.10</u>. Notice of any such transfer shall be forwarded to the Debtors by registered or certified mail, as set forth in <u>Section 12.11</u> hereof. Both the transferee and transferor shall execute any notice, and the signatures of the parties shall be acknowledged before a notary public. The notice must clearly describe the interest in the Claim to be transferred. No transfer of a partial Claim shall be allowed. All transfers must be of one hundred percent (100%) of the transferor's interest in the Claim.

#### **12.11.** Notices

All notices, requests, and demands required or permitted to be provided to the Debtors, the Purchaser, the Reorganized Debtors, the Official Committee, or the Distribution Trust under the Plan shall be in writing and shall be deemed to have been duly given or made (a) when actually delivered (i) by certified mail, return receipt requested, (ii) by hand delivery or (iii) by mail, postage prepaid or, (b) in the case of notice by facsimile transmission, when received and confirmed, addressed (in all instances, with a simultaneous copy by electronic mail, which shall not independently constitute notice) as follows:

#### (a) If to the Debtors, at:

Scottish Holdings, Inc. Scottish Annuity & Life Insurance Company (Cayman) Ltd. 14120 Ballantyne Corporate Place, Suite 300 Charlotte, NC 28277 Facsimile: (704) 752-7736 Attn: Gregg Klingenberg, Chief Executive Officer Gregg.Klingenberg@scottishre.com

with copies to:

Hogan Lovells US LLP 875 Third Avenue New York, NY 10022 Facsimile: (212) 918-3100 Attn: Peter Ivanick, Esq. Lynn W. Holbert, Esq. John D. Beck, Esq. Email: peter.ivanick@hoganlovells.com lynn.holbert@hoganlovells.com

-and-

Morris, Nichols, Arsht & Tunnell LLP 1201 N. Market St., 16th Floor PO Box 1347 Wilmington, DE 19899-1347 Facsimile: (302) 658-3989 Attn: Eric D. Schwartz, Esq. Gregory W. Werkheiser, Esq. Matthew B. Harvey, Esq. Email: eschwartz@mnat.com gwerkheiser@mnat.com mharvey@mnat.com

-and-

Mayer Brown LLP 1221 Avenue of the Americas New York, New York 10020 Telephone: (212) 506-2227 Facsimile: (212) 262-1910 Attn: Stephen G. Rooney, Esq. Francis R. Monaco, Esq. Email: fmonaco@mayerbrown.com srooney@mayerbrown.com

(b) If to the Purchaser or the Reorganized Debtors, at:

Hildene Re Holdings, LLC c/o Hildene Capital Management, LLC 700 Canal Street, Suite 12C Stamford, CT 06902 Telephone: (203) 517-2500

Attention:David Hoffman, General Counsel<br/>Jennifer Nam, Deputy General CounselEmail:dhoffman@hildenecap.com

jnam@hildenecap.com

with a copy to:

Kramer Levin Naftalis & Frankel LLP 1177 Avenue of the Americas New York, New York 10036 Telephone: (212) 715-9100 Facsimile: (212) 715-8000 Attention: Stephen Zide, Esq. Ernest S. Wechsler, Esq. Anupama Yerramalli, Esq. Seth R. Merl, Esq. Email: szide@kramerlevin.com ewechsler@kramerlevin.com ayerramalli@kramerlevin.com smerl@kramerlevin.com

(c) If to the Official Committee, at:

Pepper Hamilton LLP Hercules Plaza, Suite 5100 1313 Market Street P.O. Box 1709 Wilmington, DE 19899-1709 (Courier Deliveries: 19801-1151) Facsimile: (302) 421-8390 Attn: David M. Fournier, Esq. H. Peter Haveles Jr., Esq. John Henry Schanne II, Esq. Email: fournierd@pepperlaw.com havelesp@pepperlaw.com schannej@pepperlaw.com

(d) If to the Distribution Trustee, at the contact information to be supplied in the notice of the occurrence of the Effective Date.

### 12.12. U.S. Trustee Fees and Reports

All pre-confirmation fees due and payable pursuant to section 1930 of Title 28 of the U.S. Code prior to the Effective Date shall be paid by the Debtors. On and after the Effective Date, the Distribution Trustee shall cause the Distribution Trust to pay any and all such fees when due and payable from the Trust Administration Reserve, and shall file with the Bankruptcy Court quarterly reports in a form reasonably acceptable to the United States Trustee. The Distribution Trust shall remain obligated to pay quarterly fees to the Office of the United States Trustee for each of the Chapter 11 Cases until such Chapter 11 Case has been closed, dismissed, or converted to a case under Chapter 7 of the Bankruptcy Code.

#### 12.13. Implementation

The Debtors, the Reorganized Debtors, the Purchaser, and the Distribution Trustee shall be authorized to perform all reasonable, necessary and authorized acts to consummate the terms and conditions of the Plan and the Plan Documents.

#### 12.14. No Admissions

Notwithstanding anything herein to the contrary, nothing contained in the Plan shall be deemed an admission by the Debtors with respect to any matter set forth herein, including, without limitation, liability on any Claim or Interest or the propriety of the classification of any Claim or Interest.

### 12.15. SRGL Consent Rights Reserved

For the avoidance of doubt, all SRGL Consent Rights relating to Specified Restructuring Documents are reserved in their entirety. Nothing herein shall affect SRGL's rights under the Restructuring Implementation Agreement, all of which are incorporated herein by reference, including in respect of the Restructuring Documents and the Specified Restructuring Documents. Without limiting the generality of the foregoing, (a) the Restructuring Documents shall, unless otherwise expressly indicated in the Restructuring Implementation Agreement, be consistent in all material respects with the Restructuring Implementation Agreement, and (b) the Specified Restructuring Documents shall be consistent in all material respects with the Restructuring Implementation Agreement, nothing set forth in the Restructuring Implementation Agreement or this Plan shall operate as a waiver or release of (i) the Admitted SALIC/SRGL Revolver Claim; (ii) the SALIC Claims; or (iii) any Causes of Action against the SRGL Equity Holders.

### 12.16. Substantial Consummation

The Plan shall be deemed substantially consummated on the Effective Date.

### 12.17. Final Decree

On full consummation and performance of the Plan and Plan Documents, the Distribution Trustee may request the Bankruptcy Court to enter a final decree closing the Chapter 11 Cases and such other orders that may be necessary and appropriate.

## Scottish Holdings, Inc. Scottish Annuity & Life Insurance Company (Cayman) Ltd.

Gregg Klingenberg

Gregg Klingenberg Chief Executive Officer

## <u>Exhibit A</u>

**Glossary of Defined Terms**
#### EXHIBIT A Glossary of Defined Terms

1.1 "<u>Administrative Claim</u>" means a Claim for any costs or expenses of administration of the Estates under sections 503(b), 507(b) or 1114(e)(2) of the Bankruptcy Code, including, without limitation, for: (a) any actual and necessary costs and expenses incurred after the Petition Date of preserving the Estates and operating the businesses of the Debtors; (b) any payment to be made under the Plan to cure a default on an assumed Executory Contract or assumed Unexpired Lease; (c) any postpetition cost, indebtedness or contractual obligation duly and validly incurred or assumed by the Debtors in the ordinary course of its business or by order of the Bankruptcy Court; (d) any Allowed Claims that are entitled to be treated as Administrative Claims pursuant to a Final Order under section 546(c)(2)(A) of the Bankruptcy Code; (e) any Allowed Claims of Professionals in the Chapter 11 Cases; and (f) any fees and charges assessed against the Estates under chapter 123 of title 28 of the United States Code, 28 U.S.C. §§ 1911–1930.

1.2 "<u>Administrative Claims Bar Date</u>" means, (i) for any Administrative Claim (other than a Professional Fee Claim) incurred prior to the Confirmation Date, the date that is thirty (30) days after the Confirmation Date or such earlier deadline applicable to such Administrative Claim as established by order of the Bankruptcy Court entered before the Confirmation Date ; or (ii) for any Administrative Claim (other than a Professional Fee Claim) incurred between the Confirmation Date and the Effective Date, the date that is thirty (30) days after the Effective Date or such earlier deadline applicable to such Administrative Claim as established by order of the Bankruptcy Court entered before the Effective Date or such earlier deadline applicable to such Administrative Claim as established by order of the Bankruptcy Court entered before the Effective Date.

1.1 "<u>Admitted SALIC/SRGL Revolver Claim</u>" shall have the meaning set forth in paragraph 11 of the RIA Order.

1.2 "<u>Affiliate</u>" means "affiliate" as defined in section 101(2) of the Bankruptcy Code. With respect to any Person that is not a Debtor, the term "Affiliate" shall apply to such Person as if the Person were a Debtor.

1.3 "<u>Allocated Portion</u>" means with respect to a particular issuance of TruPS, the portion of the TruPS Claim allocable on a pro rata basis to a Beneficial Holder on account of its beneficial ownership of such TruPS, as determined in accordance with the applicable TruPS Declaration.

1.4 "<u>Allocation/Use Priorities</u>" means the waterfall described in Section 6.1(b) of the Plan.

1.5 "<u>Allowed</u>" or "<u>Allowance</u>" means with respect to any Claim (including any Administrative Claim) or portion thereof (to the extent such Claim is not Disputed or Disallowed) or any Interest (a) any Claim or Interest, proof of which (i) was timely Filed with the Bankruptcy Court or its duly appointed claims agent, (ii) was deemed timely filed pursuant to section 1111(a) of the Bankruptcy Code, or (iii) by the Bar Date Order or other Final Order, was not required to be Filed; (b) any Claim or Interest that has been, or hereafter is, listed in the Schedules as liquidated in an amount other than zero or unknown and not Disputed or

Contingent (or as to which the applicable Proof of Claim has been withdrawn or Disallowed); and (c) any Claim or Interest which has been allowed (whether in whole or in part) by the Plan, the Restructuring Implementation Agreement or the RIA Order or other Final Order (but only to the extent so allowed), and, in (a) and (b) above, as to which no objection to the allowance thereof, or action to subordinate, avoid, classify, reclassify, expunge, estimate or otherwise limit recovery with respect thereto, has been Filed within the applicable period of limitation fixed by the Plan, the Bankruptcy Code, the Bankruptcy Rules or a Final Order; (d) any Claim or Interest allowed under or pursuant to the terms of the Plan; (e) any Claim arising from the recovery of property under sections 550 or 553 of the Bankruptcy Code which has been allowed in accordance with section 502(h) of the Bankruptcy Code; (f) a Claim relating to a rejected Executory Contract or Unexpired Lease that either (i) is not a Disputed Claim or (ii) has been allowed by a Final Order, in either case only if a Proof of Claim has been Filed by the applicable Bar Date or has otherwise been deemed timely Filed under Applicable Law; or (g) which is a Professional Claim for which a fee award amount has been approved by order of the Bankruptcy Court; provided, however, that Claims or Interests allowed solely for the purpose of voting to accept or reject the Plan pursuant to an order of the Bankruptcy Court shall not be considered "Allowed" hereunder. For the avoidance of doubt, a Proof of Claim Filed after the Bar Date or a request for payment of an Administrative Claim Filed after the Administrative Claims Bar Date, as applicable, shall not be Allowed for any purposes whatsoever absent entry of a Final Order allowing such late-filed Claim.

1.6 "<u>Applicable Law</u>" means any foreign, domestic, state, federal, national, international, multinational, regional or local law, statute, ordinance, rule, regulation, writ, directive, treaty, order, judgment, decree, injunction or other legally binding obligation imposed by or on behalf of a Governmental Unit.

1.7 "<u>Assumption Dispute</u>" means a pending objection relating to the assumption of an Executory Contract or Unexpired Lease pursuant to section 365 of the Bankruptcy Code, including, but not limited to, as to the Cure Amount.

1.8 "<u>Available Plan Distribution Funding Amount</u>" means all of the Plan Funding Payment *less* the amounts set forth in Sections 6.1(b)(1) through (4) of the Plan.

1.9 "<u>Ballantyne</u>" means Ballantyne Re II plc.

1.10 "<u>Ballot</u>" has the meaning set forth in paragraph D of the Disclosure Statement Order.

1.11 "<u>Bankruptcy Code</u>" means the Bankruptcy Reform Act of 1978, as codified in title 11 of the United States Code, 11 U.S.C. §§ 101-1532, as in effect on the Petition Date, together with all amendments and modifications thereto that were subsequently made applicable to the Chapter 11 Cases.

1.12 "<u>Bankruptcy Court</u>" means the United States Bankruptcy Court for the District of Delaware or such other court as may have jurisdiction over the Chapter 11 Cases.

1.13 "<u>Bankruptcy Rules</u>" means, when referenced generally, (i) the Federal Rules of Bankruptcy Procedure and the Official Bankruptcy Forms, as amended and promulgated under section 2075 of title 28 of the United States Code, (ii) the applicable Federal Rules of Civil Procedure, as amended and promulgated under section 2072 of title 28 of the United States Code, (iii) the applicable Local Rules of Bankruptcy Practice and Procedure for the United States Bankruptcy Court for the District of Delaware, and (iv) any standing orders governing practice and procedure issued by the Bankruptcy Court, each as in effect on the Petition Date, together with all amendments and modifications thereto that were subsequently made applicable to the Chapter 11 Cases or proceedings therein, as the case may be; provided, however, when a specific Bankruptcy Rule is referenced (e.g., Bankruptcy Rule 9019), such reference shall be to such Rule under the Federal Rules of Bankruptcy Procedure.

1.14 "<u>Bar Date</u>" means, for any Claim, the date set by the Bankruptcy Court by the Bar Date Order as the last day for Filing a Proof of Claim on account of such Claim against the Debtors in the Chapter 11 Cases.

1.15 "<u>Bar Date Order</u>" means the Order (A) Establishing Bar Dates For Filing Proofs Of Claim, (B) Approving The Form And Manner For Filing Proofs Of Claim, (C) Approving Notice Thereof, And (D) Granting Related Relief entered by the Bankruptcy Court on March 28, 2018, at Docket No. 189.

1.16 "<u>Beneficial Holder</u>" means, with respect to any TruPS, the person or entity having "beneficial ownership" of such TruPS (as determined pursuant to Rule 13d-3 of the Securities Exchange Act of 1934). For the avoidance of doubt, the term Beneficial Holder includes SRGL with respect to its TruPS holdings.

1.17 "<u>BNYM</u>" means The Bank of New York Mellon Trust Company, N.A., as trustee.

1.18 "<u>Business Day</u>" means any day, excluding Saturdays, Sundays, or "legal holidays" (as defined in Bankruptcy Rule 9006(a)), on which commercial banks are open for business in Wilmington, Delaware.

1.19 "<u>Cash</u>" or "<u>\$</u>" means legal tender of the United States of America or the equivalent thereof, including bank deposits, checks and cash equivalents.

1.20 "<u>Cash Election</u>" means an election made by a New Equity Eligible Holder to take its Distribution from the Plan Funding Payment under the Plan in the form of Cash, which shall be made by not checking the "New Equity Election" box on its timely completed and submitted Ballot. In the event that a New Equity Eligible Holder fails to make an election or fails to submit a Ballot, then such New Equity Eligible Holder shall be deemed to have made the Cash Election.

1.21 "<u>Causes of Action</u>" means any and all Claims, causes of action, controversies, obligations, suits, judgments, damages, demands, debts, rights, preference actions, fraudulent conveyance actions and other claims or causes of action under sections 510, 544, 545, 546, 547, 548, 549, 550 and 553 of the Bankruptcy Code and other similar state law claims and causes of action, Liens, indemnities, guaranties, suits, liabilities, judgments, accounts, defenses, offsets, powers, privileges, licenses and franchises of any kind or character whatsoever, whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, suspected or unsuspected, disputed or undisputed, secured or unsecured, assertable directly or derivatively, whether arising before, on, or after the Petition Date, in

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 74 of 241

contract or in tort, arising in law, equity or pursuant to any other theory of law. For the avoidance of doubt, Causes of Action also includes: (a) any right of setoff, counterclaim or recoupment and any claim on contracts or for breaches of duties imposed by law or in equity; (b) the right to object to Claims or Interests; (c) any claim pursuant to sections 362 or chapter 5 of the Bankruptcy Code; (d) any claim or defense including fraud, mistake, duress and usury, and any other defenses set forth in section 558 of the Bankruptcy Code; and (e) any state law fraudulent transfer claim.

1.22 "<u>Cayman Islands Court</u>" means the Grand Court of the Cayman Islands, Financial Services Division.

1.23 "<u>Cayman Islands Winding Up Proceedings</u>" means winding up proceedings of SRGL in the Cayman Islands Court.

1.24 "Cerberus" means Cerberus Capital Management, L.P., and SRGL Acquisition.

1.25 "<u>Certificate</u>" means any instrument evidencing a Claim or Interest.

1.26 "<u>Chapter 11 Cases</u>" mean the jointly administered chapter 11 cases of Scottish Holdings, Inc., and Scottish Annuity & Life Insurance Company (Cayman) Ltd., Case No. 18-10160 (LSS) in the United States Bankruptcy Court for the District of Delaware.

1.27 "<u>Charging Lien</u>" means any Lien or other priority in payment to which a TruPS Indenture Trustee is entitled under the terms of a TruPS Indenture to assert against Distributions to be made to Holders of Claims under such TruPS Indenture.

1.28 "<u>Claim</u>" has the same meaning as "claim" under section 101(5) of the Bankruptcy Code.

1.29 "<u>Claims Objection Deadline</u>" means the date that is one hundred eighty (180) days after the Effective Date or such later date as may be extended by order of the Bankruptcy Court.

1.30 "<u>Class</u>" means a category of Holders of Claims or Interests pursuant to section 1122(a) of the Bankruptcy Code, as described in Articles III and IV of the Plan.

1.31 "<u>Closing</u>" has the meaning as defined in the Stock Purchase Agreement.

1.32 "<u>Closing Date</u>" has the meaning as defined in the Stock Purchase Agreement.

1.33 "<u>Closing Date Plan Distributions</u>" means Distributions, and the Cash to be distributed on account thereof, to the Holders of Secured Claims, Administrative Claims, Priority Claims, and Priority Non-Tax Claims all to the extent Allowed as of the Effective Date, <u>provided</u> that in no event shall the aggregate of Closing Date Plan Distributions exceed the Plan Funding Payment.

1.34 "<u>Confirmation</u>" means the entry of the Confirmation Order on the docket of the Chapter 11 Cases, subject to all conditions specified having been (a) satisfied, or (b) waived.

### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 75 of 241

1.35 "<u>Confirmation Date</u>" means the date upon which the Bankruptcy Court enters the Confirmation Order on the docket of the Chapter 11 Cases, within the meaning of Bankruptcy Rules 5003 and 9021.

1.36 "<u>Confirmation Hearing</u>" means the hearing held by the Bankruptcy Court pursuant to section 1128 of the Bankruptcy Code to consider Confirmation of the Plan, as such hearing may be adjourned or continued from time to time.

1.37 "<u>Confirmation Order</u>" means the order of the Bankruptcy Court confirming the Plan pursuant to, among others, section 1129 of the Bankruptcy Code, which shall be in form and substance reasonably acceptable to the Purchaser.

1.38 "<u>Contract Objection</u>" has the meaning set forth in <u>Section 8.3(b)</u> of the Plan.

1.39 "<u>Contract Objection Deadline</u>" has the meaning set forth in <u>Section 8.3(b)</u> of the Plan.

1.40 "<u>Creditor</u>" means any Holder of a Claim.

1.41 "<u>Cure</u>" means the Distribution of Cash, or such other property as may be agreed upon by the parties or ordered by the Bankruptcy Court, with respect to the assumption or assumption and assignment of an Executory Contract or Unexpired Lease, pursuant to Bankruptcy Code section 365(b), in an amount equal to all unpaid monetary obligations, without interest, or such other amount as may be agreed upon by the parties, under such Executory Contract or Unexpired Lease, to the extent such obligations are enforceable under the Bankruptcy Code and applicable bankruptcy law and as determined pursuant to Section 8.3 of the Plan.

1.42 "<u>Cure Amount</u>" means, for any Executory Contract or Unexpired Lease, the amount of the Cure asserted by the Debtors or the counterparty, as applicable.

1.43 "<u>Debtors</u>" means Scottish Holdings, Inc., and Scottish Annuity & Life Insurance Company (Cayman) Ltd., the debtors and debtors in possession in these Chapter 11 Cases.

1.44 "<u>Delaware DOI</u>" means the Delaware Department of Insurance.

1.45 "<u>Disallowed</u>" means, with respect to any Claim or Interest or portion thereof, any Claim against or Interest in a Debtor which is not Allowed and: (i) has been disallowed, in whole or part, by a Final Order; (ii) has been withdrawn by agreement of the Holder thereof and the applicable Debtor(s), in whole or in part; (iii) has been withdrawn, in whole or in part, by the Holder thereof; (iv) if listed in the Schedules as zero or as Disputed, contingent or unliquidated and in respect of which a Proof of Claim or a Proof of Interest, as applicable, has not been timely Filed or deemed timely Filed pursuant to the Plan, the Bankruptcy Code or any Final Order or other Applicable Law; (v) has been reclassified, expunged, subordinated or estimated to the extent that such reclassification, expungement, subordination or estimation results in a reduction in the Filed amount of any Proof of Claim or Proof of Interest; (vi) is evidenced by a Proof of Claim or a Proof of Interest which has been Filed, or which has been deemed to be Filed under Applicable Law or order of the Bankruptcy Court or which is required to be Filed by order of the Bankruptcy Court but as to which such Proof of Claim or Proof of Interest was not timely or properly Filed; (vii) is unenforceable to the extent provided in section 502(b) of the Bankruptcy Code; (viii) where the Holder of a Claim is a Person or Entity from which property is recoverable under sections 542, 543, 550, or 553 of the Bankruptcy Code or that is a transferee of a transfer avoidable under sections 522(f), 522(h), 544, 545, 547, 548, 549, or 724(a) of the Bankruptcy Code, unless such Person, Entity or transferee has paid the amount, or turned over any such asset or property, for which such Person, Entity or transferee is liable under section 522(i), 542, 543, 550, or 553 of the Bankruptcy Code; or (ix) is for reimbursement or contribution that is contingent as of the time of allowance or disallowance of such Claim. In each case a Disallowed Claim is disallowed only to the extent of disallowance, withdrawal, reclassification, expungement, subordination or estimation.

1.46 "<u>Disallowed Claim</u>" means a Claim, or any portion thereof, that is Disallowed.

1.47 "<u>Disclosure Statement</u>" means the disclosure statement for the Plan, as amended, supplemented or modified from time to time, describing the Plan, that is prepared and distributed in accordance with, among others, sections 1125, 1126(b) and 1145 of the Bankruptcy Code, Bankruptcy Rule 3018 and other Applicable Law.

1.48 "<u>Disclosure Statement Order</u>" means the order of the Bankruptcy Court approving the Disclosure Statement and solicitation procedures with respect to the Plan, which can be found at Docket No. 380.

"Disputed" means, with respect to any Claim, including Priority Claims and 1.49 Administrative Claims, that has not been Allowed, (a) if no Proof of Claim has been Filed by the applicable Bar Date or has otherwise been deemed timely Filed under Applicable Law: (i) a Claim that is listed on the Debtor's Schedules as other than disputed, contingent or unliquidated, but as to which the Debtors or Reorganized Debtors or, prior to the Effective Date, any other party in interest, has Filed an objection by the Claims Objection Deadline, and such objection has not been withdrawn or denied by a Final Order; or (ii) a Claim that is listed on the Debtors' Schedules as disputed, contingent or unliquidated; or (b) if a Proof of Claim or request for payment of an Administrative Claim has been Filed by the applicable Bar Date or has otherwise been deemed timely Filed under Applicable Law: (i) a Claim for which no corresponding Claim is listed on the Debtor's Schedules; (ii) a Claim for which a corresponding Claim is listed on the Debtor's Schedules as other than disputed, contingent or unliquidated, but the nature or amount of the Claim as asserted in the Proof of Claim varies from the nature and amount of such Claim as it is listed on the Schedules; (iii) a Claim for which a corresponding Claim is listed on the Debtor's Schedules as disputed, contingent or unliquidated; (iv) a Claim for which an objection has been Filed by the Debtors or Reorganized Debtors or, prior to the Effective Date, any other party in interest, by the Claims Objection Deadline, and such objection has not been withdrawn or denied by a Final Order; or (v) a tort claim.

1.50 "<u>Disputed Claim Amount</u>" means (a) if a liquidated amount is set forth in the Proof of Claim relating to a Disputed Claim that has not been Allowed, (i) the liquidated amount set forth in the Proof of Claim relating to the Disputed Claim; (ii) an amount agreed to by the Debtors or the Reorganized Debtors, as applicable, and the Holder of such Disputed Claim; or (iii) if a request for estimation is Filed by any party, the amount at which such Disputed Claim is

estimated by the Bankruptcy Court; (b) if no liquidated amount is set forth in the Proof of Claim relating to a Disputed Claim, (i) an amount agreed to by the Debtors or the Reorganized Debtors, as applicable, and the Holder of such Disputed Claim or (ii) the amount estimated by the Bankruptcy Court with respect to such Disputed Claim; or (c) zero, if the Disputed Claim was listed on the Schedules as unliquidated, contingent or disputed and no Proof of Claim was Filed, or deemed to have been Filed, by the applicable Bar Date and the Claim has not been resolved by written agreement of the parties or an order of the Bankruptcy Court.

1.51 "Disputed Claims Reserve" means, Cash set aside by the Distribution Trustee in the amount required pursuant to Section 6.3(f)(i) of the Plan for each Disputed Claim that, as of the Effective Date, is neither an Allowed Claim nor a Disallowed Claim, and includes, without limitation, a Claim that is the subject of a timely objection or request for estimation with the Bankruptcy Court, which has not been withdrawn, settled or overruled by a Final Order; *provided, however,* if any dispute arises regarding any increase or reduction of the Disputed Claims Reserve, the Distribution Trustee shall consult with the Purchaser and the Reorganized Debtors, and obtain approval of the Bankruptcy Court, which shall have jurisdiction and power to set the amount of the reserve applying the principals of section 502(c) of the Bankruptcy Code to estimate any claim.

1.52 "<u>Distribution</u>" means any distribution pursuant to the Plan to the Holders of Allowed Claims or Interests.

1.53 "<u>Distribution Date</u>" means (a) with respect to Distributions made as Closing Date Plan Distributions, the Effective Date, (b) with respect to a DT Initial Distribution, the DT Initial Distribution Date, and (c) with respect to a DT Subsequent Distribution, the DT Subsequent Distribution Date.

1.54 "<u>Distribution Trust</u>" means the trust to be formed on or prior to the Effective Date in accordance with the provisions of <u>Section 6.3</u> of the Plan and the Distribution Trust Agreement for the payment of Allowed Claims and for the benefit of the Distribution Trust Beneficiaries as set forth herein.

1.55 "<u>Distribution Trust Agreement</u>" means the trust agreement that establishes the Distribution Trust and governs the powers, duties, and responsibilities of the Distribution Trustee. The Distribution Trust Agreement shall be part of the Plan Supplement, and shall be in form and substance reasonably acceptable to the Official Committee, the Debtors and the Purchaser.

1.56 "<u>Distribution Trust Assets</u>" means: (a) the Admitted SALIC/SRGL Revolver Claim; (b) all SALIC Group Services Claims incurred prior to the Petition Date; (c) any distributions or rights to distribution from SRGL or its estate that are made in connection with the Cayman Islands Winding Up Proceeding at any time on account of (i) the Admitted SALIC/SRGL Revolver Claim and/or (ii) any SALIC Group Services Claims incurred prior to the Petition Date; (d) all DT Post-Closing Rights; (e) the SFL Shares; and (f) any and all Causes of Action against the SRGL Equity Holders. For the avoidance of doubt, any SALIC Group Services Claims incurred on or after the Petition Date are not Distribution Trust Assets and, instead, shall re-vest in the Reorganized Debtors.

1.57 "<u>Distribution Trust Asset Proceeds</u>" means the proceeds of the Distribution Trust Assets, net of expenses incurred in preserving and/or monetizing such Distribution Trust Assets.

1.58 "<u>Distribution Trust Beneficiaries</u>" means the Holders of Allowed Claims in Classes 4, 5, 6, and 7, whether Allowed on or after the Effective Date.

1.59 "<u>Distribution Trust Reserves</u>" means the Disputed Claims Reserve, the Professional Fee Reserve, and the Trust Administration Reserve.

1.60 "<u>Distribution Trustee</u>" the Person appointed under the Plan and Distribution Trust Agreement to administer the Distribution Trust, which Person shall be selected by the Official Committee and reasonably acceptable to the Debtors and the Purchaser; *provided, however*, that the Distribution Trustee shall not take any action inconsistent with the purposes of the Distribution Trust and the qualification of the Distribution Trust as a "liquidating trust" for U.S. federal income tax purposes.

1.61 "<u>DT Initial Distribution</u>" means the Distribution set forth in section 6.3(i)(i) of the Plan.

1.62 "<u>DT Initial Distribution Date</u>" means the date that is as soon as reasonably practicable after the Administrative Claims Bar Date, but in no event later than thirty (30) days after the Administrative Claims Bar Date, when Distributions of New Equity and Cash (as applicable) shall commence to Holders of Allowed Claims.

1.63 "<u>DT Subsequent Distribution</u>" means after the completion of the DT Initial Distribution, the Distributions to the Holders of Allowed Claims in Classes 4, 5, 6 and 7 on account of their Allowed Claims from Distribution Trust Asset Proceeds; *provided, however*, as set forth in the Plan, that the Distribution Trustee shall not be required to make a Distribution pursuant to this <u>Section 6.3(i)</u> of the Plan if the Distribution Trustee determines that the expense associated with making the Distribution would likely utilize a substantial portion of the amount to be distributed, thus making the Distribution impracticable.

1.64 "<u>DT Subsequent Distribution Date</u>" means any later date that a Distribution not made on the DT Initial Distribution Date is made.

1.65 "<u>DT Post-Closing Rights</u>" has the meaning as defined in the Stock Purchase Agreement.

1.66 "<u>Effective Date</u>" means any Business Day following the date on which all conditions to consummation set forth in <u>Section 9.1</u> of the Plan have been satisfied or, if capable of being duly and expressly waived, as provided in <u>Section 9.2</u> of the Plan, any conditions to the occurrence of consummation set forth in the Plan have been satisfied or waived.

1.67 "<u>Eligible SALIC TruPS Claims</u>" means all TruPS Claims against SALIC arising from or relating to the SHST I TruPS, the SHST III TruPS or the SFLST I TruPS.

1.68 "<u>Eligible SHI TruPS Claims</u>" means all TruPS Claims against SHI arising from or relating to the SHST I TruPS or the SHST III TruPS.

1.69 "<u>Entity</u>" means a Person, estate, trust, Governmental Unit, and U.S. Trustee, within the meaning of Bankruptcy Code section 101(15).

1.70 "<u>Estates</u>" means the estates of the Debtors in the Chapter 11 Cases created pursuant to section 541 of the Bankruptcy Code.

1.71 "<u>Exculpated Parties</u>" means (a) the Debtors, (b) the Reorganized Debtors, (c) the Official Committee (including the current and former Official Committee members in their capacities as such); (d) the Voting Agent; and (e) for each of the foregoing, their respective Representatives, each to the extent that they held such office or capacity during the pendency of the Chapter 11 Cases; *provided, however*, that, the SRGL Equity Holders shall not be an Exculpated Party or a Representative of an Exculpated Party.

1.72 "<u>Executory Contract</u>" means any contract or Unexpired Lease to which the Debtor is a party that is subject to assumption or rejection under section 365 of the Bankruptcy Code.

1.73 "Face Amount" means the outstanding amount of any Certificate or Claim.

1.74 "<u>File</u>," "<u>Filed</u>" or "<u>Filing</u>," means, respectively, file, filed or filing with the Bankruptcy Court in the Chapter 11 Cases; <u>provided</u>, <u>however</u>, that with respect to Proofs of Claim and Proofs of Interest only, "Filed" shall mean delivered and received in the manner provided by the Bar Date Order or as otherwise established by order of the Bankruptcy Court.

1.75 "<u>Final Order</u>" means an order, ruling, judgment, the operation or effect of a judgment or other decree issued and entered by the Bankruptcy Court or by any state or other federal court or other court of competent jurisdiction which has not been reversed, vacated, stayed, modified or amended and as to which (i) the time to appeal or petition for review, rehearing, certiorari, reargument or retrial has expired and as to which no appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal and provided, however, that the possibility that a motion under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules, may be filed with respect to such order or judgment shall not cause such order or judgment not to be a Final Order.

1.76 "<u>General Unsecured Claim</u>" means any unsecured Claim that is not an Administrative Claim, Priority Claim, Priority Tax Claim, Professional Fee Claim, Secured Claim, a claim for U.S. Trustee Fees, Intercompany Claim, or Subordinated Claim.

1.77 "<u>General Unsecured Creditor</u>" means any Holder of a General Unsecured Claim.

1.78 "Governmental Unit" has the meaning of such term under Bankruptcy Code section 101(27).

1.79 "<u>GPIC TruPS</u>" has the meaning ascribed to such term in the definition of "TruPS".

### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 80 of 241

1.80 "<u>GPIC TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

1.81 "<u>Hildene Master Funds</u>" means Hildene Opportunities Master Fund, Ltd. and Hildene Opportunities Master Fund II, Ltd.

1.82 "<u>Holder</u>" means the legal or beneficial holder of a Claim or Interest (and, when used in conjunction with a Class or type of Claim or Interest, means a Holder of a Claim or Interest in such Class or of such type). Additionally, in reference to the Distribution Trust, the term "Holder" means the legal or beneficial holder of an Allowed Claim.

1.83 "<u>Impaired</u>" means, when used with reference to a Claim or Interest, a Claim or Interest that is impaired within the meaning of section 1124 of the Bankruptcy Code.

1.84 "<u>Impaired Class</u>" means a Class of Claims or Interests that are Impaired.

1.85 "<u>Indemnified D&O Parties</u>" means each present and former director and officer of each of the SALIC Group Companies, as set forth in <u>section 5.8(a) of the Stock Purchase</u> <u>Agreement.</u>

1.86 "<u>Indemnification Obligation</u>" means any Claim against or obligation of the Debtors to indemnify, reimburse, advance expenses or provide contribution to or with respect to any present or former officers, directors or employees pursuant to by-laws, articles of incorporation, agreements, contracts, common law or otherwise as may be in existence immediately prior to the Effective Date.

1.87 "<u>Indenture Trustee Fees</u>" means the reasonable and documented fees and out-ofpocket expenses of the TruPS Indenture Trustees that are recoverable by such TruPS Indenture Trustees in accordance with the terms of their respective TruPS Indentures.

1.88 "Insider" has the meaning set forth in section 101(31) of the Bankruptcy Code.

1.89 "<u>Intercompany Claims</u>" means all Claims between or among any Debtors and any other SALIC Group Company. For the avoidance of doubt, Claims by or against SFL and SRGL are not Intercompany Claims for purposes of the Plan.

1.90 "Interest" means the legal, equitable, contractual interests, equity interests or ownership interests, or other rights of any Person in the Debtors including all capital stock, stock certificates, common stock, preferred stock, partnership interests, limited liability company or membership interests, rights, treasury stock, options, warrants, contingent warrants, convertible or exchangeable securities, investment securities, subscriptions or other agreements and contractual rights to acquire or obtain such an interest or share in the Debtors, partnership interests in the Debtors' stock appreciation rights, conversion rights, repurchase rights, redemption rights, dividend rights, preemptive rights, subscription rights and liquidation preferences, puts, calls, awards or commitments of any character whatsoever relating to any such equity, common stock, preferred stock, ownership interests or other shares of capital stock of the Debtors or obligating the Debtors to issue, transfer or sell any shares of capital stock whether or not certificated, transferable, voting or denominated "stock" or a similar security.

## Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 81 of 241

1.91 "<u>IT Fee Notice Parties</u>" means (a) for any notice to be provided or service to be made as of a date prior to the Effective Date, (i) the Debtors, (ii) the Official Committee, (iii) the Purchaser, and (iv) the U.S. Trustee, and (b) for any notice to be provided or service to be made as of a date on or after the Effective Date, (i) the Reorganized Debtors, (ii) the Distribution Trustee, (iii) the Purchaser, and (iv) the U.S. Trustee.

1.92 "Joint Liquidators" has the meaning set forth in the Restructuring Implementation Agreement.

1.93 "<u>Lien</u>" means, with respect to any asset or property (or the rents, revenues, income, profits or proceeds therefrom), and in each case, whether the same is consensual or nonconsensual or arises by contract, operation of law, legal process or otherwise: (a) any and all mortgages or hypothecation to secure payment of a debt or performance of an obligation, liens, pledges, attachments, charges, leases evidencing a capitalizable lease obligation, conditional sale or other title retention agreement, or other security interest or encumbrance or other legally cognizable security devices of any kind in respect of any asset or property, or upon the rents, revenues, income, profits or proceeds therefrom; or (b) any arrangement, express or implied, under which any asset or property is transferred, sequestered or otherwise identified for the purpose of subjecting or making available the same for the payment of debt or performance of any other obligation in priority to the payment of General <u>Unsecured Creditors</u>.

1.94 "<u>Mass Mutual</u>" means Massachusetts Mutual Life Insurance Company, MassMutual Capital Partners LLC, Benton Street Partners I, L.P., Benton Street Partners III, L.P., and Benton Street Partners III, L.P.

1.95 "<u>New Corporate Governance Documents</u>" means the corporate governance documents for the Reorganized Debtors and New Holdco (if applicable), including charters, bylaws, memoranda and article of association, operating agreements, or other organization or formation documents, as applicable, including, but not limited to, the Stockholders Agreement. For the avoidance of doubt, all New Corporate Governance Documents shall be in form and substance acceptable to the Purchaser in the Purchaser's sole discretion (other than that such New Corporate Governance Documents must not be materially inconsistent with the terms of the Plan).

1.96 "<u>New Equity</u>" means the voting capital stock of or membership interests in New Holdco.

1.97 "<u>New Equity Election</u>" means an election made by a New Equity Eligible Holder to take its Distribution in the form of Offered New Equity, which shall be made by checking the "New Equity Election" box on its timely completed and submitted Ballot. In the event that a New Equity Eligible Holder fails to make a New Equity Election or fails to submit a Ballot, then such New Equity Eligible Holder shall be deemed to have made the Cash Election.

1.98 "<u>New Equity Eligible Beneficial Holder</u>" means a Beneficial Holder of TruPS other than SRGL.

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 82 of 241

1.99 "<u>New Equity Eligible Holders</u>" means (i) all New Equity Eligible Beneficial Holders and (ii) the SFL Receiver with respect to the SFL Note Claim, the extent that the SFL Note Claim Allowance Conditions have been satisfied.

1.100 "<u>New Holdco</u>" means a newly organized Entity, shall have 100% of the New SALIC Equity as its primary asset and shall engage in no business other than to act as a holding company for such New SALIC Equity.

1.101 "<u>New SALIC Equity</u>" means 19,999,999 Ordinary Shares of SALIC to be issued under the terms of the Plan, the Restructuring Implementation Agreement, the Share Surrender Documents, the New SALIC Shares Issuance Documents, and the Stock Purchase Agreement.

1.102 "<u>New SHI Equity</u>" means 100% of the new common stock of SHI to be issued to, or at the direction of, the Purchaser.

1.103 "<u>New SALIC Shares Issuance Documents</u>" means each of the documents, directions and resolutions reasonably required of SALIC to effectuate the New SALIC Shares Issuance (as such term is defined in the Restructuring Implementation Agreement). The New SALIC Shares Issuance Documents shall be in form and substance reasonably satisfactory to the Purchaser, and otherwise subject to the SRGL Consent Rights.

1.104 "<u>Offered New Equity</u>" refers to the thirty percent (30%) of the New Equity to be offered to New Equity Eligible Holders pursuant to the Plan.

1.105 "<u>Official Committee</u>" means the Official Committee of Unsecured Creditors appointed by the U.S. Trustee in the Chapter 11 Cases pursuant to section 1102(a) of the Bankruptcy Code, as the membership of such committee is constituted and reconstituted from time to time.

1.106 "<u>Ordinary Shares</u>" shall have the meaning ascribed to such term in the Stock Purchase Agreement.

1.107 "Orkney Re II" means Orkney Re II plc.

1.108 "Person" means and includes a natural person, individual, partnership, corporation (as defined in section 101(a) of the Bankruptcy Code), or organization including, without limitation, corporations, limited partnerships, limited liability companies, general partnerships, joint ventures, joint stock companies, trusts, land trusts, business trusts, unincorporated organizations or associations, Official Committee, or ad hoc committee, or other organizations, irrespective of whether they are legal entities, governmental bodies (or any agency, instrumentality or political subdivision thereof), or any other form of legal entities; provided, however, the term "Person" does not include Governmental Units, except that a Governmental Unit that (a) acquires an asset from a Person (i) as a result of the operation of a loan guarantee agreement or (ii) as receiver or liquidating agent of a Person; (b) is a guarantor of a pension benefit payable by or on behalf of a Debtor or an Affiliate of a Debtor; or (c) is the legal or beneficial owner of an asset of (i) an employee pension benefit plan that is a governmental plan, as defined in section 414(d) of the Internal Revenue Code of 1986 or (ii) an eligible deferred

compensation plan, as defined in section 457(b) of the Internal Revenue Code of 1986, shall be considered for purposes of section 1102 of the Bankruptcy Code to be a Person with respect to such asset or such benefit.

1.109 "<u>Petition Date</u>" means January 28, 2018, the date on which each Debtor Filed its petition for relief commencing the Chapter 11 Cases.

1.110 "<u>Plan</u>" means the Third Amended Joint Chapter 11 Plan of Reorganization of Scottish Holdings, Inc., and Scottish Annuity & Life Insurance Company (Cayman) Ltd., as it may be altered, amended, modified or supplemented from time to time including in accordance with its terms, any Plan Supplement, the Bankruptcy Code or the Bankruptcy Rules.

1.111 "<u>Plan Documents</u>" means, collectively, those material documents executed or to be executed in order to consummate the transactions contemplated under the Plan, including without limitation the Plan Supplement, which shall be Filed with the Bankruptcy Court in advance of the Confirmation Hearing, which shall be in form and substance reasonably acceptable to the Purchaser.

1.112 "<u>Plan Funding Payment</u>" means an amount equal to twenty-one million five hundred thousand dollars (\$21,500,000), subject to downward adjustment in an amount equal to the TruPS Returned Cash, which shall be funded by Purchaser to the Reorganized SALIC which shall deliver such amount to the Distribution Trust (subject to the payments and reserves provided for in Sections 6.1(b)(1) through (5) of the Plan on the Effective Date as provided in the Stock Purchase Agreement and as a condition to occurrence of the Effective Date).

1.113 "<u>Plan Supplement</u>" means the supplement to the Plan to be Filed hereafter to supplement or clarify aspects of the Plan, and which shall include, among others: (a) the following documents, each of which shall be in form and substance acceptable to the Purchaser, (i) the Rejection Schedule, (ii) the New Corporate Governance Documents, (iii) the identity of any individual proposed to serve, after Confirmation of the Plan, as a director, officer, or voting trustee of the Reorganized Debtors, (iv) the identity of any insider that will be employed or retained by the Reorganized Debtors and the nature of any compensation for such insider, (v) a notice stating whether the New Equity will be issued by Reorganized SALIC or New Holdco, and (vi) the Stockholders Agreement; and (b) the Distribution Trust Agreement, which shall be in form and substance reasonably acceptable to the Debtors and the Purchaser; and (c) the identity of the Distribution Trustee (who shall be a Person selected by the Official Committee that is reasonably acceptable to the Debtors and the Purchaser) and the material terms of the Distribution Trustee's proposed compensation if not already set forth in the Distribution Trust Agreement.

1.114 "<u>Priority Claims</u>" means any and all Priority Tax Claims and Priority Non-Tax Claims.

1.115 "<u>Priority Non-Tax Claim</u>" means any and all Allowed Claims accorded priority in right of payment under section 507(a) of the Bankruptcy Code, other than a Priority Tax Claim.

1.116 "<u>Priority Tax Claim</u>" means any and all Claims of a Governmental Unit of the kind specified in section 507(a)(8) of the Bankruptcy Code.

1.117 "<u>Post-Effective Date Indenture Trustee Fee Reserve</u>" means the reserve to be established by the Distribution Trust in the following amounts: (a) the amount of any disputed Indenture Trustee Fees incurred through the Effective Date and (b) \$50,000 in aggregate for Indenture Trustee Fees incurred by the TruPS Indenture Trustees post-Effective Date with respect the SHST I TruPS, the SHST III TruPS, and the SFLST I TruPS as provided for in <u>Section 4.1(d)(ii)</u> of the Plan. For the avoidance of doubt, the Distribution Trustee shall be under no obligation to reserve any amount in the Post-Effective Date Indenture Trustee Fee Reserve on account of post-Effective Date Indenture Trustee Fees that may be incurred by the TruPS Indenture Trustees for the SHST II Debentures or the GPIC Debentures.

1.118 "<u>Professional</u>" means any professional employed in these Chapter 11 Cases pursuant to Bankruptcy Code sections 327, 328, or 1103, or for which compensation and reimbursement has been Allowed by the Bankruptcy Court pursuant to section 503(b)(4) of the Bankruptcy Code.

1.119 "<u>Professional Fee Claim</u>" means a Claim of a Professional for compensation for services rendered or reimbursement of costs, expenses, or other charges incurred after the Petition Date and on or before the Effective Date.

1.120 "<u>Professional Fee Reserve</u>" means a professional fee reserve to be maintained by the Distribution Trust in an amount mutually agreed by the Debtors, the Official Committee, and the Purchaser, and estimated in good faith, after consultation with the relevant Professionals, to be necessary to pay in full all amounts then owing or that may later become owing to such Professionals for professional fees and expenses incurred through the Effective Date.

1.121 "<u>Proof of Claim</u>" means a proof of claim Filed with the Bankruptcy Court or its duly appointed claims agent in connection with the Chapter 11 Cases pursuant to section 501 of the Bankruptcy Code.

1.122 "<u>Purchaser</u>" means Hildene Re Holdings, LLC, and its permitted designees, successors and assigns.

1.123 "<u>Recapitalization Funding Payment</u>" means an amount equal to twelve million five hundred thousand dollars (\$12,500,000), which shall be utilized by Reorganized SALIC (x) to recapitalize the SALIC Group Companies after Closing and (y) used by the Debtors to payCure Amounts on behalf of the Purchaser in excess of \$100,000 in accordance with section 2.2(b) of the Stock Purchase Agreement.

1.124 "<u>Reinstate</u>," "<u>Reinstated</u>" or "<u>Reinstatement</u>" means with respect to Claims and Interests, that the Claim or Interest shall be rendered unimpaired in accordance with section 1124 of the Bankruptcy Code.

1.125 "<u>Rejection Schedule</u>" means that certain schedule initially to be Filed as part of the Plan Supplement and identifying any Executory Contract or Unexpired Lease to be rejected by the Debtors effective as of the Effective Date. Except as otherwise provided in the Stock Purchase Agreement, the Rejection Schedule may be modified at any time prior to the Effective Date, including to add or remove Executory Contracts or Unexpired Leases to be rejected by the Debtors.

### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 85 of 241

1.126 "<u>Rejection Claim Bar Date</u>" means the date that is twenty-one (21) days after the Effective Date; *provided, however*, to the extent that the Bankruptcy Court grants a motion to reject an Executory Contract or Unexpired Lease Filed after the Effective Date as permitted by <u>Section 8.3(b)</u> of the Plan, the Rejection Claim Bar Date for such contract or lease shall be thirty (30) days after the date of entry of the Bankruptcy Court's order granting the post-Effective Date rejection of such contract or lease.

1.127 "<u>Released Parties</u>" means: (a) the Debtors; (b) the Reorganized Debtors; (c) the Official Committee (including the current and former Official Committee members in their capacities as such); (d) the Purchaser; (e) the Hildene Master Funds; (f) the Stalking Horse; (g) the Joint Liquidators; (h) SRGL; (i) the Voting Agent; and (j) for each of the foregoing, all of their respective Representatives; *provided, however*, that "Released Parties" specifically excludes (x) the SRGL Equity Holders and (y) the Specified SALIC Officers and Directors.

1.128 "<u>Reorganized Debtors</u>" means Reorganized SALIC and Reorganized SHI, and for the avoidance of doubt, includes New Holdco.

1.129 "<u>Reorganized SALIC</u>" means SALIC as reorganized upon the Effective Date pursuant to the Plan.

1.130 "<u>Reorganized SHI</u>" means SHI as reorganized upon the Effective Date pursuant to the Plan.

1.131 "<u>Representatives</u>" means, with respect to an Entity, all of that Entity's current and former managed and controlled affiliates, subsidiaries, officers, directors, managers, managing members, principals, shareholders, members, partners, employees, agents, advisors, attorneys, professionals, accountants, investment bankers, consultants and other representatives and such persons' respective heirs, executors, estates, servants and nominees, in each case in their capacity as such.

1.132 "<u>Restructuring Documents</u>" has the meaning set forth in the Restructuring Implementation Agreement.

1.133 "<u>Restructuring Implementation Agreement</u>" means that certain Restructuring Implementation Agreement among the Debtors and SRGL, made January 28, 2018, and approved by the RIA Order, as such Restructuring Implementation Agreement may be amended from time to time and as confirmed by SRGL in accordance with section 6.1(c) of the Stock Purchase Agreement. A copy of the Restructuring Implementation Agreement is annexed to the Plan as <u>Exhibit C-1</u> and the evidence of such confirmation is annexed to the Plan as <u>Exhibit C-1</u>.

1.134 "<u>Retained Causes of Action</u>" shall have the meaning set forth in <u>Section 6.5</u> of the Plan.

1.135 "<u>RIA Order</u>" means the *Order Authorizing Debtors To Assume Restructuring Implementation Agreement And Related Relief* entered by the Bankruptcy Court on March 19, 2018, at Docket No. 170 (as may be amended).

### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 86 of 241

1.136 "SALIC" means Scottish Annuity & Life Insurance Company (Cayman) Ltd.

1.137 "<u>SALIC Existing Equity Interests</u>" means all issued and outstanding Ordinary Shares of SALIC existing prior to the Effective Date.

1.138 "<u>SALIC General Unsecured Claim</u>" means any Claim against SALIC that is not an Administrative Claim, Secured Claim, Priority Claim, Intercompany Claim, TruPS Claim, SFL Claim, SFL Note Claim, or Subordinated Claim.

1.139 "SALIC Group Companies" means SALIC, SHI, SRD, SRUS and SRLB.

1.140 "<u>SALIC Group Services Claims</u>" has the meaning ascribed to such term in the Restructuring Implementation Agreement.

1.141 "<u>SALIC Net Worth Maintenance Agreements</u>" means (i) that certain Net Worth Maintenance Agreement, dated as of January 1, 2002, by and between SRD and SALIC, as amended from time to time, and (ii) that certain Net Worth Maintenance Agreement, restated as of February 1, 2002, by and between SRUS and SALIC, as amended from time to time.

1.142 "<u>SALIC TruPS Claim</u>" means, other than any Subordinated Claim, any and all TruPS Claims related to the TruPS Parent Guarantees, but does not include Indenture Trustee Fees.

1.143 "<u>SALIC Claims</u>" has the meaning as set forth in the Restructuring Implementation Agreement.

1.144 "<u>Schedules</u>" means the schedules of assets and liabilities, schedules of Executory Contracts and Unexpired Leases and statements of financial affairs Filed by the Debtors pursuant to section 521 of the Bankruptcy Code and in substantial accordance with the Official Bankruptcy Forms, as the same may have been amended, modified or supplemented from time to time.

1.145 "Scottish Re" means the Debtors collectively with their non-debtor affiliates.

1.146 "Secured Claim" means a Claim that is secured by a Lien which is not subject to avoidance under the Bankruptcy Code or otherwise invalid under the Bankruptcy Code or applicable state law, on property in which the Estate has an interest, or a Claim that is subject to setoff under section 553 of the Bankruptcy Code; to the extent of the value of the Holder's interest in the Estate's interest in such property or to the extent of the amount subject to setoff, as applicable; as determined by a Final Order pursuant to section 506(a) of the Bankruptcy Code, or in the case of setoff, pursuant to section 553 of the Bankruptcy Code, or in either case as otherwise agreed upon in writing by the Debtors or the Reorganized Debtors and the Holder of such Claim. The amount of any Claim that exceeds the value of the Holder's interest in the Estate's interest in property or the amount subject to setoff shall be treated as an SHI General Unsecured Claim or SALIC General Unsecured Claim, as applicable.

1.147 "Security Depository or Custodian" means a securities depository or custodian or any nominee for such a securities depository or custodian, *e.g.*, Depository Trust Company and its nominee Cede & Co.

1.148 "<u>SFL</u>" means Scottish Financial (Luxembourg) S.á r.l.

1.149 "SFL Bankruptcy Administration Senior Cash Election" means election exercisable by the SFL Receiver on or before the Voting Deadline applicable to the SFL Note Claim to designate that a portion of the SFL Note Claim should receive its Distribution in the form of Cash. The SFL Bankruptcy Administration Senior Cash Election is made available to the SFL Receiver to ensure that the SFL Receiver, notwithstanding any inconsistent New Equity Election having been made by one or more of the SFLST I TruPS CDO Facility Holders, will have sufficient Cash available to fulfill the SFL Receiver's duties as the insolvency receiver for SFL in its bankruptcy proceeding in Luxembourg. Once the initial SFL Bankruptcy Administration Senior Cash Election, it may not thereafter be modified, except with the prior written consent of each of the SFLST I TruPS CDO Facility Holders.

1.150 "<u>SFL Claims</u>" means any and all Claims and Causes of Action owned or held by SFL or the SFL Receiver on SFL's behalf against SALIC or SHI, including, but not limited, to the SFL Note Claim.

1.151 "<u>SFL Distribution Agreement</u>" means that certain agreement entered into between SFL and the Debtors on or prior to the Effective Date setting forth the mechanics for distributions made on account of the SFL Note Claim.

1.152 "<u>SFL Note</u>" means that certain Floating Rate Junior Subordinated Deferrable Interest Debenture of Scottish Re (Dublin) Limited, dated December 15, 2004, as amended, in the original principal amount of fifty-one million five hundred and forty-seven thousand dollars (\$51,547,000) issued by SRD, as obligor, to SFL, as obligee, and subsequently assigned to, and assumed by, SALIC, as obligor, as such note may have been amended from time to time.

1.153 "<u>SFL Note Claim</u>" means any and all Claims and Causes of Action against any of the Debtors arising out of or relating to the SFL Note, including, but not limited to, any amendment to the SFL Note or any payment made or not made on account of the SFL Note.

1.154 "SFL Note Claim Allowance Conditions" has the meaning ascribed to such term in Section 4.3(c)(ii)(B) of the Plan.

1.155 "<u>SFL Receiver</u>" means Max Mailliet, as insolvency receiver for SFL in its bankruptcy proceeding in Luxembourg or any successor thereto.

1.156 "<u>SFL Shares</u>" means the 47,046 shares of SFL, representing all of the issued and outstanding equity interest in SFL.

1.157 "<u>SFLST I TruPS</u>" has the meaning ascribed to such term in the definition of "TruPS".

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 88 of 241

1.158 "<u>SFLST I TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

1.159 <u>"SFLST I TruPS CDO Facility Holders</u>" means the Preferred Term Securities XVI, the Preferred Term Securities XVII, and the Preferred Term Securities XVIII collateraized debt obligation facilities, each as a Beneficial Holder of SFLST I TruPS, holding through BNYM, as nominee.

1.160 "Share Surrender Documents" has the meaning set forth in the Restructuring Implementation Agreement.

1.161 "SHI" means Scottish Holdings, Inc.

1.162 "<u>SHI Existing Equity Interests</u>" means all issued and outstanding common shares of SHI existing prior to the Effective Date.

1.163 "<u>SHI General Unsecured Claim</u>" means any Claim against SHI that is not an Administrative Claim, Secured Claim, Priority Claim, Intercompany Claim, TruPS Claim, SFL Claim, SFL Note Claim, or Subordinated Claim.

1.164 "<u>SHI TruPS Claim</u>" means, other than any Subordinated Claim, any and all TruPS Claims related to the SHST I TruPS, the SHST II TruPS, the GPIC TruPS, and the SHST III TruPS but does not include Indenture Trustee Fees.

1.165 "<u>SHST I TruPS</u>" has the meaning ascribed to such term in the definition of "<u>TruPS</u>".

1.166 "<u>SHST I TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

1.167 "<u>SHST II TruPS</u>" has the meaning ascribed to such term in the definition of "TruPS".

1.168 "<u>SHST II TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

1.169 "<u>SHST III TruPS</u>" has the meaning ascribed to such term in the definition of "TruPS".

1.170 "<u>SHST III TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

1.171 "<u>SLD</u>" means Security Life of Denver Insurance Company, a claimant in these Chapter 11 Cases.

1.172 "Specified Restructuring Documents" has the meaning set forth in the Restructuring Implementation Agreement.

1.173 "Specified SALIC Officers and Directors" means any former officer or director of SALIC who held said position at any point during the period January 1, 2011 through December 31, 2013, exclusive of Gregg Klingenberg and Mark Harris (who are each a Released Party).

1.174 "<u>SRD</u>" means Scottish Re (Dublin) dac.

1.175 "SRGL" means Scottish Re Group Limited.

1.176 "<u>SRGL Consent Rights</u>" has the meaning set forth in the Restructuring Implementation Agreement.

1.177 "<u>SRGL Equity Holders</u>" means each Person that (i) holds or at any time held any direct or indirect equity interest of any kind in SRGL, including without limitation Cerberus, MassMutual and their current and former affiliates, subsidiaries and assigns, in each case, holding or having held a direct or indirect equity interest of any kind in SRGL, or (ii) at any time received directly or indirectly any Transfer (or proceeds thereof) from any Debtors pursuant to, or on account of any such direct or indirect equity interest in SRGL, including without limitation any transfere of a kind referenced in section 550 of the Bankruptcy Code.

1.178 "<u>SRGL Exclusively Held SALIC TruPS Claims</u>" means all SALIC TruPS Claim arising from or relating to the SHST II TruPS or the GPIC TruPS, of which SRGL is the sole Beneficial Holder.

1.179 "<u>SRGL Exclusively Held SHI TruPS Claims</u>" means all SHI TruPS Claim arising from or relating to the SHST II TruPS or the GPIC TruPS, of which SRGL is the sole Beneficial Holder.

1.180 "<u>SRGL Exclusively Held TruPS Claims</u>" means all SRGL Exclusively Held SHI TruPS Claims and all SRGL Exclusively Held SALIC TruPS Claims.

1.181 "<u>SRGL Revolving Credit Agreement</u>" means that certain Revolving Credit Agreement, dated as of September 20, 2009, by and between SALIC, as lender, and SRGL, as borrower, as amended, novated, supplemented, extended, or restated from time to time.

1.182 "<u>SRGL Revolver Facility</u>" means the credit facility provided for under the SRGL Revolving Credit Agreement.

1.183 "<u>SRGL Revolver Facility Documents</u>" means the SRGL Revolving Credit Agreement, together with any related agreement, instrument, report or other document executed in connection therewith or otherwise evidencing Claims or other obligations arising thereunder, each as amended, novated, supplemented, extended, or restated from time to time.

1.184 "<u>SRGL TruPS Claims</u>" has the meaning set forth in the Restructuring Implementation Agreement.

1.185 "<u>SRLB</u>" means Scottish Re Life (Bermuda) Limited.

1.186 "<u>SRUS</u>" means Scottish Re (U.S.), Inc.

### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 90 of 241

1.187 "SRUS Retrocession Agreements" means all retrocession agreements to which SALIC and SRUS were party as of the Petition Date, including the following: (a) Coinsurance/Modified Coinsurance Reinsurance Agreement by and between SRUS (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocessionaire) and SRUS (as reinsurer), effective January 1, 2001; (b) Coinsurance Reinsurance Agreement by and between SRUS (as retrocedent) and SALIC (coinsurance Agreement) (as retrocessionaire), effective January 1, 2001; (c) Coinsurance Reinsurance Agreement by and between SRUS (as retrocedent) and SALIC (coinsurance Agreement) (as retrocessionaire), effective January 1, 2001; (c) Coinsurance Reinsurance Agreement by and between SRUS (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocessionaire), effective September 30, 2001; (d) Coinsurance Reinsurance Agreement by and between SRUS (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocessionaire), effective March 31, 2005; and (e) Modco Reinsurance Agreement by and between SRUS (as reinsurance Agreement by and SALIC (as retrocessionaire), as of December 31, 2015.

1.188 "Stalking Horse" means HSCM Bermuda Fund Ltd.

1.189 "<u>Stock Purchase Agreement</u>" means that certain Stock Purchase Agreement by and among SALIC, SHI and the Purchaser, dated as of June 11, 2018, and Filed with the Bankruptcy Court on June 11, 2018, at Docket No. 342, as such Stock Purchase Agreement may be amended from time to time. A copy of the Stock Purchase Agreement is annexed to the Plan as **Exhibit B**.

1.190 "<u>Stockholders Agreement</u>" means the agreement among (x) New Holdco and the Reorganized Debtors, on the one hand, and (y) each holder of New Equity, on the other hand, whether as of the Effective Date or subsequent thereto, to be set forth in the Plan Supplement.

1.191 "<u>Subordinated Claims</u>" means any Claims arising under section 510(b) of the Bankruptcy Code or other Claims that are subordinated to general unsecured claims under the Bankruptcy Code; *provided, however,* for the avoidance of doubt, that the SHI TruPS Claims, the SALIC TruPS Claims and the SFL Note Claim shall not be classified or treated as Subordinated Claims for purposes of the Plan.

1.192 "<u>Third-Party Reinsurance Agreements</u>" means all reinsurance type agreements to which SALIC and any third-party unaffiliated insurer were party as of the Petition Date, including the following: (a) Coinsurance Agreement, dated as of July 28, 1999, by and between Highmark Life Insurance Co., as ceding company, and SALIC, as reinsurer; (b) Coinsurance Agreement, dated as of October 21, 1999, by and between Highmark Life Insurance Co., as ceding company, and SALIC, as reinsurer; (c) Coinsurance Agreement, dated as of April 1, 2000, by and between Investors Heritage Life Insurance Co., as ceding company, and SALIC, as reinsurer; (d) Coinsurance Agreement, dated as of April 1, 2000, by and between Lincoln Heritage Life Insurance Co., as ceding company, and SALIC, as reinsurer; (e) Modco Agreement, dated as of February 15, 2000, by and between Lincoln National Life Ins. Co., as ceding company, and SALIC, as reinsurer; and (f) Automatic Monthly Renewable Term Reinsurance Agreement, dated as of June 1, 2016, between C.M. Life Insurance Co. and Mass Mutual Life Insurance Co., each a ceding insurer, and SALIC, as reinsurer.

### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 91 of 241

1.193 "<u>Transfer</u>" means to sell, resell, reallocate, use, pledge, assign, transfer, hypothecate, participate, donate, or otherwise encumber or dispose of, directly or indirectly (including through derivatives, options, swaps, pledges, forward sales, or other transactions); provided, however, that holding Certificates in an account with a broker-dealer where the brokerdealer holds a security interest or other encumbrance over property in the account generally, which security interest or other encumbrance is released upon transfer of such securities, shall not constitute a "Transfer" under the Plan.

1.194 "<u>TruPS</u>" means:

(i) the undivided beneficial interests, having an aggregate liquidation amount of \$17,500,000.00, in Scottish Holdings Statutory Trust I, a Connecticut statutory trust ("<u>SHST I</u>"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of December 4, 2002, among State Street Bank and Trust Company of Connecticut, National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>SHST I TruPS</u>");

(ii) the undivided beneficial interests, having an aggregate liquidation amount of \$20,000,000.00, in Scottish Holdings, Inc. Statutory Trust II, a Connecticut statutory trust ("<u>SHST II</u>"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of October 29, 2003, among U.S. Bank National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>SHST II</u> <u>TruPS</u>");

(iii) the undivided preferred beneficial interests, having an aggregate liquidation amount of \$10,000,000.00, in GPIC Holdings Inc. Statutory Trust, a Delaware statutory trust ("<u>GPIC</u>"), issued pursuant to that certain Amended and Restated Trust Agreement dated as of November 14, 2003, among JPMorgan Chase Bank as property trustee, Chase Manhattan Bank USA, National Association as Delaware trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>GPIC TruPS</u>");

(iv) the undivided beneficial interests, having an aggregate liquidation amount of \$32,000,000.00, in Scottish Holdings, Inc. Statutory Trust III, a Connecticut statutory trust ("<u>SHST III</u>"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of May 12, 2004, among U.S. Bank National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>SHST III</u> <u>TruPS</u>"); and

(v) the undivided beneficial interests, having an aggregate liquidation amount of \$50,000,000.00, in SFL Statutory Trust I, a Delaware statutory trust ("<u>SFLST I</u>," and collectively with SHST I, SHST II, GPIC, and SHST III, the "<u>TruPS Trusts</u>"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of December 15, 2004, among Wilmington Trust Company, as institutional trustee, the administrators of the issuer

named therein, SFL as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>SFLST I TruPS</u>").

1.195 "<u>TruPS Claims</u>" means Claims of any Person relating to or arising out of any TruPS, TruPS Junior Subordinated Debentures or TruPS Documents, including any Claims relating to or arising out of any TruPS Documents. For the avoidance of doubt, TruPS Claims shall include all of the Claims set forth in the preceding sentence that could be asserted by one or more of the several parties thereto without duplication. When referring to TruPS Claims arising out of a particular TruPS issuance, the words "TruPS Claims" will be preceded by the name of the applicable TruPS issuance (*e.g.*, "SHST I TruPS Claims").

1.196 "<u>TruPS Claims Aggregate Amount</u>" means (i) all Allowed SHI TruPS Claims, (ii) all Allowed SALIC TruPS Claims, and (iii) \$63,536,014.32 on account of the SFL Note Claim.

1.197 "<u>TruPS Claims Equity Distribution Amount</u>" means (i) with respect to a Beneficial Holder of TruPS, the amount of the Offered New Equity to be distributed to a Beneficial Holder that elects to receive New Equity under Section 4.3 of the Plan, calculated based on such Beneficial Holder's Allocated Portion of the Allowed TruPS Claim in respect of which it is a Beneficial Holder divided by the TruPS Claims Aggregate Amount; and (ii) with respect to the Holder of the SFL Note Claim, for which the SFL Claims Allowance Conditions have been satisfied or deemed satisfied, the amount of the Offered New Equity to be distributed on account of the Allowed SFL Note Claim, if the Holder thereof elects or is deemed to have elected to receive New Equity under section 4.3 of the Plan, calculated based on the portion of the Allowed SFL Note Claim for which the New Equity Election has been made or has been deemed to have been made divided by the TruPS Claims Aggregate Amount.

1.198 "<u>TruPS Debentures</u>" means:

(i) with respect to the SHST I TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Holdings, Inc., and State Street Bank and Trust Company of Connecticut, National Association, due 2032 (the "<u>SHST I TruPS</u> <u>Debentures</u>");

(ii) with respect to the SHST II TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Holdings, Inc., and U.S. Bank National Association, due 2033 (the "<u>SHST II TruPS Debentures</u>");

(iii) with respect to the GPIC TruPS, that certain Floating Rate Junior Subordinated Note Due 2033 between Scottish Holdings, Inc., and JPMorgan Chase Bank (the "<u>GPIC TruPS Debentures</u>");

(iv) with respect to the SHST III TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Holdings, Inc., and U.S. Bank National Association, due 2034 (the "<u>SHST III TruPS Debentures</u>"); and

(v) with respect to the SFLST I TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Financial (Luxembourg) S.à r.l. and Wilmington Trust Company, due 2034 (the "<u>SFLST I TruPS Debentures</u>").

## 1.199 "TruPS Declarations" means:

(i) with respect to the SHST I TruPS, that certain Amended and Restated Declaration of Trust by and among State Street Bank and Trust Company of Connecticut, National Association, as Institutional Trustee, Scottish Holdings, Inc., as Sponsor, and Paul Goldean and Oscar R. Scofield, as Administrators, dated as of December 4, 2002 (the "<u>SHST I Trust Declaration</u>").

(ii) with respect to the SHST II TruPS, that certain Amended and Restated Declaration of Trust by and among U.S. Bank National Association, as Institutional Trustee, Scottish Holdings, Inc., as Sponsor, and Paul Goldean and Oscar R. Scofield, as Administrators, dated as of October 29, 2003 (the "<u>SHST II Trust Declaration</u>").

(iii) with respect to the GPIC TruPS, that certain Amended and Restated Declaration of Trust by and among Scottish Holdings, Inc., as Depositor, JPMorgan Chase Bank, as Property Trustee, Chase Manhattan Bank USA, National Association, as Delaware Trustee, and The Administrative Trustees Named [T]Herein, as Administrative Trustees, dated as of November 14, 2003 (the "<u>GPIC Trust Declaration</u>").

(iv) with respect to the SHST III TruPS, that certain Amended and Restated Declaration of Trust by and among U.S. Bank National Association, as Institutional Trustee, Scottish Holdings, Inc., as Sponsor, and Paul Goldean and Oscar R. Scofield, as Administrators, dated as of May 12, 2004 (the "<u>SHST III Trust Declaration</u>").

(v) with respect to the SFLST I TruPS, that certain Amended and Restated Declaration of Trust by and among Wilmington Trust Company, as Institutional Trustee, Wilmington Trust Company, as Delaware Trustee, Scottish Financial (Luxembourg) S.à r.l., as Sponsor, and Paul Goldean and George Scott, as Administrators, dated as of December 15, 2004 (the "SFLST Trust Declaration").

1.200 "<u>TruPS Documents</u>" means collectively all TruPS Indentures, all documents evidencing TruPS Debentures, all TruPS Declarations, all TruPS Sponsor Guarantees, all TruPS Parent Guarantees and all related and ancillary documents and instruments, each as altered, amended, modified or supplemented from time to time and including all exhibits and schedules thereto.

1.201 "<u>TruPS/GUC Claims Aggregate Amount</u>" means (i) all Allowed SHI TruPS Claims, (ii) all Allowed SALIC TruPS Claims, (iii) \$63,536,014.32 on account of the SFL Note Claim. (iv) all Allowed SHI General Unsecured Claims, (v) all Allowed SALIC General Unsecured Claims, and (vi) the Disputed Claim Reserve Amount on account of all Disputed General Unsecured Claims.

1.202 "<u>TruPS/GUC Claims Cash Distribution Amount</u>" means the pro rata amount of the Available Plan Distribution Funding Amount to be distributed to a Beneficial Holder, Holder of the SFL Note Claim or Holder of a General Unsecured Claim, as applicable, calculated based on: (i) with respect to a Beneficial Holder of TruPS, such Beneficial Holder's Allocated Portion of the Allowed TruPS Claim in respect of which it is a Beneficial Holder, divided by the TruPS/GUC Claims Aggregate Amount; (ii) with respect to the Allowed SFL Note Claim, the

portion of such Holder's Allowed SFL Note Claim for which the Cash Election has been made (or deemed made), divided by the TruPS/GUC Claims Aggregate Amount; and (iii) with respect to an Allowed General Unsecured Claim, the Allowed amount of such General Unsecured Claim, divided by the TruPS/GUC Claims Aggregate Amount.

1.203 "<u>TruPS Indentures</u>" means:

(i) with respect to the SHST I TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2032, between Scottish Holdings, Inc. as Issuer, and State Street Bank and Trust Company of Connecticut, National Association, as Trustee, dated as of December 4, 2002;

(ii) with respect to the SHST II TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2033, between Scottish Holdings, Inc., as Issuer, and U.S. Bank National Association, as Trustee, dated as of October 29, 2003;

(iii) with respect to the GPIC TruPS, that certain Junior Subordinated Indenture, due 2033, between Scottish Holdings, Inc., as Issuer, and JPMorgan Chase Bank, as Trustee, dated as of November 14, 2003;

(iv) with respect to the SHST III TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2034, between Scottish Holdings, Inc., as Issuer, and U.S. Bank National Association, as Trustee, dated as of May 12, 2004; and

(v) with respect to the SFLST I TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2034, between Scottish Financial (Luxembourg) S.à r.l., as Issuer, and Wilmington Trust Company, as Trustee, dated as of December 15, 2004.

1.204 "<u>TruPS Indenture Trustee</u>" means an Entity serving as an indenture trustee under a TruPS Indenture during the period from the Petition Date through the Effective Date.

1.205 "<u>TruPS Institutional Trustee</u>" means an Entity serving as the trustee pursuant to a TruPS Declaration, a TruPS Parent Guarantee or a TruPS Sponsor Guarantee.

1.206 "TruPS Parent Guarantees" means:

(i) with respect to the SHST I TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and State Street Bank and Trust Company of Connecticut, National Association, dated as of December 4, 2002;

(ii) with respect to the SHST II TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and U.S. Bank National Association, dated as of October 29, 2003;

(iii) with respect to the GPIC TruPS, that certain Parent Guarantee Agreement

by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and JPMorgan Chase Bank, dated as of November 14, 2003;

(iv) with respect to the SHST III TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and U.S. Bank National Association, dated as of May 12, 2004; and

(v) with respect to the SFLST I TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and Wilmington Trust Company, dated as of December 15, 2004.

1.207 "<u>TruPS Returned Cash</u>" means the amount of the Available Plan Distribution Funding Amount that is not required to be funded by the Purchaser or is returned to Purchaser on account of the Beneficial Holders of TruPS that make the New Equity Election, calculated based on the aggregate Beneficial Holders who made, or are deemed to make, the New Equity Election, divided by the TruPS/GUC Claims Aggregate Amount.

1.208 "<u>TruPS Shares</u>" means the Offered New Equity. Any reference to the TruPS Shares in the Stock Purchase Agreement shall mean the Offered New Equity.

1.209 "TruPS Sponsor Guarantees" means:

(i) with respect to the SHST I TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and State Street Bank and Trust Company of Connecticut, National Association, dated as of December 4, 2002;

(ii) with respect to the SHST II TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and U.S. Bank National Association, dated as of October 29, 2003;

(iii) with respect to the GPIC TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and JPMorgan Chase Bank, dated as of November 14, 2003;

(iv) with respect to SHST III TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and U.S. Bank National Association, dated as of May 12, 2004; and

(v) with respect to the SFLST I TruPS, that certain Guarantee Agreement by and between Scottish Financial (Luxembourg) S.à r.l and Wilmington Trust Company, dated as of December 15, 2004.

1.210 "<u>Trust Administration Reserve</u>" means a reserve to be maintained by the Distribution Trust in an amount, mutually agreed by the Debtors, the Official Committee, and the Purchaser, estimated in good faith to be necessary to cover the costs of administration of the Distribution Trust, as further defined in Section 6.3(f)(iii) of the Plan.

1.211 "<u>Trust Administration Reserve Cap</u>" means the cap on the aggregate amount to be funded into the Trust Administration Reserve, which shall be mutually acceptable to the Debtors,

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 96 of 241

the Purchaser and the Official Committee. The Trust Administration Reserve Cap shall be disclosed at or before the commencement of the Confirmation Hearing.

1.212 "Trust Agreements" means all trust agreements relating to Third-Party Reinsurance Agreements to which SALIC, as grantor, and any ceding company were party as of the Petition Date, including the following: (a) Trust Agreement dated as of July 8, 1999, by and among SALIC, as grantor, Highmark Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (b) Trust Agreement dated as of October 21, 1999, by and among SALIC, as grantor, Highmark Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (c) Trust Agreement dated as of April 1, 2000, by and among SALIC, as grantor, Investors Heritage Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (d) Trust Agreement dated as of April 1, 2000, by and among SALIC, as grantor, Lincoln Heritage Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (e) Trust Agreement dated as of July 12, 2016, by and among SALIC, as grantor, C.M. Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (f) Trust Agreement dated as of July 12, 2016, by and among SALIC, as grantor, Massachusetts Mutual Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; and (g) Security Trust Agreement dated as of July 12, 2016, by and among SALIC, as grantor, Massachusetts Mutual Life Insurance Company and C.M. Life Insurance Company, each as beneficiary, and The Bank of New York Mellon, as trustee.

1.213 "<u>Unexpired Lease</u>" means a lease of non-residential real property to which any Debtor is a party that is subject to assumption or rejection under section 365 of the Bankruptcy Code.

1.214 "<u>Unimpaired</u>" means, when used with reference to a Claim or Interest, a Claim or Interest that is not impaired within the meaning of section 1124 of the Bankruptcy Code.

1.215 "<u>Unimpaired Class</u>" means a Class of Claims that are not impaired within the meaning of section 1124 of the Bankruptcy Code.

1.216 "U.S. Bank" means U.S. Bank National Association, as trustee.

1.217 "<u>U.S. Trustee</u>" means the Office of the United States Trustee for the District of Delaware.

1.218 "<u>Voting Agent</u>" means Prime Clerk, LLC, the Entity the Bankruptcy Court has authorized to serve as the "Voting Agent" pursuant to the Disclosure Statement Order.

1.219 "Voting Deadline" means August 13, 2018 at 4:00 p.m. (Eastern Time), which date and time has been established by the Bankruptcy Court pursuant to the Disclosure Statement Order as the deadline by which all Ballots to accept or reject the Plan must be received in order to be counted for purposes of section 1126 of the Bankruptcy Code; *provided, however*, that the Voting Deadline for the SFL Receiver as the Holder of the SFL Note Claim shall be the later of (x) August 13, 2018 at 4:00 p.m. (Eastern Time), and (y) two (2) Business Days after the Debtors or the Voting Agent has provided counsel for the SFL Receiver with notice in writing (which may take the form of an email to the SFL Receiver's counsel) of the outcome of voting on the Plan by the SFLST I TruPS CDO Facility Holders.

1.220 "<u>WTC</u>" means Wilmington Trust Company, as trustee.

# <u>Exhibit B</u>

**Stock Purchase Agreement** 

# **EXECUTION VERSION**

## STOCK PURCHASE AGREEMENT

#### **BY AND AMONG**

## SCOTTISH ANNUITY & LIFE INSURANCE COMPANY (CAYMAN) LTD.,

# SCOTTISH HOLDINGS, INC.

#### AND

## HILDENE RE HOLDINGS, LLC

## DATED AS OF JUNE 11, 2018

# TABLE OF CONTENTS

# Page

ARTICLE I DEFINITIONS AND TERMS					
Section 1.1	Definitions	3			
Section 1.2	Interpretation				
ARTICLE II ISSUANCE AND PURCHASE OF ORDINARY SHARES					
Section 2.1	Deposit	23			
Section 2.2	Purchase of Ordinary Shares				
Section 2.3	Assumption of Contracts				
Section 2.4	Closing				
Section 2.5	Withholding				
ARTICLE III REPRESENTATIONS AND WARRANTIES OF SALIC					
Section 3.1	Organization and Authority of SALIC				
Section 3.2 Section 3.3	Binding Effect Organization and Authority of the SALIC Subsidiaries				
Section 3.4	•				
	Capital Structure; Ownership of the SALIC Group Subsidiaries				
Section 3.5	Governmental Filings and Consents				
Section 3.6 Section 3.7	No Violations				
Section 3.7	Financial and Statutory Statements; No Undisclosed Liabilities				
Section 3.9	Absence of Certain Changes Litigation; Orders				
Section 3.10	•				
Section 3.11	Taxes.				
	Employee Benefits	33			
Section 3.12	Compliance with Applicable Laws; Governmental	26			
Section 2.12	Authorizations; Regulatory Filings				
Section 3.13	Intellectual Property				
Section 3.14	Actuarial Reports; Reserves				
Section 3.15	Insurance				
Section 3.16	Material Contracts				
Section 3.17	Assets; Real Property				
Section 3.18 Section 3.19	Finders' Fees				
	Investment Assets				
Section 3.20	Labor Matters.				
Section 3.21	Reinsurance				
Section 3.22	Affiliate Transactions				
Section 3.23	Environmental Matters				
Section 3.24	Information Technology, Data Security and Privacy				
Section 3.25	Third-Party Administrators				
Section 3.26 Section 3.27	Orders NO OTHER REPRESENTATIONS OR WARRANTIES				
ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PURCHASER					
Section 4.1	Organization and Authority	45			
Section 4.2	Binding Effect				

# **TABLE OF CONTENTS**

(continued)

## Page

Section 4.3	Governmental Filings and Consents	46
Section 4.4	No Violations	46
Section 4.5	Purchaser Impediments	46
Section 4.6	Finders' Fees	47
Section 4.7	Financial Capability	47
Section 4.8	Purchase for Own Account	47
Section 4.9	NO OTHER REPRESENTATIONS OR WARRANTIES	47
ARTICLE V COVENA	NTS	47
Section 5.1	Access; Confidentiality	47
Section 5.2	Conduct of Business	48
Section 5.3	Reasonable Best Efforts; Regulatory Matters; Third-Party Consents	
Section 5.4	Employee Matters.	
Section 5.5	Further Assurances	
Section 5.6	Resignations	
Section 5.7	Insurance	
Section 5.8	D&O Liabilities	56
Section 5.9	Orkney Re II	57
Section 5.10	Abandonment of SFL Capital Stock	
Section 5.11	Notification of Certain Matters	
Section 5.12	Subsequent Financial Statements	58
Section 5.13	Terminating Intercompany Agreements	
Section 5.14	Scottish Re Trademarks	
ARTICLE VI BANKRUPTCY MATTERS		
Section 6.1	Bankruptcy Court Filings and Approvals	59
ARTICLE VII CONDIT	TONS TO CLOSING	62
Section 7.1	Conditions to the Obligations of Purchaser and SALIC	62
Section 7.2	Conditions to the Obligations of Purchaser	
Section 7.3	Conditions to the Obligations of SALIC	65
ARTICLE VIII TERMINATION		
Section 8.1	Termination	65
Section 8.2	Effect of Termination	
Section 8.3	Fees and Expenses	
ARTICLE IX MISCELI	LANEOUS	
Section 9.1	No Survival of Representations and Warranties.	69
Section 9.2	Notices	
Section 9.2 Section 9.3	Amendment; Modification and Waiver	
Section 9.4	Assignment.	
		/ 1

# **TABLE OF CONTENTS**

(continued)

## Page

Section 9.5	Entire Agreement	
Section 9.6	No Third Party Beneficiaries	
Section 9.7	Public Disclosure	
Section 9.8	Disclosure Schedules	
Section 9.9	Governing Applicable Law and Jurisdiction	
Section 9.10	WAIVER OF JURY TRIAL	
Section 9.11	Counterparts	
Section 9.12	Severability	
Section 9.13	Specific Performance	
Section 9.14	Non-Recourse	

<u>Exhibits</u>

Exhibit A Plan Term Sheet

Schedules

SALIC Disclosure Schedule Purchaser Disclosure Schedule This STOCK PURCHASE AGREEMENT (this "<u>Agreement</u>") is dated as of June 11, 2018 (the "<u>Agreement Date</u>"), by and among Scottish Annuity & Life Insurance Company (Cayman) Ltd., an exempted company limited by shares organized and existing under the laws of the Cayman Islands ("<u>SALIC</u>"), Scottish Holdings, Inc., a corporation organized and existing under the laws of the state of Delaware ("<u>SHI</u>"), and Hildene Re Holdings, LLC, a Delaware limited liability company ("<u>Purchaser</u>").

#### RECITALS

WHEREAS, SALIC is engaged, through itself and the Insurance Company Subsidiaries, in the reinsurance of, among other products, accident and health insurance, disability insurance, life insurance, annuities and annuity-type products (the "<u>SALIC Group Business</u>");

WHEREAS, SALIC is a wholly owned direct subsidiary of Scottish Re Group Limited ("<u>SRGL</u>"), an exempted company limited by shares incorporated and existing under the laws of the Cayman Islands;

WHEREAS, on January 28, 2018, SALIC and SHI filed voluntary petitions for relief (each a "<u>Chapter 11 Case</u>," and collectively, the "<u>Chapter 11 Cases</u>") under Chapter 11 of the United States Bankruptcy Code (the "<u>Bankruptcy Code</u>") in the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>");

WHEREAS, on May 17, 2017, SRGL (a) commenced a winding up proceeding (the "<u>Bermuda Winding Up Proceeding</u>") before the Supreme Court of Bermuda (the "<u>Bermuda Court</u>") by filing a petition (the "<u>Bermuda Winding Up Petition</u>") for its winding up pursuant to Part XIII of the Bermuda Companies Act 1981 (the "<u>Bermuda Act</u>") and (b) commenced a parallel winding up proceeding (the "<u>Cayman Islands Winding Up Proceeding</u>," and together with the Bermuda Winding Up Proceeding, the "<u>Winding Up Proceedings</u>") before the Grand Court of the Cayman Islands, Financial Services Division (the "<u>Cayman Islands Court</u>") by filing a petition (the "<u>Cayman Islands Winding Up Petition</u>," and together with the Bermuda Winding Up Petition, "Islands Winding Up Petition," and together with the Bermuda Winding Up Petition," and together with the Bermuda Winding Up Petition, "Islands Winding Up Petition," and together with the Bermuda Winding Up Petition, "Islands Winding Up Petition, "Islands Islands Islands (The "<u>Winding Up Petition</u>") by filing a petition, the "<u>Winding Up Petitions</u>") for its winding up under Cayman Islands law;

WHEREAS, simultaneous with SRGL's filing of the Bermuda Winding Up Petition, SRGL filed an application to the Bermuda Court for the appointment of joint provisional liquidators in connection with the Bermuda Winding Up Proceeding;

WHEREAS, on May 18, 2017, the Bermuda Court appointed John C. McKenna of Finance & Risk Services Ltd. and Eleanor Fisher of Kalo (Cayman) Limited as joint provisional liquidators (with limited powers) for SRGL in connection with the Bermuda Winding Up Proceeding (the "Joint Provisional Liquidators");

WHEREAS, on January 30, 2018, the Bermuda Court entered a winding up Order for SRGL and expanded the powers of the Joint Provisional Liquidators under the Bermuda Act such that they became full powers liquidators in the Bermuda Winding Up Proceeding (the "<u>Bermuda Full Powers Liquidators</u>"), and on May 18, 2018 the Cayman Islands Court entered a winding up Order and appointed Mr. McKenna and Ms. Fisher as full powers liquidators in the Cayman Islands Winding Up Proceeding (the "<u>Cayman Islands Full Powers Liquidators</u>," and together with the Bermuda Full Powers Liquidators, the "<u>Full Powers Liquidators</u>");

WHEREAS, SALIC is authorized to issue 20,000,000 ordinary shares, par value \$0.001 (the "<u>Ordinary Shares</u>"), of which 20,000,000 Ordinary Shares are outstanding;

WHEREAS, the Board of Directors of SALIC has approved the execution of this Agreement and the consummation of the Transactions;

WHEREAS, on January 28, 2018, SRGL, SHI and SALIC entered into the Restructuring Implementation Agreement (as defined below), pursuant to which, among other things, the parties thereto have agreed, subject in all respects to the terms and conditions set forth in the Restructuring Implementation Agreement, (a) that SRGL will provide SALIC with an irrevocable letter of surrender for all but one of the Ordinary Shares held by SRGL and (b) consent to the issuance of the Purchaser Shares to Purchaser upon the terms and subject to the conditions set forth in the Restructuring Implementation Agreement and this Agreement will be granted, and the Joint Provisional Liquidators have reviewed the Restructuring Implementation Agreement and support the actions thereunder as representing the best plan available in the best interests of SRGL's general body of unsecured creditors;

WHEREAS, on January 28, 2018, SALIC, SHI and HSCM Bermuda Fund Ltd. ("<u>HSCM</u>") entered into that certain Stock Purchase Agreement (the "<u>Stalking Horse SPA</u>") providing for, among other things, the purchase by HSCM of 19,999,999,999 Ordinary Shares of SALIC;

WHEREAS, on January 31, 2018 the Debtors filed with the Bankruptcy Court a motion seeking (a) the approval and assumption of the Restructuring Implementation Agreement pursuant to Section 365 of the Bankruptcy Code; (b) the approval and assumption of the Plan Sponsorship Agreement (as defined below) pursuant to Section 365 of the Bankruptcy Code; (c) the approval of the Bidding Procedures (including the Break-Up Fee and Expense Reimbursement Amount) (each, as defined below) and scheduling certain dates, deadlines and forms of notice in connection therewith; and (d) the grant of other related relief (the "<u>Bidding Procedures Motion</u>"), which motion was approved by Order of the Bankruptcy Court dated February 28, 2018 (the "<u>Bidding Procedures Order</u>");

WHEREAS, at the conclusion of the Auction (as defined below) held on May 30, 2018, the Debtors declared Purchaser to be the Winning Bidder (as defined below) and HSCM to be the Back-Up Bidder in accordance with the Bidding Procedures;

WHEREAS, Purchaser wishes to purchase, and SALIC wishes to issue and sell, that number of Ordinary Shares of SALIC equal to 19,999,999,999 Ordinary Shares less the TruPS Shares (as defined below) (the "<u>Purchaser Shares</u>"), upon the terms and conditions stated herein, in the Chapter 11 Plan (as defined below); and

WHEREAS, in connection with the issuance and sale to Purchaser of the Purchaser Shares, (a) SALIC will issue to the TruPS Holders the TruPS Shares upon the terms and conditions stated in the Chapter 11 Plan and (b) SRGL will surrender to SALIC for cancellation the SRGL Shares (as defined below) as described, and on the terms and conditions set forth, in this Agreement and the Chapter 11 Plan.

NOW, THEREFORE, in consideration of the mutual covenants and agreements contained herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and upon the terms and subject to the conditions set forth herein, the parties hereto hereby agree as follows:

#### **ARTICLE I**

#### **DEFINITIONS AND TERMS**

Section 1.1 <u>Definitions</u>. As used in this Agreement, the following terms have the meanings set forth or as referenced below:

"<u>Action</u>" means any claim, action, suit, litigation, audit, assessment, arbitration, investigation, inquiry, hearing, investigation, charge, complaint, demand, notice or proceeding of any nature, whether civil, criminal, administrative, regulatory, investigative or otherwise, in law or in equity.

"<u>Administrative Expense Claim</u>" means any right to payment constituting a cost or expense of administration of the Chapter 11 Cases as it relates to the Debtors under Sections 503(b) and 507(a)(2) of the Bankruptcy Code including, claims under Section 503(b)(9) of the Bankruptcy Code, any actual and necessary costs and expenses of preserving the Debtors' estates, any actual and necessary costs and expenses of operating either Debtor's business, any indebtedness or obligations incurred by either Debtor after the Petition Date in connection with the conduct of their businesses, all compensation and reimbursement of expenses awarded or otherwise approved for payment by Final Order of the Bankruptcy Court under Section 330, 503(b) or 1129(a)(4) of the Bankruptcy Code, any fees or charges assessed against either Debtor's estate under Section 1930 of chapter 123 of title 28 of the United States Code, all wages, salaries and health and other benefits on account of services rendered after the Petition Date taxes, and all other claims entitled to administrative expense status pursuant to a Final Order of the Bankruptcy Court, in each case relating to the period from the Petition Date to the Effective Date but not beyond (but excluding the Break-Up Fee, Expense Reimbursement Amount and any Intercompany Claims).

"<u>Affiliate</u>" means, with respect to any specified Person, any other Person that, at the time of determination, directly or indirectly through one or more intermediaries, Controls, is Controlled by, or is under common Control with, such first Person.

"<u>Agreement</u>" has the meaning set forth in the Preamble.

"<u>Agreement Date</u>" has the meaning set forth in the Preamble.

"<u>Allowed</u>" means, in reference to a Claim against a Debtor, a Claim that is an "Allowed Claim" pursuant to the terms of the Chapter 11 Plan or that otherwise has been allowed against such Debtor by a Final Order of the Bankruptcy Court.

"<u>Alternative Transaction</u>" mean any transaction or transactions pursuant to which any Person or group of Persons other than Purchaser and its Affiliates acquire, directly or indirectly,
all or any part (directly or through a proposed investment in equity securities, debt securities or claims of creditors) of the business, assets or properties, capital stock or capital stock equivalents of SRGL or any SALIC Group Company, whether by merger, recapitalization, reinsurance, share exchange, stock purchase (including a rights offering with respect to such entity's securities), purchase of assets, tender offer, debt-for-equity exchange, distribution of securities for the benefit of the stockholders of such entity, consolidation or similar transaction, or Debt Financing; <u>provided</u>, <u>however</u>, that in no event shall an "Alternative Transaction" include any transaction (including any reinsurance or retrocession transaction) undertaken by any SALIC Group Company in the Ordinary Course of Business.

"<u>Amended RIA Order</u>" means an Order of the Bankruptcy Court granting such relief as is necessary for the Cayman Islands Full Powers Liquidators on behalf of SRGL to provide the RIA Winning Bidder Confirmation, in form and substance reasonably acceptable to Purchaser.

"<u>Ancillary Agreements</u>" means the other agreements and instruments executed and delivered in connection with this Agreement.

"<u>Applicable Bermuda Law</u>" means all statutes, uncodified laws, regulations and rules of Bermuda (including the Bermuda Act and Orders of the Bermuda Court) that may be applicable to any SRGL Group Company or SALIC Group Company.

"<u>Applicable Cayman Islands Law</u>" means all statutes, uncodified laws, regulations and rules of the Cayman Islands (including the Companies Winding Up Rules, the Insolvency Practitioners' Regulations, the Foreign Bankruptcy Proceedings Rules (International Cooperation Rules), CIMA regulations and agency rulings and Orders of the Cayman Islands Court) that may be applicable to any SRGL Group Company or SALIC Group Company.

"<u>Applicable Law</u>" means any foreign, domestic, state, federal, national, international, multinational, regional or local law, statute, ordinance, rule, regulation, writ, directive, treaty, Order, judgment, decree, injunction or other legally binding obligation imposed by or on behalf of a Governmental Authority.

"<u>Assumed Contracts</u>" means all Contracts (a) that are Executory Contracts, (b) to which one or both of SALIC or SHI is party, and (c) that are to be assumed by the applicable Debtor pursuant to section 365 or 1123 of the Bankruptcy Code (i) by reason of their inclusion as Material Contracts on Section 3.16(a) of the SALIC Disclosure Schedule, but subject to Purchaser's rights under Section 2.3(c) and Section 2.3(f), (ii) by reason of Purchaser's inclusion of such Contracts on the schedule described in Section 2.3(a), (iii) by other agreement in writing of either Debtor (as applicable) and Purchaser, or (iv) otherwise in accordance with the Chapter 11 Plan, the Confirmation Order or another Final Order of the Bankruptcy Court. For the avoidance of doubt, the Assumed Contracts shall include all Reinsurance Contracts that are Executory Contracts and to which SALIC is a party.

"<u>Auction</u>" has the meaning set forth in the Bidding Procedures.

"<u>Back-Up Bidder</u>" has the meaning set forth in the Bidding Procedures.

"Bankruptcy and Equity Exceptions" has the meaning set forth in Section 3.2.

"<u>Bankruptcy Code</u>" has the meaning set forth in the Recitals.

"Bankruptcy Court" has the meaning set forth in the Recitals.

"<u>Bankruptcy Court Approvals</u>" means the Amended RIA Order (if necessary pursuant to Section 6.1(c)), the Winning Bidder Order, the Disclosure Statement Order, the Sale Order and the Confirmation Order.

"Bankruptcy Rules" means the Federal Rules of Bankruptcy Procedure.

"<u>Benefit Plans</u>" has the meaning set forth in Section 3.11(a).

"<u>Bermuda Act</u>" has the meaning set forth in the Recitals.

"Bermuda Court" has the meaning set forth in the Recitals.

"Bermuda Full Powers Liquidators" has the meaning set forth in the Recitals.

"Bermuda Winding Up Petition" has the meaning set forth in the Recitals.

"Bermuda Winding Up Proceeding" has the meaning set forth in the Recitals.

"<u>Bidding Procedures</u>" means the Bidding Procedures annexed as <u>Exhibit A</u> to the Bidding Procedures Order.

"<u>Bidding Procedures Motion</u>" has the meaning set forth in the Recitals.

"Bidding Procedures Order" has the meaning set forth in the Recitals.

"Books and Records" means originals or copies of all books, documents, accounts, ledgers and records (including computer generated, recorded or stored records) of, maintained by, or relating to the business of the SALIC Group Companies, including contract forms, policy information, policyholder information, claim records, underwriting records, administrative, pricing, underwriting, claims handling and reserving records and manuals, corporate and accounting and other records (including the books of account and other records), Tax records (including Tax Returns), disclosure and other documents and filings required under Applicable Law, financial records, and compliance records relating to the business of the SALIC Group Companies, including any database, magnetic or optical media and any other form of recorded, computer generated or stored information or process relating to the operations of the SALIC Group Companies.

"Break-Up Fee" has the meaning set forth in the Bidding Procedures Order.

"<u>Burdensome Condition</u>" has the meaning set forth in Section 5.3(d).

"<u>Business Day</u>" means any day other than a Saturday, a Sunday or a day on which banks in New York, New York or the Cayman Islands are authorized or obligated by Applicable Law to close.

"Cayman Islands Court" has the meaning set forth in the Recitals.

"Cayman Islands Full Powers Liquidators" has the meaning set forth in the Recitals.

"Cayman Islands Winding Up Petition" has the meaning set forth in the Recitals.

"Cayman Islands Winding Up Proceeding" has the meaning set forth in the Recitals.

"Chapter 11 Case" has the meaning set forth in the Recitals.

"<u>Chapter 11 Plan</u>" means the plan of reorganization under Chapter 11 of the Bankruptcy Code, either in the form as contemplated by the Plan Term Sheet as of the Agreement Date or as it may be altered, amended, modified, or supplemented from time to time in accordance with the Bankruptcy Code or the Bankruptcy Rules, as the case may be, and the Plan Supplement, including, all exhibits and schedules thereto. Unless otherwise agreed in writing by the parties hereto, the Chapter 11 Plan shall be consistent in all material respects with this Agreement and the Plan Term Sheet and otherwise in form and substance reasonably acceptable to the SALIC Parties and Purchaser. For the avoidance of doubt, the Chapter 11 Plan shall include provisions providing for, among other things, (a) the establishment of the Distribution Trust, (b) the funding of the Distribution Trust Reserve to the Distribution Trust, (c) the contribution and transfer of all SALIC/SRGL Claims to the Distribution Trust, and (d) the contribution and transfer of all DT Post-Closing Rights to the Distribution Trust.

"<u>Claim</u>" has the meaning set forth in Section 101(5) of the Bankruptcy Code.

"<u>Closing</u>" means the closing of the purchase and sale of the Purchaser Shares.

"<u>Closing Date</u>" has the meaning set forth in Section 2.4(a).

"<u>Closing Date Cure Amounts</u>" means all pre- and post-petition amounts payable pursuant to Section 365(b)(l)(A) or (B) of the Bankruptcy Code in order to effectuate the Debtors' assumption of Assumed Contracts pursuant to the Chapter 11 Plan, either (a) as determined by an Order of the Bankruptcy Court entered prior to the Closing Date or (b) agreed to between the applicable parties as of a date prior to the Closing Date.

"<u>Closing Date Plan Distributions</u>" means cash in an amount equal to the aggregate face amount of Allowed Other Secured Claims, Allowed Administrative Expense Claims, Allowed Priority Claims, Allowed Priority Tax Claims and TruPS Trustee Fees and Expenses for which distributions will be made on the Effective Date in accordance with the terms of the Chapter 11 Plan; <u>provided</u> that in no event shall the Closing Date Plan Distributions exceed the Plan Funding Payment. "<u>Code</u>" means the Internal Revenue Code of 1986, as amended from time to time, or any successor statute.

"<u>Company Group</u>" means any "affiliated group" (as defined in Section 1504(a) of the Code without regard to the limitations contained in Section 1504(b) of the Code) that, at any time on or before the Closing Date, included or has included any SALIC Group Company or any direct or indirect predecessor of any SALIC Group Company, or any other group of corporations filing Tax Returns on a combined, consolidated, unitary or similar basis that, at any time on or before the Closing Date, includes or has included any SALIC Group Company or any direct or indirect predecessor of any SALIC Group Company.

"<u>Condition Satisfaction</u>" has the meaning set forth in <u>Section 2.4(a)</u>.

"<u>Confidentiality Agreement</u>" means the Confidentiality Agreement, dated April 18, 2018, between SALIC and Hildene Opportunities Master Fund Ltd., as amended, modified or supplemented. Following the Auction, the parties may modify the Confidentiality Agreement as mutually agreed in order to appropriately reflect the receipt of confidential information by Purchaser in its capacity as such under this Agreement.

"<u>Confirmation Hearing</u>" means the hearing held by the Bankruptcy Court pursuant to Section 1128 of the Bankruptcy Code for the purpose of confirming the Chapter 11 Plan.

"<u>Confirmation Order</u>" means the Order of the Bankruptcy Court confirming the Chapter 11 Plan pursuant to section 1129 of the Bankruptcy Code, which Order (and any exhibits, appendices and related documents) shall be in form and substance reasonably acceptable to Purchaser and the SALIC Parties. For the avoidance of doubt, the Confirmation Order may include findings and conclusions approving and authorizing this Agreement and the Transactions on a final basis, in which case entry of the Confirmation Order shall also constitute entry of the Sale Order.

"<u>Contract</u>" means, with respect to any Person, any agreement, contract, lease, license, indenture, commitment, instrument or other legally binding obligation to which such Person is a party or is otherwise subject or bound.

"<u>Contract Information Deadline</u>" has the meaning set forth in Section 2.3(a).

"Contracting Parties" has the meaning set forth in Section 9.14.

"<u>Control</u>," "<u>Controlled</u>" or "<u>Controlling</u>" means, as to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by Contract or otherwise. The terms "Controlled by" and "under common Control with" shall have correlative meanings.

"<u>Cure Amount Schedule</u>" has the meaning set forth in Section 2.3(a).

"Cure Amounts" means all Closing Date Cure Amounts and Post-Closing Cure Amounts.

"<u>DDOI Proceeding</u>" means a proceeding commenced by the Delaware Insurance Commissioner or another Governmental Authority with jurisdiction over SRUS or its assets that constitutes a "Delinquency proceeding" within the meaning of such term as used in 18 Del. C. § 5901(3).

"<u>Debt Financing</u>" means any transaction or transactions pursuant to which any Person or group of Persons other than Purchaser and its Affiliates provides SRGL and/or any SALIC Group Company with financing sufficient to stabilize the financial condition of the SALIC Group Business.

"<u>Debtors</u>" means, collectively, SALIC and SHI, each in their respective capacities as debtors in the Chapter 11 Case.

"Deposit Escrow Agent" means Citibank, N.A.

"<u>Deposit Escrow Agreement</u>" means that certain Escrow Agreement, dated as of May 25, 2018, by and among Purchaser, SALIC and the Deposit Escrow Agent, pursuant to which the Deposit Escrow Amount is escrowed.

"Deposit Escrow Amount" has the meaning set forth in Section 2.1.

"<u>Disclosure Schedule</u>" means the Purchaser Disclosure Schedule and/or the SALIC Disclosure Schedule, as applicable.

"<u>Disclosure Statement</u>" means the Debtors' disclosure statement, including any exhibits, appendices, related documents, ballots and procedures related to the solicitation of votes on the Chapter 11 Plan, in each case, as amended, supplemented or otherwise modified from time to time in accordance with the terms hereof, in respect of the Chapter 11 Plan and that is prepared and distributed in accordance with, among other things, Sections 1125, 1126(b), and 1145 of the Bankruptcy Code, Bankruptcy Rule 3018 and other Applicable Law, and which shall be in form and substance reasonably acceptable to Purchaser and the SALIC Parties.

"<u>Disclosure Statement Motion</u>" means the motion filed by the Debtors with the Bankruptcy Court on May 4, 2018 seeking entry of the Disclosure Statement Order, which motion (including any exhibits, appendices or related documents) shall be amended following the Agreement Date to reflect the transactions set forth in this Agreement and the Plan Term Sheet in form and substance reasonably acceptable to Purchaser and the SALIC Parties.

"<u>Disclosure Statement Order</u>" means an Order of the Bankruptcy Court (a) approving the Disclosure Statement as containing "adequate information" under Section 1125 of the Bankruptcy Code, (b) approving the solicitation of votes on the Chapter 11 Plan, including procedures relating thereto and (c) granting related relief (including any exhibits, appendices or related documents), which Order shall be in form and substance reasonably acceptable to Purchaser and the SALIC Parties. For the avoidance of doubt, the Disclosure Statement Order may include findings and conclusions establishing Purchaser as the Winning Bidder and Purchaser's bid as the Winning Bid, in which case entry of the Disclosure Statement Order also shall constitute entry of the Winning Bidder Order.

"<u>Distribution Trust</u>" means the trust to be established pursuant to the Chapter 11 Plan on the Effective Date into which: (a) the Distribution Trust Reserves shall be funded; and (b) to which the following shall be contributed and transferred, among other things, as described in the Chapter 11 Plan and the Distribution Trust Agreement: (i) all SALIC/SRGL Claims; and (ii) all DT Post-Closing Rights. The Distribution Trust shall have the rights, duties and obligations, through a Distribution Trustee, as set forth in the Distribution Trust Agreement and in the Chapter 11 Plan, which shall be in form and substance reasonably acceptable to Purchaser and the SALIC Parties.

"<u>Distribution Trust Agreement</u>" means the agreement establishing the Distribution Trust, appointing the Distribution Trustee, establishing the rights and obligations of the Distribution Trustee, and governing the administration of the Distribution Trust.

"<u>Distribution Trust Reserves</u>" means the reserves to be established for the Distribution Trust and funded at Closing in accordance with Section 2.4(g) of this Agreement, the Chapter 11 Plan, the Confirmation Order and the Distribution Trust Agreement.

"<u>Distribution Trustee</u>" means the Person appointed under the Chapter 11 Plan and Distribution Trust Agreement to administer the Distribution Trust. The Distribution Trustee shall be a representative of each SALIC Party's bankruptcy estate in accordance with section 1123(b)(3)(B) of the Bankruptcy Code for the purposes of the DT Post-Closing Rights.

"DT Post-Closing Rights" means: (a) all rights, remedies, privileges, Claims, counterclaims and defenses of the SALIC Parties arising under or in connection with this Agreement with respect to (i) the Post-Closing Cure Estimate, the Post-Closing Cure Reserve and the payment of Post-Closing Cure Amounts, including Purchaser's obligations in connection with any of the foregoing items pursuant to Section 2.3(f) and Section 2.4(d), (ii) the Indemnified D&O Parties, (iii) Books and Records, including Purchaser's obligations pursuant to Section 5.1(c) of this Agreement, (iv) employee matters set forth in Section 5.4 and (v) the covenants and agreements of Purchaser under Section 5.1(b) and the Confidentiality Agreement; (b) all rights, Claims, counterclaims and defenses based on setoff, recoupment or similar doctrines under Applicable Law that either SALIC Party had at any time prior to the Closing in connection with any Claim of any Person asserted or capable of being asserted against a SALIC Party in its capacity as a Debtor in its Chapter 11 Case or against the bankruptcy estate of either Debtor; and (c) with respect to any Action brought after the Closing by, for or on behalf of Purchaser, Reorganized SALIC, Reorganized SHI or any assignee of or successor to any of the foregoing Persons, by merger, consolidation, or otherwise, all counterclaims or defenses that would otherwise be available to either SALIC Party arising under or in connection with this Agreement. For the avoidance of doubt, the Distribution Trustee retains and may assert, and by virtue of this Agreement, the Ancillary Agreements and the Transactions there shall be no waiver of, attorneyclient privilege, work product or any other evidentiary privilege, as applicable, with respect to matters within the scope of any of the foregoing DT Post-Closing Rights.

"<u>Effective Date</u>" means the date on which the Chapter 11 Plan becomes effective in accordance with its terms, which date shall be the Closing Date.

"<u>Electronic Data Rooms</u>" means, collectively, (i) the electronic data room established by or on behalf of SALIC with respect to the SALIC Group Companies, and (ii) the electronic data room established by or on behalf of SALIC with respect to the Reinsurance Contracts and Reserve Financing Contracts to which the Scottish Insurance Companies are a party, in each case maintained by ShareFile.

"<u>Employees</u>" means each individual who immediately prior to the Closing is employed by one or more of the SALIC Group Companies (including those individuals on approved leaves of absence at such time).

"<u>Employment Agreements</u>" has the meaning set forth in Section 2.4(b)(v).

"<u>Encumbrance</u>" means any security interest, pledge, mortgage, lien, equitable interest, easement, lease, sublease, covenant, right of first refusal, hypothecation, option, restriction (including any restriction on use, voting, transfer, alienation, receipt of income or exercise of any other attribute of ownership), encumbrance, deed of trust, hypothecation or charge of any kind.

"<u>Environmental Law</u>" means any Applicable Law relating to pollution or protection of the environment, including the use, handling, transportation, treatment, storage, disposal, release or discharge of hazardous chemicals, materials or substances.

"<u>ERISA</u>" means the Employee Retirement Income Security Act of 1974, or any successor statute.

"<u>ERISA Affiliate</u>" means any entity which is considered a single employer with a SALIC Group Company under Section 414(b) or (c) of the Code.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"<u>Executory Contract</u>" means any Contract to which one or more of the Debtors is a party that is subject to assumption or rejection under sections 365 or 1123 of the Bankruptcy Code.

"<u>Expense Reimbursement Amount</u>" has the meaning set forth in the Bidding Procedures Order.

"<u>Final Order</u>" means an Order that has not been reversed, stayed, modified, vacated or amended and as to which the time to appeal or seek certiorari or move for a new trial, reargument, or rehearing has expired, and no appeal or petition for certiorari or other proceedings for a new trial, reargument, or rehearing has been timely taken, or as to which any appeal that has been taken or any petition for certiorari that has been timely filed has been withdrawn or resolved by the highest court to which the order or judgment was appealed or from which certiorari was sought or the new trial, reargument or rehearing has been denied or resulted in no modification of such order or judgment; provided, however, that the possibility that a motion under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Federal Rules of Bankruptcy Procedure or rules of procedure applicable in the court that made or issued the Order or that may review such Order, may be filed with respect to such Order, shall not cause such Order not to be a Final Order.

"<u>Foreign Court Approvals</u>" means (a) if required under law applicable to the Bermuda Winding Up Proceedings, the Order or Orders by the Bermuda Court sanctioning entry into and performance under the Restructuring Implementation Agreement and the actions by SRGL set forth in Section 7.2(c) of this Agreement, (b) if required under law applicable to the Cayman Islands Winding Up Proceedings, the Order or Orders by the Cayman Islands Court sanctioning entry into and performance under the Restructuring Implementation Agreement and the actions by SRGL set forth in Section 7.2(c) of this Agreement, and (c) if necessary pursuant to Section 6.1(c), the Supplemental Cayman Islands Court Order.

"<u>FRS 101</u>" means the applicable accounting standards issued by the Financial Reporting Council and promulgated by the Institute of Chartered Accountants in Ireland, including FRS 101 Reduced Disclosure Framework (Generally Accepted Accounting Practice in Ireland) and Irish statute comprising the Companies Act 2014, including the European Union (Insurance Undertakings: Financial Statements) Regulations 2015.

"Full Powers Liquidators" has the meaning set forth in the Recitals.

"GAAP" means generally accepted accounting principles and practices in the United States.

"<u>Governmental Authority</u>" means any foreign, domestic, national, federal, state, regional, municipal or local governmental, legislative, judicial, administrative, regulatory, quasigovernmental or self-regulatory authority, agency, commission, body or court or any arbitral body or other similar entity, including the Bankruptcy Court, the Bermuda Court and the Cayman Islands Court.

"<u>Governmental Authorizations</u>" means all licenses, permits (including insurance permits), variances, waivers, Orders, registrations, consents, certificates, qualifications and other authorizations, actions and approvals of or by a Governmental Authority required (a) with respect to SALIC or Purchaser, to perform their respective obligations hereunder and (b) with respect to the SALIC Group Companies, to carry on their business and operations substantially as currently conducted under Applicable Law.

"<u>HSCM</u>" has the meaning set forth in the Recitals.

"<u>IFRS</u>" means the International Financial Reporting Standards as set forth by the International Accounting Standards Board.

"<u>In-force YRT Reinsurance Contract</u>" has the meaning set forth in Section 5.11(b).

"<u>Indemnified D&O Parties</u>" has the meaning set forth in Section 5.8(a).

"<u>Information Technology</u>" means Software and any tangible or digital computer systems (including computers, servers, workstations, routers, hubs, switches, networks, data communications lines and hardware), data or information subscription or access agreements, Internet-related information technology infrastructure, telecommunications systems and other hardware, owned or leased by, or licensed to, any of the SALIC Group Companies.

"Insurance Company Subsidiary" means each of SRUS, SRD and SRLB.

"<u>Insurance Laws</u>" means the Applicable Law relating to or regulating the business and products of insurance, including all applicable Orders and directives of Insurance Regulatory Authorities.

"<u>Insurance Regulatory Authority</u>" means, with respect to any jurisdiction, the Governmental Authority responsible for administering the Insurance Laws of such jurisdiction and regulating insurance companies domiciled or doing business in such jurisdiction.

"<u>Intellectual Property</u>" means, collectively, all United States and foreign registered and unregistered (a) patents and pending patent applications, (b) Trademarks, (c) copyrights, (d) trade secrets and (e) tangible embodiments of any of the foregoing.

"<u>Investment Assets</u>" means any investment assets (whether or not required by GAAP or SAP to be reflected on a balance sheet) beneficially owned (within the meaning of Rule 13d-3 under the Exchange Act) by any SALIC Group Company, including bonds, notes, debentures, mortgage loans, real estate and all other instruments of indebtedness, stocks, partnership or joint venture interests and all other equity interests, certificates issued by or interests in trusts, derivatives and all other assets acquired for investment purposes.

"Investment Guidelines and Policies" has the meaning set forth in Section 3.19(d).

"<u>IRS</u>" means the United States Internal Revenue Service.

"Joint Provisional Liquidators" has the meaning set forth in the Recitals.

"Key Employee" means an individual who is employed by SALIC and holds the position of Vice President or higher.

"<u>Knowledge</u>" means with respect to: (a) SALIC as it relates to any fact or other matter, the actual knowledge of the natural Persons listed in Section 1.1(a) of the SALIC Disclosure Schedule of such fact or matter and (b) Purchaser as it relates to any fact or other matter, the actual knowledge of the natural Persons listed in Section 1.1(b) of the Purchaser Disclosure Schedule of such fact or matter and, in each case, the knowledge that such Persons would have if such Persons had conducted a reasonable inquiry of the Persons having primary responsibility of such fact or matter.

"<u>Malware</u>" means any virus, Trojan horse, time bomb, key-lock, spyware, worm, malicious code or other software program designed to or able to, without the knowledge and authorization of the SALIC Group Companies, disrupt, disable, harm, interfere with the operation of or install itself within or on any Software, computer data, network memory or hardware.

"<u>Material Contract</u>" has the meaning set forth in Section 3.16(a).

"<u>Nonparty Affiliates</u>" has the meaning set forth in Section 9.14.

"<u>Order</u>" means any order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority, whether interlocutory or final, that has not been reversed, stayed, modified, or amended and, in the case of any order of the Bankruptcy Court, as entered on the docket of the Bankruptcy Court.

"<u>Ordinary Course of Business</u>" with respect to a Person means the ordinary course of business of such Person, consistent with past practice, subject to such changes by such Person and/or its Affiliates as are reasonably necessary or appropriate in light of the Bankruptcy Proceedings or Winding Up Proceedings.

"Ordinary Shares" has the meaning set forth in the Recitals.

"<u>Orkney Re II</u>" means Orkney Re II plc, a public limited company incorporated under the laws of Ireland.

"<u>Other Secured Claims</u>" means any prepetition secured Claims against a Debtor.

"<u>Outside Closing Date</u>" has the meaning set forth in Section 8.1(b).

"<u>Permitted Encumbrances</u>" means (a) liens for Taxes, assessments and other governmental charges not yet due and payable or due and being contested in good faith and for which accruals or reserves have been established against the full amount of such liability in accordance with GAAP, SAP, IFRS or FRS 101, as applicable, (b) mechanics', workmen's, repairmen's, warehousemen's, carriers' or other like liens arising or incurred in the Ordinary Course of Business or pursuant to original purchase price conditional sales contracts and equipment leases with third parties entered into in the Ordinary Course of Business, (c) pledges or deposits made to comply with workers' compensation, unemployment insurance or other social security laws or regulations, (d) Encumbrances related to deposits to secure policyholders' obligations as required by the Insurance Regulatory Authorities, (e) Encumbrances described in Section 3.17(b) recorded against the leases described in Section 3.17(b), (f) statutory landlords' or lessors' liens under the leases, (g) any other imperfection of title or encumbrance that does not materially detract from the current value or materially interfere with the current use by the SALIC Group Companies of the assets, properties or rights affected thereby and (h) any Encumbrance set forth on Section 1.2 of the SALIC Disclosure Schedule.

"<u>Person</u>" means an individual, a corporation, a partnership, an association, a limited liability company, a joint venture, a trust or other entity or organization, including a Governmental Authority.

"<u>Petition Date</u>" means the date of the filing of the Chapter 11 petitions of the Debtors.

"<u>Plan Funding Payment</u>" means an amount in cash equal to the difference of (a) twenty one million five hundred thousand dollars (\$21,500,000) less (b) the TruPS Returned Cash.

"<u>Plan Sponsorship Agreement</u>" means the Plan Sponsorship Agreement, dated as of January 28, 2018, by and among HSCM and the SALIC Parties.

"<u>Plan Supplement</u>" means the compilation of documents and forms of documents, schedules, and exhibits to be filed in the Chapter 11 Cases pursuant to and as contemplated in the Chapter 11 Plan, as amended, modified or supplemented from time to time in accordance with the terms hereof and in accordance with the Bankruptcy Code and the Bankruptcy Rules, which shall be in form and substance reasonably acceptable to Purchaser and the SALIC Parties.

"<u>Plan Term Sheet</u>" means the SALIC and SHI Term Sheet for the Chapter 11 Plan, dated as of the Agreement Date, and attached hereto as <u>Exhibit A</u>.

"<u>Post-Closing Cure Amounts</u>" means all pre- and post-petition amounts payable pursuant to Section 365(b)(l)(A) or (B) of the Bankruptcy Code in order to effectuate the Debtors' assumption of Assumed Contracts pursuant to the Chapter 11 Plan, either (a) as determined by an Order of the Bankruptcy Court entered on or after the Closing Date or (b) agreed to between the applicable parties as of a date occurring on or after the Closing Date.

"<u>Post-Closing Cure Estimate</u>" means the aggregate amount, as estimated by the applicable Debtor as of the Closing Date in consultation with Purchaser, of the cure obligations under Sections 365(b) and 1123(b)(2) of the Bankruptcy Code that will be owed to counterparties of Assumed Contracts as Post-Closing Cure Amounts.

"<u>Post-Closing Cure Reserve</u>" means an account to be established for the Distribution Trust and funded by Purchaser at Closing with cash in the amount of the Post-Closing Cure Estimate.

"<u>Priority Claims</u>" means any Claims entitled to priority in right of payment under Section 507(a) of the Bankruptcy Code, other than Priority Tax Claims.

"<u>Priority Tax Claims</u>" means any Claims entitled to priority in right of payment under Section 507(a)(8) of the Bankruptcy Code.

"<u>Purchase Price</u>" has the meaning set forth in Section 2.2(b).

"<u>Purchaser</u>" has the meaning set forth in the Preamble.

"<u>Purchaser Disclosure Schedule</u>" means the disclosure schedule delivered by Purchaser to SALIC in connection with the execution and delivery of this Agreement.

"<u>Purchaser Fundamental Reps</u>" has the meaning set forth in Section 7.3(a).

"<u>Purchaser Material Adverse Effect</u>" means a material adverse effect on the ability of Purchaser to consummate the Transactions.

"<u>Purchaser Rejection Schedule</u>" has the meaning set forth in Section 2.3(c).

"<u>Purchaser Shares</u>" has the meaning set forth in the Recitals.

"Qualified Bid" has the meaning set forth in the Bidding Procedures.

"<u>Recapitalization Funding Payment</u>" means an amount equal to Twelve Million Five Hundred Thousand Dollars (\$12,500,000) which shall be utilized by Reorganized SALIC to recapitalize the SALIC Group Companies after the Closing and in accordance with Section 2.4(k) and the proviso in Section 2.2(b) in respect of Cure Amounts in excess of \$100,000.

"<u>Reinsurance Contracts</u>" has the meaning set forth in Section 3.21(a).

"<u>Remaining SRGL Share</u>" has the meaning set forth in Section 7.2(c).

"<u>Reorganized SALIC</u>" shall mean SALIC, as reorganized pursuant to and under the Chapter 11 Plan, or any successor thereto, by merger, consolidation, or otherwise, on or after the effective date of such Chapter 11 Plan.

"<u>Reorganized SHI</u>" shall mean SHI, as reorganized pursuant to and under the Chapter 11 Plan, or any successor thereto, by merger, consolidation, or otherwise, on or after the effective date of such Chapter 11 Plan.

"<u>Representatives</u>" means, with respect to any Person, the officers, directors, principals, employees, agents, auditors, advisors, bankers and other representatives of such Person.

"<u>Required Approvals</u>" means the consents, approvals, waivers, authorizations, notices and filings relating to any Governmental Authority referred to in Sections 3.5, 4.3 and 5.3(b), including the Bankruptcy Court Approvals and Foreign Court Approvals.

"Reserve Financing Contracts" has the meaning set forth in Section 3.21.

"<u>Resignations</u>" has the meaning set forth in Section 5.6.

"<u>Restructuring Documents</u>" has the meaning set forth in Section 6.1(k).

"<u>Restructuring Implementation Agreement</u>" means that certain Restructuring Implementation Agreement, dated as of January 28, 2018, by and among SALIC, SRGL and SHI, as amended, modified or supplemented.

"<u>RIA Order</u>" means the Order of the Bankruptcy Court entered on March 19, 2018 granting the RIA/PSA Assumption Motion with respect to the Restructuring Implementation Agreement and authorizing the Debtors' assumption of the Restructuring Implementation Agreement.

"<u>RIA/PSA Assumption Motion</u>" means the motion filed by the Debtors with the Bankruptcy Court on January 31, 2018 seeking the assumption of the Restructuring Implementation Agreement and the Plan Sponsorship Agreement pursuant to Section 365 of the Bankruptcy Code.

"<u>RIA Winning Bidder Confirmation</u>" means confirmation in writing, in form and substance reasonably acceptable to Purchaser, from the Cayman Islands Full Powers Liquidators on behalf of SRGL that, after giving effect to this Agreement and the entry of the Winning Bidder Order, (x) the Restructuring Implementation Agreement remains in full force and effect and enforceable against SRGL in accordance with its terms, (y) that Purchaser shall have the rights of HSCM under such agreement and (z) that Purchaser, and not HSCM, is an express third party beneficiary of the Restructuring Implementation Agreement as set forth therein.

"<u>Sale Order</u>" means an Order of the Bankruptcy Court approving the Transactions and this Agreement, and authorizing the Debtors to consummate the Transactions and this Agreement in accordance with the terms thereof. For the avoidance of doubt, the inclusion of findings and conclusions granting the foregoing relief in the Confirmation Order as entered by the Bankruptcy Court shall satisfy any requirement in this Agreement for entry of the Sale Order.

"SALIC" has the meaning set forth in the Preamble. For the avoidance of doubt, the term "SALIC" also includes "SALIC" in its capacity as a debtor in its Chapter 11 Case.

"SALIC Audited Financial Statements" has the meaning set forth in Section 3.7(a).

"<u>SALIC Disclosure Schedule</u>" means the disclosure schedule delivered by SALIC to Purchaser in connection with the execution and delivery of this Agreement.

"<u>SALIC Financial Statements</u>" has the meaning set forth in Section 3.7(a).

"<u>SALIC Fundamental Reps</u>" has the meaning set forth in Section 7.2(a).

"<u>SALIC Group</u>" means, collectively, SALIC and the SALIC Subsidiaries.

"<u>SALIC Group Business</u>" has the meaning set forth in the Recitals.

"<u>SALIC Group Company</u>" means SALIC or a SALIC Subsidiary, and "<u>SALIC Group</u> <u>Companies</u>" means all of them.

"SALIC Intercompany Agreements" has the meaning set forth in Section 3.22.

"SALIC Material Adverse Effect" means a material adverse effect (i) on the business, operations, assets, liabilities, results of operations or condition (financial or otherwise) of the SALIC Group Companies (taken as a whole), or (ii) on the ability of SALIC and SHI to consummate the Transactions; provided, however, that with respect to the preceding clause (i) none of the following, and with respect to the preceding clause (ii) nothing in the following clause (h), shall constitute or be deemed to constitute a SALIC Material Adverse Effect, and otherwise shall not be taken into account in determining whether a SALIC Material Adverse Effect has occurred or would reasonably be likely to occur: any adverse effect arising out of, resulting from or attributable to (a) any changes or proposed changes in Applicable Laws, GAAP, IFRS, FRS 101 or SAP or in the interpretation or enforcement thereof; (b) changes occurring after the date of this Agreement in economic conditions in the United States generally affecting participants in the industries in which the SALIC Group Companies operate; (c) changes in United States or global financial or securities markets or conditions, including changes in prevailing interest rates, currency exchange rates or price levels or trading volumes in the United States or foreign securities markets; (d) any changes in global or national political

conditions (including, arising out of any election or the outbreak or escalation of war, military action, sabotage or acts of terrorism) or changes due to any pandemic natural disaster or other act of nature; (e) the effects of the actions or omissions of SALIC under this Agreement which actions or omissions to act are expressly requested or consented to in writing by Purchaser; (f) the effects of any breach, violation or non-performance of any provision of this Agreement by Purchaser or any of its Affiliates; (g) the announcement to the public of this Agreement and the Transactions or the identity of or facts relating to Purchaser or its Affiliates or the effects of any action taken by Purchaser or its Affiliates, to the extent directly attributable thereto; (h) any reasonably anticipated effects of the filing, commencement, pendency or prosecution of the Chapter 11 Cases and/or the Winding Up Proceedings; (i) any increases in claims or reserves of the SALIC Group Companies following the date of this Agreement; or (j) any failure (in and of itself) of the SALIC Group Companies to meet any projections or forecasts of earnings, claims paid or loss reserves; provided, further, that (x) clauses (i) and (j) in the preceding proviso shall not by themselves exclude the underlying causes of any such change, and (y) with respect to clauses (a), (b), (c) and (d) in the preceding proviso, such change will be taken into account in determining whether a SALIC Material Adverse Effect has occurred or would reasonably be expected to occur to the extent such change disproportionately affects the SALIC Group Companies relative to other Persons engaged in the industries in which the SALIC Group Companies operate.

"<u>SALIC Parties</u>" means, collectively, SALIC and SHI, including, as appropriate to context, as debtors in their respective Chapter 11 Cases.

"SALIC Subsidiary" means each of (i) SHI, (ii) SRD, (iii) SRUS, and (iv) SRLB.

"SALIC Unaudited Financial Statements" has the meaning set forth in Section 3.7(a).

"<u>SALIC/SRGL Claims</u>" means all Claims that SALIC holds against SRGL, including all Claims that SALIC holds against SRGL under or in connection with the SRGL Revolver Facility or the SRGL Revolver Facility Documents for principal, interest, charges, fees and expenses of attorneys and other professionals, and any other obligations arising thereunder or in connection therewith. The "SALIC/SRGL Claims" further include all rights, claims, counterclaims and defenses SALIC may have against SRGL based on setoff, recoupment or similar doctrines under Applicable Law.

"<u>SAP</u>" means, with respect to SRUS, the statutory accounting principles and practices prescribed or permitted by the Insurance Regulatory Authority in the jurisdiction in which SRUS is domiciled, consistently applied.

"Scottish Insurance Company" means each of (i) SALIC, (ii) SRUS, (iii) SRLB and (iv) SRD.

"Securities Act" means the Securities Act of 1933.

"<u>SFL</u>" means Scottish Financial (Luxembourg) S.á.r.l., a private limited liability company organized under the laws of Luxembourg.

"<u>SHI</u>" has the meaning set forth in the Preamble. For the avoidance of doubt, the term "SHI" also includes SHI in its capacity as a debtor in its Chapter 11 Case.

"<u>Software</u>" means all computer software, including assemblers, applets, compilers, source code, object code, binary libraries, development tools, design tools and user interfaces, in any form or format, however fixed, and all associated documentation.

"SRD" means Scottish Re (Dublin) DAC, an Ireland insurance company.

"SRE Marks" has the meaning set forth in Section 5.14.

"<u>SRGL</u>" has the meaning set forth in the Recitals.

"<u>SRGL Group Company</u>" means SRGL and each Affiliate of SRGL other than any SALIC Group Company.

"<u>SRGL Revolver Facility</u>" means the credit facility provided for under the SRGL Revolving Credit Agreement.

"<u>SRGL Revolver Facility Documents</u>" means the SRGL Revolving Credit Agreement, together with any related agreement, instrument, report or other document executed in connection therewith or otherwise evidencing Claims or other obligations arising thereunder, each as amended, novated, supplemented, extended, or restated from time to time.

"<u>SRGL Revolving Credit Agreement</u>" means that certain Revolving Credit Agreement, dated as of September 20, 2009, by and between SALIC, as lender, and SRGL, as borrower, as amended, novated, supplemented, extended, or restated from time to time.

"<u>SRGL Shares</u>" has the meaning set forth in Section 3.4(a).

"SRLB" means Scottish Re Life (Bermuda) Limited, a Bermuda insurance company.

"<u>SRUS</u>" means Scottish Re (U.S.), Inc., a Delaware insurance company.

"Stalking Horse Protection Payment" has the meaning set forth in Section 2.2(b).

"<u>Stalking Horse SPA</u>" has the meaning set forth in the Recitals.

"<u>Statutory Statements</u>" has the meaning set forth in Section 3.7(b).

"<u>Subsequent Audited GAAP Financial Statements</u>" has the meaning set forth in Section 5.12(a).

"Subsequent GAAP Financial Statements" has the meaning set forth in Section 5.12(a).

"<u>Subsidiary</u>" means with respect to any entity, any other entity as to which it owns, directly or indirectly, or otherwise controls, directly or indirectly, an amount of the voting securities, other voting ownership or voting partnership interests of which is sufficient to elect at

### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 122 of 241

least a majority of its board of directors or other governing body or more than fifty percent (50%) of the voting shares or other similar interests.

"<u>Supplemental Cayman Islands Court Order</u>" means an Order of the Cayman Islands Court granting such relief as is necessary for the Cayman Islands Full Powers Liquidators on behalf of SRGL to provide the RIA Winning Bidder Confirmation, in form and substance reasonably acceptable to Purchaser.

"<u>Tax</u>" or "<u>Taxes</u>" means (i) any and all taxes, including any interest, penalties or other additions to tax that may become payable in respect thereof, imposed by any Governmental Authority, which taxes shall include all income, profits, alternative minimum, estimated, payroll, withholding, social security, sales, use, ad valorem, value added, excise, escheat, unclaimed property, franchise, premium, gross receipts, stamp, transfer, net worth, and other taxes, fees, duties, levies, customs, tariffs, imposts, assessments, obligations and charges of the same or of a similar nature to any of the foregoing, and (ii) any liability for the payment of amounts determined by reference to amounts described in clause (i) as a result of being or having been a member of any group of corporations that files, will file, or has filed Tax Returns on a combined, consolidated, unitary or similar basis, as a result of any obligation under any agreement or arrangement (including any Tax Sharing Arrangement), as a result of being a transferee or successor, or by contract or otherwise.

"<u>Tax Returns</u>" means any and all returns (including information returns), reports, statements, certificates, schedules or claims for refund of or with respect to any Tax, including any and all attachments, amendments and supplements thereto.

"<u>Tax Sharing Arrangement</u>" means any written or unwritten agreement or arrangement providing for the allocation or payment of Tax liabilities or for Tax benefits between or among members of any group of corporations that file, or has filed, Tax Returns on a combined, consolidated or unitary basis.

"Terminating Intercompany Agreements" has the meaning set forth in Section 3.22.

"<u>Termination Date</u>" has the meaning set forth in Section 8.1.

"Third Party Consent Contracts" has the meaning set forth in Section 5.3(a).

"<u>Trademarks</u>" means all trademarks, trade names, trade dress, service marks, assumed names, business names and logos, slogans and Internet domain names, together with all goodwill of the businesses symbolized thereby, and all current registrations and applications for any of the foregoing.

"<u>Transactions</u>" means the transactions between SALIC and Purchaser, as contemplated by this Agreement, the Restructuring Implementation Agreement, and all other Ancillary Agreements, including the surrender by SRGL of the SRGL Shares to SALIC, SALIC's cancellation of the SRGL Shares, SALIC's issuance of the TruPS Shares to the TruPS Holders and Purchaser's acquisition of the Purchaser Shares.

### "TruPS" means:

(i) the undivided beneficial interests, having an aggregate liquidation amount of \$17,500,000.00, in Scottish Holdings Statutory Trust I, a Connecticut statutory trust, issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of December 4, 2002, among State Street Bank and Trust Company of Connecticut, National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer;

(ii) the undivided beneficial interests, having an aggregate liquidation amount of \$20,000,000.00, in Scottish Holdings, Inc. Statutory Trust II, a Connecticut statutory trust, issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of October 29, 2003, among U.S. Bank National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer;

(iii) the undivided preferred beneficial interests, having an aggregate liquidation amount of \$10,000,000.00, in GPIC Holdings Inc. Statutory Trust, a Delaware statutory trust, issued pursuant to that certain Amended and Restated Trust Agreement dated as of November 14, 2003, among JPMorgan Chase Bank as property trustee, Chase Manhattan Bank USA, National Association as Delaware trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer;

(iv) the undivided beneficial interests, having an aggregate liquidation amount of \$32,000,000.00, in Scottish Holdings, Inc. Statutory Trust III, a Connecticut statutory trust, issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of May 12, 2004, among U.S. Bank National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer; and

(v) the undivided beneficial interests, having an aggregate liquidation amount of \$50,000,000.00, in SFL Statutory Trust I, a Delaware statutory trust, issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of December 15, 2004, among Wilmington Trust Company, as institutional trustee, the administrators of the issuer named therein, SFL as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer.

# "TruPS Holders" means holders of TruPS.

"<u>TruPS Returned Cash</u>" has the meaning set forth in the Plan Term Sheet and as may be subsequently defined in the Chapter 11 Plan with the consent of the Purchaser.

"<u>TruPS Shares</u>" has the meaning set forth in the Plan Term Sheet and as may be subsequently defined in the Chapter 11 Plan with the consent of the Purchaser.

"<u>TruPS Trustee Fees and Expenses</u>" means the reasonable and documented fees and outof-pocket expenses of the indenture trustees of the TruPS incurred through the Closing Date, which shall consist of reasonable and documented fees and expenses incurred during the period ending thirty (30) days prior to the Closing Date, as well as a good faith estimate of such expenses expected to be incurred during the thirty (30) day period prior to the Closing Date, in each case, subject to an aggregate cap to be mutually agreed by the Debtors and Purchaser.

"<u>Winding Up Petitions</u>" has the meaning set forth in the Recitals.

"<u>Winding Up Proceedings</u>" has the meaning set forth in the Recitals.

"<u>Winning Bid</u>" has the meaning set forth in the Bidding Procedures.

"<u>Winning Bidder</u>" has the meaning set forth in the Bidding Procedures.

"<u>Winning Bidder Order</u>" means an Order of the Bankruptcy Court declaring the Winning Bidder and the Winning Bid in accordance with the Bidding Procedures. For the avoidance of doubt, the inclusion of findings and conclusions granting the foregoing relief in the Disclosure Statement Order as entered by the Bankruptcy Court shall satisfy any requirement in this Agreement for entry of the Winning Bidder Order.

"<u>Wire Transfer</u>" means a payment in immediately available funds by wire transfer in lawful money of the United States of America to such account or accounts as shall have been designated by notice from or on behalf of the receiving party to the paying party not less than two (2) Business Days prior to the scheduled date of payment.

Section 1.2 <u>Interpretation</u>.

(a) As used in this Agreement, references to the following terms have the meanings indicated:

(i) To the Preamble or to the Recitals, Sections, Articles or Schedules are to the Preamble or a Recital, Section or Article of, or a Schedule to, this Agreement unless otherwise clearly indicated to the contrary.

(ii) To any agreement or other document are to such agreement or other document (together with the schedules, exhibits and other attachments thereto) as it may have been or may hereafter be amended, modified, supplemented, waived or restated from time to time in accordance with its terms and the terms hereof (if applicable thereto), except with respect to any disclosure made in the SALIC Disclosure Schedule or the Purchaser Disclosure Schedule.

(iii) To any "statute" or "regulation" are to the statute or regulation as amended, modified, supplemented or replaced from time to time (and, in the case of statutes, include any rules and regulations promulgated under the statute) and to any "section of any statute or regulation" include any successor to the section. (iv) To any Governmental Authority include any successor to the Governmental Authority and to any Affiliate include any successor to the Affiliate.

(v) To any "copy" of any Contract or other document or instrument are to a true and complete copy.

(vi) To "hereof," "herein," "hereunder," "hereby," "herewith" and words of similar import refer to this Agreement as a whole and not to any particular Article, Section or clause of this Agreement, unless otherwise clearly indicated to the contrary.

(vii) To the "date of this Agreement," "the date hereof" and words of similar import refer to June 11, 2018.

(viii) To "this Agreement" includes the Schedules (including the Purchaser Disclosure Schedule and the SALIC Disclosure Schedule) to this Agreement.

(b) Whenever the words "include," "includes" or "including" are used in this Agreement, they will be deemed to be followed by the words "without limitation." The word "or" shall not be exclusive. Any singular term in this Agreement will be deemed to include the plural, and any plural term the singular. All pronouns and variations of pronouns will be deemed to refer to the feminine, masculine or neuter, singular or plural, as the identity of the Person referred to may require. Where a word or phrase is defined herein, each of its other grammatical forms shall have a corresponding meaning.

(c) Whenever the last day for the exercise of any right or the discharge of any duty under this Agreement falls on other than a Business Day, the party hereto having such right or duty shall have until the next Business Day to exercise such right or discharge such duty. Unless otherwise indicated, the word "day" shall be interpreted as a calendar day.

(d) The table of contents and headings contained in this Agreement are for reference purposes only and will not affect in any way the meaning or interpretation of this Agreement.

(e) References to "party" or "parties" hereto mean the parties to this Agreement.

(f) References to "dollars" or "\$" mean United States dollars, unless otherwise clearly indicated to the contrary.

(g) The parties have participated jointly in the negotiation and drafting of this Agreement; consequently, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as having been jointly drafted by the parties and no presumption or burden of proof shall arise favoring or disfavoring any party hereto by virtue of the authorship of any provision of this Agreement.

(h) No summary of this Agreement prepared by or on behalf of any party hereto shall affect the meaning or interpretation of this Agreement.

(i) Any document, list or other item shall be deemed to have been "made available" to Purchaser for all purposes of this Agreement only if such document, list or other item was posted before the date hereof in the Electronic Data Rooms or was delivered in writing (including electronic mail) to Purchaser or an Affiliate thereof in response to Purchaser's due diligence requests.

(j) All capitalized terms used without definition in the Schedules (including the Purchaser Disclosure Schedule and the SALIC Disclosure Schedule) to this Agreement shall have the meanings ascribed to such terms in this Agreement.

#### **ARTICLE II**

# ISSUANCE AND PURCHASE OF ORDINARY SHARES

Section 2.1 <u>Deposit</u>. On May 25, 2018, Purchaser deposited with the Deposit Escrow Agent an amount equal to two million six hundred seventy thousand dollars (\$2,670,000) as an earnest money deposit (together with all interest earned thereon but net of any fees, costs or other charges of the Deposit Escrow Agent, the "<u>Deposit Escrow Amount</u>") in accordance with the Bidding Procedures, to be held by the Deposit Escrow Agent in accordance with the terms of the Deposit Escrow Agreement. SALIC shall be responsible for any fees due and payable to the Deposit Escrow Agent under the Deposit Escrow Agreement. At the Closing, the Deposit Escrow Amount shall be credited against, and reduce dollar-for-dollar, the Recapitalization Funding Payment as set forth in <u>Section 2.4(c)(iv)</u>.

Section 2.2 <u>Purchase of Ordinary Shares</u>.

(a) Upon the terms and subject to the conditions set forth in this Agreement, at the Closing, SALIC shall issue and sell to Purchaser and Purchaser shall purchase from SALIC, the Purchaser Shares in exchange for the consideration set forth in Section 2.2(b).

(b) The purchase price to be paid or made available by Purchaser to SALIC at the Closing as consideration for the Purchaser Shares shall be: (i) the Plan Funding Payment; (ii) the Recapitalization Funding Payment; (iii) an aggregate amount equal to the Expense Reimbursement Amount and the Break-Up Fee (up to a maximum aggregate amount of one million two hundred fifty thousand dollars (\$1,250,000)) (the "<u>Stalking Horse Protection</u> <u>Payment</u>"); and (iv) all Cure Amounts (and together with the Plan Funding Payment, the Recapitalization Funding Payment and the Stalking Horse Protection Payment, collectively, the "<u>Purchase Price</u>"); provided, however, that the amount delivered by Purchaser at the Closing in respect of the Cure Amounts shall not exceed \$100,000 and the Recapitalization Funding Payment to be paid by Purchaser pursuant to this Agreement in respect of the Cure Amounts in excess of \$100,000.

# Section 2.3 <u>Assumption of Contracts.</u>

(a) The Debtors filed their Schedules of Assets and Liabilities on March 23, 2018 listing each Executory Contract on Schedule G and have previously delivered to HSCM the

schedule pursuant to Section 2.3(a) of the Stalking Horse SPA setting forth for each Executory Contract the amount the Debtors reasonably believe will be required to be paid in order to cure any defaults under any such Executory Contract pursuant to section 365(b)(l)(A) or (B) of the Bankruptcy Code (the "<u>Cure Amount Schedule</u>"). Within ten (10) Business Days after conclusion of the Auction ("<u>Contract Information Deadline</u>"), the Debtors shall deliver to Purchaser an update of the Cure Amount Schedule as of such date in order to assist Purchaser in making determinations as to assumption or rejection of such Executory Contracts.

(b) The applicable Debtor shall assume all Material Contracts identified on Section 3.16(a) of the SALIC Disclosure Schedule that are Executory Contracts to which one or both of the Debtors is a party, and any other Executory Contracts designated by Purchaser for assumption by the applicable Debtor pursuant to the Chapter 11 Plan, but excluding (i) any Contracts that are Terminating Intercompany Agreements, and (ii) any Executory Contracts identified on the Purchaser Rejection Schedule and any Contracts subject to Section 2.3(f). The Confirmation Order or Sale Order, as applicable, shall approve, among other things, the applicable Debtor's (i) assumption as of the Effective Date of the Assumed Contracts and (ii) rejection as of the Effective Date of the Executory Contracts that are not designated by Purchaser to be assumed pursuant to this Agreement or the Chapter 11 Plan, to the extent not previously rejected by an Order of the Bankruptcy Court at Purchaser's direction. The Debtors' obligations set forth in this Section 2.3(b) are subject to: (x) Purchaser's right to deliver a schedule (which schedule shall set forth any additional Assumed Contracts designated by Purchaser) on or before the date which is thirty (30) calendar days following the Contract Information Deadline; and (y) Debtors' obligation to reject any Executory Contracts identified on the Purchaser Rejection Schedule.

(c) On or before the date which is thirty (30) calendar days following the Contract Information Deadline, Purchaser shall deliver a schedule (the "<u>Purchaser Rejection Schedule</u>"), which shall be a schedule of the Executory Contracts to be rejected on the Effective Date. The Purchaser Rejection Schedule shall not include (i) any Reinsurance Contract or Reserve Financing Contract to which SALIC is a party, or (ii) any other material Contract comprising part of an excess reserve financing arrangement to which SHI is a party.

(d) The Debtors shall provide sufficient notice under the Bankruptcy Rules and local rules of the Bankruptcy Court to all counterparties to the Executory Contracts of their potential assumption or rejection and, with respect to the Executory Contracts to be assumed, also provide a schedule of Cure Amounts.

(e) In the event a Debtor is unable to assume any Material Contract that is an Executory Contract because any required consent is not obviated by an Order of the Bankruptcy Court, then such Debtor (as applicable), on the one hand, and Purchaser, on the other hand, shall use their commercially reasonable efforts to obtain, and to cooperate in obtaining, all Bankruptcy Court Approvals and all other Required Approvals from any third parties or Governmental Authorities necessary to assume such Material Contracts.

(f) On the Effective Date or as soon thereafter as is practicable, Purchaser shall pay, or Reorganized SALIC or Reorganized SHI shall pay on behalf of Purchaser in accordance with

Section 2.2(b), all Closing Date Cure Amounts to the applicable counterparty. Purchaser shall pay, or Reorganized SALIC or Reorganized SHI shall pay on behalf of Purchaser in accordance with Section 2.2(b), such Post-Closing Cure Amounts to the applicable counterparty promptly after such Post-Closing Cure Amount has been (i) determined by an Order of the Bankruptcy Court, or (ii) agreed to between Purchaser and the applicable counterparty. With respect to any cure obligations under Sections 365(b) or 1123(b)(2) of the Bankruptcy Code, as applicable, that are disputed by a Debtor, the Distribution Trustee or Purchaser, such Debtor or the Distribution Trustee, as applicable, and Purchaser shall reasonably cooperate and diligently pursue resolution of such disputes through the Effective Date. From and after the Effective Date, the Distribution Trustee shall reasonably cooperate with Purchaser and diligently pursue, at Purchaser's cost, resolution and, to the extent necessary, adjudication by the Bankruptcy Court of any thenoutstanding disputes of such cure obligations, and upon resolution of any such dispute pay the Cure Amount payable to any counterparty in accordance with this Section 2.3(f). Any Executory Contract proposed to be assumed hereunder that is the subject of a dispute over the amount or manner of cure (other than (A) any Reinsurance Contract or Reserve Financing Contract to which SALIC is a party, and (B) any other material Contract comprising part of an excess reserve financing arrangement to which SHI is a party) may be rejected upon a motion made by the Debtors, at the direction of Purchaser, or, after the Effective Date, by Reorganized SALIC or Reorganized SHI. Any order entered after the Confirmation Date by the Court, after notice and a hearing, authorizing the rejection of an Executory Contract or unexpired lease shall cause such rejection to be a prepetition breach under Sections 365(g) and 502(g) of the Bankruptcy Code, as if such relief had been granted and such order entered prior to the Confirmation Date. Upon payment by Purchaser, or the Reorganized SALIC or Reorganized SHI on behalf of Purchaser, of the aforesaid Cure Amounts, all defaults under the Assumed Contracts (monetary or otherwise) and all actual or pecuniary losses that have or may have resulted from such defaults shall be deemed cured, whether or not such obligation became due, or accrued, after the effective date of the assignment of such Assumed Contract, and Purchaser shall have no further liability thereunder relating to the Cure Amounts.

#### Section 2.4 <u>Closing</u>.

Unless another date, time or place is mutually agreed to in writing by the parties (a) hereto, the Closing shall take place at the offices of Mayer Brown LLP, 1221 Avenue of the Americas, New York, New York 10020 at 10:00 a.m., New York City time, on a Business Day to be specified by the parties, which shall be no later than the tenth (10<sup>th</sup>) Business Day following the date upon which the satisfaction or waiver (to the extent permitted by Applicable Law) of the last of the conditions set forth in Article VII (other than conditions that by their terms are to be satisfied at the Closing, but subject to the satisfaction or waiver (to the extent permitted by Applicable Law) at or prior to the Closing of all such conditions) in accordance with this Agreement (the "Condition Satisfaction") or at such other time, date and place as the parties may mutually agree in writing; provided, however, that the parties will work together to close on or about the first Business Day of the first calendar month following the calendar month in which the Condition Satisfaction occurs, including, to the extent possible, by Purchaser waiving the ten (10) Business Day requirement above. The date on which the Closing occurs is referred to herein as the "Closing Date". The parties agree that effectiveness of the Closing shall be as of 12:01 a.m., New York City time, on the Closing Date.

(b) At the Closing, SALIC shall deliver or cause to be delivered to Purchaser the following:

(i) the executed certificate(s) described in Sections 7.2(a), (b) and (e);

(ii) a certificate representing the Purchaser Shares to be newly issued by SALIC in the name of Purchaser, and if applicable, duly endorsed in blank, or accompanied by stock powers duly executed in blank, in proper form for transfer on the stock transfer books of SALIC;

(iii) an executed cross-receipt for the Purchase Price paid by Purchaser at the Closing;

(iv) the duly tendered Resignations;

(v) employment agreements executed by each of Gregg Klingenberg and Thomas J. Keller and any other individual agreed upon by the parties (the "<u>Employment</u> <u>Agreements</u>"), in form and substance reasonably satisfactory to Purchaser and containing terms no less favorable to the employee than the terms of his or her employment package (including base salary, bonus potential, benefits and severance) in effect as of the date hereof;

(vi) certified copies of the Amended RIA Order and the Supplemental Cayman Islands Court Order (in each case, if applicable pursuant to Section 6.1(c)), the Sale Order, the Disclosure Statement Order, the Winning Bidder Order and the Confirmation Order; and

(vii) a certificate executed by a duly authorized officer of SALIC certifying that any required Foreign Court Approvals have been obtained and that the Orders granting the Foreign Court Approvals as of the Closing Date have not been modified, stayed, reversed, revoked or vacated.

(c) At the Closing, Purchaser shall deliver or cause to be delivered to SALIC the following:

(i) the executed certificate(s) described in Sections 7.3(a), (b) and (c);

- (ii) the Plan Funding Payment by Wire Transfer;
- (iii) the Deposit Escrow Amount by Wire Transfer from the Escrow Agent;

(iv) an amount equal to the difference of (x) the Recapitalization Funding Payment less (y) the Deposit Escrow Amount, by Wire Transfer;

(v) the Stalking Horse Protection Payment (which amount may be delivered directly by Purchaser to HSCM as contemplated by Section 2.4(i) below); and

(vi) an executed cross-receipt for the Purchaser Shares to be delivered at the Closing by SALIC.

(d) At the Closing, Purchaser shall deliver or cause to be delivered to the counterparties of Assumed Contracts for which Closing Date Cure Amounts then exist, by Wire Transfer (or by check if Wire Transfer is not feasible), the Closing Date Cure Amounts; <u>provided</u> that the aggregate amount to be delivered by Purchaser pursuant to Sections 2.4(d) and 2.4(e) shall in no event be greater than \$100,000 (it being understood that any Cure Amounts in excess of \$100,000 shall be treated in accordance with the proviso in Section 2.2(b)).

(e) At the Closing, Purchaser shall deliver or cause to be delivered to the Post-Closing Cure Reserve, cash in an amount equal to the Post-Closing Cure Estimate; <u>provided</u> that the aggregate amount to be delivered by Purchaser pursuant to Sections 2.4(d) and 2.4(e) shall in no event be greater than \$100,000 (it being understood that any Cure Amounts in excess of \$100,000 shall be treated in accordance with the proviso in Section 2.2(b)).

(f) At the Closing, *first* from the unrestricted cash then available to SALIC and SHI and *thereafter*, to the extent that the unrestricted cash then available to SALIC and SHI is insufficient, from the Plan Funding Payment, SALIC shall make the Closing Date Plan Distributions to the holders of Allowed Other Secured Claims, Allowed Administrative Expense Claims, Allowed Priority Claims, Allowed Priority Tax Claims and TruPS Trustee Fees and Expenses entitled to receive such Closing Date Plan Distributions on the Effective Date in accordance with the terms of the Chapter 11 Plan.

(g) At the Closing, the remaining Plan Funding Payment, net of the Closing Date Plan Distributions that are paid out of the Plan Funding Payment at Closing, shall be deposited by SALIC into the Distribution Trust and thereafter utilized or disbursed in accordance with the terms of the Chapter 11 Plan, the Confirmation Order and the Distribution Trust Agreement.

(h) At the Closing, the following shall be contributed and transferred to the Distribution Trust and thereafter utilized or disbursed in accordance with the terms of the Chapter 11 Plan, the Confirmation Order and the Distribution Trust Agreement: (i) to the extent contemplated by the Confirmation Order, all capital stock or share capital of SFL held by any SALIC Group Company; (ii) all SALIC/SRGL Claims; and (iii) all DT Post-Closing Rights.

(i) At the Closing, SALIC shall pay (or SALIC shall direct Purchaser to pay directly on behalf of SALIC) the Stalking Horse Protection Payment to HSCM in accordance with the Stalking Horse SPA (it being understood that such payment shall be made from cash on hand resulting from the payment received at Closing from Purchaser pursuant to Section 2.4(c)(v) above, and shall not be paid from the Plan Funding Payment or the Recapitalization Funding Payment).

(j) At the Closing, each party hereto shall deliver to the other party hereto copies (or other evidence) of all of its Required Approvals (other than the Foreign Court Approvals) as evidence of the satisfaction of the condition set forth in Section 7.1(h).

(k) For the avoidance of doubt, the Recapitalization Funding Payment (less Cure Amounts paid or deposited in accordance with the provisions of this Section 2.4 above) shall be retained by Reorganized SALIC and held or used in its operation of the SALIC Group Business and for other permissible business purposes after the Closing.

(l) Immediately following the Closing, SALIC shall deliver to Purchaser evidence satisfactory to Purchaser of the cancellation of the Remaining SRGL Share.

Section 2.5 <u>Withholding</u>. Purchaser and its Affiliates shall be entitled to deduct and withhold from any amount otherwise payable pursuant to this Agreement such amounts as it is required to deduct and withhold with respect to the making of such payment under any provision of federal, state, local or foreign law. If any amount is so withheld, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction or withholding was imposed.

#### **ARTICLE III**

#### **REPRESENTATIONS AND WARRANTIES OF SALIC**

SALIC represents and warrants to Purchaser, as of the date hereof and as of the Closing Date, as follows. Each such representation and warranty is qualified by and includes the disclosure set forth in the numbered or lettered sections or subsections of the SALIC Disclosure Schedule that correspond to such representation and warranty and shall be deemed to be qualified by and include any disclosure in any other section or subsection of the SALIC Disclosure Disclosure Schedule to which the relevance of such disclosure to such representation and warranty is readily apparent.

Section 3.1 Organization and Authority of SALIC. SALIC is an entity duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization. SALIC has all requisite corporate power and authority to execute and deliver this Agreement, and, subject to obtaining the Required Approvals as contemplated by this Agreement, to perform its obligations hereunder and to consummate the Transactions. The execution, delivery and performance of this Agreement by SALIC and the consummation by SALIC of the Transactions have been duly authorized by all necessary corporate action on the part of SALIC and no additional corporate proceedings on the part of SALIC are necessary to approve or authorize this Agreement, the consummation by SALIC of the Transactions or the performance by SALIC of its obligations under this Agreement or the Transactions. SALIC (i) has all requisite corporate power and authority to own, lease, or otherwise hold its assets and to carry on its business as currently conducted and (ii) is qualified to do business in each jurisdiction where the ownership or operation of its assets or the conduct of its business requires such qualification, except where the failure to be so qualified would not, individually or in the aggregate, reasonably be likely to have a SALIC Material Adverse Effect.

Section 3.2 <u>Binding Effect</u>. Assuming the due authorization, execution and delivery by Purchaser, this Agreement constitutes the valid and binding obligation of SALIC, enforceable against SALIC in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent transfer, preference and other similar laws affecting

creditors' rights generally, and by general principles of equity (regardless of whether enforcement is sought in equity or at law) (the "<u>Bankruptcy and Equity Exceptions</u>").

#### Section 3.3 Organization and Authority of the SALIC Subsidiaries.

(a) Each of the SALIC Subsidiaries is a company duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization. Except as set forth in Section 3.3(a) of the SALIC Disclosure Schedule, each of the SALIC Subsidiaries (i) has all requisite corporate power and authority to own, lease or otherwise hold its assets and to carry on its business as currently conducted and (ii) is duly qualified to do business and is in good standing (if applicable) in each jurisdiction where the ownership or operation of its assets or the conduct of its business requires such qualification, except, in the cases of clauses (i) and (ii) above, where the failure to have such power and authority or to be so qualified would not, individually or in the aggregate, reasonably be likely to have a SALIC Material Adverse Effect.

(b) SALIC has made available to Purchaser copies of the organizational documents of each of the SALIC Group Companies, in each case as amended to the date of this Agreement.

#### Section 3.4 Capital Structure; Ownership of the SALIC Group Subsidiaries.

As of the Agreement Date, SALIC is authorized to issue 20,000,000,000 Ordinary (a) Shares, of which 20,000,000,000 are issued and outstanding and held by SRGL (the "SRGL Shares"). Immediately prior to the Closing, after giving effect to the surrender and cancellation of all SRGL Shares other than the Remaining SRGL Share as contemplated by Sections 2.4(1) and 7.2(c), SALIC will be authorized to issue 20,000,000 Ordinary Shares and the Remaining SRGL Share will be the only Ordinary Share issued and outstanding, and the Remaining SRGL Share will be held by SRGL. Immediately following the Closing and the cancellation of the Remaining SRGL Share, the only Ordinary Shares of SALIC issued and outstanding will be the Purchaser Shares and the TruPS Shares. The authorized capital stock or share capital of each of the SALIC Group Companies, including the record owners of such capital stock or share capital, is set forth on Section 3.4(a) of SALIC Disclosure Schedule, and there are no other shares of capital stock or other equity securities of any of the SALIC Group Companies issued, reserved for issuance or outstanding. All of the outstanding shares of the SALIC Group Companies' capital stock or share capital have been duly authorized and validly issued, and are fully paid and nonassessable. Except as set forth above or in Section 3.4(a) of the SALIC Disclosure Schedule, there are no options, puts, tag-alongs, drag-alongs, warrants, rights, convertible securities or other agreements or commitments obligating a SALIC Group Company to issue, transfer or sell, or cause the issuance, transfer or sale of, any shares of capital stock or share capital of such SALIC Group Company or to make any payments in respect of the value of any shares of such SALIC Group Company. There are no outstanding arrangements, agreements, obligations, commitments or other rights or obligations of any of the SALIC Group Companies to repurchase, redeem, or otherwise acquire any capital stock, membership interests, partnership interests, joint venture interests or other equity interests of any of the SALIC Group Companies. There are no bonds, debentures, notes or other indebtedness of the SALIC Group Companies having voting rights (or convertible into securities having voting rights). There are no voting trusts, stockholder agreements, proxies or other rights or agreements in effect with

respect to the voting, transfer or dividend rights of the securities of the SALIC Group Companies. There are no capital appreciation rights, phantom stock plans, securities with participation rights or features or similar obligations or commitments of the SALIC Group Companies.

(b) SFL and the SALIC Subsidiaries are the only Subsidiaries of SALIC. Except as set forth on Section 3.4(b) of the SALIC Disclosure Schedule, SALIC owns all of the capital stock of each SALIC Subsidiary and SFL, in each case of record and beneficially, free and clear of all Encumbrances (other than restrictions on transfer imposed by any foreign, federal and state insurance and securities Applicable Laws).

Section 3.5 <u>Governmental Filings and Consents</u>. Except as may result from any facts or circumstances relating to the identity or regulatory status of Purchaser or its Affiliates, no consents or approvals of, waivers from or filings or registrations with, any Governmental Authority are or will be required to be made or obtained at or prior to the Closing by SRGL, SALIC or any SALIC Group Company in connection with the execution, delivery or performance by SALIC of this Agreement or to consummate the Transactions, except for (a) the approvals, filings and notices required under the Insurance Laws set forth in Section 3.5(a) of the SALIC Disclosure Schedule, (b) such other matters set forth on Section 3.5(b) of the SALIC Disclosure Schedule and (c) consents, approvals, waivers, filings or registrations, the failure of which to make with or obtain from the applicable Governmental Authorities would not, individually or in the aggregate, reasonably be likely to have a SALIC Material Adverse Effect.

No Violations. Subject to the making of the filings and registrations and Section 3.6 receipt of the consents, approvals and waivers referred to in Section 3.5 and the expiration of related waiting periods, except as may result from any facts or circumstances relating to the identity or regulatory status of Purchaser or its Affiliates, the execution, delivery and performance of this Agreement by SALIC and the consummation of the Transactions do not and will not (a) conflict with, constitute a breach or violation of, or a default under, or give rise to any Encumbrance (other than Permitted Encumbrances) or any acceleration of remedies, penalty, increase in benefit payable or right of termination, impairment, alteration, suspension, revocation, cancellation or other change of any right or obligation or the loss of any benefit under, or forfeiture of any Benefit Plan or Contract to which any of the SALIC Group Companies are party or by which any of their assets, properties or rights is bound or subject, except as would not, individually or in the aggregate, reasonably be likely to have a SALIC Material Adverse Effect, (b) conflict with or result in a breach or violation of any Applicable Law, or (c) conflict with, constitute a breach or violation of, or a default under, the organizational documents of any SALIC Group Company.

# Section 3.7 Financial and Statutory Statements; No Undisclosed Liabilities.

(a) SALIC has made available to Purchaser in the Electronic Data Rooms as of the date hereof copies of (i) the audited consolidated balance sheets of SALIC and its Subsidiaries as of December 31, 2016 and December 31, 2015 and the related consolidated statements of operations, changes in stockholders' equity and cash flows for the fiscal years then ended, together with the report of SALIC's independent auditors thereon and all exhibits, schedules and

notes thereto (collectively, the "SALIC Audited Financial Statements"), and (ii) the unaudited consolidated balance sheets of SALIC and its Subsidiaries as of March 31, 2017, June 30, 2017, September 30, 2017, December 31, 2017 and March 31, 2018 and the related consolidated statements of operations and changes in stockholders' equity for the periods then ended (the "SALIC Unaudited Financial Statements" and, collectively with the SALIC Audited Financial Statements, the "SALIC Financial Statements"). The SALIC Financial Statements (A) were derived from and are consistent with the Books and Records, (B) were prepared in accordance with GAAP consistently applied during the periods presented (except as may be indicated in the notes thereto) and (C) present fairly, in all material respects, the consolidated financial position, results of operations, cash flows and changes in stockholders' equity of SALIC and its Subsidiaries as of the respective dates and for the respective periods referred to in the SALIC Financial Statements, subject, in the case of the SALIC Unaudited Financial Statements, to normal year-end adjustment and to the absence of footnotes. The Subsequent GAAP Financial Statements required to be delivered after the date hereof pursuant to Section 5.12(a) will (1) be derived from and be consistent with the Books and Records, (2) be prepared in accordance with GAAP consistently applied during the periods presented (except as may be indicated in the notes thereto) and (3) present fairly, in all material respects, the consolidated financial position, results of operations, cash flows and changes in stockholders' equity of SALIC and its Subsidiaries at and for the respective periods indicated (subject, in the case of interim financial statements, to normal year-end adjustments, and to the absence of footnotes).

(b) SALIC has made available to Purchaser in the Electronic Data Rooms as of the date hereof copies of the following statements, in each case together with the exhibits, schedules and notes thereto (the "Statutory Statements"): (i) the annual statement of SRUS, as filed with the Delaware Department of Insurance, as of and for the years ended December 31, 2017 and 2016; (ii) the audited statutory financial statements of SRUS, as filed with the Delaware Department of Insurance, as of and for the years ended December 31, 2017 and 2016; (iii) the quarterly financial statements of SRUS, as filed with the Delaware Department of Insurance, for the quarterly period ended March 31, 2018; (iv) the audited annual financial statements of SRD as of and for the years ended December 31, 2017 and 2016; and (v) the unaudited quarterly statements of SRD as of and for the quarterly period ended March 31, 2018. Such Statutory Statements (A) were derived from and consistent with the Books and Records, (B) were prepared, in all material respects, in accordance with all Applicable Laws and (x) in the case of the Statutory Statements of SRUS, SAP consistently applied during the periods presented and (y) in the case of the financial statements of SRD, FRS 101 consistently applied during the periods presented and (C) present fairly, in all material respects, the statutory financial position and the statutory results of operations, capital and surplus of such Scottish Insurance Company as of the respective dates and for the respective periods referred to in the Statutory Statements, subject in the case of any interim financial statements included in the Statutory Statements, to normal yearend adjustments and to the absence of footnotes). No material weakness or deficiency has been asserted by any Governmental Authority with respect to any of the Statutory Statements and there are no permitted practices utilized in the preparation of the Statutory Statements. No Governmental Authority has requested the refiling or amendment of any Statutory Statement. The Statutory Statements required to be delivered after the date hereof pursuant to Section 5.12(b) will (1) be prepared from, and be consistent with, the Books and Records, (2) be prepared in accordance with all Applicable Laws and with SAP (in the case of SRUS) or FRS

101 (in the case of SRD), applied on a consistent basis during the periods presented and (3) present fairly, in all material respects, the respective statutory financial position of the Scottish Insurance Companies at the respective dates thereof, and the statutory results of their operations and cash flows for the periods then ended (subject, in the case of any interim financial statements included in such Statutory Statements, to normal year-end adjustments and to the absence of footnotes).

(c) SALIC and its Subsidiaries maintain a system of internal controls that provide reasonable assurance that: (i) records are maintained in reasonable detail and accurately and fairly reflect the transactions and dispositions of the assets of SALIC and its Subsidiaries; (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP, SAP or FRS 101, as applicable, and that receipts and expenditures of SALIC and its Subsidiaries are being made only in accordance with authorizations of management and directors of SALIC or any of its Subsidiaries, as applicable; (iii) controls prevent or timely detect unauthorized acquisition, use or disposition of SALIC's or its Subsidiaries' assets, as applicable, that could have a material effect on the financial statements of SALIC or any of its Subsidiaries; and (iv) the recorded accountability for its assets is compared with its existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

(d) The Books and Records (i) have been maintained in all material respects in accordance with industry customary business practices and in accordance with Applicable Law, (ii) accurately represent and reflect, in all material respects, the business of the SALIC and its Subsidiaries and all transactions and actions related thereto, (iii) are true, complete and correct in all material respects, and (iv) constitute all of the files and data necessary for the operation of the business of SALIC and its Subsidiaries.

(e) Except for those liabilities (i) that are reflected or reserved against in the SALIC Financial Statements, (ii) incurred in the Ordinary Course of Business since December 31, 2016, (iii) incurred by or on behalf of SALIC in connection with this Agreement or the Transactions or (iv) that would not, individually or in the aggregate, reasonably be likely to have a SALIC Material Adverse Effect, as of the date hereof, the SALIC Group Companies have no liabilities that would be required by GAAP to be reflected on an unaudited balance sheet of SALIC (or disclosed in the notes thereto) (excluding any liabilities of the type covered by the representations and warranties set forth in this Agreement other than this Section 3.7(e)).

Section 3.8 <u>Absence of Certain Changes</u>. Except to the extent arising out of or relating to the Transactions, since December 31, 2016, (a) the business of the SALIC Group has been operated in all material respects in the Ordinary Course of Business, (b) there has not occurred any event or events that, individually or in the aggregate, have had, or would reasonably be expected to have, a SALIC Material Adverse Effect, (c) other than as set forth in Section 3.8(c) of the SALIC Disclosure Schedule, there has not occurred any declaration, setting aside or payment of any dividend or other distribution (whether in cash, stock or property) with respect to any of SALIC's outstanding capital stock or the outstanding capital stock of the SALIC Group Companies other than regular dividend payments and (d) there has not occurred any change in accounting methods, principles or practices by any SALIC Group Company materially affecting its assets or liabilities, except insofar as may have been required by

Applicable Law or required or permitted by a change in applicable GAAP or SAP. Except with regard to Material Contracts (which are addressed in Section 3.16) and Reinsurance Contracts and Reserve Financing Contracts (which are addressed in Section 3.21), other than as set forth in Section 3.8(e) of the SALIC Disclosure Schedule, since December 31, 2016 to the date hereof, no SALIC Group Company has taken any action or omitted to take any action, which action or omission, if occurring after the date hereof without the consent of Purchaser, had Section 5.2 been in effect from December 31, 2016 to the date hereof, would constitute (x) a breach of clause (i) through (iv), (vii), (viii), (ix), (x), (xii), (xv), (xvi), (xviii), (xix), (xx) or (xxi) (or clause (xxii) with respect to the foregoing clauses) of Section 5.2; or (y) a breach in any material respect of clause (v), (xii), (xiv), or (xvii) (or clause (xxii) thereof with respect to such other clauses) of Section 5.2.

# Section 3.9 Litigation; Orders.

(a) Other than as set forth in Section 3.9(a) of the SALIC Disclosure Schedule, as of the date hereof, there is no material Action pending or, to the Knowledge of SALIC, threatened against any SALIC Group Company or its business or any of its properties or assets.

(b) Other than as set forth in Section 3.9(b) of the SALIC Disclosure Schedule, as of the date hereof, none of the SALIC Group Companies is a party or subject to any material Order applicable to that SALIC Group Company, its business or any of its properties or assets other than any Order that is generally applicable to all Persons in businesses similar to that of the SALIC Group.

(c) There are no Actions pending or, to the Knowledge of SALIC, threatened against any SALIC Group Company or its business or any of its properties or assets that question the validity of, or seek injunctive relief with respect to, this Agreement or the right of any of the SALIC Group Companies to enter into, or consummate the Transactions.

# Section 3.10 <u>Taxes.</u>

(a) Other than as set forth in Section 3.10(a) of the SALIC Disclosure Schedule, (i) all Tax Returns required to be filed by or with respect to the SALIC Group Companies have been timely filed and no extension of time within which to file any such Tax Return is in effect, (ii) all such Tax Returns were true, complete, and correct in all respects and disclose all Taxes required to be paid by or with respect to the SALIC Group Companies for the periods covered thereby and (iii) all Taxes (whether or not shown as due on any such Tax Returns) for which any of the SALIC Group Companies may be liable have been timely paid.

(b) Other than as set forth in Section 3.10(b) of the SALIC Disclosure Schedule, (i) there are no audits, claims, actions or assessments regarding Taxes pending or proposed or threatened in writing with respect to Taxes for which any of the SALIC Group Companies may be liable, and (ii) all deficiencies asserted or assessments made as a result of any examination of the Tax Returns described in Section 3.10(a) have been paid in full or otherwise finally resolved. (c) No waiver of any statute of limitations relating to Taxes for which any of the SALIC Group Companies may be liable is in effect, and no written request for such a waiver is outstanding.

(d) There are no Tax rulings, requests for rulings, or closing agreements relating to Taxes for which any of the SALIC Group Companies may be liable that could affect the SALIC Group Companies' liability for Taxes for any taxable period ending after the Closing Date. None of the SALIC Group Companies will be required to include or accelerate the recognition of any item in income, or exclude or defer any deduction or other tax benefit, in each case in any taxable period (or portion thereof) after Closing, as a result of any change in method of accounting, closing agreement, intercompany transaction, installment sale, or the receipt of any prepaid amount, in each case prior to Closing.

(e) No Encumbrances (other than Permitted Encumbrances) for Taxes exist with respect to any of the assets or properties of the SALIC Group Companies.

(f) All Taxes that any of the SALIC Group Companies is required by law to withhold or to collect for payment have been duly withheld and collected and have been paid to the appropriate Governmental Authority.

(g) No SALIC Group Company is a party to or bound by any Tax allocation, sharing, indemnity (entered into in connection with a material transaction with a third party) or similar agreement. Except as set forth in Section 3.10(g) of the SALIC Disclosure Schedule, no SALIC Group Company has been a member of any Company Group other than each Company Group of which it is presently a member, and no SALIC Group Company presently has or has had any direct or indirect ownership interest in any corporation, partnership, joint venture or other entity (other than the SALIC Subsidiaries). None of the SALIC Group Companies has any liability for Taxes of another Person under Treasury Regulation § 1.1502-6 (or any similar provision of state, local or foreign law), as a transferee or successor, or otherwise.

(h) None of the SALIC Group Companies has participated in any "listed transaction" within the meaning of Treasury Regulation § 1.6011-4(b)(2) and, with respect to each transaction in which any SALIC Group Company has participated that is a "reportable transaction" within the meaning of Treasury Regulation § 1.6011-4(b)(1), such participation has been properly disclosed on IRS Form 8886 (Reportable Transaction Disclosure Statement) and on any corresponding form required under state, local or other law.

(i) During the last three years, no SALIC Group Company has been a party to any transaction treated by the parties thereto as one to which Section 355 of the Code (or any similar provision of state, local or foreign law) applied.

(j) None of the SALIC Group Companies (other than SRLB) has made an election under Section 953(d) of the Code to be treated as a domestic corporation.

(k) Other than as set forth on Section 3.10(k) of the SALIC Disclosure Schedule, (i) none of the SALIC Group Companies has unutilized net operating losses the use of which to offset U.S. federal income tax liability would be restricted or prohibited under Section 1503 of

the Code or Treasury Regulations thereunder, (ii) each Company Group of which any of the SALIC Group Companies is a member does not have and will not have on the Closing Date a consolidated Section 382 limitation (as determined under Treasury Regulation § 1.1502-93) and (iii) other than as a consequence of the Transactions, immediately after the Closing Date no limitation under Section 382 of the Code will apply to the utilization of any net operating loss (or alternative minimum tax net operating loss) carryover or capital loss carryover of any of the SALIC Group Companies carried from a period ending on or prior to the Closing Date.

(l) None of the Transactions are subject to withholding under Section 1445 of the Code or similar provision of state, local or foreign law.

### Section 3.11 Employee Benefits.

(a) Section 3.11(a) of the SALIC Disclosure Schedule sets forth a true, complete, and correct list of each material employee benefit plan (as such term is defined in Section 3(3) of ERISA, whether or not such plans are subject to ERISA) and each other material plan, program or policy providing for equity-based compensation, bonuses, incentive compensation, retention, severance, change in control or fringe benefits, (i) that is sponsored, maintained or contributed to by any of the SALIC Group Companies or any of the ERISA Affiliates immediately prior to the Closing for the benefit of any Employee or (ii) with respect to which Purchaser may have any liability or contingent liability on account of the execution of this Agreement or any Transactions (collectively, the "<u>Benefit Plans</u>"). SALIC has made available to Purchaser prior to the date hereof copies of each Benefit Plan.

(b) Except as would not, individually or in the aggregate, reasonably be likely to have a SALIC Material Adverse Effect, all Benefit Plans comply in form with all requirements of Applicable Law and have been operated and administered in all material respects in accordance with their terms and with all requirements of Applicable Law.

(c) Each Benefit Plan that is an "employee pension benefit plan" within the meaning of Section 3(2) of ERISA that is intended to be qualified under Section 401(a) of the Code, has received a favorable determination or opinion letter from the IRS or has applied to the IRS for a favorable determination letter, and there are no circumstances reasonably likely to result in the loss of the qualification of such plan under Section 401(a) of the Code.

(d) No Benefit Plan is (i) subject to Section 412 or 430 of the Code or Title IV of ERISA, or (ii) a multiemployer plan as defined in Section 3(37) of ERISA.

(e) Other than as set forth in Section 3.11(e) of the SALIC Disclosure Schedule, there are no Actions pending or, to the Knowledge of SALIC, threatened involving any Benefit Plan (other than with respect to routine claims for benefits), by any Person covered thereby or otherwise, which would, individually or in the aggregate, reasonably be likely to have a SALIC Material Adverse Effect.

(f) Other than as set forth in Section 3.11(f) of the SALIC Disclosure Schedule, neither the execution and delivery of this Agreement nor the consummation of the Transactions, either alone or together with subsequent events, will (i) entitle any Employee, director or

independent contractor to any payments or benefits (including severance pay or any increase in severance pay or other compensation or benefits upon any termination of employment or service after the date hereof); (ii) accelerate the time of payment or vesting or result in any payment or funding (through a grantor trust or otherwise) of compensation or benefits under, increase the amount payable or result in any other material obligation pursuant to, any of the Benefit Plans; or (iii) result in payments or benefits under any of the Benefit Plans that would not be deductible under Section 280G of the Code, in each case, for which any SALIC Group Company or Purchaser would reasonably be expected to have any liability.

### Section 3.12 <u>Compliance with Applicable Laws; Governmental Authorizations;</u> <u>Regulatory Filings</u>.

(a) Since January 1, 2014, each of the SALIC Group Companies has been, and the SALIC Group Business has been conducted, in compliance with all Applicable Law in all material respects. Since January 1, 2014, none of the SALIC Group Companies (i) has received any written or, to the Knowledge of SALIC, oral notice from any Governmental Authority regarding any actual or alleged violation of, or failure on the part of such SALIC Group Company to comply with, any Applicable Law that has not been remedied or (ii) to the Knowledge of SALIC, has been placed under investigation with respect to any material violation of or non-compliance with any Applicable Law. None of the SALIC Group Companies is a party to, or bound by, any material Governmental Order applicable to it or its assets, properties, rights or business.

Other than as set forth on Section 3.12(b) of the SALIC Disclosure Schedule, (i) (b) each SALIC Group Company holds all material Governmental Authorizations required to conduct its business in the manner and in all such jurisdictions as it is currently conducted and all such Governmental Authorizations are valid and in full force and effect, (ii) each SALIC Group Company is in compliance in all material respects with all such Governmental Authorizations, (iii) none of the SALIC Group Companies has received, at any time since January 1, 2014, any written notice from any Governmental Authority regarding any actual or alleged violation of, or failure on the part of such SALIC Group Company to comply with, any term or requirement of any such Governmental Authorization that has not been remedied, (iv) no such Governmental Authorization will be terminated as a result of the Transaction, and (v) no SALIC Group Company is the subject of any pending or, to the Knowledge of SALIC, threatened Action seeking the revocation, suspension, termination, modification or impairment of any such Governmental Authorization. Section 3.12(b) of the SALIC Disclosure Schedule sets forth a true, complete and correct list of all licenses and permits to conduct the business of insurance or reinsurance held by any SALIC Group Company. None of the SALIC Group Companies has received any written notice, or to the Knowledge of SALIC, oral or any other communication from any Governmental Authority regarding any material violation of any Governmental Authorization by any SALIC Group Company.

(c) No director or officer of any of the SALIC Group Companies, and to SALIC's Knowledge, no Employee or other employee or Representative of the SALIC Group Companies or any of their Affiliates acting for or on behalf of any SALIC Group Company has, directly or indirectly (i) used any funds for contributions, gifts, gratuities, entertainment or other expenses

related to political activity, in each case in violation of any Applicable Laws, (ii) made any payment in violation of any Applicable Laws or offered, promised or authorized the payment of anything of value, regardless of form, whether in money, property or services, to or for the benefit of any U.S. or non-U.S. government official or employee, any official or employee of a public international organization, or any political party or candidate for political office, in each case in violation of any Applicable Laws and for the purpose of influencing any act or decision of such individual or of any governmental body or public international organization, or securing any improper advantage, in order to obtain or retain business or direct business to any Person in violation of any Applicable Laws, (iii) made any other payment, regardless of form, whether in money, property or services which constitutes criminal bribery under any Applicable Laws, or (iv) violated any applicable export control, money laundering or anti-terrorism law or regulation, the U.S. Foreign Corrupt Practices Act of 1977, as amended, or any other applicable anti-bribery law or regulation, of any applicable jurisdiction, or any Applicable Laws of similar effect.

(d) Since January 1, 2014, the SALIC Group Companies have filed all material reports, statements, documents, registrations, filings and submissions required to be filed with any Governmental Authority, and all such reports, statements, documents, registrations, filings and submissions complied in all material respects with Applicable Law in effect when filed and no material deficiencies have been asserted by, nor any material penalties imposed by, any such Governmental Authorities with respect to such reports, statements, documents, registrations, filings or submissions.

(e) SALIC has made available to Purchaser (i) any material reports of examination (including financial, market conduct and similar examinations) of any Scottish Insurance Company issued by any Insurance Regulatory Authority, in any case, since January 1, 2014, (ii) all material Insurance Holding Company System Act filings or submissions made by any Scottish Insurance Company with any Insurance Regulatory Authority since January 1, 2014 through the date hereof and (iii) all analyses and reports submitted by any Scottish Insurance Company to the Insurance Regulatory Authority in its state or jurisdiction of domicile since January 1, 2014 relating to its risk-based capital calculations. As of the date hereof, all material deficiencies or violations noted in the examination reports described in clause (i) above have been resolved to the reasonable satisfaction of the Insurance Regulatory Authority that noted such deficiencies or violations.

(f) Except for the Scottish Insurance Companies, no Subsidiary of SALIC is a regulated insurance company. None of the Scottish Insurance Companies is "commercially domiciled" under the Applicable Laws of any jurisdiction or is otherwise treated as domiciled in a jurisdiction other than its jurisdiction of organization.

# Section 3.13 Intellectual Property.

(a) Section 3.13(a) of the SALIC Disclosure Schedule sets forth, as of the date hereof, a list of issued or registered Intellectual Property owned by the SALIC Group Companies, including all Trademarks, copyrights and patents owned by any SALIC Group Company that is the subject of a registration or pending application.

(b) Each SALIC Group Company owns or has rights or licenses to use, the Intellectual Property used in the business of such SALIC Group Company as currently conducted, free and clear of all Encumbrances, other than Permitted Encumbrances. Such Intellectual Property is subsisting and, to the Knowledge of SALIC, valid and enforceable. None of the SALIC Group Companies has conducted or is conducting its business in a manner that would reasonably be expected to result in the cancellation or unenforceability of such Intellectual Property.

(c) To the Knowledge of SALIC, the conduct of the business of each SALIC Group Company as currently conducted does not infringe upon, misappropriate, dilute or otherwise violate the Intellectual Property rights of any third party, and none of the SALIC Group Companies has received any written notice of any alleged breach, infringement, misappropriation or dilution or other violation by any SALIC Group Company of the Intellectual Property rights of any third party, in each case where such breach, infringement, misappropriation or dilution is pending and not resolved and except as would not, individually or in the aggregate, reasonably be likely to have a SALIC Material Adverse Effect. To the Knowledge of SALIC, as of the date hereof, no third party is breaching, infringing upon, misappropriating or otherwise violating any material Intellectual Property owned by any SALIC Group Company and no such claims have been made or threatened by any SALIC Group Company.

### Section 3.14 <u>Actuarial Reports; Reserves</u>.

(a) Section 3.14(a) of the SALIC Disclosure Schedule lists (and SALIC has made available to Purchaser true, complete and correct copies of) all material actuarial reports prepared by opining actuaries, independent or otherwise, from and after January 1, 2017, with respect to the business of any Scottish Insurance Company (including all material attachments, addenda, supplements and modifications thereto). There have been no actuarial reports of a similar nature covering any Scottish Insurance Company in respect of any period subsequent to the latest period covered in such actuarial reports. The information and data furnished by Scottish Insurance Companies to their respective independent actuaries in connection with the preparation of any such actuarial reports were compiled from the Books and Records and as of the dates so furnished were true, complete and correct in all material respects for the periods covered in such actuarial reports. To the Knowledge of SALIC, no data provided in connection with the preparation of such actuarial reports was materially inaccurate.

(b) The policy reserves of the Scottish Insurance Companies recorded in the SALIC Financial Statements and the Statutory Statements, as of the date thereof: (a) have been computed in all material respects in accordance with presently accepted actuarial standards consistently applied and prepared in accordance with SAP (in the case of SRUS) and GAAP (in the case of SALIC and SRLB) and FRS 101 (in the case of SRD), consistently applied; (b) have been based on actuarial assumptions that are consistent in all material respects with applicable contract provisions; and (c) meet the requirements of applicable Insurance Laws and SAP in all material respects. Notwithstanding the foregoing provisions of this Section 3.14, SALIC is not

making any representations, express or implied, in or pursuant to this Agreement concerning the adequacy or sufficiency of reserves.

Section 3.15 <u>Insurance</u>. Section 3.15 of the SALIC Disclosure Schedule contains a true, complete and correct list of all currently in effect and most recently expired liability, property and casualty, employee liability, directors and officers liability, surety bonds, key man life insurance and other similar insurance contracts of the SALIC Group Companies that insure the business, properties, operations or affairs of the SALIC Group Companies or affect or relate to the ownership, use or operations of the SALIC Group Companies' assets or properties, and the amount of coverage, insurance carrier, policy number and deductible or self-insured retention under each such insurance contract. To the Knowledge of SALIC, no event has occurred that, with notice or the lapse of time or both, would constitute a breach or default under, or permit termination of, any insurance policy of SALIC, and except as set forth on Section 3.15 of the SALIC Disclosure Schedule, there has been no threatened termination or non-renewal of, or material premium increase with respect to, any insurance policy of SALIC. All premiums due thereunder have been paid when due in all material respects and all such policies are in full force and effect.

Section 3.16 <u>Material Contracts</u>.

(a) Section 3.16(a) of the SALIC Disclosure Schedule contains a true, complete and correct list of each Contract in force as of the date hereof (other than and excluding any Reinsurance Contract or Reserve Financing Contract (which are addressed in Section 3.21)) to which any SALIC Group Company is a party or under which any of the SALIC Group Companies has material continuing obligations as of the date hereof that meets any of the following criteria (each, a "<u>Material Contract</u>"):

(i) requires expenditures by a SALIC Group Company involving consideration in excess of One Hundred Thousand Dollars (\$100,000) in any twelve (12)-month period or the delivery by the SALIC Group Company or their Affiliates of goods or services with a fair market value in excess of One Hundred Thousand Dollars (\$100,000);

(ii) provides for payments or goods or services to be received by a SALIC Group Company in excess of One Hundred Thousand Dollars (\$100,000) in any twelve (12)-month period;

(iii) relates to the incurrence by a SALIC Group Company of any indebtedness, other than such Contracts entailing past or reasonably expected future amounts less than One Hundred Thousand Dollars (\$100,000) in the aggregate;

(iv) relates to the acquisition or disposition by a SALIC Group Company outside the Ordinary Course of Business of any material assets or any material business (whether by merger, sale or purchase of stock, sale or purchase of assets or otherwise) to the extent any actual or contingent material obligations of the SALIC Group Company thereunder remain in effect, other than transactions involving Investment Assets;
(v) grants a right of first refusal or first offer or similar right or materially restricts or limits a SALIC Group Company's ability to freely engage in any business, compete with other entities, market any product or solicit employees or customers, or provides for "exclusivity" or any similar requirement, in each case in favor of any Person other than the subject SALIC Group Company;

(vi) is a capital maintenance Contract, keepwell or similar agreement pursuant to which any Person has agreed to contribute capital or surplus to the SALIC Group Companies or any capital maintenance Contract or similar agreement pursuant to which the SALIC Group Companies have agreed to contribute capital or surplus to any Person or guarantee the obligations of any Person under any insurance Contract;

(vii) is a collective bargaining agreement or other Contract with any labor organization, union or association;

(viii) relates to the license or sublicense to a SALIC Group Company of any material Intellectual Property or the license from a SALIC Group Company of any material Intellectual Property, other than "shrink wrap" or "click through" licenses or licenses of generally-available "off the shelf" computer software or databases;

(ix) is a written Contract with any Employee related to such Employee's employment with any of the SALIC Group Companies;

(x) is a Contract for any joint venture, partnership or similar arrangement, or any Contract involving a sharing of profits, losses, costs or liabilities by the SALIC Group Companies with any other Person or relating to the formation, creation, operation, management or control of any partnership or joint venture in respect of the business of the SALIC Group Companies;

(xi) is an investment management agreement;

(xii) is a material indemnification agreement or guarantee in respect of the business of the SALIC Group Companies;

(xiii) is entered into with any Governmental Authority;

(xiv) provides for exclusivity or any similar requirement or includes a "most favored nation" provision;

(xv) contains change of control provisions;

(xvi) contains notification or termination provisions related to the insolvency of a Debtor, other than to the extent such provision would be unenforceable pursuant to Section 365(e)(1) of the Bankruptcy Code;

(xvii) requires any of the SALIC Group Companies to maintain a minimum rating or has a ratings trigger;

(xviii) provides for any obligation to loan or contribute funds to, or make investments in, another Person;

(xix) is a SALIC Intercompany Agreement or a Terminating Intercompany Agreement;

(xx) is a mortgage, indenture, loan or credit agreement, security agreement or other agreement or instrument relating to the borrowing of money or extension of credit or the direct or indirect guarantee of any obligation for borrowed money of any Person or any other liability in respect of indebtedness for borrowed money of any Person, in each case, involving liabilities in excess of One Hundred Thousand Dollars (\$100,000) or any direct or indirect guarantee of any obligation or liability in respect of a Benefit Plan;

(xxi) provides for a material operation or function of the business of the SALIC Group Companies to be outsourced to, or otherwise performed by, a third Person; or

(xxii) is an obligation to enter into any of the foregoing.

(b) Except as set forth on Section 3.16(b) of the SALIC Disclosure Schedule, with respect to each Material Contract, assuming the due authorization, execution and delivery thereof by the other party or parties thereto, (i) each Material Contract is a valid and binding obligation of the applicable SALIC Group Company and, to the Knowledge of SALIC, as of the date hereof, each other party or parties thereto, in accordance with its terms and is in full force and effect, subject to the Bankruptcy and Equity Exceptions, and is enforceable against the applicable SALIC Group Company and, to the Knowledge of SALIC, each other party thereto in accordance with its terms, subject to the Bankruptcy and Equity Exceptions, (ii) the applicable SALIC Group Company is not, and, to the Knowledge of SALIC, no other party thereto is, in default or breach in any material respect in the performance, observance or fulfillment of any obligation, covenant or condition contained in each of the Material Contracts, and (iii) to the Knowledge of SALIC, there does not exist any event, condition or omission that would constitute such a default or breach (with or without lapse of time or notice or both); provided, however, that this Section 3.16(b) does not apply to any Terminating Intercompany Agreement or any other Material Contract to be rejected or terminated pursuant to this Agreement.

(c) Prior to the date of this Agreement, copies of each Material Contract have been made available to Purchaser.

### Section 3.17 <u>Assets; Real Property</u>.

(a) Except as disclosed in Section 3.17(a) of the SALIC Disclosure Schedule, each of SALIC and the SALIC Subsidiaries has good title to, or a valid leasehold interest in, each of its material assets reflected in the SALIC Financial Statements and the Statutory Statements, as applicable, free and clear of any Encumbrance, except for Permitted Encumbrances.

(b) Section 3.17(b) of the SALIC Disclosure Schedule contains a true, complete and correct list of all leases pursuant to which any SALIC Group Company leases real property as tenant.

(c) Section 3.17(c) of the SALIC Disclosure Schedule contains a true, complete and correct list of all subleases pursuant to which any SALIC Group Company subleases real property as subtenant.

(d) Except as set forth on Section 3.17(d) of the SALIC Disclosure Schedule, immediately following the Closing, neither SRGL nor any of its Affiliates will own or hold, directly or indirectly, an interest in any material asset, property or right of any kind or nature (real or personal, tangible or intangible, including Intellectual Property) necessary to enable the SALIC Group Companies to continue to operate their business as owned by Purchaser after the Closing in substantially the same manner as the business of the SALIC Group Companies has been conducted by the SALIC Group Companies prior to the Closing Date.

Section 3.18 <u>Finders' Fees</u>. Except as set forth on Section 3.18 of the SALIC Disclosure Schedule, there is no investment banker, broker, financial adviser, finder or other intermediary who is or might be entitled to any fee or commission in connection with the Transactions, based on arrangements made by or on behalf of any SALIC Group Company or any of their respective Affiliates.

Section 3.19 <u>Investment Assets</u>.

(a) SALIC has provided to Purchaser a true, complete and correct list of the Investment Assets as of March 31, 2018. The SALIC Group Companies hold valid title to all Investment Assets free and clear of all Encumbrances other than Permitted Encumbrances.

(b) Except as set forth on Section 3.19(b) of the SALIC Disclosure Schedule, to the Knowledge of SALIC, none of the SALIC Group Companies or any of their Affiliates (i) has received written notice that any of the Investment Assets is in default in any payment of principal, distributions, interest, dividends or any other material payment or performance obligation thereunder or (ii) is aware of any breach of, or default under, any covenants of any of the Investment Assets.

(c) Except as set forth on Section 3.19(c) of the SALIC Disclosure Schedule, none of the SALIC Group Companies has any material funding obligations of any kind, or material obligation to make any additional advances or investments (including any obligation relating to any currency or interest rate swap, hedge or similar arrangement), in respect of any of the Investment Assets. There are no material outstanding commitments, options, put agreements or other arrangements relating to the Investment Assets to which any of the SALIC Group Companies may be subject upon or after the Closing.

(d) SALIC has made available to Purchaser true, complete and correct copies of the investment guidelines and policies applicable to the SALIC Group Companies in effect as of the date hereof (the "<u>Investment Guidelines and Policies</u>"). No material changes have been made to such Investment Guidelines and Policies since December 31, 2016.

Section 3.20 Labor Matters.

(a) None of the SALIC Group Companies is a party to, or bound by, any agreement with respect to the Employees with any labor union or any other employee organization, group or association organized for purposes of collective bargaining. To the Knowledge of SALIC, there are, and since January 1, 2014 there have been, (i) no activities or proceedings of any labor union to organize any Employees or employees of SALIC dedicated to the business of the SALIC Group Companies.

(b) Section 3.20(b) of the SALIC Disclosure Schedule sets forth a true, complete, and correct list of each employee of the SALIC Group Companies and in the case of each such employee, the following information, if applicable, as of the date hereof: (i) title or position; (ii) date of hire; (iii) whether full-time or part-time and whether exempt or non-exempt; and (iv) whether absent from active employment or service and if so, the date such absence commenced, and the anticipated date of return to active employment or active service.

SALIC has provided or otherwise made available to Section 3.21 Reinsurance. Purchaser true, complete and correct copies of (a) each material reinsurance or retrocessional treaty, coinsurance, yearly or monthly renewable term, modified coinsurance, excess insurance, ceding of insurance, assumption of reinsurance or similar arrangements, placements or Contracts (together with all amendments, extensions, renewals, guaranties, modifications, waivers, supplements and other agreements, if any, related thereto) to which any Scottish Insurance Company are parties, whether as ceding company, retrocedent, reinsurer or retrocessionaire (the "Reinsurance Contracts") and (b) each material Contract, other than a Reinsurance Contract, comprising an excess reserve financing arrangement to which any of the Scottish Insurance Companies are parties (the "Reserve Financing Contracts"). Each of the Reinsurance Contracts and the Reserve Financing Contracts constitutes a valid and binding obligation of the Scottish Insurance Companies and, to the Knowledge of SALIC, each other party thereto, is enforceable against the Scottish Insurance Companies and, to the Knowledge of SALIC, each other party thereto in accordance with its terms (subject to the Bankruptcy and Equity Exceptions), and is in full force and effect, except for such failures to be valid and binding as are not, individually or in the aggregate, reasonably likely to result in a SALIC Material Adverse Effect. None of the applicable Scottish Insurance Companies or, to the Knowledge of SALIC, any counterparty to any Reinsurance Contract or Reserve Financing Contract is (with or without notice or lapse of time or both) in default or breach under the terms of such Reinsurance Contract or Reserve Financing Contract in any material respect. Except as set forth on Section 3.21(a) of the SALIC Disclosure Schedule, there are no pending or, to the Knowledge of SALIC, threatened Actions with respect to any Reinsurance Contract or Reserve Financing Contract. Except as set forth on Section 3.21(b) of the SALIC Disclosure Schedule, as of the date hereof, no party to any Reinsurance Contract or Reserve Financing Contract has given notice of termination (provisional or otherwise) in respect of any such Contract. Except as set forth on Section 3.21(c) of the SALIC Disclosure Schedule, since January 1, 2014, there has not been any dispute with respect to any material amounts recoverable or payable by any of the Scottish Insurance Companies pursuant to any Reinsurance Contract or Reserve Financing Contract and no reinsurer or ceding party has sought to deny or limit coverage or revoke, terminate, rescind or change, in accordance with the terms of any Reinsurance Contract, reinsurance premiums or expense allowances. Except as set forth on Section 3.21(d) of the SALIC Disclosure Schedule, no Reinsurance Contract or Reserve Financing Contract contains any provision providing that the other party

thereto may terminate or otherwise modify such Reinsurance Contract or Reserve Financing Contract by reason of the Transactions. No Reinsurance Contract or Reserve Financing Contract contains any provision which by its own terms would result in a modification in the operation of such Reinsurance Contract or Reserve Financing Contract by reason of the Transactions.

Section 3.22 <u>Affiliate Transactions</u>. Section 3.22 of the SALIC Disclosure Schedule sets forth a true, complete, and correct list, as of the date hereof, of all Contracts, agreements, leases, licenses and other instruments (whether or not reduced to writing) (i) between any of the SALIC Group Companies, on the one hand, and any other SALIC Group Companies, on the other hand (the "<u>SALIC Intercompany Agreements</u>") and (ii) between any SALIC Group Company, on the one hand, and any Affiliate thereof that is not a SALIC Group Company, on the other hand (the "<u>Terminating Intercompany Agreements</u>") and, with respect to such Terminating Intercompany Agreements, Section 3.22 of the SALIC Disclosure Schedule shall also set forth SALIC's good faith estimate of the settlement or termination amounts due and payable in connection with the termination thereof in accordance with Section 5.13.

Section 3.23 <u>Environmental Matters</u>. None of the SALIC Group Companies has received a written notice, request for information, claim or demand from any Governmental Authority or third party alleging liability in connection with the violation of any Environmental Law, there are no material judicial or administrative proceedings pending or threatened against the SALIC Group Companies arising under or relating to an Environmental Law, and each of the SALIC Group Companies is and has been in compliance in all material respects with any applicable Environmental Laws.

### Section 3.24 Information Technology, Data Security and Privacy.

(a) The Information Technology (i) is in good repair and operating condition and is adequate and suitable (including with respect to working condition, security, performance and capacity) for the purposes for which it is being used or held for use and (ii) does not contain any Malware that would reasonably be expected to interfere with the conduct of the business of the SALIC Group Companies or present a material risk of unauthorized access, disclosure, use, corruption, destruction or loss of any personally identifiable information, data or non-public information.

(b) The SALIC Group Companies (i) have implemented, maintain, and comply with commercially reasonable written security, business continuity and backup and disaster recovery plans and procedures with respect to the Information Technology and (ii) have taken commercially reasonable steps to test such plans and procedures on no less than an annual basis, and such plans and procedures have been proven effective upon such testing in all material respects.

(c) The SALIC Group Companies have been and are in compliance with any such privacy statement (as applicable to any given set of personally identifiable information, data and non-public information collected by or on behalf of any of them) and with any and all Applicable Law, regulatory guidelines, PCI-DSS standards, contractual requirements and terms of use pertaining to such personally identifiable information, data and non-public information.

(d) Since January 1, 2014, the SALIC Group Companies have not, nor, to the Knowledge of SALIC or any of its Affiliates, has any third Person working on behalf of any of them, in each case with respect to the business of the SALIC Group Companies, received any written claims, notices or complaints regarding their information practices or the use of any personally identifiable information, data and non-public information of individuals, or alleging a violation of any individual's privacy, personal or confidentiality rights under any applicable privacy statement or otherwise from any Person.

Section 3.25 <u>Third-Party Administrators</u>. To the Knowledge of SALIC, from January 1, 2014 to the date hereof, each third-party administrator that managed or administered insurance business for the SALIC Group Companies, at the time such Person managed or administered such business, was duly licensed or registered as required by Applicable Law (for the type of business managed or administered on behalf of the Scottish Insurance Companies), and to the Knowledge of SALIC, no such third-party administrator is in violation (or with or without notice or lapse of time or both, would be in violation) of any term or provision of any Applicable Law related to the administration or management of insurance business for the Scottish Insurance Companies, except for such failures to be licensed or such violations which have been cured, resolved or settled through agreements with applicable Governmental Authorities or are barred by an applicable statute of limitations or, individually or in the aggregate, are not material.

Section 3.26 <u>Orders</u>. Prior to the date hereof, the Bermuda Court and/or the Cayman Court have approved SRGL's entry into the Restructuring Implementation Agreement to the extent such approvals are required in connection with the Winding Up Proceedings.

Section 3.27 <u>NO OTHER REPRESENTATIONS OR WARRANTIES</u>. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS <u>ARTICLE III</u> AND IN <u>ARTICLE VI</u>, SALIC DOES NOT MAKE ANY OTHER EXPRESS OR IMPLIED REPRESENTATION OR WARRANTY WITH RESPECT TO SALIC, THE SALIC SUBSIDIARIES OR THE SALIC GROUP BUSINESS OR WITH RESPECT TO ANY OTHER MATTER OR ANY OTHER INFORMATION PROVIDED TO PURCHASER OR ITS REPRESENTATIVES OR AFFILIATES IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY.

#### **ARTICLE IV**

#### **REPRESENTATIONS AND WARRANTIES OF PURCHASER**

Purchaser represents and warrants to SALIC, as of the date hereof and as of the Closing Date, as follows. Each such representation and warranty is qualified by and includes the disclosure set forth in the numbered or lettered sections or subsections of the Purchaser Disclosure Schedule that correspond to such representation and warranty and shall be deemed to be qualified by and include any disclosure in any other section or subsection of the Purchaser Disclosure Schedule to which the relevance of such disclosure to such representation and warranty is readily apparent.

Section 4.1 <u>Organization and Authority</u>. Purchaser is a limited liability company duly organized, validly existing and in good standing under the laws of Delaware. Purchaser has all

requisite corporate power and authority to execute and deliver this Agreement and to perform its obligations hereunder.

Section 4.2 <u>Binding Effect</u>. The execution and delivery of this Agreement by Purchaser, the performance of its obligations hereunder and the consummation of the Transactions have been duly and validly approved by all requisite corporate action on the part of Purchaser and no additional corporate proceedings on the part of Purchaser or any Affiliate thereof or any of their respective securityholders are necessary to approve or authorize, as applicable, this Agreement, the performance of Purchaser's obligations hereunder or the consummation of the Transactions. Assuming the due authorization, execution and delivery by SALIC, this Agreement constitutes the valid and binding obligation of Purchaser, enforceable against Purchaser in accordance with its terms, subject to the Bankruptcy and Equity Exceptions.

Section 4.3 <u>Governmental Filings and Consents</u>. No consents or approvals of, waivers from, or filings or registrations with, any Governmental Authority are required to be made or obtained at or prior to the Closing by Purchaser or any of its Affiliates in connection with the execution, delivery or performance by Purchaser of this Agreement or to consummate the Transactions, except for the consents, approvals, waivers, filings and registrations described in Section 4.3 of the Purchaser Disclosure Schedule, except as would not, individually or in the aggregate, reasonably be likely to have a Purchaser Material Adverse Effect.

Section 4.4 <u>No Violations</u>. Subject to the making of the filings and registrations and receipt of the consents, approvals and waivers referred to in Section 4.3 and the expiration of related waiting periods, the execution, delivery and performance of this Agreement by Purchaser and the consummation of the Transactions do not and will not (a) conflict with, constitute a breach or violation of, or a default under, or give rise to any Encumbrance (other than Permitted Encumbrances) or any acceleration of remedies, penalty, increase in benefit payable or right of termination, suspension, revocation or cancellation under, or forfeiture of, as applicable, any Applicable Law, Order or Governmental Authorization or Contract of Purchaser, except as would not, individually or in the aggregate, reasonably be likely to have a Purchaser Material Adverse Effect or (b) constitute a breach or violation of, or a default under, the organizational documents of Purchaser.

Section 4.5 <u>Purchaser Impediments</u>. As of the date hereof, there is no Action pending or, to the Knowledge of Purchaser, threatened, or any outstanding Order against Purchaser or any of its Affiliates which (a) would affect in any material respect the validity or enforceability of this Agreement, (b) seeks to enjoin or prohibit the consummation of the Transactions or (c) would (i) materially impair or materially delay the ability of Purchaser to obtain Required Approvals (other than the Bankruptcy Court Approvals and Foreign Court Approvals) or (ii) individually or in the aggregate, reasonably be likely to have a Purchaser Material Adverse Effect. As of the date hereof, to the Knowledge of Purchaser, there are no facts or circumstances involving Purchaser or its Affiliates (or involving any permitted assignee of the rights of Purchaser hereunder pursuant to <u>Section 9.4</u>) that would reasonably be expected to prohibit or materially delay the granting of any Required Approval (other than the Bankruptcy Court Approvals and Foreign Court Approvals). Section 4.6 <u>Finders' Fees</u>. There is no investment banker, broker, financial advisor, finder or other intermediary who is or might be entitled to any fee or commission in connection with the Transactions, based on arrangements made by or on behalf of Purchaser or its Affiliates.

Section 4.7 <u>Financial Capability</u>. Purchaser has sufficient capital commitments, and will have at the Closing, sufficient funds to complete the Transactions on the terms and subject to the conditions set forth in this Agreement.

Section 4.8 <u>Purchase for Own Account</u>.

(a) Purchaser is an "accredited investor" as that term is defined in Rule 501(a) of Regulation D under the Securities Act.

(b) Purchaser is acquiring the Purchaser Shares for investment and not with a view toward, or for sale in connection with, any distribution thereof, nor with any present intention of distributing or selling the Purchaser Shares. Purchaser agrees that the Purchaser Shares may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act and any applicable state securities laws, except pursuant to an exemption from such registration under such Applicable Laws.

(c) Purchaser is able to bear the economic risk of holding the Purchaser Shares for an indefinite period, including a complete loss of its investment in the Purchaser Shares, and has knowledge and experience in financial and business matters such that it is capable of evaluating the risks of an investment in the Purchaser Shares.

Section 4.9 <u>NO OTHER REPRESENTATIONS OR WARRANTIES</u>. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS <u>ARTICLE IV</u>, NEITHER PURCHASER NOR ANY OTHER PERSON MAKES ANY OTHER EXPRESS OR IMPLIED REPRESENTATION OR WARRANTY WITH RESPECT TO PURCHASER OR ANY OF ITS AFFILIATES OR WITH RESPECT TO ANY OTHER MATTER OR ANY OTHER INFORMATION PROVIDED TO SALIC OR THE SALIC GROUP COMPANIES OR THEIR REPRESENTATIVES OR AFFILIATES IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY.

### ARTICLE V

### **COVENANTS**

### Section 5.1 <u>Access; Confidentiality</u>.

(a) Prior to the Closing, SALIC shall permit Purchaser and its representatives to have reasonable access, during regular business hours and upon reasonable advance notice to SALIC, to the Books and Records to the extent not prohibited by Applicable Law, for any reasonable business purpose relating to this Agreement, except that SALIC shall use reasonable best efforts to provide access to Books and Records maintained by any Person acting as a third party administrator for any SALIC Group Business; <u>provided</u>, that to the extent any Books and Records or other information that is subject to an attorney-client or other legal privilege or

KL2 3073884.8

obligation of confidentiality or non-disclosure, it shall not be made so accessible, it being understood that SALIC shall use its reasonable best efforts to make other arrangements (including redacting information or entering into joint defense agreements), in each case, that would enable any otherwise required disclosure to Purchaser to occur without so jeopardizing privilege or contravening such privilege or obligation. Such access shall be at Purchaser's sole cost and expense and may not unreasonably interfere with the conduct of the SALIC Group's or its Affiliates' businesses.

(b) Purchaser acknowledges that the information and access provided to it pursuant to Section 5.1(a) shall be subject to the terms and conditions of the Confidentiality Agreement.

Following the Closing Date, to the extent not prohibited by Applicable Law, (c) Purchaser shall (i) (A) permit the SRGL Group Companies, the Distribution Trustee and their respective Representatives, during regular business hours and upon reasonable advance notice to Purchaser, to examine and make copies of the Books and Records and (B) make the Employees available to the SRGL Group Companies and their respective representatives as the SRGL Companies and their representatives shall from time to time reasonably request, in each case, for any reasonable business purpose, including the preparing or examination of the SRGL Group Companies regulatory and Tax filings and financial statements and the conduct of any third party litigation or dispute resolution (not involving Purchaser or any of its Affiliates), or regulatory dispute, whether pending or threatened, concerning the business of the SALIC Group Companies prior to the Closing; and (ii) maintain the Books and Records for the foregoing examination and copying for a period of not less than six (6) years following the Closing Date or such longer period as may be required by Applicable Law. Such access to the Books and Records shall be at the sole cost and expense of the applicable SRGL Group Companies and may not unreasonably interfere with the conduct of Purchaser's or its Affiliates' businesses.

### Section 5.2 <u>Conduct of Business</u>.

(a) Except (A) as required by the Bankruptcy Code, the Bankruptcy Rules, pursuant to an Order of the Bankruptcy Court, Applicable Bermuda Law or Applicable Cayman Islands Law, (B) as expressly required by this Agreement, (C) as set forth on Section 5.2(a)(C) of the SALIC Disclosure Schedule or (D) with the prior written consent of Purchaser (such consent not to be unreasonably withheld, conditioned or delayed), from the date of this Agreement until the Closing Date, SALIC shall cause the SALIC Group to be operated in the Ordinary Course of Business, and to the extent consistent therewith, use reasonable best efforts to preserve intact its current business organization and its material relationships, including by (x) preserving relationships with third parties (including agents, brokers, producers, Governmental Authorities, reinsurance and retrocession counterparties, suppliers and others having business dealings with them) and employees, consultants and agents (including by keeping available the services of officers and key employees) (y) maintaining the Assumed Contracts and Governmental Authorizations, and (z) complying with all obligations under all Material Contracts, Reinsurance Contracts, Reserve Financing Contracts, including providing any required notifications under such Contracts as a result of the Transactions, Chapter 11 Cases or otherwise. Without limiting the generality of the foregoing, from the date hereof until the Closing Date, SALIC shall not, and shall cause the SALIC Subsidiaries not to, except as permitted by the preceding sentence, take any of the following actions:

(i) (A) issue, sell, transfer, pledge, grant, dispose of, encumber or deliver any equity securities of any class or any securities convertible into or exercisable or exchangeable for voting or equity securities of any class of stock of the SALIC Group Companies or grant options, warrants, calls or other rights to purchase or otherwise acquire equity securities of any class of stock of the SALIC Group Companies, (B) adjust, split, combine, recapitalize or reclassify any of its equity securities, in each case of any of the SALIC Group Companies or (C) declare, set aside or pay any dividend or other distribution (whether in cash, stock or property) with respect to any outstanding capital stock;

(ii) acquire (by merger, consolidation, acquisition of stock or assets, bulk reinsurance or otherwise) any corporation, partnership, joint venture, association or other business organization or division thereof, or substantially all of the assets of any of the foregoing except for acquisitions of investment assets in the Ordinary Course of Business, in accordance with the Investment Guidelines and Policies;

(iii) amend the organizational documents of any SALIC Group Company;

(iv) adopt a plan of complete or partial liquidation or rehabilitation or authorize or undertake a merger, dissolution, rehabilitation, consolidation, restructuring, recapitalization, or other reorganization;

(v) other than in the Ordinary Course of Business, sell, lease, transfer, assign or otherwise dispose of any material assets in any individual transaction in excess of One Hundred Thousand Dollars (\$100,000) or in the aggregate in excess of Two Hundred Thousand Dollars (\$200,000), or permit any such material assets to become subject to any Encumbrance other than Permitted Encumbrances;

(vi) amend (in any material respect), or terminate (other than at its stated expiry date) any Material Contract or Reinsurance Contract or commute or recapture any Reinsurance Contract or enter into any Contract which, if entered into prior to the date hereof, would have been a Material Contract or Reinsurance Contract;

(vii) except as required by GAAP, SAP, FRS 101 or by Applicable Law, change any of the accounting principles or practices used by any of the SALIC Group Companies;

(viii) prepare or file any income or other material Tax Return inconsistent with past practice, make, change or revoke any income or other material Tax election, change any annual Tax accounting period, adopt or change any accounting method with respect to income or other material Taxes, file any amended income or other material Tax Return, enter into any closing agreement with respect to income or other material Taxes, settle or compromise any proceeding with respect to any income or other material Tax claim or assessment relating to any of the SALIC Group Companies, surrender any right to claim a material refund of Taxes, consent to any extension or waiver of the limitation period applicable to any income or other material Tax claim or assessment relating to any of the SALIC Group Companies, or request any ruling or similar guidance with respect to Taxes;

(ix) institute or promise to institute any new Benefit Plan for any Employee or increase or grant, accelerate, or promise to increase the annual level of compensation of any Employee, increase the coverage or benefits available under any Benefit Plan or otherwise modify or amend or terminate any such Benefit Plan, in each case, in any material respect and other than in the Ordinary Course of Business (including any merit increases awarded in accordance with past practice), as required by Applicable Law, or pursuant to the terms of an existing Contract or Benefit Plan;

(x) (A) terminate the employment relationship with any Key Employee other than for cause, enter into or amend in any material respect any employment contracts with Key Employees, (B) hire any Employee or promote any Employee other than in the Ordinary Course of Business, (C) increase or promise to increase the base salary or target bonus percentage or opportunity of any Employee except as required by the terms of any Benefit Plan as in effect on the date hereof, or in the Ordinary Course of Business;

(xi) subject any of the properties or assets (whether tangible or intangible) of the SALIC Group Companies to any Encumbrance (except for Permitted Encumbrances);

(xii) enter into any Contract for expenditures in excess of One Hundred Thousand Dollars (\$100,000) for any individual Contract or Two Hundred Thousand Dollars (\$200,000) for all Contracts and commitments in the aggregate;

(xiii) enter into any Contract or commitment which restrains, restricts, limits or impedes the ability of the SALIC Group Companies to compete with or conduct any business or line of business in any geographic area; and

(xiv) abandon, modify, waive, surrender, terminate or withdraw any Governmental Authorization, or change any Scottish Insurance Company's state of domicile;

(xv) make any material change to the Investment Guidelines and Policies or in the underwriting, claims administration, investment, reserving, hedging or financial accounting policies, practices or principles of the SALIC Group Companies, as applicable, in effect on the date hereof (other than any change required by GAAP, SAP or FRS 101), or fail to comply with the Investment Guidelines and Policies;

(xvi) incur any indebtedness to any Person for borrowed money (other than current trade accounts payable incurred in respect of property or services purchased in the Ordinary Course of Business) or assume, grant, guarantee or endorse, or otherwise as an accommodation become responsible for, the obligations of any Person, or make any loans or advances (other than, in each case, in respect of transactions in the Investment Assets, in the Ordinary Course of Business and in accordance with the Investment Guidelines and Policies);

(xvii) settle any Action involving any SALIC Group Companies, other than any such settlement that is solely a monetary settlement that requires payment by any of the SALIC Group Companies of less than Two Hundred Thousand Dollars (\$200,000); provided that none of the SALIC Group Companies admits a breach or violation of Applicable Law or any contractual obligation;

(xviii) modify the terms of, or default under, any indebtedness or, other than in the Ordinary Course of Business, cancel or compromise any material indebtedness or waive any material rights without receiving a realizable benefit of similar or greater value;

(xix) fail to pay or satisfy when due any material liability of any of the SALIC Group Companies (other than any such liability that is being contested in good faith);

(xx) enter into any new line of business, introduce any new products, or change in any material respect existing products;

(xxi) fail to timely file with Governmental Authorities all required annual and quarterly statutory financial statements and other material insurance regulatory filings;

(xxii) make any material change to any of the actuarial, investment, reserving, hedging, underwriting, claims administration, marketing or reinsurance policies, practices, guidelines or principles of any SALIC Group Companies, in each case, other than as required by GAAP or SAP; or

(xxiii) agree or commit to do, or resolve, authorize or approve any action to do, any of the foregoing.

(b) From the date hereof until the Closing, SALIC shall deliver to Purchaser copies of any material correspondence received by any of the SALIC Group Companies from any Governmental Authority (other than routine administrative matters), including any such correspondence that contains any assertion of any failure of any of the SALIC Group Companies to comply with Applicable Law in any material respect, notice of any pending audit, examination or investigation, or any findings or conclusions arising out of any such audit, examination or investigation, and copies of any material correspondence sent by any of the SALIC Group Companies to any Governmental Authority relating to any such matters; provided that all such correspondence provided to Purchaser pursuant to this Section 5.2(b) shall be subject to the terms and conditions of the Confidentiality Agreement.

# Section 5.3 <u>Reasonable Best Efforts; Regulatory Matters; Third-Party Consents</u>.

(a) Subject to the terms and conditions of this Agreement, each party agrees to use its reasonable best efforts to take, or cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper or advisable to consummate and make effective, as soon as practicable

after the date of this Agreement, the Transactions, including using reasonable best efforts to (i) lift or rescind any injunction or restraining order or other Order adversely affecting the ability of the parties to consummate the Transactions, and (ii) prevent the initiation of any Action seeking to enjoin, prevent or delay the consummation of the Transactions. Each party shall use its reasonable best efforts, and shall cooperate fully with each other, (i) to comply as promptly as practicable with all requirements of Government Authorities applicable to the Transactions and (ii) to obtain as promptly as practicable all necessary permits, Orders or other consents, approvals or Governmental Authorizations and consents or waivers of all other third parties necessary in connection with the consummation of the Transactions (including those set forth in Section 3.5 of the SALIC Disclosure Schedule or Section 4.3 of the Purchaser Disclosure Schedule), subject, in the case of third party consents, to the last sentence of this Section 5.3(a). In connection therewith, subject to the terms and conditions of this Agreement, SALIC and Purchaser each shall make and shall cause its Affiliates to make all legally required filings as promptly as practicable in order to facilitate prompt consummation of the Transactions, shall provide and shall cause their respective Affiliates to provide such information and communications to Governmental Authorities as such Governmental Authorities may request to the extent permitted by Applicable Law, and shall consent to and comply with any condition imposed by any Governmental Authority on its grant of any such permit, Order, consent, approval or authorization. The SALIC Parties shall use commercially reasonable efforts (taking into account their status as Debtors) to obtain the consent or waiver of any Person that is a counter-party to any Material Contract, Reinsurance Contract or Reserve Financing Contract to extent such Contract contains an automatic termination or such Person has a right to take an action adverse to any SALIC Group Company or Purchaser as a result of the Transactions or Chapter 11 Cases ("Third Party Consent Contracts"), it being understood that no such consent or waiver shall be a condition to the Closing. If the SALIC Parties fail to obtain the consent or waiver of any Person that is a counter-party to a Third Party Consent Contract, the SALIC Parties shall use commercially reasonable efforts and shall work with Purchaser in good faith to procure a replacement Contract having substantially equivalent terms and conditions.

(b) No later than ten (10) Business Days after the issuance of the Winning Bidder Order (but subject to the proviso below), each party shall, and shall cause its Affiliates to, make all filings and notifications with all Governmental Authorities that may be or may become reasonably necessary, proper or advisable under this Agreement and Applicable Law to consummate and make effective the Transactions, including: (i) Purchaser causing a "Form A" or a similar change of control application to be filed in each jurisdiction where required by applicable Insurance Laws with respect to the Transactions; (ii) Purchaser causing "Form E" or similar market share notifications to be filed in each jurisdiction where required by applicable Insurance Laws with respect to the Transactions; (iii) each party making any other filing that may be required under any other antitrust or competition laws or by any Governmental Authority with jurisdiction over enforcement of any applicable antitrust or competition laws; and (iv) each party making any other filing that may be required under any insurance, financial services or other Applicable Law or by any Governmental Authority with jurisdiction over enforcement of any such Applicable Law; provided that (A) with respect to the filings to be made by Purchaser and its Affiliates pursuant to this Section 5.3(b), Purchaser and its Affiliates shall use their reasonable best efforts to submit such filings no later than ten (10) Business Days after the issuance of the Winning Bidder Order and in any case shall submit such filings no later than fifteen (15) Business Days after issuance of the Winning Bidder Order, (B) with respect to the filings to be made by SALIC and its Affiliates listed in Section 3.5(a)(1) the SALIC Disclosure Schedule, SALIC and its Affiliates shall make such filings no later than five (5) Business Days after the date on which Purchaser and its Affiliates submit the filings described in clause (i) above, and (C) with respect to the filings to be made by SALIC and its Affiliates listed in Section 3.5(a)(3) of the SALIC Disclosure Schedule, SALIC and its Affiliates shall make such filings no later than five (5) Business Days after the date on which Purchaser and its Affiliates submit the filings to be made by Purchaser and its Affiliates listed in Section 4.3(3) of the Purchaser Each party shall promptly supply any additional information and Disclosure Schedule. documentary material that may be requested by any Governmental Authority pursuant to Applicable Law. Notwithstanding anything to the contrary in this Agreement, Purchaser shall have responsibility for the filing and other fees associated with its "Form A" or similar change of control applications and its "Form E" or similar market share notifications, and SALIC and Purchaser shall have responsibility for their other respective filing fees associated with any other filings required to be made in connection with the Transactions.

Subject to Applicable Law relating to the sharing of information, each party shall (c) promptly notify the other parties of any communication it or any of its Affiliates receives from any Governmental Authority relating to the matters that are the subject of this Agreement, and shall permit counsel for the other parties to review in advance, and consider in good faith the views of the other parties in connection with, any proposed written communication to any Governmental Authority relating to the Transactions, and provide each other party with copies of all correspondence, filings or communications between such party or any of its Representatives, on the one hand, and any Governmental Authority or members of the staff of any Governmental Authority, on the other hand, subject to Section 5.1; provided that no party shall be required to share with or disclose to the other any of its or its Affiliates' confidential competitive information or any personal identifiable information of their respective officers, directors or other applicable individuals. Neither SALIC nor Purchaser may participate or agree to participate in any meeting with any Governmental Authority relating to the matters that are the subject of this Agreement (other than telephone calls initiated by such Governmental Authority and not scheduled in advance or ministerial telephone calls not expected to involve a substantive discussion of the Transactions) unless it consults with the other party in advance and, to the extent permitted by such Governmental Authority, gives the other party a reasonable opportunity to attend and participate at such meeting. Subject to the Confidentiality Agreement and Section 5.1, each party shall coordinate and cooperate fully with each other party in exchanging such information and providing such assistance as such other party may reasonably request in connection with the foregoing; provided, however, that the foregoing shall not require either party or any of their Affiliates to (i) disclose any information that in the reasonable judgment of such party or its Affiliates, as the case may be, would result in the disclosure of any trade secrets of third parties or violate any of its contractual obligations or obligations with respect to confidentiality or (ii) disclose any privileged information or confidential competitive information of such party or its Affiliates, or personally identifiable information of their respective officers, directors or other applicable individuals. Further, no party shall be required to comply with any of the foregoing provisions of this Section 5.3(c) to the extent that such compliance would be prohibited by Applicable Law. The parties further covenant and agree not to extend any waiting period associated with any Governmental Authorization or enter into any agreement with any

Governmental Authority not to consummate the Transactions, except with the prior written consent of the other party.

Notwithstanding anything herein to the contrary, in connection with any Required (d) Approval (other than the Bankruptcy Court Approvals and Foreign Court Approvals), Purchaser shall not be obligated to take or refrain from taking or to agree to it or its Affiliates taking or refraining from taking, any action or to permit or suffer to exist any restriction, condition, limitation or requirement imposed by a Governmental Authority that, individually or in the aggregate with any other actions, conditions, limitations, restrictions or requirements imposed by Governmental Authorities, would or would reasonably be likely to result in a Burdensome Condition. A "Burdensome Condition" means: (i) a material negative effect on the business or the assets, liabilities, properties, operations, results of operations or condition (financial or otherwise) of any of the SALIC Group Companies or Purchaser or any of their respective Affiliates; (ii) any requirement to sell, divest, operate in a specified manner, hold separate or discontinue or limit, before or after the Closing Date, any material assets, liabilities, businesses, operations, or interest in any assets or businesses of Purchaser, any SALIC Group Company or any of their respective Affiliates; or (iii) any requirement relating to contribution of capital, keep-well or capital maintenance arrangements or maintaining risk based capital level or any restrictions on dividends or distributions; provided, however, that neither (x) any requirement to allocate all or any portion of the Recapitalization Funding Payment to one or more particular SALIC Group Companies, nor (y) any restriction, condition, limitation or requirement imposed by a Governmental Authority, in effect as of the date hereof and listed in Section 5.3(d) of the SALIC Disclosure Schedule, shall be considered a Burdensome Condition. Without the prior written consent of Purchaser, SALIC shall not, and shall cause its Affiliates not to, take or refrain from or to agree to the taking or refraining from any action or to permit or suffer to exist any restriction, condition, limitation or requirement that would or would reasonably be expected to result, individually or in the aggregate, in a Burdensome Condition being imposed by a Governmental Authority.

(e) Except as set forth in Section 5.3(a), each party shall use its commercially reasonable efforts to obtain, as promptly as practicable after the date hereof, all consents and approvals, including any Governmental Authorization contemplated by this Section 5.3, and to make any notifications, in each case that may be required by or of it in connection with the Transactions, including consenting to and complying with any condition imposed by any Governmental Authority on its grant of any such Governmental Authorization; provided that no SALIC Group Company shall be required to compensate any third party, commence or participate in litigation or offer or grant any accommodation (financial or otherwise) to any third party to obtain any such consent or approval; and provided, further, that for purposes of this Section 5.3(e), SALIC shall not be required to take any action with respect to any third party unless the taking of such action is required in order to satisfy any of the conditions set forth in Section 7.1 or 7.2 and such condition has not been waived by Purchaser.

Section 5.4 <u>Employee Matters</u>.On the Closing and for the remainder of 2018, Purchaser shall not take any action directly prohibiting the SALIC Group Companies' ability to provide to each Employee while in the employ of a SALIC Group Company (i) base compensation and annual bonus potential and (ii) Benefit Plans (other than severance benefits, which are addressed in <u>Section 5.4(b)</u>), which are no less favorable in the aggregate, in a material manner, than that which was provided to such Employee immediately prior to the Closing.

(b) Purchaser shall not take any action directly prohibiting the SALIC Group Companies' ability to provide any Employee whose employment is terminated for any reason, other than for cause, within twenty-four (24) months after the Closing a severance benefit package substantially similar to the severance benefit package that would be provided under the applicable severance plan for the SALIC Group Companies or any other contractual severance obligations as in effect for such Employee immediately prior to the Closing and disclosed in Section 5.4(b) of the SALIC Disclosure Letter.

(c) From and after the Closing, Purchaser shall not take any action directly prohibiting the SALIC Group Company's ability to honor, pay, perform and satisfy any and all liabilities, obligations and responsibilities to or in respect of each Employee or former Employee under the terms of each employment or service agreement, retention plan and each other arrangement between such SALIC Group Company and any such Employee or former Employee, in each case as in effect or existing immediately prior to the Closing and disclosed in Section 5.4(c) of the SALIC Disclosure Letter.

(d) Prior to Closing, Purchaser shall consult with, and seek the recommendations of, the Chief Executive Officer and Chief Financial Officer of SALIC, with respect to (i) a plan for the retention, following the Closing, of Employees and (ii) the terms and conditions of the employment packages and employee benefits to be made available to Employees the Purchaser elects to retain.

(e) Notwithstanding the foregoing, this Section 5.4 is not intended to and shall not give any third party, including any Employee or any agent or representative thereof any right to (i) enforce the provisions of this Section 5.4, (ii) require Purchaser or the SALIC Group Companies to continue to employ any Employee, or (iii) require Purchaser or the SALIC Group Companies to continue any Benefit Plan beyond the time when it otherwise lawfully could be terminated or modified.

(f) Notwithstanding the foregoing, each of the parties acknowledges and agrees that nothing in this Section 5.4 is intended to or shall (i) without limiting Purchaser's obligations above, limit Purchaser's ability to manage the SALIC Group Companies, in consultation with management of the SALIC Group Companies, in Purchaser's sole discretion on and after the Closing, or (ii) require Purchaser to contribute any funds to any of the SALIC Group Companies or incur any liability in satisfaction of the obligations set forth in this Section 5.4.

Section 5.5 <u>Further Assurances</u>. The parties shall, subject to the terms and conditions of this Agreement (a) execute and deliver, or shall cause to be executed and delivered, such documents, certificates, agreements and other writings and shall take, or shall cause to be taken, such further actions as may be reasonably required or requested by either party to carry out the provisions of this Agreement and consummate or implement expeditiously the Transactions, and (b) shall refrain from taking any actions that could reasonably be expected to impair, delay or impede the Closing.

Section 5.6 <u>Resignations</u>. SALIC shall cause the directors and, to the extent specified in writing by Purchaser at least thirty (30) calendar days prior to the Closing Date, other officers, of each of the SALIC Group Companies, to resign such position or positions, effective as of the Closing (the "<u>Resignations</u>"). Any such resignation that is not furthered by a reappointment to the same or similar position shall be deemed an involuntary resignation for purposes of severance and employment agreement terms.

Section 5.7 <u>Insurance</u> Prior to the Closing, SALIC shall obtain a replacement to SALIC's existing policy of directors' and officers' liability insurance identified in Section 3.15 of the SALIC Disclosure Schedules (Policy No. ELU154535-18) on terms reasonably acceptable to Purchaser, to be effective as of the Closing Date and providing coverage to the directors and officers of each of the SALIC Group Companies immediately following the Closing.

## Section 5.8 <u>D&O Liabilities</u>.

(a) From and after the Closing Date, to the extent permitted by Applicable Law, the certificate of incorporation, certificate of formation, bylaws or limited liability company operating agreement (or similar organizational documents) of each SALIC Group Company shall continue to contain provisions no less favorable with respect to indemnification, advancement of expenses and exculpation of each present and former director and officer of each of the SALIC Group Companies (collectively, the "<u>Indemnified D&O Parties</u>") than are set forth in the organizational documents of the SALIC Group Companies as of the date hereof, which provisions shall not be amended, repealed or otherwise modified for a period of six (6) years after the Closing Date in any manner that would adversely affect the rights thereunder of any such individual.

For a period of six (6) years from and after the Closing Date, to the extent that the (b) Indemnified D&O Parties are not otherwise covered as insureds under an existing policy of directors' and officers' liability insurance in accordance with the requirements set forth in this Section 5.8(b), Purchaser shall cause the SALIC Group Companies to maintain in effect policies of directors' and officers' liability insurance comparable to those maintained by the SALIC Group Companies or SRGL with respect to matters existing or occurring at or prior to the Closing Date; provided, that Purchaser or the SALIC Group Companies may substitute therefor policies of at least the same coverage containing terms and conditions that are not less advantageous than the existing policies (including with respect to the period covered); provided, further, that in lieu of maintaining the current policies of directors' and officers' liability insurance, Purchaser may cause the SALIC Group Companies to purchase "tail" coverage or otherwise replace such policies with coverage with a scope, policy limits and retained coverage not less favorable than the scope, policy limits and retained coverage currently provided. Notwithstanding the foregoing, at Purchaser's direction and in satisfaction of Purchaser's obligations under this Section 5.8(b), SALIC shall obtain such "tail" coverage in respect of SALIC's existing policy of directors' and officers' liability insurance identified in Section 3.15 of the SALIC Disclosure Schedules (Policy No. ELU154535-18) on terms reasonably acceptable to Purchaser, to be effective as of the Closing Date.

Section 5.9 <u>Orkney Re II</u>. Purchaser and SALIC shall cooperate in good faith to complete with Assured Guaranty (UK) Ltd., prior to Closing, the winding up, restructuring or other termination relating to the Orkney Re II facility on terms and conditions reasonably acceptable to Purchaser (it being understood that such winding up, restructuring or other termination shall not be a condition to the Closing).

Section 5.10 <u>Abandonment of SFL Capital Stock</u> SALIC shall use commercially reasonable efforts to cause any capital stock or share capital of SFL held by any SALIC Group Company to no longer be held by any SALIC Group Company as of a date no later than the Closing Date. Purchaser acknowledges and agrees that the transfer of any capital stock or share capital of SFL held by any SALIC Group Company to the Distribution Trust on or prior to the Closing Date with the approval of the Bankruptcy Court pursuant to the Confirmation Order or another Final Order will satisfy the SALIC Parties' obligations as set forth in this Section 5.10.

### Section 5.11 <u>Notification of Certain Matters</u>.

From the date hereof until the Closing, to the extent the SALIC Parties have such (a) Knowledge, the SALIC Parties shall promptly notify Purchaser in writing of: (i) any circumstance, event or action the existence, occurrence or taking of which (a) has had or could reasonably be expected to have, individually or in the aggregate, a SALIC Material Adverse Effect, (b) has resulted in or could reasonably be expected to result in any representation or warranty made by SALIC hereunder or under any other Ancillary Agreement not being true and correct or (c) could reasonably be expected to result in the failure of any of the conditions set forth in Article VII to be satisfied; (ii) any notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the Transactions, provided that, if the SALIC Parties are restricted by any confidentiality or nondisclosure obligation that would prevent the SALIC Parties from providing such notice, the SALIC Parties will provide Purchaser with sufficient information, including, the nature of the confidentiality or non-disclosure obligation, the nature of such consent, the type of transaction to which such consent relates, the nature of the counterparty alleging to have such consent right, and any other information reasonably necessary for Purchaser to evaluate such Person's alleged consent right; (iii) any notice or other communication from any Governmental Authority in connection with the Transactions; and (iv) any Actions commenced or threatened against, relating to or involving or otherwise affecting any SALIC Group Company that, if pending on the date of this Agreement, would have been required to have been disclosed pursuant to Section 3.9 or that relates to the consummation of the Transactions. Purchaser's receipt of information pursuant to this Section 5.11(a) or otherwise shall not operate as a waiver or otherwise affect any representation, warranty, covenant or agreement given or made by SALIC in this Agreement or the other Ancillary Agreement.

(b) With respect to any periodic reporting period, but no less frequently than monthly, the applicable SALIC Group Company shall, as promptly as practicable after such reporting period, notify Purchaser in writing if both (i) loss development under an In-force YRT Reinsurance Contract for such period was adverse and (ii) such adverse development was, in the reasonable judgment of the SALIC Group Company, caused in part or in whole by higher-than-expected mortality for such period for such In-force YRT Reinsurance Contract. Purchaser's

receipt of information pursuant to this Section 5.11(b) or otherwise shall not operate as a waiver or otherwise affect any representation, warranty, covenant or agreement given or made by SALIC in this Agreement or the other Ancillary Agreement. For purposes of this Section 5.11(b), an "<u>In-force YRT Reinsurance Contract</u>" means a Contract of yearly renewable term reinsurance assumed from an unrelated third-party ceding company by any SALIC Group Company that is in force at any time between the date hereof and the Closing.

### Section 5.12 <u>Subsequent Financial Statements.</u>

(a) SALIC shall prepare and deliver to Purchaser as soon as reasonably practicable audited consolidated financial statements of SALIC and its Subsidiaries at and for the 12-month periods ended December 31, 2018, 2017 and 2016, together with the report of the independent auditor of SALIC thereon, including a balance sheet and the related statements of operations, cash flows and changes in stockholders' equity (the "Subsequent Audited GAAP Financial Statements"), and if the report of the independent auditor of SALIC on the Subsequent Audited GAAP Financial Statements identifies any material weakness, SALIC shall remedy or resolve such material weakness to the reasonable satisfaction of Purchaser prior to the Closing. From the date hereof until the Closing, SALIC shall deliver to Buyer, as soon as reasonably practicable after the end of the applicable quarter, unaudited quarterly financial statements of the type and scope described in Section 3.7(a) (together with the Subsequent Audited GAAP Financial Statements, the "Subsequent GAAP Financial Statements").

(b) From the date hereof until the Closing, SALIC shall deliver to Purchaser reasonably promptly following the filing thereof, all Statutory Statements, in each case prepared and/or filed after the date hereof and prior to the Closing Date.

(c) From the date hereof until the Closing, SALIC shall, and shall cause each of the SALIC Subsidiaries to, (i) provide to Purchaser, as soon as reasonably practicable after the end of each fiscal month, a monthly management report in scope and detail consistent with those monthly management reports that have been historically prepared by such SALIC Group Company and delivered to SALIC or SRGL, as applicable, and (ii) prepare, and deliver to Purchaser, as soon as reasonably practicable after the end of each fiscal month, a monthly balance sheet as of the last day of such month, in scope and detail consistent with the monthly balance sheets that have been historically prepared by such SALIC Group Company.

Section 5.13 <u>Terminating Intercompany Agreements</u>. Subject to the right of the SALIC Parties to reject pursuant to section 365 of the Bankruptcy Code those Terminating Intercompany Agreements to which a Debtor is a party and that are Executory Contracts, the SALIC Parties shall, and shall cause the SALIC Group Companies, as applicable, to terminate any Terminating Intercompany Agreement to which any SALIC Group Company is a party on or prior to Closing. Notwithstanding anything to the contrary contained herein, Purchaser may determine prior to the Closing to have that certain Investment Management Agreement between SRGL and Barings LLC (f/k/a Babson Capital Management LLC), dated October 14, 2009, remain in place at the Closing and not be terminated as a Terminating Intercompany Agreement on or prior to the Closing pursuant to this Section 5.13.

Section 5.14 <u>Scottish Re Trademarks</u>. Between the date hereof and the Closing Date, SALIC shall use reasonable best efforts to obtain from SRGL in writing, and effective as of the Closing, the commitment of SRGL to provide a license at the Closing to SALIC (and/or another SALIC Group Company as identified by SALIC) in respect of, the marks "Scottish Re" (US Registration Nos. 3263875 and 3254382 and any corresponding marks throughout the world, including the European Union) (collectively, the "<u>SRE Marks</u>"), in form and on terms reasonably acceptable to Purchaser.

## **ARTICLE VI**

### **BANKRUPTCY MATTERS**

#### Section 6.1 <u>Bankruptcy Court Filings and Approvals</u>.

(a) The Debtors shall use reasonable best efforts to cause the Bankruptcy Court to enter the Winning Bidder Order on or prior to June 12, 2018, in form and substance reasonably acceptable to the Debtors and Purchaser.

(b) Prior to or concurrent with the entry of the Winning Bidder Order, the Debtors shall terminate the Plan Sponsorship Agreement in accordance with its terms and take all other actions reasonably necessary in connection with the Chapter 11 Cases to effectuate such termination, and deliver to Purchaser evidence of such termination in form and substance reasonably acceptable to Purchaser.

(c) On or prior to the date that is ten (10) calendar days after the date of entry of the Winning Bidder Order, the Debtors shall use reasonable best efforts to obtain the RIA Winning Bidder Confirmation and provide same to Purchaser; provided that in the event the Debtors are unable to obtain the RIA Winning Bidder Confirmation within such ten (10) calendar day period due to the requirement of the Cayman Islands Court and/or the Bankruptcy Court to direct or permit the Cayman Islands Full Powers Liquidators on behalf of SRGL to provide the RIA Winning Bidder Confirmation, then the Debtors shall use reasonable best efforts to (i) cause the Cayman Islands Court to make the Supplemental Cayman Islands Court Order, if applicable, (ii) cause the Bankruptcy Court to enter the Amended RIA Order, if applicable and (iii) promptly thereafter, obtain the RIA Winning Bidder Confirmation and provide same to Purchaser, in each case on or prior to the date that is the earlier of (x) thirty-five (35) calendar days from the entry of the Winning Bidder Order and (y) the date of entry of the Disclosure Statement Order.

(d) The Debtors shall amend the Chapter 11 Plan and the Disclosure Statement to reflect this Agreement and the Plan Term Sheet, in each case in form and substance reasonably acceptable to the Debtors and Purchaser on or prior to the date that is fifteen (15) Business Days from the conclusion of the Auction.

(e) The Debtors shall use reasonable best efforts to cause the Bankruptcy Court to enter the Disclosure Statement Order (pursuant to an amendment to the Disclosure Statement Motion reflecting the transactions set forth in this Agreement and the Plan Term Sheet in form and substance reasonably acceptable to the Debtors and Purchaser) in form and substance reasonably acceptable to the Debtors and Purchaser on or prior to the date that is thirty (30) calendar days from the conclusion of the Auction; <u>provided</u> that if the Debtors or SRGL (at the Debtors' request) are seeking the Supplemental Cayman Islands Court Order pursuant to the proviso in Section 6.1(c), such deadline shall be extended to such date that is thirty-five (35) calendar days from the entry of the Winning Bidder Order.

(f) The Debtors shall use reasonable best efforts to cause the Bankruptcy Court to enter the Confirmation Order in form and substance reasonably acceptable to the Debtors and Purchaser on or prior to the date that is sixty (60) days following the entry of the Disclosure Statement Order.

(g) The Debtors shall give notice under the Bankruptcy Code of the request for the relief specified in the Bidding Procedures Motion, the RIA/PSA Assumption Motion, the Disclosure Statement Motion, of the Confirmation Hearing and the Effective Date, any request to assume or reject all Contracts, and any bar date motions, to all creditors and parties in interest entitled to notice thereof pursuant to the Bankruptcy Code, the Bankruptcy Rules, the local rules of the Bankruptcy Court, and Orders of the Bankruptcy Court, including all Persons that have asserted Encumbrances against each of either Debtor's assets, and all non-debtor parties to the Assumed Contracts.

(h) Each of the SALIC Parties and Purchaser agrees that it will promptly take such actions as are reasonably requested by the other parties to assist in obtaining (i) the RIA Winning Bidder Confirmation from SRGL, (ii) if necessary pursuant to Section 6.1(c), entry of the Amended RIA Order and the Supplemental Cayman Islands Court Order, and (iii) the Winning Bidder Order, the Disclosure Statement Order, the Sale Order and the Confirmation Order, in each case, within the applicable period referenced in Section 8.1.

(i) The SALIC Parties and Purchaser acknowledge that this Agreement and the Transactions are subject to Bankruptcy Court approval.

SALIC agrees that from the conclusion of the Auction through the Closing Date, (i) neither it nor any of its Affiliates nor any of their respective Representatives shall, and that it shall direct its and their respective key employees and Representatives (including any investment banker, attorney or accountant retained by it or any of its Affiliates) not to, directly or indirectly, initiate, solicit, entertain, encourage, negotiate, accept or discuss any proposal or offer of an Alternative Transaction from any Person or group of Persons other than Purchaser and its Affiliates, or provided any non-public or other information to any third party in connection therewith, or enter into any agreement, arrangement or understanding requiring it to abandon, terminate, delay or fail to consummate the Transactions. Immediately upon the execution of this Agreement, SALIC and its Affiliates, and any of their Representatives shall, and SALIC shall direct its and their respective key employees and Representatives (including any investment banker, attorney or accountant retained by it or any of its Affiliates) to, suspend any and all existing discussions or negotiations with any Person or group of Persons regarding an Alternative Transaction, in each case except as permitted by the immediately preceding sentence. Additionally, SALIC represents and warrants to Purchaser that it is not party to or bound by any agreement with respect to an Alternative Transaction as of the date hereof, other than the Stalking Horse SPA. The SALIC Parties shall, within two (2) Business Days of receipt by the

SALIC Parties or their Affiliates of an Alternative Transaction, provide Purchaser with the material terms and conditions of any such Alternative Transaction. Following entry of the Bidding Procedures Order, any such contacts, solicitations or encouragements shall be in accordance with the terms of the Bidding Procedures Order.

(k) The SALIC Parties shall use reasonable best efforts to give Purchaser at least three (3) Business Days advance notice of proposed drafts of all pleadings, motions, Orders, other papers, hearings, and other proceedings relating to this Agreement and the Transactions (collectively, "<u>Restructuring Documents</u>") (other than the Chapter 11 Plan, Disclosure Statement, Confirmation Order, Supplemental Cayman Islands Court Order, Amended RIA Order and Plan Supplement, each of which shall be provided by the SALIC Parties to Purchaser within a reasonable period of time prior to the deadline for its submission but in each case, no less than five (5) Business Days prior to the deadline for its submission), and each such Restructuring Document shall be consistent with this Agreement and the Transactions and subject to the consent of the Purchaser, not to be unreasonably withheld, conditioned or delayed.

(1) After entry of the Sale Order, neither the Debtors nor Purchaser shall take any action which is intended to, or fail to take any action the intent of which failure to act is to, result in the reversal, voiding, modification or staying of the Sale Order.

(m) After entry of the Confirmation Order, neither the Debtors nor Purchaser shall take any action which is intended to, or fail to take any action the intent of which failure to act is to, result in the reversal, voiding, modification or staying of the Confirmation Order.

(n) In the event an appeal is taken or a stay pending appeal is requested from any Bankruptcy Court Approval or Foreign Court Approval, Debtors shall immediately notify Purchaser of such appeal or stay request. Debtors agree to take all action as may be reasonable and appropriate to defend against any such appeal or stay request, and Purchaser agrees to cooperate with, and support such efforts by, the Debtors. Additionally, the Debtors and Purchaser agree to use their reasonable efforts to obtain an expedited resolution of such appeal or stay request. For the avoidance of doubt, nothing herein shall preclude the parties hereto from consummating the Transactions if the Confirmation Order and Sale Order shall have been entered and have not been stayed and the applicable Debtor and Purchaser, each acting in their respective sole and absolute discretion, waive in writing the condition that the Confirmation Order and Sale Order be a Final Order.

(o) The SALIC Parties agree that, from the date hereof until the Closing Date, unless otherwise consented to in advance in writing by Purchaser, the SALIC Parties shall:

(i) Convene conference calls, on reasonable advance notice, with Purchaser and its Representatives, which calls will include the management and the legal advisors of the SALIC Parties;

(ii) Provide reasonable access to the respective management and advisors of the SALIC Parties for the purposes of evaluating the finances and operations of the SALIC Group Companies, ensuring the SALIC Parties' compliance with this Agreement and the Restructuring Implementation Agreement and participating in the planning and implementation aspects of the Transaction;

(iii) Timely file a formal objection to any motion filed with the Bankruptcy Court or any other proceeding commenced by any party seeking (provided, that a draft of any such objection shall be provided to Purchaser at least two (2) Business Days in advance of filing, to the extent reasonably practicable):

(1) Entry of an order (A) directing the appointment of an examiner or a trustee, (B) converting the Chapter 11 Cases to cases under chapter 7 of the Bankruptcy Code or (C) dismissing the Chapter 11 Cases;

(2) Entry of an order modifying or terminating the SALIC Parties' exclusive right to file and/or solicit acceptances of a plan of reorganization; or

(3) Other relief that would be inconsistent with the Chapter 11 Plan, the Plan Term Sheet, this Agreement, the Restructuring Implementation Agreement or other Restructuring Document; and

(iv) Not dispute that the giving of notice of termination by any party pursuant to this Agreement shall not be a violation of the automatic stay of section 362 of the Bankruptcy Code (and the SALIC Parties hereby waive, the applicability of the automatic stay to the giving of such notice).

# ARTICLE VII

# CONDITIONS TO CLOSING

Section 7.1 <u>Conditions to the Obligations of Purchaser and SALIC</u>. The obligations of SALIC and Purchaser to effect the Closing are subject to the satisfaction (or written waiver by each party hereto) as of the Closing of the following conditions:

(a) No Injunction or Prohibition. No Governmental Authority of competent jurisdiction shall have enacted, enforced or entered any Applicable Law or Order that is in effect on the Closing Date and prohibits the consummation of the Transactions, and no Action by any Governmental Authority seeking the imposition of any such Order shall be pending or threatened in writing by any Governmental Authority.

(b) Entry of the Confirmation Order. The Confirmation Order shall have been entered by the Bankruptcy Court in form and substance reasonably acceptable to the Debtors and Purchaser and shall be a Final Order.

(c) Entry of the Sale Order. The Sale Order shall have been entered by the Bankruptcy Court in form and substance reasonably acceptable to the Debtors and Purchaser and shall be a Final Order.

(d) Receipt of the RIA Winning Bidder Confirmation. The RIA Winning Bidder Confirmation shall have been received from SRGL in form and substance reasonably acceptable to Purchaser.

(e) Entry of the Disclosure Statement Order. The Disclosure Statement Order shall have been entered by the Bankruptcy Court in form and substance reasonably acceptable to the Debtors and Purchaser.

(f) Entry of the Winning Bidder Order. The Winning Bidder Order shall have been entered by the Bankruptcy Court in form and substance reasonably acceptable to the Debtors and Purchaser.

(g) Foreign Court Approvals. All Foreign Court Approvals shall have been obtained in form and substance reasonably acceptable to SALIC and Purchaser and any Orders granting the Foreign Court Approvals shall not have been modified, stayed, reversed, revoked or vacated by the court which made such Order or any court with jurisdiction to grant such relief.

(h) Required Approvals. All Required Approvals (other than the Bankruptcy Court Approvals and Foreign Court Approvals) shall have been obtained and be in full force and effect (or any waiting period applicable thereto shall have been terminated or otherwise expired), in each case without the imposition of a Burdensome Condition.

Section 7.2 <u>Conditions to the Obligations of Purchaser</u>. The obligation of Purchaser to effect the Closing is subject to the satisfaction (or written waiver by Purchaser) as of the Closing of the following conditions:

(a) Warranties True as of the Closing Date. Except for the representations and warranties set forth in Sections 3.1, 3.2, 3.3(a), 3.4, 3.18 and 3.26 (the "<u>SALIC Fundamental Reps</u>"), each of the representations and warranties of SALIC contained in Article III shall be true and correct (without giving effect to any limitations as to materiality or SALIC Material Adverse Effect) as of the date hereof and as of the Closing Date as if made anew as of such date (except to the extent such representations and warranties expressly relate to an earlier date (in which case, as of such earlier date)), except for such failures of the representations and warranties to be true and correct as do not and would not reasonably be expected to have, in the aggregate, a SALIC Material Adverse Effect. The SALIC Fundamental Reps shall be true and correct (without giving effect to any limitations as to materiality or SALIC Material Adverse Effect) as of the date hereof and as of the Closing Date as if made anew as of such date (except to the true and correct as do not and would not reasonably be expected to have, in the aggregate, a SALIC Material Adverse Effect. The SALIC Fundamental Reps shall be true and correct (without giving effect to any limitations as to materiality or SALIC Material Adverse Effect) as of the date hereof and as of the Closing Date as if made anew as of such date (except to the extent such representations and warranties expressly relate to an earlier date (in which case, as of such earlier date)). Purchaser shall have received a certificate to such effect dated the Closing Date and executed by a duly authorized officer of SALIC.

(b) Compliance with Agreements and Covenants. (i) SALIC shall have performed and complied with all of the covenants, obligations and agreements contained in this Agreement to be performed and complied with by it on or prior to the Closing Date, in all material respects and (ii) SHI shall have performed and complied with all of the covenants, obligations and agreements of SHI contained in this Agreement and any other Ancillary Agreement to be performed and complied with by it on or prior to the Closing Date, in all material respects.

## Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 167 of 241

Purchaser shall have received a certificate to such effect dated the Closing Date and executed by a duly authorized officer of SALIC.

(c) SRGL Shares. SRGL shall have surrendered to SALIC for cancellation all SRGL Shares and all such SRGL Shares so surrendered shall have been cancelled, except for one such SRGL Share (the "<u>Remaining SRGL Share</u>"), which shall not be surrendered for cancellation or otherwise cancelled until immediately following the Closing as contemplated by Section 2.4(1).

(d) SALIC Material Adverse Effect. There shall not have occurred a SALIC Material Adverse Effect which has not been cured within fifteen (15) Business Days after SALIC's receipt of written notice thereof from Purchaser.

(e) Secretary's Certificate. A certificate of the secretary, assistant secretary or equivalent Person of SALIC, dated the Closing Date, certifying resolutions of the board of directors of SALIC approving and authorizing the execution, delivery and performance of this Agreement and the consummation of the Transactions (together with an incumbency and signature certificate regarding the officer(s) signing on behalf of SALIC).

(f) Removal of SFL. The Confirmation Order shall provide for all capital stock or share capital of SFL held by any SALIC Group Company to be transferred to the Distribution Trust upon the occurrence of the Effective Date.

(g) Employment Agreements. The Employment Agreements shall have been executed and delivered by the parties thereto and shall be valid and in full force and effect as of the Closing.

(h) Closing Date Plan Distributions. The Closing Date Plan Distributions shall not exceed the Plan Funding Payment.

(i) Restructuring Implementation Agreement Amendment. The Restructuring Implementation Agreement shall have been amended in form and substance reasonably acceptable to Purchaser.

(j) Termination of Plan Sponsorship Agreement. The Plan Sponsorship Agreement shall have been terminated in accordance with its terms, and Purchaser shall have received evidence of such termination in form and substance reasonably acceptable to Purchaser.

(k) Amendment of Chapter 11 Plan and Disclosure Statement. The Chapter 11 Plan and the Disclosure Statement shall have been amended to reflect this Agreement and the Plan Term Sheet, in each case in form and substance reasonably acceptable to Purchaser.

(1) Stalking Horse SPA. The SALIC Parties have terminated the Stalking Horse SPA in accordance with its terms and delivered to Purchaser evidence of such termination in form and substance reasonably acceptable to Purchaser.

(m) SRGL Matters. SRGL shall have confirmed in writing its intention not to take any action that could cause the "tail" coverage obtained by SRGL in respect of SRGL's three (3)

policies of directors' and officers' liability insurance identified in Section 3.15 of the SALIC Disclosure Schedules to terminate prior to the end of its term, in form and substance reasonably acceptable to Purchaser.

Section 7.3 <u>Conditions to the Obligations of SALIC</u>. The obligations of SALIC to effect the Closing are subject to the satisfaction (or written waiver by SALIC) as of the Closing of the following conditions:

(a) Warranties True as of the Closing Date. Except for the representations and warranties set forth in Sections 4.1, 4.2, and 4.7 (the "<u>Purchaser Fundamental Reps</u>"), each of the representations and warranties of Purchaser contained in Article IV shall be true and correct (without giving effect to any limitations as to materiality or Purchaser Material Adverse Effect) as of the date hereof and as of the Closing Date as if made anew as of such date (except to the extent such representations and warranties expressly relate to an earlier date (in which case, as of such earlier date)), except for such failures of the representations and warranties to be true and correct as do not and would not reasonably be expected to have, in the aggregate, a Purchaser Material Adverse Effect. The Purchaser Fundamental Reps shall be true and correct (without giving effect to any limitation as to materiality or Purchaser Material Adverse Effect) as of the date hereof and as of the Closing Date as if made anew as of such date (except to the extent such representations and warranties expressly relate to an earlier date). SALIC shall have received a certificate to such effect dated the Closing Date and executed by a duly authorized officer of Purchaser.

(b) Covenants. Purchaser shall have performed and complied with all of its covenants, obligations and agreements contained in this Agreement to be performed and complied with by it on or prior to the Closing Date, in all material respects. SALIC shall have received a certificate to such effect dated the Closing Date and executed by a duly authorized officer of Purchaser.

(c) Secretary's Certificate. A certificate of the secretary, assistant secretary or equivalent Person of Purchaser, dated the Closing Date, certifying resolutions of the members of Purchaser approving and authorizing the execution, delivery and performance of this Agreement and the consummation of the Transactions (together with an incumbency and signature certificate regarding the officer(s) signing on behalf of Purchaser).

# **ARTICLE VIII**

### TERMINATION

Section 8.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Closing (the date on which this Agreement terminates in accordance with its terms, the "<u>Termination Date</u>"):

- (a) with the mutual written consent of Purchaser and SALIC;
- (b) by either the SALIC Parties or Purchaser:

(i) if the Closing shall not have occurred on or before the one hundred eightieth (180<sup>th</sup>) calendar day after the entry of the Winning Bidder Order (the "<u>Outside</u> <u>Closing Date</u>"); provided, that if, as of the Outside Closing Date, all conditions to this Agreement shall have been satisfied or waived (other than those that are satisfied by action taken at the Closing) other than the conditions set forth in Section 7.1(h), then Purchaser, by written notice to the SALIC Parties, may extend the Outside Closing Date to a date no later than sixty (60) calendar days after the initial Outside Closing Date; <u>provided further</u>, that the right to terminate this Agreement under this Section 8.1(b)(i) shall not be available to any party whose failure to fulfill any obligation under this Agreement has been the cause of, or resulted in, the failure of the Closing to occur on or before the Outside Closing Date or such later date;

(ii) if any Governmental Authority shall have issued an Order, decree or ruling or taken any other action (and such Order, decree, ruling or other action shall have become final and nonappealable), or if any Applicable Law shall have been enacted, that permanently restrains, enjoins or otherwise prohibits the Transactions; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement under this Section 8.1(b)(ii) shall not be available to any party whose failure to comply with Section 5.3 has caused or resulted in such action or inaction;

(iii) upon entry of the Winning Bidder Order, if at the end of the Auction Purchaser was not determined to be the Winning Bidder or Back-Up Bidder;

(iv) upon termination of the Restructuring Implementation Agreement; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement under this Section 8.1(b)(iv) shall not be available to the SALIC Parties if either SALIC Party breached its obligations under the Restructuring Implementation Agreement and such breach caused or resulted in the termination of the Restructuring Implementation Agreement;

(v) if either Chapter 11 Case is dismissed or converted to a case under Chapter 7 of the Bankruptcy Code, and neither such dismissal nor conversion expressly contemplates the Transactions provided for in this Agreement; or

(vi) if a Governmental Authority shall have commenced a DDOI Proceeding against SRUS;

(c) by the SALIC Parties, if:

(i) Purchaser shall have breached or failed to perform any of its representations, warranties, covenants or other agreements contained in this Agreement, which breach or failure to perform (A) would give rise to the failure of a condition set forth in <u>Article VII</u> and (B) has not been or is incapable of being cured by Purchaser within ten (10) calendar days after its receipt of written notice thereof from the SALIC Parties; or

(ii) (A) all of the conditions set forth in <u>Article VII</u> (other than conditions that by their terms are to be satisfied at the Closing) have been satisfied or waived (it being agreed that any conditions that are not so satisfied due to a breach by Purchaser of any representation, warranty or covenant contained in this Agreement shall be deemed to be satisfied for purposes of this clause (A)), (B) Purchaser fails to consummate the Closing within three (3) Business Days after the date on which the Closing is required to occur pursuant to Section 2.4 (assuming the satisfaction of conditions deemed satisfied in accordance with the immediately preceding clause (A)) and (C) the SALIC Parties have notified Purchaser in writing that the SALIC Parties are ready, willing and able to consummate the Closing;

(d) by Purchaser, if:

(i) SALIC shall have breached or failed to perform any of its representations, warranties, covenants or other agreements contained in this Agreement, which breach or failure to perform (A) would give rise to the failure of a condition set forth in <u>Article VII</u> and (B) has not been or is incapable of being cured by SALIC within ten (10) calendar days after its receipt of written notice thereof from Purchaser;

(ii) The Bankruptcy Court shall not have entered the Winning Bidder Order on or prior to June 12, 2018;

(iii) The RIA Winning Bidder Confirmation shall not have been received by Purchaser within ten (10) calendar days after the date of entry of the Winning Bidder Order; <u>provided</u> that if the Debtors or SRGL (at the Debtors' request) are seeking the Supplemental Cayman Islands Court Order and/or Amended RIA Order pursuant to the proviso in Section 6.1(c), such ten (10) calendar day period shall be extended to such date that is the earlier of (x) thirty-five (35) calendar days from the entry of the Winning Bidder Order and (y) the date of entry of the Disclosure Statement Order;

(iv) The Bankruptcy Court shall not have entered the Disclosure Statement Order on or prior to the date that is thirty (30) calendar days after the conclusion of the Auction; <u>provided</u> that if the Debtors or SRGL (at the Debtors' request) are seeking the Supplemental Cayman Islands Court Order pursuant to the proviso in Section 6.1(c), such deadline shall be extended to such date that is thirty-five (35) calendar days from the entry of the Winning Bidder Order;

(v) The Bankruptcy Court has not entered the Confirmation Order and the Sale Order on or prior to the date that is sixty (60) days after the entry of the Disclosure Statement Order;

(vi) A trustee or examiner with expanded powers is appointed in either of the Chapter 11 Cases;

(vii) The Debtors withdraw, or modify in any respect that is materially adverse to Purchaser's rights under this Agreement or the Chapter 11 Plan without the consent of Purchaser; or (viii) There shall have occurred any event, occurrence, condition or change that, individually or in the aggregate, has had or would reasonably be expected to have a SALIC Material Adverse Effect.

(e) The party seeking to terminate this Agreement pursuant to this Section 8.1 (other than Section 8.1(a)) shall give prompt written notice of such termination to the other party hereto.

## Section 8.2 <u>Effect of Termination</u>.

(a) If this Agreement is terminated pursuant to Section 8.1, this Agreement shall become void and of no effect with no liability on the part of any party hereto; <u>provided</u>, <u>however</u>, that if (i) Purchaser shall have terminated this Agreement and shall have suffered loss as a result of a material breach of this Agreement by SALIC or SHI and (ii) such loss is the result of a third party claim against Purchaser, then such termination shall not relieve SALIC or SHI of liability for such breach of this Agreement with respect to such third party claim.

(b) To the extent permitted by Applicable Law, no party hereto shall assert, and each party hereto waives, any Claim against any party hereto, such party's Affiliates or any Representatives of such party or such party's Affiliate (solely in such capacity) on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, the Ancillary Agreements, the Transactions or any agreement or instrument contemplated by any of the foregoing.

(c) Notwithstanding Section 8.2(a), if this Agreement is terminated pursuant to Section 8.1, then:

(i) each party will return or destroy all documents, workpapers and other material of any other party relating to the Transactions, whether so obtained before or after the execution of this Agreement, to the party furnishing the same;

(ii) if this Agreement is terminated (i) jointly by Purchaser and SALIC pursuant to Section 8.1(a), (ii) by Purchaser or the SALIC Parties pursuant to Sections 8.1(b), or (iii) by Purchaser pursuant to Sections 8.1(d), then Purchaser shall be entitled to receive, and, to the extent necessary the SALIC Parties shall cause the Deposit Escrow Agent to return to Purchaser, the Deposit Escrow Amount within five (5) Business Days of the Termination Date; and

(iii) if this Agreement is terminated by the SALIC Parties pursuant to Section 8.1(c), then SALIC shall be entitled to receive, and, to the extent necessary, Purchaser shall cause the Deposit Escrow Agent to deliver to SALIC, the Deposit Escrow Amount within five (5) Business Days of the Termination Date, and receipt of the Deposit Escrow Amount shall be the sole and exclusive remedy of the SALIC Parties and their Affiliates, on the one hand, against Purchaser and its Affiliates, on the other hand, for any loss suffered as a result of any breach by Purchaser or any of its Affiliates of any

representation, warranty, covenant or agreement in this Agreement, the Ancillary Agreements or the transactions contemplated hereby or thereby.

Section 8.3 <u>Fees and Expenses</u>. Except as otherwise expressly provided in this Agreement whether or not the Transactions are consummated, all direct and indirect costs and expenses incurred in connection with this Agreement and the Transactions shall be borne by the party incurring such expenses; provided, however, that Purchaser shall bear the costs described in Section 5.3(b).

#### MISCELLANEOUS

Section 9.1 <u>No Survival of Representations and Warranties</u>. Except as otherwise provided in this Agreement, the representations, warranties, covenants and agreements of SALIC and Purchaser contained in or made pursuant to this Agreement or in any certificate furnished pursuant to this Agreement (other than those covenants and agreements contained herein that by their terms apply or are to be performed in whole or in part after the Closing Date, which shall survive in accordance with their terms) shall terminate at, and be of no further force and effect following, the Closing.

Section 9.2 <u>Notices</u>. All notices, requests, claims, demands or other communications hereunder shall be deemed to have been duly given and made if in writing and (a) at the time personally delivered if served by personal delivery upon the party for whom it is intended, (b) at the time received if delivered by registered or certified mail (postage prepaid, return receipt requested) or by a national courier service (delivery of which is confirmed), or (c) upon confirmation of receipt if sent by facsimile or email; in each case to the Person at the address set forth below, or such other address as may be designated in writing hereafter, in the same manner, by such Person:

To SALIC or SHI:

Post Office Box HM 2939		
Hamilton HM MX, Bermuda		
Telephone:	(441) 298-4375	
Facsimile:	(441) 295-7576	
Email:	Gregg.Klingenberg@scottishre.com	
Attention:	Gregg Klingenberg	

with copies (which shall not constitute notice to SALIC for the purposes of this Section 9.2) to:

Mayer Brown LLP 1221 Avenue of the Americas New York, New York 10020 Telephone: (212) 506-2227 Facsimile: (212) 262-1910 Email: fmonaco@mayerbrown.com

	srooney@mayerbrown.com
Attention:	Stephen G. Rooney, Esq.
	Francis R. Monaco, Esq.

Hogan Lovells US LLP 875 Third Avenue New York, New York 10022 Telephone: (212) 918-5560 Facsimile: (212) 918-3100 Email: peter.ivanick@hoganlovells.com Attention: Peter Ivanick, Esq.

Morris, Nichols, Arsht & Tunnell LLP 1201 North Market Street, 16<sup>th</sup> Floor P.O. Box 1347 Wilmington, Delaware 19899-1347 Telephone: (302) 351-9229 Facsimile: (302) 425-4663 Email: eschwartz@MNAT.com gwerkheiser@MNAT.com Attention: Eric D. Schwartz, Esq. Gregory W. Werkheiser, Esq.

To Purchaser:

Hildene Re Holdings, LLC		
c/o Hildene Capital Management, LLC		
700 Canal Street, Suite 12C		
Stamford, CT 06902		
Telephone:	(203) 517-2500	
Email:	dhoffman@hildenecap.com	
	jnam@hildenecap.com	
Attention:	David Hoffman, General Counsel	
	Jennifer Nam, Deputy General Counsel	

with a copy (which shall not constitute notice to Purchaser for the purposes of this Section 9.2) to:

Kramer Levin Naftalis & Frankel LLP 1177 Avenue of the Americas New York, New York 10036 Telephone: (212) 715-9100 Facsimile: (212) 715-8000 Email: szide@kramerlevin.com ewechsler@kramerlevin.com Attention:

averramalli@kramerlevin.com smerl@kramerlevin.com Stephen Zide, Esq. Ernest S. Wechsler, Esq. Anupama Yerramalli, Esq. Seth R. Merl, Esq.

Amendment; Modification and Waiver. Any provision of this Agreement Section 9.3 may be amended, modified or waived if, and only if, such amendment, modification or waiver is in writing and signed, in the case of an amendment, by the parties, or in the case of a waiver, by the party hereto against whom the waiver is to be effective. No failure or delay by any party hereto in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege.

Assignment. Neither this Agreement nor any of the rights, interests or Section 9.4 obligations under it may be directly or indirectly assigned, delegated, sublicensed or transferred by any of the parties, in whole or in part, to any other Person (including any bankruptcy trustee) by operation of law or otherwise, whether voluntarily or involuntarily, without the prior written consent of the other party, and any attempted or purported assignment in violation of this Section 9.4 will be null and void. Notwithstanding the foregoing, (a) Purchaser may assign all or any portion of its rights hereunder to an Affiliate of Purchaser (including the equityholders of Purchaser or their Affiliates) without the prior written consent of SALIC or SHI, including transactions prior to the Closing pursuant to which Purchaser redomiciles or reorganizes in another jurisdiction and (b) Purchaser may assign its rights hereunder by way of security in connection with any financing obtained by Purchaser or its Affiliates in connection with the Transactions and such secured party may assign such rights by way of exercise of remedies, in each case, without the prior written consent of SALIC or SHI, provided that no such assignment will relieve Purchaser of its obligations hereunder. Subject to the two immediately preceding sentences, this Agreement will be binding upon, inure to the benefit of, and be enforceable by the parties and their respective heirs, executors, administrators, successors, legal representatives and permitted assigns. Notwithstanding anything to the contrary contained herein or in the Plan Term Sheet, it is understood that in the event Purchaser determines prior to the Closing to have SALIC owned one hundred percent (100%) by a newly formed holding company immediately following the Closing as contemplated by footnote 6 of the Plan Term Sheet, all Ordinary Shares of SALIC to be issued in connection with the Closing in respect of the Purchaser Shares and the TruPS Shares as contemplated by Sections 2.4(b)(ii) and the Chapter 11 Plan shall be issued at Closing to such newly formed holding company, as assignee of Purchaser under this Section 9.4, and all applicable TruPS Holders shall receive their respective TruPS Shares directly from such newly formed holding company in accordance with the Chapter 11 Plan. All rights of the Debtors and Purchaser are reserved with respect to the specific terms and conditions relating to any contemplated issuance of the TruPS Shares by a newly formed holding company instead of Reorganized SALIC.

Entire Agreement. This Agreement, the Ancillary Agreements, along with Section 9.5 the Schedules and Exhibits hereto and thereto, constitute the entire agreement between the

parties with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral or written, with respect to such matters, except for the Confidentiality Agreement which will remain in full force and effect until the Closing at which point it shall automatically terminate.

Section 9.6 <u>No Third Party Beneficiaries</u>. Other than the parties, the Distribution Trustee and the Distribution Trust (as to the Distribution Trustee and the Distribution Trust, solely as to the DT Post-Closing Rights), and the Indemnified D&O Parties (as to such Indemnified D&O Parties solely as to Section 5.12 hereof), and their respective heirs, executors, administrators, successors, legal representatives and permitted assigns, nothing expressed or implied in this Agreement is intended to confer any rights, benefits, remedies, obligations or liabilities upon any Person.

Section 9.7 <u>Public Disclosure</u>. The parties shall agree in writing on the form and content of any initial press release and, except with the prior written consent of the other party hereto (which consent shall not be unreasonably withheld, delayed or conditioned), shall not issue any other press release or other public statement or communication with respect to this Agreement or the Transactions other than filings made in the Bankruptcy Court, Bermuda Court or Cayman Islands Court; <u>provided</u> that each party hereto may, without the prior written consent of the other party hereto, issue such communication or make such public statement as may be required by Applicable Law or stock exchange rules, in which case the party required to make the communication or statement shall allow the other party reasonable time to comment on such release or announcement in advance of such issuance.

Section 9.8 Disclosure Schedules. Disclosures on the Purchaser Disclosure Schedule or the SALIC Disclosure Schedule shall be arranged in sections corresponding to the numbered and lettered sections of this Agreement, and any disclosure set forth on any section of a Disclosure Schedule shall be deemed to be disclosed by the party hereto delivering such Disclosure Schedule for all sections of this Agreement and all other sections of such Disclosure Schedule to the extent that it is readily apparent that such disclosure is applicable to such other sections of this Agreement or such other sections of such Disclosure Schedule; provided, however, that no disclosure shall qualify any SALIC Fundamental Rep or Purchaser Fundamental Rep unless it is set forth in the specific Disclosure Schedule, or the section or subsection of the Disclosure Schedule, corresponding to such SALIC Fundamental Rep or Purchaser Fundamental Rep. The headings contained in a Disclosure Schedule are for convenience of reference only and shall not be deemed to modify or influence the interpretation of the information contained in such Disclosure Schedule or this Agreement. The inclusion of any information in any section of a Disclosure Schedule shall not be deemed to be an admission or acknowledgment by the party hereto delivering such Disclosure Schedule or otherwise imply that such information is required to be listed in any section of such Disclosure Schedule or that any such matter rises to a SALIC Material Adverse Effect or Purchaser Material Adverse Effect, as applicable, or is material to or outside the Ordinary Course of Business. Matters reflected in a Disclosure Schedule are not necessarily limited to matters required by this Agreement to be reflected in such Disclosure Schedule. Such additional matters are set forth for informational purposes and do not necessarily include other matters of a similar nature. All references in a Disclosure Schedule to the enforceability of agreements with third parties, the existence or nonexistence of third-party rights, the absence of breaches or defaults by third parties, or similar matters or statements, are intended only to allocate rights and risks between Purchaser and SALIC and were not intended to be admissions against interests, give rise to any inference or proof of accuracy, be admissible against either party to this Agreement by any Person who is not a party to this Agreement, or give rise to any claim or benefit to any Person who is not a party to this Agreement. The disclosure in a Disclosure Schedule of any allegation, threat, notice or other communication shall not be deemed to include disclosure of the truth of the matter communicated. In addition, the disclosure of any matter in a Disclosure Schedule is not to be deemed an admission that such matter actually constitutes noncompliance with, or a violation of Applicable Law, any Order or Governmental Authorization or Contract or other topic to which such disclosure is applicable. In no event shall the disclosure of matters disclosed in a Disclosure Schedule or, in the case of SALIC, the Electronic Data Rooms, be deemed or interpreted to constitute or broaden a representation, warranty, obligation, covenant, condition or agreement of the party hereto delivering such Disclosure Schedule except to the extent provided in this Agreement.

### Section 9.9 <u>Governing Applicable Law and Jurisdiction</u>.

(a) Except to the extent governed by the Bankruptcy Code, this Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware.

(b) Each of the parties hereto hereby irrevocably and unconditionally submits, for itself and its property, to the exclusive jurisdiction of the Bankruptcy Court and any appellate court thereof, in any action or proceeding arising out of or relating to this Agreement or the Ancillary Agreements, or for recognition or enforcement of any judgment, and each of the parties hereto hereby irrevocably and unconditionally agrees that all Claims in respect of any such action or proceeding may be heard and determined in such court(s); provided that if the Bankruptcy Court does not have (or abstains from exercising) jurisdiction, then the courts of the State of Delaware, sitting in New Castle County, and of the United States District Court for the District of Delaware, and any appellate court from any thereof shall have exclusive jurisdiction with respect to any of the foregoing and each of the parties hereto irrevocably and unconditionally submits to the jurisdiction of such courts and agrees that all claims in respect of any such action, litigation or proceeding may be heard and determined in such Delaware state court or, to the fullest extent permitted by applicable law, in such Delaware venued federal court. In addition, each of the parties hereto (i) agrees that it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court; (ii) agrees that it will not bring any Proceeding relating to this Agreement in any court other than the abovenamed courts; and (iii) agrees that it will not seek to assert by way of motion, as a defense or otherwise, that any such Proceeding (A) is brought in an inconvenient forum, (B) should be transferred or removed to any court other than one of the above-named courts, (C) should be stayed by reason of the pendency of some other proceeding in any court other than one of the above-named court, or (D) that this Agreement or the subject matter hereof may not be enforced in or by the above-named courts. Each party hereto agrees that service of process upon such party in any such Proceeding shall be effective if notice is given in accordance with Section 9.2.

Section 9.10 WAIVER OF PARTY JURY TRIAL. EACH HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES AND, THEREFORE, EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY FOR ANY DISPUTE ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE BREACH, TERMINATION OR VALIDITY THEREOF OR ANY TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. EACH PARTY HERETO CERTIFIES AND ACKNOWLEDGES THAT (A) NEITHER THE OTHER PARTY HERETO NOR ITS REPRESENTATIVES, AGENTS OR ATTORNEYS HAVE REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH PARTY HERETO UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH PARTY HERETO MAKES THIS WAIVER VOLUNTARILY AND (D) EACH PARTY HERETO HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS OF THIS SECTION 9.10. ANY PARTY HERETO MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS AGREEMENT WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE PARTIES TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

Section 9.11 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which will be deemed to constitute an original, but all of which shall constitute one and the same agreement, and may be delivered by facsimile or other electronic means intended to preserve the original graphic or pictorial appearance of a document.

Section 9.12 <u>Severability</u>. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced under any Applicable Law, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the Transactions is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties to this Agreement shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in a mutually acceptable manner in order that the Transactions be consummated as originally contemplated to the greatest extent possible.

Section 9.13 <u>Specific Performance</u>. The parties agree that irreparable harm would occur and that the parties would not have an adequate remedy at law if any of the provisions of this Agreement were not performed in accordance with their specific terms on a timely basis or were otherwise breached. It is accordingly agreed that, without posting bond or other undertaking, the parties shall be entitled to injunctive or other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in any court of competent jurisdiction, this being in addition to any other remedy to which they are entitled at law or in equity. If any such action is brought in equity to enforce the provisions of

this Agreement, no party hereto will allege, and each party hereto hereby waives the defense or counterclaim, that there is an adequate remedy at law. The parties further agree that (a) by seeking any remedy provided for in this Section 9.13, a party hereto shall not in any respect waive its right to seek any other form of relief that may be available to such party hereto under this Agreement and (b) nothing contained in this Section 9.13 shall require any party hereto to institute any action for (or limit such party's right to institute any action for) specific performance under this Section 9.13 before exercising any other right under this Agreement.

Section 9.14 Non-Recourse. Except to the extent otherwise expressly set forth in the Ancillary Agreements, all claims, obligations, liabilities, or causes of action (whether in Contract or in tort, in law or in equity, or granted by statute) that may be based upon, in respect of, arise under, out or by reason of, be connected with, or relate in any manner to this Agreement, or the negotiation, execution or performance of this Agreement (including any representation or warranty made in, in connection with, or as an inducement to, this Agreement), may be made only against (and such representations and warranties are those solely of) the Persons that are expressly identified as parties in the preamble to this Agreement (the "Contracting Parties"). No Person who is not a Contracting Party, including any current, former or future director, officer, employee, incorporator, member, partner, manager, stockholder, Affiliate, agent, attorney, other Representative or assignee of, and any financial advisor or lender to, any Contracting Party, or any current, former or future director, officer, employee, incorporator, member, partner, manager, stockholder, Affiliate, agent, attorney, other Representative or assignee of, and any financial advisor or lender to, any of the foregoing (collectively, the "Nonparty Affiliates"), shall have any liability (whether in Contract or in tort, in law or in equity, or granted by statute) for any claims, causes of action, obligations or liabilities arising under, out of, in connection with, or related in any manner to this Agreement or based on, in respect of, or by reason of this Agreement or its negotiation, execution, performance or breach (other than as expressly set forth in the Ancillary Agreements), and, to the maximum extent permitted by law, each Contracting Party hereby waives and releases all such liabilities, claims, causes of action and obligations against any such Nonparty Affiliates. For the avoidance of doubt, nothing in this Section 9.14 or elsewhere in this Agreement shall limit any rights of any party to this Agreement in the case of fraud.

(The remainder of this page is intentionally left blank.)
IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

## SCOTTISH ANNUITY & LIFE INSURANCE COMPANY (CAYMAN) LTD.

١ By: Name: Gregg Klingenberg Title: Chief Executive Officer

## SCOTTISH HOLDINGS, INC.

By: <u>Thomas</u> Pelly Name: Thomas J. Keller

Name: Thomas J. Keller Title: Executive Vice President, Chief Financial Officer

Signature Page to Stock Purchase Agreement

# HILDENE RE HOLDINGS, LLC

By:

Name: Brett Jefferson Title: President

Signature Page to Stock Purchase Agreement

# EXHIBIT A

Plan Term Sheet

See attached.

#### IN RE SCOTTISH HOLDINGS, INC., et al. 18-10160 (LSS)

## CHAPTER 11 PLAN TERM SHEET

This Plan Term Sheet (together with the exhibits and schedules attached hereto, as each may be amended, restated, supplemented, or otherwise modified from time to time, the "<u>Term Sheet</u>") describes a proposed chapter 11 plan of reorganization("<u>Plan</u>") for Scottish Holdings, Inc. ("<u>SHI</u>") and Scottish Annuity & Life Insurance Company (Cayman) Ltd. ("<u>SALIC</u>"), and together with SHI, the "<u>Debtors</u>"). This term sheet should be read in conjunction with the proposed bid submitted by Hildene Re Holdings, LLC, a Delaware Limited Liability Company (the "<u>Purchaser</u>"), which contains that certain stock purchase agreement (together with all exhibits, schedules, and other ancillary documents, as each may be amended, restated, supplemented, or otherwise modified from time to time, in accordance with the terms thereof, the "<u>Stock Purchase Agreement</u>"), to which this term sheet is an exhibit. This term sheet uses the proposed chapter 11 plan found at docket number 213 ("<u>Filed Plan</u>") as a basis to describe the transactions proposed in this Term Sheet and such Filed Plan will have to be modified to reflect the terms of this Term Sheet or as otherwise agreed to between the Parties. Capitalized terms used herein but not defined are ascribed the meanings set forth in the Filed Plan.

THIS TERM SHEET SHALL NOT CONSTITUTE AN ADMISSION BY ANY PERSON OR ENTITY, AND IS NOT INTENDED TO AND DOES NOT CREATE ANY LEGAL OR EQUITABLE OBLIGATIONS ON ANY PARTY. THIS TERM SHEET HAS BEEN PRODUCED IN CONNECTION WITH SETTLEMENT DISCUSSIONS AND IS SUBJECT TO THE PROVISIONS OF RULE 408 OF THE FEDERAL RULES OF EVIDENCE AND OTHER SIMILAR APPLICABLE STATE AND FEDERAL RULES.

THIS TERM SHEET IS NOT AN OFFER OR A SOLICITATION WITH RESPECT TO ANY SECURITIES OF ANY PARTY OR WITH RESPECT TO A CHAPTER 11 PLAN OF THE DEBTORS. ANY SUCH OFFER OR SOLICITATION SHALL COMPLY WITH ALL APPLICABLE SECURITIES LAWS AND/OR PROVISIONS OF THE BANKRUPTCY CODE AND WILL BE SUBJECT TO THE FINALIZATION OF A CHAPTER 11 PLAN AND ACCOMPANYING DISCLOSURE STATEMENT.

	CHAPTER 11 PLAN TERM SHEET OVERVIEW
Party Definitions:	" <u>Debtors</u> " has the meaning ascribed to such term in the Preamble to this Term Sheet.
	"Joint Liquidators" means the duly appointed liquidators in the winding up proceeding SRGL has commenced before the Supreme Court of Bermuda and/or any liquidators that may be duly appointed for SRGL in the parallel winding up proceeding SRGL has commenced before Grand Court of the Cayman Islands, Financial Services Division.
	" <u>Purchaser</u> " has the meaning ascribed to such term in the Preamble to this Term Sheet.
	" <u>SALIC</u> " has the meaning ascribed to such term in the Preamble to this Term Sheet.
	" <u>SALIC Non-Debtor Subsidiaries</u> " means SFL and SALIC Subsidiaries (other than SHI).
	" <u>SHI</u> " has the meaning ascribed to such term in the Preamble to this Term Sheet.
	" <u>SRGL</u> " has the meaning ascribed to such term in the Preamble to this Term Sheet.
Overview of Alternative Transaction: <sup>1</sup>	As contemplated by the Bidding Procedures Order, the proposed Alternative Transaction (as such term is defined in the Bidding Procedures Order) embodied in the Stock Purchase Agreement and the Term Sheet provides the following: (a) Hildene Re Holdings, LLC shall receive at least 70% of the equity of the Reorganized SALIC; (b) holders of Allowed TruPS Claims can elect to receive their Pro Rata share of either (i) 30% of the equity of the Reorganized SALIC (the " <u>TruPS Offered Shares</u> " and such amount of TruPS Offered Shares to be distributed to all holders of Allowed TruPS Claims electing for TruPS Offered Shares, the " <u>TruPS Shares</u> ") or (ii) the Plan Funding Payment available for distribution to unsecured creditors (the " <u>TruPS</u> <u>Offered Cash</u> " and such amount of cash to be distributed to all holders of Allowed TruPS Claims electing for TruPS Offered Cash, the " <u>TruPS</u> <u>Cash</u> ") <sup>2</sup> .

<sup>&</sup>lt;sup>1</sup> This Term Sheet outlines the proposed chapter 11 plan to implement the transaction embodied in the Stock Purchase Agreement and the Filed Plan shall be modified to implement the transaction set forth therein and herein. The Term Sheet does not include a description of all of the terms, conditions, and other provisions that are to be contained in the definitive documentation governing the Alternative Transaction (e.g. the Plan Supplement), which remain subject to further discussion and negotiation.

 $<sup>^{2}</sup>$  The default election for the holders of TruPS will be determined by the Purchaser prior to filing the amended chapter 11 plan.

	As set forth herein, (a) any TruPS Offered Shares not distributed to the holders of TruPS shall be Purchaser Shares and shall be distributed to the Purchaser and (b) any TruPS Offered Cash not distributed to holders of the TruPS shall reduce the Plan Funding Payment (the " <u>TruPS Returned Cash</u> ").	
Parties:	<ul> <li>The Debtors and their non-debtor subsidiaries;</li> <li>Certain fund(s) managed by Hildene Capital Management, LLC ("<u>Hildene</u>"); and</li> <li>Hildene Re Holdings, LLC (the "<u>Purchaser</u>").</li> </ul>	
Additional Definitions:	Capitalized terms not otherwise defined in this Term Sheet (including its Preamble and Annex 1) have the meanings ascribed to such terms in the Stock Purchase Agreement or, if not defined therein, the Restructuring Implementation Agreement or the Filed Plan.	
CLASSIFICA	CLASSIFICATION AND TREATMENT OF CLAIMS AND INTERESTS	
Administrative Expense Claims, Priority Tax Claims and Priority Claims:	Same treatment as the Filed Plan.	
Other Secured Claims:	Same treatment as the Filed Plan.	
SHI TruPS and General Unsecured Claims:	SHI TruPS Claims <sup>3</sup> : Unless a holder of an Allowed SHI TruPS Claim agrees to a lesser treatment, on or as soon as reasonably practicable following the Effective Date, each holder of an Allowed SHI TruPS Claim will receive, in full and final satisfaction, compromise, settlement, release, and discharge of and in exchange for such Allowed SHI TruPS Claim, (a) its Pro Rata share (calculated based on the aggregate amount of Allowed SHI TruPS Claims and Allowed SHI General Unsecured Claims) of either the TruPS Shares or the TruPS Cash and (b) its Pro Rata share of Distribution Trust Interests.	
	SHI General Unsecured Claims: Unless a holder of an Allowed SHI General Unsecured Claim agrees to a lesser treatment, on or as soon as reasonably practicable following the Effective Date, each holder of an Allowed SHI General Unsecured Claim will receive, in full and final satisfaction, compromise, settlement, release, and discharge of and in	

<sup>&</sup>lt;sup>3</sup> The SHI TruPS Claims shall include the TruPS held by SRGL against SHI. The Joint Liquidators have confirmed in writing that they will elect cash, thus, SRGL will only be able to receive cash on account of the TruPS held by SRGL against SHI, unless the Purchaser and SRGL agree otherwise pursuant to mutually agreeable terms to be included in the Plan.

In addition, the SHI TruPS Claims shall not include the fees and expenses of indenture trustees or other agents. The Charging Lien of the Indenture Trustees shall be satisfied through the TruPS Trustee Fees and Expenses (as such term is defined in the Stock Purchase Agreement).

	<ul> <li>exchange for such Allowed SHI General Unsecured Claim, (a) its Pro Rata share of the Plan Funding Payment available for distribution to unsecured creditors and (b) its Pro Rata share of Distribution Trust Interests.</li> <li><u>Classification</u>: To the extent permissible under the Bankruptcy Code and reasonably determined by the Debtors and the Purchaser to be conducive to obtaining confirmation of the Chapter 11 Plan, the Chapter 11 Plan should provide for separate classification of SHI TruPS Claims and any SHI General Unsecured Claims.</li> </ul>
SALIC TruPS and General Unsecured Claims:	SALIC TruPS Claims <sup>4</sup> : Unless a holder of an Allowed SALIC TruPS Claim agrees to a lesser treatment, on or as soon as reasonably practicable following the Effective Date, each holder of an Allowed SALIC TruPS Claim will receive, in full and final satisfaction, compromise, settlement, release, and discharge of and in exchange for such Allowed SALIC TruPS Claim, (a) its Pro Rata share of either the TruPS Shares or the TruPS Cash and (b) its Pro Rata share of Distribution Trust Interests.
	SALIC General Unsecured Claim agrees to a lesser treatment, on or as soon as reasonably practicable following the Effective Date, each holder of an Allowed SALIC General Unsecured Claim will receive, in full and final satisfaction, compromise, settlement, release, and discharge of and in exchange for such Allowed SALIC General Unsecured Claim, (a) its Pro Rata share of the Plan Funding Payment available for distribution to unsecured creditors and (b) its Pro Rata share of Distribution Trust Interests.
	<u>Classification</u> : To the extent permissible under the Bankruptcy Code and reasonably determined by the Debtors and the Purchaser to be conducive to obtaining confirmation of the Chapter 11 Plan, the Chapter 11 Plan should provide for separate classification of SALIC TruPS Claims and any SALIC General Unsecured Claims.
Inter-Debtor Claims:	Same treatment as Filed Plan.

<sup>&</sup>lt;sup>4</sup> The SALIC TruPS Claims include the TruPS held by SRGL against SALIC. The Joint Liquidators have confirmed in writing that they will elect cash, thus, SRGL will only be able to receive cash on account of the TruPS held by SRGL against SALIC, unless the Purchaser and SRGL agree otherwise pursuant to mutually agreeable terms to be included in the Plan.

The classification and treatment of any claims held by SFL against SALIC is subject to further discussion among the Debtors, the Purchaser and SFL. Any such classification and treatment of SFL's claims shall be consistent with this Plan Term Sheet and the Stock Purchase Agreement and on terms acceptable to Purchaser.

Further, the SALIC TruPS Claims shall not include the fees and expenses of the indenture trustees or other agents. The Charging Lien of the Indenture Trustees shall be satisfied through the TruPS Trustee Fees and Expenses (as such term is defined in the Stock Purchase Agreement).

Intercompany Non-	Same treatment as Filed Plan.
Debtor Claims:	
SHI Existing Equity Interests:	Same treatment as Filed Plan.
Interests:	
SALIC Existing	Same treatment as Filed Plan. <sup>5</sup>
<b>Equity Interests:</b>	
	GENERAL PLAN PROVISIONS
Sale of Reorganized	Pursuant to the Chapter 11 Plan and the Stock Purchase Agreement, on
SALIC Equity	the Effective Date, Reorganized SALIC will issue to the Purchaser
Interests:	70% of 19,999,999,999 shares of US\$0.001 par value New SALIC
	Shares, which shall be deemed fully paid and non-assessable upon issuance. The remaining 30% of the equity of the Reorganized SALIC shall be issued to either the Purchaser or the holders of TruPS (as applicable) in accordance with the Chapter 11 Plan. <sup>6</sup>
	Pursuant to and subject to the terms and conditions set forth in the Stock Purchase Agreement, in consideration of its contemplated receipt of the New SALIC Shares and other good and valuable consideration, the Purchaser will pay, on the Effective Date, the Purchase Price (as defined in the Stock Purchase Agreement).
	The Chapter 11 Plan and the Confirmation Order shall each provide that the issuance of any equity securities in connection therewith, including the New SALIC Shares, will be exempt from securities laws to the fullest extent permitted by section 1145 of the Bankruptcy Code.
Plan Distribution	Pursuant to the Chapter 11 Plan and a liquidation trust agreement (the

<sup>5</sup> Purchaser reserves the right to amend the treatment based upon applicable Cayman law.

<sup>&</sup>lt;sup>6</sup> Notwithstanding anything to the contrary in this Plan Term Sheet or the Stock Purchase Agreement to the contrary, in connection with any assignment by Purchaser of its rights under the Stock Purchase Agreement (including any transaction prior to the Closing pursuant to which Purchaser redomiciles or reorganizes in another jurisdiction) pursuant to Section 9.4 of the Stock Purchase Agreement, Purchaser may determine to have the TruPS Shares issued by a newly formed holding company organized for the purpose of directly or indirectly holding all of the issued and outstanding Ordinary Shares of SALIC immediately following the Closing, through which newly formed holding company Purchaser and the TruPS Holders who receive TruPS Shares at the Closing in accordance with the Chapter 11 Plan indirectly hold their equity interests in Reorganized SALIC (it being understood that if Purchaser determines to modify the Plan Term Sheet and/or the Chapter 11 Plan in accordance with the foregoing, in all instances, Reorganized SALIC will be a directly or indirectly wholly-owned subsidiary of such newly formed holding company through which Purchaser and such TruPS Holders hold their equity interests in Reorganized SALIC). Capitalized terms in this footnote shall have the meaning ascribed to them in the Stock Purchase Agreement. For the avoidance of doubt, although this footnote sets forth the agreement in concept of the Debtors and Purchaser that Purchaser may determine to have the TruPS Shares issues by a newly formed holding company as set forth above, all rights of the Debtors and the Purchaser are reserved with respect to the specific terms and conditions relating to any contemplated issuance of the TruPS Shares by a newly formed holding company instead of Reorganized SALIC.

Trust and Application of the Plan Funding Payment:	terms of which are to be determined, but which shall be acceptable to the Debtors and reasonably acceptable to the Purchaser), on the Effective Date, a liquidating trust (the " <u>Distribution Trust</u> ") will be established on terms and conditions intended to qualify and be treated as a grantor trust for U.S. federal income tax purposes pursuant to Sections 671 through 677 of the Internal Revenue Code of 1986. The Debtors shall appoint the initial trustee of the Distribution Trust (the " <u>Distribution Trustee</u> "), which person or entity shall be reasonably acceptable to the Purchaser.
	The beneficiaries of the Distribution Trust shall be (i) the holders of Other Secured Claims, Administrative Expense Claims, Priority Claims and Priority Tax Claims that have not received through the Closing Date Plan Distributions made on account of such Claims all distributions to which the holders of such Claims are entitled under the Chapter 11 Plan, and (ii) the holders of Allowed SHI TruPS Claims, Allowed SHI General Unsecured Claims, Allowed SALIC TruPS Claims and Allowed SALIC General Unsecured Claims.
	Upon the Effective Date, the Plan Funding Payment shall be allocated as follows:
	<ul> <li>(a) <u>First</u>, to fund all Closing Date Plan Distributions to the extent that such Closing Date Plan Distributions are not fully funded from the unrestricted Cash available to SALIC and SHI;</li> </ul>
	(b) <u>Second</u> , to fund the Professional Fee Reserve;
	(c) <u>Third</u> , to fund the Trustee Administration Reserve;
	(d) <u>Fourth</u> , to fund the TruPS Trustee Fees and Expenses;
	(e) <u>Fifth</u> , to fund the Disputed Claims Reserve.
	<ul> <li>(f) <u>Sixth</u>, to fund distributions Pro Rata to Allowed SHI General Unsecured Claims, Allowed SALIC General Unsecured Claims, Allowed SHI TruPS Claims (that elect to receive TruPS Cash) Allowed SALIC TruPS Claims (that elect to receive TruPS Cash).</li> </ul>
	For the avoidance of doubt, any Plan Funding Payment not allocated to the Distribution Trust on account of the distributions to be made to Allowed SHI TruPS Claim and Allowed SALIC TruPS Claims electing to receive equity shall reduce the Plan Funding Payment made by Purchaser.
	In addition, upon the Effective Date, the following shall be contributed and transferred to the Distribution Trust: (i) to the extent contemplated by the Confirmation Order, all capital stock or share capital of SFL held by any SALIC Group Company; (ii) all SALIC/SRGL Claims; (iii) all DT Post-Closing Rights and (iv) causes of action against the equity holders of SRGL. For the avoidance of doubt, these assets

	transferred to the Distribution Trust shall solely be for the benefit of all holders of Allowed SHI TruPS Claims, Allowed SHI General Unsecured Claims, Allowed SALIC TruPS Claims, and Allowed SALIC General Unsecured Claims; <u>provided</u> , that to the extent the Closing Date Plan Distributions are not satisfied in full, then such assets described in this paragraph may be used to fulfill such Closing Date Plan Distributions.
Tax Matters:	To the extent possible, the Restructuring contemplated by this Term Sheet shall be structured so as to obtain the most beneficial tax structure for the Reorganized Debtors and the holders of the Reorganized SALIC Equity Interests as reasonably determined by the Debtors and the Purchaser.
Claims Resolution Matters:	Other than as set forth in the Restructuring Implementation Agreement, prior to the Effective Date of the Chapter 11 Plan, the Debtors shall not enter into any agreements with holders of Claims or Equity Interests relating to the allowance, estimation, validity, extent or priority of such Claims or Equity Interests, or the classification and treatment of such Claims or Equity Interests under the Chapter 11 Plan, without the consent of the Purchaser (such consent not to be unreasonably withheld), except for (i) claims which the Debtors are authorized to pay pursuant to an applicable first day order, (ii) undisputed Administrative Expense Claims arising postpetition in the ordinary course of business, and (iii) Claims for which the allowed amount is less than US\$100,000. For the avoidance of doubt, on and after the Effective Date, the Distribution Trustee may enter into any such agreements with holders of Claims and Equity Interests without the consent of the Purchaser.
Executory Contracts:	Except as otherwise ordered by the Bankruptcy Court, on the Effective Date, the Debtors: (a) will assume all Executory Contracts required or otherwise designated by the Purchaser to be assumed pursuant to the Stock Purchase Agreement; and (b) will reject all Executory Contracts required or otherwise designated by the Purchaser to be rejected pursuant to the Stock Purchase Agreement.
Board of Reorganized SALIC:	The initial board of directors of Reorganized SALIC shall be selected by the Purchaser in its sole discretion.
Other Governance Matters/Reporting Obligations:	Pursuant to the Chapter 11 Plan, all corporate governance documents related to each of the Reorganized Debtors, including, but not limited to, the charter, bylaws and/or other organizational documents, shall be in form and substance acceptable to the Purchaser in its sole discretion.
Retained Causes of Action:	The Chapter 11 Plan shall contain customary provisions regarding revesting in the Reorganized Debtors of all causes of action, other than those expressly released pursuant to the Chapter 11 Plan, subject to the consent of the Purchaser. Any causes of action against Released Parties shall not be transferred or assigned to the Distribution Trust.

Releases and	The Chapter 11 Plan shall include, to the extent permitted by law,
Exculpation:	customary release and exculpation provisions in favor of (i) the Debtors, (ii) Hildene, (iii) the Purchaser, (iv) the trustees of the trusts established pursuant to the TruPS Declarations, (v) the indenture trustees under the TruPS Indentures, (vi) the Joint Liquidators and SRGL, (vii) any official committee of creditors appointed by the United States Trustee in the Chapter 11 Cases and the members thereof (in such capacity only), (viii) the Plan Sponsor (as such term is defined in the Filed Plan), and (ix) for each of the entities identified in the preceding clauses (i) through (viii), each entity's respective current and former affiliates, current and former officers, managers, directors, equity holders (regardless of whether such interests are held directly or indirectly), predecessors, successors, assigns, subsidiaries, principals, members, employees, agents, independent contractors, managed accounts or funds, management companies, fund advisors, advisory board members, financial advisors, partners, attorneys, accountants, investment bankers, consultants, representatives, and other professionals, each in their capacity as such in their capacities as such; <u>provided</u> , <u>however</u> , that, the equity holders of SRGL shall not be a Released Party or a Representative of a Released Party (as such terms are defined in the Filed Plan).
Injunction/Discharge:	The Chapter 11 Plan and Confirmation Order shall contain discharge and injunction provisions that are satisfactory to the Debtors and the Purchaser.
Conditional Transfer of SFL Equity Interests to Distribution Trust:	On the Effective Date, to the extent not previously abandoned by order of the Bankruptcy Court, all Equity Interests held by SALIC in SFL shall be contributed to the Distribution Trust.
Conditions to Confirmation and Effectiveness:	The Chapter 11 Plan shall contain and be subject to usual and customary conditions to confirmation and effectiveness (as applicable), as well as such other conditions that are reasonably satisfactory to the Debtors and the Purchaser, including the following:
	• The satisfaction or waiver in accordance with the Stock Purchase Agreement of all conditions to closing of the Stock Purchase Agreement.
	• The Bankruptcy Court shall have entered a Disclosure Statement Order in form and substance reasonably acceptable to the Debtors and the Purchaser.
	• Except as provided in the immediately succeeding bullet, the Chapter 11 Plan and all documents contained in any Plan Supplement (which will include the management agreement with AlesiaRe), including any exhibits, schedules, amendments, modifications or supplements thereto, and all other Restructuring Documents shall have been negotiated,

	<ul> <li>executed, delivered and filed with the Bankruptcy Court in substantially final form and in form and substance reasonably acceptable to the Debtors and Purchaser.</li> <li>With respect to the amended Restructuring Implementation Agreement, (i) the amended RIA Order shall have been entered by the Bankruptcy Court and be a Final Order, (ii) all necessary Foreign Court Approvals shall have been obtained and remain in full force and effect, and (iii) the amended Restructuring Implementation Agreement shall not have been terminated and shall be in full force and effect.</li> </ul>
	• The Bankruptcy Court shall have entered a Confirmation Order in form and substance reasonably acceptable to the Debtors and the Purchaser and the Confirmation Order shall be a Final Order.
	• The Debtors and Purchaser shall have obtained all authorizations, consents, regulatory approvals, rulings, waivers or other documents that are necessary to implement and effectuate the Chapter 11 Plan and Stock Purchase Agreement.
	• SRGL's winding up proceedings before the Supreme Court of Bermuda and its parallel winding up proceeding before the Grand Court of the Cayman Islands, Financial Services Division shall not have been dismissed without the consent of the Debtors and the Purchaser.
	• SRUS shall not have become the subject of a pending DDOI Proceeding.
OTHER PROVISIONS	
Director and Officer Indemnification:	To be treated in accordance with the Stock Purchase Agreement.
Definitive Documents:	As soon as reasonably practicable, the Debtors and Purchaser will execute definitive documentation implementing the Restructuring in form and substance materially consistent with this Term Sheet.

# Term Sheet Annex 1

### Additional Definitions

"<u>Entity</u>" means an "entity" as defined in section 101(15) of the Bankruptcy Code.

"<u>Equity Interest</u>" means all outstanding ownership interests in an Entity, including any interest evidenced by common or preferred stock, a limited liability company or other membership or partnership interest or unit, a warrant, an option, or any other right to acquire or otherwise receive any ownership interest in such Entity, or any right to payment or compensation based upon any such interest, whether or not such interest is owned by the holder of such right to payment or compensation.

"<u>Professionals</u>" means (a) any Entity employed in the Chapter 11 Cases pursuant to section 327, 328, 363 or 1103 of the Bankruptcy Code or otherwise and (b) any Entity seeking compensation or reimbursement of expenses in connection with the Chapter 11 Cases pursuant to section 503(b)(4) of the Bankruptcy Code.

"<u>Reorganized</u>" means, in reference to a Debtor, such Debtor from and after the Effective Date.

"<u>Restructuring Implementation Agreement</u>" has the meaning ascribed to such term in the Preamble to this Term Sheet.

"<u>SALIC General Unsecured Claim</u>" means any Claim against SALIC that is not an Other Secured Claim, Administrative Expense Claim, Priority Tax Claim, Priority Claim, Inter-Debtor Claim or Intercompany Non-Debtor Claim.

"<u>SALIC Share Subdivision</u>" means the subdivision by SALIC of each US\$1.00 par value Existing SALIC Share into 1,000 US\$0.001 par value shares. The SALIC Share Subdivision was made on January 19, 2018, pursuant to a duly authorized written shareholder resolution from SRGL.

"SALIC TruPS Claim" means any TruPS Claim against SALIC.

"<u>SHI General Unsecured Claim</u>" means any Claim against SHI that is not an Other Secured Claim, Administrative Expense Claim, Priority Tax Claim, Priority Claim, Inter-Debtor Claim or Intercompany Non-Debtor Claim.

"SHI TruPS Claim" means any TruPS Claim against SHI.

"<u>Stock Purchase Agreement</u>" has the meaning ascribed to such term in the Preamble to the Term Sheet.

# <u>Exhibit C</u>

**Restructuring Implementation Agreement** 

#### IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re

SCOTTISH HOLDINGS, INC., et al.,

Debtors.<sup>1</sup>

Chapter 11

**Re: D.I. 29** 

Case No. 18-10160 (LSS)

Jointly Administered

#### ORDER AUTHORIZING DEBTORS TO ASSUME RESTRUCTURING IMPLEMENTATION AGREEMENT AND GRANTING RELATED RELIEF

Upon the Debtors' Motion For: (A) Order Authorizing Debtors' Assumption Of Restructuring Implementation Agreement And Granting Related Relief; And (B) Order Authorizing Debtors' Assumption Of Plan Sponsorship Agreement And Granting Related Relief [D.I. 29] (the "Motion"), insofar as such Motion seeks authorization for the above-captioned debtors and debtors-in-possession (together, the "Debtors") to assume the Restructuring Implementation Agreement attached hereto as **Exhibit 1** (the "RIA");<sup>2</sup> and the Court having jurisdiction to consider the Motion and the relief requested therein in accordance with 28 U.S.C. §§ 157 and 1334; and consideration of the Motion and the relief requested therein being a core proceeding pursuant to 28 U.S.C. § 157(b)(2); and venue being proper in this District pursuant to 28 U.S.C. §§ 1408 and 1409; and due and proper notice of the Motion being adequate and appropriate under the particular circumstances; and a hearing having been held to consider the relief requested in the Motion (the "Hearing"); and upon consideration of the First Day Declaration, the supplemental declaration of Gregg Klingenberg in support of, among other

<sup>&</sup>lt;sup>1</sup> The Debtors, along with the last four digits of their federal tax identification numbers, are as follows: Scottish Holdings, Inc. (4408) and Scottish Annuity & Life Insurance Company (Cayman) Ltd. (3285). The Debtors' mailing address for purposes of these chapter 11 cases is 14120 Ballantyne Corporate Place, Suite 300, Charlotte, NC 28277.

<sup>&</sup>lt;sup>2</sup> Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to them in the Motion.

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 195 of 241

things, the Motion [D.I. 28], the record of the Hearing and all proceedings had before the Court; the Parties (as defined in the RIA) having determined that it may be in their best interest to seek a stay of the Bermuda Winding Up Proceeding (as defined in the RIA) (the "Bermuda Stay"); and the Court having found and determined that the relief sought in the Motion is in the best interests of the Debtors' estates, their creditors and other parties in interest; and that the legal and factual bases set forth in the Motion establish just cause for the relief granted herein; and any objections to the requested relief having been withdrawn, resolved as set forth in this Order or on the record at the Hearing, or overruled on the merits; and upon all of the proceedings had before the Court and after due deliberation and sufficient cause appearing therefor,

IT IS HEREBY ORDERED THAT:

1. The Motion is GRANTED, as set forth herein.

2. The Debtors are authorized to: (a) assume the RIA, as modified by this Order, in its entirety, and, effective as of the date of entry of this Order, the RIA is hereby assumed pursuant to section 365(a) of the Bankruptcy Code; (b) comply with the terms of the RIA; and (c) take any and all actions necessary to implement the terms of the RIA (except for actions that require further orders of the Court).

3. The RIA, as modified by this Order, shall be binding and specifically enforceable against the parties thereto in accordance with its terms.

4. The failure to describe specifically or include any particular provision of the RIA in the Motion or this Order shall not diminish or impair the effectiveness of such provision, it being the intent of the Court that the RIA be assumed by the Debtors in its entirety, as modified by this Order.

5. Section 9 (Limitations) of the RIA shall be stricken in its entirety and replaced with the language set forth in the insert attached hereto as **Exhibit 2**.

2

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 196 of 241

6. Section 3.2(a) of the RIA is amended by replacing "either Winding Up Proceeding" with "the Cayman Islands Winding Up Proceeding."

7. Section 10.2(b) of the RIA is amended by replacing "either Winding Up Proceeding" with "the Cayman Islands Winding Up Proceeding."

8. Notwithstanding anything set forth in the RIA, including Section 5.4(c) of the RIA, nothing in the RIA or this Order obligates the Debtors or SRGL to pursue releases for insiders and affiliates, other than SRGL and the Joint Liquidators (to the extent that they are deemed insiders or affiliates). For the avoidance of doubt, this Order and the relief granted hereunder are without prejudice to the ability of any party in interest, including the Committee, to object to the granting of any releases, including those contemplated by Section 5.4(c) of the RIA.

9. Unless and until the RIA is terminated in accordance with Article 10 of the RIA as modified by this Order and except only as provided in Section 10.5(a)(iii) of the RIA with respect to a termination that occurs following the New SALIC Share Issuance, (a) the acknowledgements and stipulations (collectively, the "<u>SRGL TruPS Claims Stipulations</u>") set forth in the RIA relating to the SRGL TruPS Claims, including those in Section 7.1 of the RIA, are binding upon the Debtors and their estates, effective immediately upon the entry of this Order in accordance with the RIA and (b) the SRGL TruPS Claims are deemed allowed unsecured, nonpriority claims against the Debtors, without duplication, in accordance with the SRGL TruPS Claims Stipulations.

10. For the avoidance of doubt, nothing in the RIA or this Order entitles SRGL to collect duplicate distributions on account of the allowed SRGL TruPS Claim (for example, through a direct distribution on account of the allowed SRGL TruPS Claim and a distribution made through any institutional trustee or indenture trustee relating to the SHST II TruPS, the

3

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 197 of 241

GPIC TruPS or the SHST III TruPS, as applicable) that would allow SRGL a greater pro rata recovery on account of the SRGL TruPS Claims than afforded to Other TruPS Claims, whether under a Chapter 11 Plan or otherwise.

11. Section 6.1(a) of the RIA shall be and hereby is amended by adding the following after the words "January 1, 2018" at the end of the existing provision: "(the "Admitted SALIC/SRGL Revolver Claim"), and the Admitted SALIC/SRGL Revolver Claim is allowed and admitted in full in the Winding Up Proceedings;".

For the avoidance of doubt, except as provided in Section 10.5(a)(iii) of the RIA 12. with respect to a termination that occurs following the New SALIC Share Issuance, if the RIA is terminated in accordance with Article 10, as modified by this Order: (a) (i) the SRGL TruPS Claims Stipulations, including the provisions set forth in Article 7 of the RIA (SRGL TruPS Claims and Holdings; Distributions on Account Thereof in Chapter 11 Cases), shall not be binding upon the estates and (ii) the allowance of SRGL TruPS Claims as unsecured, nonpriority claims against the Debtors in accordance with the SRGL TruPS Claims Stipulations pursuant Paragraph 9, clause (b) of this Order shall no longer be effective and the SRGL TruPS Claims shall be fully subject to allowance or disallowance in these Chapter 11 Cases in accordance with the law applicable in these Chapter 11 Cases; and (b)(i) Article 6 (SALIC Claims; Distributions on Account Thereof in Winding Up Proceedings) of the RIA shall not be binding upon SRGL, and (ii) notwithstanding Section 6.1(a) of the RIA as modified by this Order, the Admitted SALIC/SRGL Revolver Claim shall cease to have the status of a finally admitted and allowed claim against SRGL in connection with Winding Up Proceedings and shall be fully subject to allowance or disallowance in the Winding Up Proceedings in accordance with the law applicable in such Winding Up Proceedings.

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 198 of 241

13. Section 10.2 of the RIA is hereby deemed amended to add thereto a new Section 10.2(h) that provides: "(h) the Restructuring Transactions with the Purchaser are not consummated and a chapter 11 plan of reorganization for SALIC and SHI is confirmed and becomes effective that does not require SRGL's surrender of Existing SALIC Shares."

14. If (x) the Restructuring Transactions with the Purchaser are not consummated and (y) a chapter 11 plan of reorganization for the Debtors is confirmed and becomes effective that does not require SRGL's surrender of Existing SALIC Shares, then, unless otherwise agreed in writing by the Committee (which may be by email from the Committee's counsel of record in the Chapter 11 Cases) and unless the RIA has been previously terminated in accordance with the terms thereof, the SALIC Parties shall deliver a Termination Notice pursuant to section 10.2(h) of the RIA, as modified by this Order, in accordance with the terms of the RIA. Upon termination of the RIA in accordance with its terms, all parties' rights and defenses are reserved regarding the allowance of or any objections to the SRGL TruPS Claims and the SRGL/SALIC Revolver Claims with respect to any proposed chapter 11 plan that does not require SRGL's surrender of Existing SALIC Shares.

15. Section 10.2(g) of the RIA is hereby deemed amended by adding immediately prior to the period at the end thereof: "and that requires SRGL's surrender of Existing SALIC Shares."

16. Reference is made to the definitions of "SALIC/SRGL Claims," "Chapter 11 Plan" and "Distribution Trust" in Section 1.1 of the Stock Purchase Agreement and Section 2.4(h) of the Stock Purchase Agreement concerning assets to be contributed and transferred to the Distribution Trust at the Closing of the Stock Purchase Agreement (each a "<u>SALIC/SRGL</u> <u>Claims-Related SPA Provision</u>"). Notwithstanding anything to the contrary set forth in the RIA, the Plan Sponsorship Agreement, or the Stock Purchase Agreement, without the prior written

5

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 199 of 241

consent of the Committee (which may be by email from the Committee's counsel of record in the Chapter 11 Cases), (a) the Debtors shall not amend, modify or waive enforcement of any SALIC/SRGL Claims-Related SPA Provision insofar as any such SALIC/SRGL Claims-Related Provision concerns the contribution or transfer of the SALIC/SRGL Claims (including the SALIC/SRGL Revolver Claim and any right to a distribution on account of the Admitted SALIC/SRGL Revolver Claim), and (b) the Debtors shall not enter into any agreement for an Alternative Transaction that does not similarly preserve such SALIC/SRGL Claims for the benefit of the Debtors' creditors under a confirmed and effective plan of reorganization.

17. For the avoidance of doubt, nothing in this Order or the RIA is intended to adjudicate whether or to what extent the SRGL TruPS Claims would be counted for purposes of determining, pursuant to section 1129(a)(10) of the Bankruptcy Code, whether at least one class of claims that is impaired under the Chapter 11 Plan has accepted the Chapter 11 Plan.

18. Subject to Paragraph 16 of this Order, the Debtors, after consultation with the Committee and the Purchaser, are authorized, but not directed, to enter into amendments to, modifications of or waivers of terms of the RIA, from time to time as necessary, including to facilitate the Bermuda Stay, subject to the terms and conditions set forth in the RIA and without further order of the Court, *provided, however*, that any such amendment, modification or waiver shall be effective only upon further order of the Court if such amendment, modification or waiver (a) is or reasonably should be anticipated by the Debtors to have a material adverse effect on the Debtors' estates, or (b) is an amendment to or modification or waiver of Section 9 (Limitations) or Section 10.2 of the RIA, as modified by this Order. Within two (2) Business Days of the effective date of each such amendment, modification or waiver for which a further order of the Court is not required, the Debtors will file a notice attaching a copy of any such amendment, modification or waiver with the Court.

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 200 of 241

19. Notice of the Motion as provided therein shall be deemed good and sufficient and the requirements of the Bankruptcy Rules and the Local Rules are satisfied by such notice.

20. To the extent the automatic stay provisions of section 362 of the Bankruptcy Code would otherwise apply, such provisions are vacated and modified to permit the delivery of any Termination Notice pursuant to the RIA.

21. Except for the rights of the Joint Liquidators and, as applicable, the Purchaser or other Plan Sponsor, and except as expressly stated within the RIA, the RIA shall be solely for the benefit of the parties thereto and no other person or entity shall be a third-party beneficiary to the RIA. Except as expressly provided in the RIA, no entity, other than the parties to the RIA or their respective successors (including any trustee appointed in these bankruptcy cases) shall have any right to seek or enforce specific performance under the RIA agreement.

22. For the avoidance of doubt, no prior consent of the Purchaser is required for the SALIC Parties to deliver a Termination Notice on the basis of the SALIC Parties' conclusion in accordance with Section 10.2(d) that continued performance under the RIA is inconsistent with the Limitations set forth in Article 9 of the RIA as modified by this Order, even if the facts and circumstances giving rise to such conclusion involve SRGL's material breach of the RIA in a way that gives rise to a termination right under Section 10.2(e); provided, that this provision does not affect Purchaser's right to notice under Section 10.2(d) or its separate consent rights under Section 10.2(e).

23. The failure of any party to seek relief or otherwise exercise its rights and remedies under this Order, the RIA, or applicable law, as the case may be, shall not constitute a waiver of any of the rights hereunder, thereunder, or otherwise of any of the parties.

7

#### Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 201 of 241

24. The RIA and this Order are not intended to, and do not, foreclose bidders from making and the Debtors from accepting bids for Alternative Restructuring Transactions as defined in the Bid Procedures Order (D.I. 119).

25. Notwithstanding the possible applicability of Bankruptcy Rules 6004(h), 7062,9014 or otherwise, this Order shall be immediately effective and enforceable upon its entry.

26. The Debtors are authorized to take all actions necessary to effectuate the relief granted pursuant to this Order.

27. The Debtors are directed to file, within three (3) days after entry of this Order, the RIA as conformed to reflect the modifications made pursuant to this Order.

28. The Court retains jurisdiction with respect to all matters arising from or related to the interpretation or implementation of this Order. For the avoidance of doubt, nothing in the RIA or this Order cedes or limits in any way (a) the jurisdiction of this Court over any claims scheduled, filed, asserted or otherwise made against either Debtor in the Chapter 11 Cases, including, but not limited to the SRGL TruPS Claims, or (b) the jurisdiction of the Cayman Islands Court over any claims filed, asserted or otherwise made against SRGL in the Winding Up Proceedings, including, but not limited to, the SALIC/SRGL Revolver Claims.

Date: Mark 19, 2018

THE HONORABLE LAURIE SELBER SILVERSTEIN UNITED STATES BANKRUPTCY JUDGE

# <u>Exhibit 1</u>

[RIA]

Case 18-10160-LSS Doc 437-1 Filed 08/09/18 Page 203 of 241

**EXECUTION VERSION** 

# **RESTRUCTURING IMPLEMENTATION AGREEMENT**

By and Among Scottish Re Group Limited (Joint Provisional Liquidators Appointed)

And

Scottish Annuity & Life Insurance Company (Cayman) Ltd. and Scottish Holdings, Inc. THIS RESTRUCTURING IMPLEMENTATION AGREEMENT IS NOT AN OFFER WITH RESPECT TO ANY SECURITIES OR A SOLICITATION OF VOTES WITH RESPECT TO A PLAN OF REORGANIZATION. ANY SUCH OFFER OR SOLICITATION WILL COMPLY WITH ALL APPLICABLE SECURITIES LAWS AND/OR PROVISIONS OF THE BANKRUPTCY CODE.

THIS RESTRUCTURING IMPLEMENTATION AGREEMENT (as may be amended, restated, supplemented, or otherwise modified from time to time in accordance with the terms hereof, together with the exhibits and schedules attached hereto (as each may be amended, restated, supplemented, or otherwise modified from time to time, in accordance with the terms hereof, the "Agreement") is made on January 28, 2018 (the "Agreement Date") between the following parties (collectively, the "Parties"):

- (a) Scottish Re Group Limited, an exempted company limited by shares incorporated and existing under the laws of the Cayman Islands ("SRGL"), with a permit to operate in Bermuda where Joint Provisional Liquidators (as defined herein) have been appointed;
- (b) Scottish Annuity & Life Insurance Company (Cayman) Ltd., an exempted company limited by shares incorporated and existing under the laws of the Cayman Islands ("SALIC"); and
- (c) Scottish Holdings, Inc., a corporation organized and existing under the laws of the state of Delaware ("SHI").

### RECITALS

- (A) SALIC and certain of its direct and indirect subsidiaries, including Scottish Re (U.S.), Inc. ("SRUS"), are engaged in the reinsurance of life insurance, annuities and annuitytype products (the "SALIC Group Business").
- (B) SRGL, SALIC, SHI and certain of their affiliates have undertaken an evaluation of strategic alternatives and have determined to effectuate a restructuring of themselves and the SALIC Group Business with the assistance of their legal and financial advisors on the terms set forth in the Restructuring Documents (as defined below) (the "Restructuring").
- (C) On May 17, 2017, SRGL (i) commenced a winding up proceeding (the "Bermuda Winding Up Proceeding") in the Supreme Court of Bermuda (the "Bermuda Court") by filing a petition (the "Bermuda Winding Up Petition") seeking its winding up pursuant to Part XIII of the Bermuda Companies Act 1981 and other applicable Bermuda law, and (ii) commenced a parallel winding up proceeding (the "Cayman Islands Winding Up Proceeding," and together with the Bermuda Winding Up Proceeding, the "Winding Up Proceedings") in the Grand Court of the Cayman Islands, Financial Services Division (the "Cayman Islands Court") by filing a petition (the "Cayman Islands Winding Up Petition," and together with the Bermuda Winding

Up Petition, the "Winding Up Petitions") seeking its winding up under Cayman Islands law.

- (D) Simultaneous with SRGL's filing of the Bermuda Winding Up Petition, SRGL filed an application to the Bermuda Court for the appointment of joint provisional liquidators in connection with the Bermuda Winding Up Proceeding.
- (E) On May 18, 2017, the Bermuda Court appointed John C. McKenna of Finance & Risk Services Ltd., and Eleanor Fisher of Kalo (Cayman) Limited as Joint Provisional Liquidators (as defined herein) for SRGL in connection with the Bermuda Winding Up Proceeding.
- (F) The Parties contemplate that, following the execution of this Agreement, (i) upon the application by SRGL, the Bermuda Court will make a winding up order for SRGL and will grant the Joint Provisional Liquidators wider powers as set forth in the Bermuda Act, and (ii) the Cayman Islands Court will make a winding up order and appoint Mr. McKenna and Ms. Fisher as joint official liquidators in the Cayman Islands Winding Up Proceeding (in such capacity following the making of the winding up orders in the Bermuda Winding Up Proceeding and the Cayman Islands Winding Up Proceeding, and without personal liability, the "Joint Official Liquidators"), the effect of which will be to displace the powers of the board of SRGL.
- (G) On January 19, 2018, pursuant to a duly authorized written shareholder resolution from SRGL, each US\$1.00 par value Ordinary Share (as defined herein) was subdivided into 1,000 US\$0.001 par value Ordinary Shares (such event, the "SALIC Share Subdivision"). After giving effect to the SALIC Share Subdivision, as of the Agreement Date, SALIC's authorized share capital is US\$20,000,000 divided into 20,000,000 Ordinary Shares having a par value of US\$0.001 each.
- (H) As of the Agreement Date, SRGL owns one-hundred percent (100%) of the Ordinary Shares.
- (I) As of the Agreement Date, SALIC owns one-hundred percent (100%) of the issued and outstanding common stock of SHI.
- (J) As of the Agreement Date, SHI owns one-hundred percent (100%) of the issued and outstanding common stock of SRUS.
- (K) In furtherance of the Restructuring, SALIC and SHI each contemplate filing a voluntary petition for relief under Chapter 11 of title 11 of the United States Code (the "Bankruptcy Code") in the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court").
- (L) In furtherance of the Restructuring, contemporaneously with the Parties' entry into this Agreement, HSCM Bermuda Fund Ltd., a Bermuda limited company ("Purchaser"),

SALIC and SHI are entering into (i) the Stock Purchase Agreement (as defined herein) and (ii) the Plan Sponsorship Agreement (as defined herein).

- (M) This Agreement sets forth the Parties' understanding concerning certain of the steps and actions relating to SRGL and its businesses and property required to effectuate the Restructuring and the Restructuring Transactions (as defined herein).
- The agreements and undertakings of the Parties made, subject to and in accordance with **(N)** the terms of this Agreement and other applicable Restructuring Documents (as defined herein), include: (i) SRGL's diligent prosecution of its requests for Orders of the Bermuda Court and the Cayman Islands Court granting the Winding Up Petitions; (ii) SRGL's consent to, support for and cooperation with the Restructuring Transactions, including the Initial Shares Surrender, the Final Share Surrender and the New SALIC Shares Issuance (each as defined herein); (iii) SRGL's diligent prosecution of an application to the Cayman Islands Court to authorize the Joint Liquidators to cause SRGL to undertake its obligations under this Agreement and the Share Surrender Documents; (iv) the SALIC Parties' timely commencement and diligent prosecution of their Chapter 11 Cases on terms consistent with this Agreement; (v) an agreement between SRGL and the SALIC Parties allowing their respective intercompany claims; (vi) agreements among SRGL (including the Joint Liquidators), the SALIC Parties and the Purchaser as to certain releases, limitations of liability and exculpatory provisions, which shall be set forth in the Chapter 11 Plan and the Confirmation Order; and (vii) additional agreements and undertakings in furtherance of the contemplated Restructuring and Restructuring Transactions.

THE PARTIES, intending to be legally bound and in consideration for the mutual undertakings provided by each of the other Parties hereto, and for other valuable consideration, the receipt and sufficiency of which is hereby acknowledged, hereby AGREE as follows:

### **1 DEFINITIONS; INTERPRETATION**

**1.1** As used in this Agreement, the following terms have the meanings set forth below:

"Agreement" has the meaning set forth in the Preamble.

"Bankruptcy Code" has the meaning set forth in the Recitals.

"Bankruptcy Court" has the meaning set forth in the Recitals.

"Bermuda Court" has the meaning set forth in the Recitals.

"Cayman Court Power" means an Order made by the Cayman Islands Court

authorizing the Joint Liquidators to cause SRGL to undertake its obligations under this Agreement and the Share Surrender Documents.

"Cayman Islands Court" has the meaning set forth in the Recitals.

"CIMA" means the Cayman Islands Monetary Authority.

"Claim" means (a) any right to payment, whether or not such right is reduced to judgment, liquidated, unliquidated, fixed, contingent, matured, unmatured, disputed, legal, equitable, secured or unsecured; or (b) any right to an equitable remedy for breach of performance if such breach gives rise to a right to payment, whether or not such right to an equitable remedy is reduced to judgment, fixed, contingent, matured, unmatured, disputed, undisputed, secured or unsecured. As against any SALIC Party, the term "Claim" shall have a meaning coextensive with that of the term "Claim" as defined in section 101(5) of the Bankruptcy Code.

"Existing SALIC Shares" means any and all Ordinary Shares that SALIC is authorized to issue, whether prior to (when such Existing SALIC Shares had a par value of US\$1.00 per share) or after (when such Existing SALIC Shares have a par value of US\$0.001 per share) giving effect to the SALIC Share Subdivision.

"Final Share Surrender" means the surrender by SRGL to Reorganized SALIC of any and all Ordinary Shares in the possession, custody or control of SRGL for cancellation, which surrender and cancellation shall occur immediately after the closing of the Stock Purchase Agreement, but in all events not later than one (1) Business Day after the closing of the Stock Purchase Agreement.

"Initial Shares Surrender" means the surrender by SRGL to SALIC of all but one (1) of the Existing SALIC Shares in accordance with the Share Surrender Documents, and after giving effect to the SALIC Share Subdivision. Following the Initial Shares Surrender, SRGL shall hold one (1) US\$0.001 par value Ordinary Share, and 19,999,999,999 unissued Ordinary Shares shall remain in SALIC's authorized capital. The Initial Shares Surrender shall be completed on or before the fifth (5th) Business Day after the latest of the date that (w) the RIA Order is entered and effective by its terms, (x) the Cayman Court Power is made and effective by its terms, (y) the Bermuda Court's Winding Up Order is made and effective by its terms, and (z) if it is determined in good faith by SALIC, after consultation with Maples and Calder, that regulatory approval of CIMA is required, the date that SALIC receives CIMA's regulatory approval for the Initial Shares Surrender. Additionally, the Initial Shares Surrender shall be completed in a manner and on terms consistent with this Agreement and the Stock Purchase Agreement in all material respects.

"JL Advisors" means Robin J. Mayor, Conyers Dill & Pearman, (441) 299-4929, robin.mayor@conyersdill.com; Aristos Galatopoulos, Maples and Calder, (345) 814-5241, aristos.galatopoulos@maplesandcalder.com; Ned S. Schodek, Shearman & Sterling LLP, (212) 848-7052, ned.schodek@shearman.com; as the same may be replaced from time to time and collectively with any other advisors retained by the Joint Liquidators.

"Joint Liquidators" means the Joint Provisional Liquidators and the Joint Official Liquidators and shall also be construed so as to include any other Person(s) who from time to time may be appointed by either the Bermuda Court or the Cayman Islands Court as a liquidator over SRGL (whether as provisional or official liquidator and whether with full or limited powers), in such capacity and without personal liability.

"Joint Official Liquidators" has the meaning set forth in the Recitals.

"Joint Provisional Liquidators" means (a) the individuals (initially John C. McKenna of Finance & Risk Services Ltd., and Eleanor Fisher of Kalo (Cayman) Limited) appointed by the Bermuda Court as joint provisional liquidators for SRGL in connection with the Bermuda Winding Up Proceeding and, (b) such individuals or their successors to whom the Bermuda Court may grant wider powers, and, in each case, in such capacity and without personal liability.

"Limitations" has the meaning ascribed to such term in Article 9 (*Limitations*) of this Agreement.

"New SALIC Shares" means 19,999,999,000 shares of US\$0.001 par value Ordinary Shares in Reorganized SALIC to be issued through the New SALIC Shares Issuance.

"New SALIC Shares Issuance" means the authorization and issuance by SALIC, upon the occurrence of the Effective Date of the Chapter 11 Plan and the closing of the Stock Purchase Agreement, of the New SALIC Shares as fully-paid, free and clear to the Plan Sponsor pursuant to the terms and conditions of the Stock Purchase Agreement, the Plan Sponsorship Agreement, the Chapter 11 Plan, the Share Surrender Documents and the New SALIC Shares Issuance Documents. The New SALIC Share Issuance shall occur in a manner and on terms consistent with this Agreement and the Stock Purchase Agreement in all material respects.

"New SALIC Shares Issuance Documents" means each of the documents, directions and resolutions reasonably required of SALIC to effectuate the New SALIC Shares Issuance. The New SALIC Shares Issuance Documents shall be in form and substance reasonably satisfactory to the Parties and the Purchaser, and otherwise subject to the SRGL Consent Rights.

"Ordinary Shares" means the ordinary shares of SALIC.

"Other TruPS Claims" means all TruPS Claims other than the SRGL TruPS Claims.

"Parties" has the meaning set forth in the Preamble; <u>provided</u>, <u>however</u>, that from and after the Effective Date of the Chapter 11 Plan, any reference to a SALIC Party as a "Party" shall be construed to include the Distribution Trustee (or similar estate representative appointed under the Chapter 11 Plan) to the extent of the DT Post-Closing Rights. "Plan Sponsor" means the Purchaser or such other Person that is determined pursuant to the Winning Bidder Order to be the Winning Bidder if such Person's Winning Bid contemplates sponsorship of a plan of reorganization for the SALIC Parties.

"Plan Sponsorship Agreement" means that certain Plan Sponsorship Agreement, dated as of January 28, 2018, by and among SALIC, SHI, and Purchaser, substantially in the form as attached hereto as **Exhibit** A, together with all exhibits, schedules and other ancillary documents, as each may be amended, restated, supplemented, or otherwise modified from time to time, in accordance with the terms thereof. In the event that the Purchaser is not the Winning Bidder, the term "Plan Sponsorship Agreement" as used in this Agreement shall be deemed to refer to the plan sponsorship agreement that is executed by and among SALIC, SHI and the Person that is the Winning Bidder and that embodies the bid that is designated as the Winning Bid. The Plan Sponsorship Agreement shall be consistent in all material respects with this Agreement and subject to the SRGL Consent Rights.

"Purchaser" has the meaning set forth in the Recitals.

"Restructuring" has the meaning set forth in the Recitals.

"Restructuring Documents" means the Specified Restructuring Documents, collectively with this Agreement, the RIA/PSA Motion, the Bidding Procedures Motion, the Bidding Procedures Order, the Disclosure Statement Motion, the Disclosure Statement, the Disclosure Statement Order, the Winning Bidder Order and the Organizational Documents. The Restructuring Documents shall, unless otherwise expressly indicated herein, be consistent in all material respects with this Agreement.

"Restructuring Steps" means the undertakings of the Parties hereto and other Persons party to applicable Restructuring Documents and the associated deadlines for completing them, each as set forth in Article 5 (*Restructuring Steps*) of this Agreement.

"Restructuring Transactions" means the transactions necessary to the Restructuring on terms set forth in the Restructuring Documents, including (i) completion of the Initial Shares Surrender, the New SALIC Shares Issuance and the Final Share Surrender, and (ii) consummation of the Stock Purchase Agreement and the Chapter 11 Plan.

"SALIC Claims" means the SALIC/SRGL Revolver Claims and the SALIC Group Services Claim.

"SALIC Group Company" means SALIC or a SALIC Subsidiary and "SALIC Group Companies" means all of them.

"SALIC Group Services Claims" means all Claims arising out of or relating to the provision of information technology, legal, administrative and other services by any SALIC Group Company to or for the benefit of SRGL. For purposes of this Agreement , the aggregate SALIC Group Services Claims shall not exceed US\$100,000.

"SALIC Parties" means SALIC and SHI, whether prior to the Petition Date or as debtors in their respective Chapter 11 Cases; <u>provided</u>, <u>however</u>, that from and after the Effective Date of the Chapter 11 Plan, any reference to a "SALIC Party" or "SALIC Parties" shall be construed to include the Distribution Trustee (or similar estate representative appointed under the Chapter 11 Plan) to the extent of the DT Post-Closing Rights.

"SALIC Share Subdivision" means the subdivision by SALIC of each US\$1.00 par value Existing SALIC Share into 1,000 US\$0.001 par value shares.

"SALIC/SRGL Revolver Claims" means all Claims against SRGL arising under or relating to the SRGL Revolver Facility and the SRGL Revolver Facility Documents, including Claims for principal, interest, charges, fees and expenses of attorneys and other professionals, and any other obligations arising thereunder or in connection therewith.

"Share Surrender Documents" means each of the documents, directions and resolutions reasonably required of SRGL to effectuate the Initial Shares Surrender and the Final Share Surrender. The Share Surrender Documents shall be in form and substance reasonably satisfactory to the Parties and otherwise subject to the SRGL Consent Rights.

"Share Surrender" means the Initial Shares Surrender together with Final Share Surrender, which will result in SRGL's surrender of all Existing SALIC Shares.

"Specified Restructuring Documents" means the Stock Purchase Agreement, the Plan Term Sheet, the RIA Order, the Plan Sponsorship Agreement, the PSA Order, the Share Surrender Documents, the New SALIC Shares Issuance Documents, the Chapter 11 Plan, the Plan Supplement (to the extent that any documents contained therein adversely affect the Joint Liquidators, SRGL or the SRGL TruPS Claims), the Sale Order and the Confirmation Order. The Specified Restructuring Documents shall be consistent in all material respects with this Agreement and subject to the SRGL Consent Rights.

"SRGL Consent Rights" means a requirement for the SALIC Parties to obtain the written consent (which may be provided by email) of SRGL (such consent not to be unreasonably withheld) to (i) any Specified Restructuring Document not attached hereto, and (ii) any amendment or modification of a Specified Restructuring Document that would adversely affect the Joint Liquidators, SRGL or the SRGL TruPS Claims; provided, however, that an amendment or modification to a Specified Restructuring Document that may have the effect of reducing, but for the avoidance of doubt not eliminating, the anticipated recovery under the Chapter 11 Plan on account of any or all SRGL TruPS Claims shall not solely or primarily as a result of such amendment or modification be subject to the SRGL Consent Rights unless such amendment or modification disproportionately and adversely affects the anticipated recovery under the Chapter 11 Plan on account of the SRGL TruPS Claims relative to anticipated recoveries under the Chapter 11 Plan for Other TruPS Claims.

"SRGL TruPS Claims" means all TruPS Claims relating to or arising out of any SRGL TruPS Holdings, whether such Claims are held by SRGL or by another Person (including any trustee) in a fiduciary capacity for SRGL's benefit on account of its SRGL TruPS Holdings.

"SRGL TruPS Holdings" means any and all TruPS owned or held by SRGL as of the Agreement Date, including: (i) approximately US\$20.0 million of SHST II TruPS; (ii) approximately US\$10.0 million of GPIC TruPS; and (iii) approximately US\$13.0 million of SHST III TruPS.

"Stock Purchase Agreement" means that certain Stock Purchase Agreement, dated as of January 28, 2018, entered into by and among SALIC, SHI and the Purchaser, substantially in the form as attached hereto as **Exhibit B**, together with all exhibits, schedules and other ancillary documents, as each may be amended, restated, supplemented, or otherwise modified from time to time, in accordance with the terms thereof. In the event that the Purchaser is not the Winning Bidder, the term "Stock **Purchase Agreement**" as used in this Agreement shall be deemed to refer to the stock purchase agreement that is executed by and among SALIC, SHI and the Person that is the Winning Bidder and that embodies the Winning Bid. The Stock Purchase Agreement shall be consistent in all material respects with this Agreement and subject to the SRGL Consent Rights.

"Termination Date" means the date any termination of this Agreement is effective in accordance with Article 10 (*Termination*) hereof.

"Termination Notice" has the meaning set forth in Section 10.2 of this Agreement.

"TruPS" means:

(i) the undivided beneficial interests, having an aggregate liquidation amount of US\$17,500,000.00, in Scottish Holdings Statutory Trust I, a Connecticut statutory trust, issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of December 4, 2002, among State Street Bank and Trust Company of Connecticut, National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "SHST I TruPS");

(ii) the undivided beneficial interests, having an aggregate liquidation amount of US\$20,000,000.00, in Scottish Holdings, Inc. Statutory Trust II, a Connecticut statutory trust ("SHST II"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of October 29, 2003, among U.S. Bank National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "SHST II TruPS");

(iii) the undivided preferred beneficial interests, having an aggregate liquidation amount of US\$10,000,000.00, in GPIC Holdings Inc. Statutory Trust, a Delaware statutory trust ("GPIC"), issued pursuant to that certain Amended and Restated Trust Agreement dated as of November 14, 2003, among JPMorgan Chase Bank as property trustee, Chase Manhattan Bank USA, National Association as Delaware trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "GPIC TruPS");

(iv) the undivided beneficial interests, having an aggregate liquidation amount of US\$32,000,000.00, in Scottish Holdings, Inc. Statutory Trust III, a Connecticut statutory trust ("SHST III"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of May 12, 2004, among U.S. Bank National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "SHST III TruPS"); and

(v) the undivided beneficial interests, having an aggregate liquidation amount of US\$50,000,000.00, in SFL Statutory Trust I, a Delaware statutory trust, issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of December 15, 2004, among Wilmington Trust Company, as institutional trustee, the administrators of the issuer named therein, SFL as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "SFL TruPS").

"**TruPS Claims**" means Claims of any Person relating to or arising out of any TruPs, TruPS Junior Subordinated Debentures or TruPS Documents, including any Claims relating to or arising out of any TruPS Documents. For the avoidance of doubt, TruPS Claims shall include all of the Claims set forth in the preceding sentence that could be asserted by one or more of the several parties thereto without duplication.

#### "TruPS Declarations" means:

(i) with respect to the SHST I TruPS, that certain Amended and Restated Declaration of Trust by and among State Street Bank and Trust Company of Connecticut, National Association, as Institutional Trustee, Scottish Holdings, Inc., as Sponsor, and Paul Goldean and Oscar R. Scofield, as Administrators, dated as of December 4, 2002.

(ii) with respect to the SHST II TruPS, that certain Amended and

Restated Declaration of Trust by and among U.S. Bank National Association, as Institutional Trustee, Scottish Holdings, Inc., as Sponsor, and Paul Goldean and Oscar R. Scofield, as Administrators, dated as of October 29, 2003.

(iii) with respect to the GPIC TruPS, that certain Amended and Restated Declaration of Trust by and among Scottish Holdings, Inc., as Depositor, JPMorgan Chase Bank, as Property Trustee, Chase Manhattan Bank USA, National Association, as Delaware Trustee, and The Administrative Trustees Named Herein, as Administrative Trustees, dated as of November 14, 2003.

(iv) with respect to the SHST III TruPS, that certain Amended and Restated Declaration of Trust by and among U.S. Bank National Association, as Institutional Trustee, Scottish Holdings, Inc., as Sponsor, and Paul Goldean and Oscar R. Scofield, as Administrators, dated as of May 12, 2004.

(v) with respect to the SFL TruPS, that certain Amended and Restated Declaration of Trust by and among Wilmington Trust Company, as Institutional Trustee, Wilmington Trust Company, as Delaware Trustee, Scottish Financial (Luxembourg) S.À.R.L., as Sponsor, and Paul Goldean and George Scott, as Administrators, dated as of December 15, 2004.

**"TruPS Documents"** means collectively all TruPS Indentures, all documents evidencing TruPS Junior Subordinated Debentures, all TruPS Declarations, all TruPS Sponsor Guarantees, all TruPS Parent Guarantees and all related and ancillary documents and instruments, each as altered, amended, modified or supplemented from time to time and including all exhibits and schedules thereto.

#### "TruPS Indentures" means:

(i) with respect to the SHST I TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2032, between Scottish Holdings, Inc. as Issuer, and State Street Bank and Trust Company of Connecticut, National Association, as Trustee, dated as of December 4, 2002;

(ii) with respect to the SHST II TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2033, between Scottish Holdings, Inc., as Issuer, and U.S. Bank National Association, as Trustee, dated as of October 29, 2003;

(iii) with respect to the GPIC TruPS, that certain Junior Subordinated Indenture, due 2033, between Scottish Holdings, Inc., as Issuer, and JPMorgan Chase Bank, as Trustee, dated as of November 14, 2003;

(iv) with respect to the SHST III TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2034, between Scottish Holdings, Inc., as Issuer, and U.S. Bank National Association, as Trustee, dated

as of May 12, 2004; and

(v) with respect to the SFL TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2034, between Scottish Financial (Luxembourg) S.À.R.L., as Issuer, and Wilmington Trust Company, as Trustee, dated as of December 15, 2004.

# "TruPS Junior Subordinated Debentures" means:

(i) with respect to the SHST I TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Holdings, Inc., and State Street Bank and Trust Company of Connecticut, National Association, due 2032;

(ii) with respect to the SHST II TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Holdings, Inc., and U.S. Bank National Association, due 2033;

(iii) with respect to the GPIC TruPS, that certain Floating Rate Junior Subordinated Note Due 2033 between Scottish Holdings, Inc., and JPMorgan Chase Bank;

(iv) with respect to the SHST III TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Holdings, Inc., and U.S. Bank National Association, due 2034; and

(v) with respect to the SFL TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Financial (Luxembourg) S.À.R.L. and Wilmington Trust Company, due 2034.

## "TruPS Parent Guarantees" means:

(i) with respect to the SHST I TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and State Street Bank and Trust Company of Connecticut, National Association, dated as of December 4, 2002;

(ii) with respect to the SHST II TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and U.S. Bank National Association, dated as of October 29, 2003;

(iii) with respect to the GPIC TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and JPMorgan Chase Bank, dated as of November 14, 2003;

(iv) with respect to the SHST III TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and U.S. Bank National Association, dated as of May 12, 2004; and
(v) with respect to the SFL TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and Wilmington Trust Company, dated as of December 15, 2004.

#### "TruPS Sponsor Guarantees" means:

(i) with respect to the SHST I TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and State Street Bank and Trust Company of Connecticut, National Association, dated as of December 4, 2002;

(ii) with respect to the SHST II TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and U.S. Bank National Association, dated as of October 29, 2003;

(iii) with respect to the GPIC TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and JPMorgan Chase Bank, dated as of November 14, 2003;

(iv) with respect to SHST III TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and U.S. Bank National Association, dated as of May 12, 2004; and

(v) with respect to the SFL TruPS, that certain Guarantee Agreement by and between Scottish Financial (Luxembourg) S.À.R.L and Wilmington Trust Company, dated as of December 15, 2004.

"Winding Up Orders" means: (a) the Order to be made by the Bermuda Court in the Bermuda Winding Up Proceedings that, among other things, will expand the powers and responsibilities of the Joint Provisional Liquidators such that they become Bermuda Full Powers Liquidators for SRGL; and (ii) the Order to be made by the Cayman Islands Court in the Cayman Islands Winding Up Proceeding that, among other things, will appoint Mr. McKenna and Ms. Fisher as the Joint Official Liquidators.

- **1.2** Capitalized terms not otherwise defined in this Agreement have the meanings ascribed to such terms in the Stock Purchase Agreement, in the form attached hereto as **Exhibit B**.
- **1.3** In interpreting this Agreement, unless the context otherwise requires:
  - (a) the headings to Articles are for convenience only and shall not affect the construction of anything in this Agreement;
  - (b) references to Articles and Schedules are to be construed as references to the Articles of, and Schedules to, this Agreement, respectively;
  - (c) references to the singular includes the plural and vice versa and

references to any gender includes the other genders;

- (d) a reference to "including" means including, without limitation;
- (e) references to a Party includes such Party's successors-in-title, but does not include the Joint Liquidators;
- (f) reference to any statutory provision includes all prior and subsequent enactments, amendments and modifications relating to that provision and any subordinate legislation made under it; and
- (g) in the event that any additional SALIC Group Companies become debtors in Chapter 11 Cases that are commenced in the Bankruptcy Court with the consent of SRGL and SALIC and that are jointly administered with the Chapter 11 Cases of SALIC and SRGL, any reference in this Agreement to a "SALIC Party" or "SALIC Parties" shall be deemed to include reference to such additional SALIC Group Companies.

## 2 **RECITALS**

The above recitals are and shall be incorporated by reference in and made a part of this Agreement.

## **3 UNDERTAKINGS BY SRGL**

Until the Termination Date:

- **3.1** SRGL shall:
  - (a) to the extent not in agreed form prior to the date of this Agreement, use its commercially reasonable efforts to reach agreement on the terms of and cooperate and assist in the preparation of the Share Surrender Documents and New SALIC Shares Issuance Documents on terms consistent in all material respects with this Agreement;
  - (b) within five (5) Business Days after the latest of the date that (w) the RIA Order is entered and effective by its terms, (x) the Cayman Court Power is obtained and effective by its terms, (y) the Bermuda Court's Winding Up Order is made and effective by its terms, and (z) if it is determined in good faith by SALIC, after consultation with Maples and Calder, that regulatory approval of CIMA is required, the date that SALIC receives CIMA's regulatory approval for the Initial Shares Surrender, execute and

deliver the Share Surrender Documents pertaining to the Initial Share Surrender;

- (c) if either Winding Up Order or the receipt of the Cayman Court Power is appealed by any Person (or if any petition for higher court review or motion for reconsideration, amendment, clarification, modification, vacation, stay, rehearing or reargument shall be filed with respect to any of such Orders) subject to rights otherwise arising from this Agreement, at SRGL's own cost and expense, use commercially reasonable efforts to defend against any such appeal, petition for higher court review, or motion for reconsideration, amendment, clarification, modification, vacation, stay, rehearing or reargument; and
- (d) execute and deliver any document or resolution, and give any notice, confirmation, consent, order, instruction or direction, make any application, filing, registration or announcement as may be reasonably necessary to support, facilitate, implement, consummate or otherwise effectuate the Share Surrender and New SALIC Shares Issuance in accordance with the Share Surrender Documents and New SALIC Shares Issuance Documents; provided in each case, that such actions are on terms in all material respects consistent with this Agreement.
- **3.2** Except as permitted by Section 3.3 of this Agreement, SRGL shall not:
  - (a) take, encourage, assist or support (or procure that any other Person takes, encourages, assists or supports) any action which would, or would reasonably be expected to, breach or be inconsistent with the Restructuring Transactions, any Restructuring Document or this Agreement, including, without limitation, dismissing either Winding Up Proceeding; or
  - (b) materially delay, impede, or prevent the implementation or consummation of the Restructuring (to the extent that the restructuring is being conducted in a manner otherwise consistent with this Agreement), including the Restructuring Steps and the Restructuring Transactions.
- **3.3** For the avoidance of doubt, this Article 3 is without prejudice to the ability of SRGL and its Representatives to participate in activities relating to a potential SRGL Debt Financing in accordance with Section 6.1(i) of the Stock Purchase Agreement, and nothing in this Agreement shall preclude SRGL or the Joint Liquidators from commencing and prosecuting any proceedings for SRGL under Chapter 15 of the Bankruptcy Code in furtherance of the Winding Up Proceedings.
- 3.4 Notwithstanding anything to the contrary in this Agreement, if the RIA Order has

not been entered and become effective by its terms by the 90th day after the Petition Date, SRGL may suspend all further performance under this Agreement until such time as the RIA Order has been entered and become effective by its terms.

### 4 UNDERTAKINGS BY SALIC PARTIES

Until the Termination Date:

- **4.1** The SALIC Parties each shall act in good faith and promptly take all actions reasonably necessary to support, facilitate, implement, consummate or otherwise effectuate the Restructuring Transactions in accordance with the terms of this Agreement and the Restructuring Documents, including:
  - (a) reasonably negotiating, preparing, and executing and delivering the Restructuring Documents on terms consistent in all material respects with this Agreement and the SRGL Consent Rights prior to entering into, amending, supplementing or otherwise modifying any Specified Restructuring Document;
  - (b) making commercially reasonable efforts, in consultation with Maples and Calder, to determine in good faith whether CIMA approval is required for the Initial Shares Surrender, and if CIMA approval is required for the Initial Shares Surrender, to obtain CIMA's approval for the Initial Shares Surrender as expeditiously as practicable under the circumstances;
  - (c) making commercially reasonable efforts (i) to provide draft copies of all documents, motions, orders, procedures, agreements and other papers the SALIC Parties intend to file with the Bankruptcy Court to the Joint Liquidators and the JL Advisors no later than three (3) calendar days prior to the date the SALIC Parties intend to file any such document, motion, order, procedure, agreement or other paper (other than the Disclosure Statement, the Chapter 11 Plan and other Restructuring Documents, which shall be provided within a longer reasonable time) and (ii) to consult in advance in good faith with the Joint Liquidators and the JL Advisors regarding the form and substance of any such proposed filing with the Bankruptcy Court and to the extent that any Joint Liquidator or JL Advisor is in attendance at the Auction, prior to designating the Winning Bidder;
  - (d) executing and delivering any document and giving any notice, confirmation, consent, order, instruction or direction, making any application, filing, registration or announcement as may be necessary or

desirable to support, facilitate, implement, consummate or otherwise effectuate the Restructuring Transactions, provided in each case, on terms that are in all material respects consistent with this Agreement;

- (e) preparing for and filing for any legal process or proceedings and any supporting petitions or applications to any Governmental Authority, to support, facilitate, implement consummate or otherwise effectuate the Restructuring Transactions and Restructuring, in each case on terms that are in all material respects consistent with this Agreement; and
- (f) upon the occurrence of the Effective Date of the Chapter 11 Plan and the closing of the Stock Purchase Agreement, effect the New SALIC Shares Issuance in accordance with the New SALIC Shares Issuance Documents.
- **4.2** The SALIC Parties shall use commercially reasonable efforts to obtain, comply with and maintain in full force and effect any necessary authorization required under any Applicable Law to:
  - (a) enable each SALIC Party to perform its obligations under the Restructuring Steps and Restructuring Documents; and
  - (b) if the RIA Order, PSA Order, Bidding Procedures Order, Disclosure Statement Order or Confirmation Order is appealed by any Person (or if any petition for higher court review or motion for reconsideration, amendment, clarification, modification, vacation, stay, rehearing or reargument shall be filed with respect to any of such Order) subject to rights otherwise arising from this Agreement, at the SALIC Parties' own cost and expense, use commercially reasonable efforts to defend against any such appeal, petition for higher court review, or motion for reconsideration, amendment, clarification, modification, vacation, stay, rehearing or reargument.
- **4.3** The SALIC Parties shall:
  - (a) cause their officers and counsel and other advisors to be available for conference calls and meetings at commercially reasonable times and on commercially reasonable notice to consult with and provide updates to the Joint Liquidators and the JL Advisors regarding the Restructuring Transactions; and
  - (b) promptly following receipt thereof and to the extent not publicly available, provide the JL Advisors with copies of all proposals, term sheets or other material correspondences related to the Restructuring Transactions.

- 4.4 Except as provided in Section 4.5 of this Agreement, the SALIC Parties shall not:
  - (a) take, encourage, assist or support (or procure that any other Person takes, encourages, assists or supports) any action which would, or would reasonably be expected to, breach or be inconsistent with the any of the Restructuring Transactions, any Restructuring Document or this Agreement;
  - (b) materially delay or impede, or prevent, the implementation or consummation of any Restructuring Transaction, the Winding Up Proceedings (to the extent that the Winding Up Proceedings are being conducted in a manner otherwise consistent with this Agreement) or any proceeding under chapter 15 of the Bankruptcy Code commenced by SRGL or the Joint Liquidators in furtherance of the Winding Up Proceedings; or
  - (c) following the Initial Shares Surrender, issue any New SALIC Shares other than in accordance with the New SALIC Shares Issuance or with prior written approval of SRGL.
- 4.5 For the avoidance of doubt, this Article 4 is without prejudice to the ability of any officer, director or employee of any SALIC Group Company that is also an officer or director of SRGL to participate in activities concerning a potential SRGL Debt Financing in accordance with Section 6.1(i) of the Stock Purchase Agreement.

### 5 **RESTRUCTURING STEPS**

- 5.1 Each of the Parties will use commercially reasonable efforts to accomplish the following:
  - (a) by no later than one (1) Business Day after execution of this Agreement, SRGL will approve SALIC filing a voluntary Chapter 11 bankruptcy petition in the Bankruptcy Court by the deadline set forth in the Stock Purchase Agreement;
  - (b) by no later than one (1) Business Day after execution of this Agreement, SALIC will approve SHI filing a voluntary Chapter 11 bankruptcy petition in the Bankruptcy Court by the deadline set forth in the Stock Purchase Agreement; and
  - (c) by no later than the date of the applicable deadline set forth in the Stock Purchase Agreement, SALIC and SHI each will file voluntary Chapter

11 bankruptcy petitions in the Bankruptcy Court.

- 5.2 Each SALIC Party will use commercially reasonable efforts:
  - (a) to cause the Bankruptcy Court to enter the RIA Order within 35 days after the Petition Date;
  - (b) to cause the Bankruptcy Court to enter the PSA Order within 35 days after the Petition Date;
  - (c) to cause the Bankruptcy Court to enter the Bidding Procedures Order within 21 days after the Petition Date;
  - (d) to cause the Bankruptcy Court to enter the Disclosure Statement Order within 90 days after the Petition Date; and
  - (e) to cause the Bankruptcy Court to enter the Confirmation Order within 45 days after the Disclosure Statement Order is entered.
- 5.3 SRGL will use commercially reasonable efforts:
  - (a) to request the Bermuda Court to make a Winding Up Order for SRGL, which shall include extending the powers of the Joint Provisional Liquidators, within 21 days after the Petition Date;
  - (b) to pursue its pending application to the Cayman Islands Court for a Winding Up Order appointing John C. McKenna of Finance & Risk Services Ltd., and Eleanor Fisher of Kalo (Cayman) Limited as Joint Official Liquidators within 21 days after the Petition Date; and
  - (c) to apply to the Cayman Islands Court to make the Cayman Court Power so that such Cayman Court Power is fully enforceable by its terms by no later than 35 days after the Petition Date.

For the avoidance of doubt, the time periods for performing any of the actions set forth in this Article 5 may be extended by written agreement of the Parties without further approval of any Governmental Authority, but only to the extent consistent with the Stock Purchase Agreement or with the consent of the Purchaser.

- 5.4 Restructuring Documents. The Parties acknowledge and agree that:
  - (a) To the fullest extent permissible under Applicable Law, the Cayman Court Power, the Winding Up Orders, the RIA Order, the Share Surrender Documents and the New SALIC Shares Issuance Documents

each must be in form and substance reasonably acceptable to each of the Parties and consistent in all material respects with this Agreement and the Stock Purchase Agreement, and otherwise subject to the SRGL Consent Rights;

- (b) All Restructuring Documents, other than those set forth in Section 5.4(a) above must be consistent in all material respects with this Agreement and the Stock Purchase Agreement;
- (c) To the fullest extent permissible under Applicable Law, as set forth more fully in the Plan Term Sheet, the Chapter 11 Plan shall contain mutual releases, limitations of liability, exculpatory provisions and related injunctive relief for the benefit of SRGL, the Joint Liquidators, the SALIC Parties, the Purchaser, the Plan Sponsor, their respective Affiliates and their respective Representatives, which provisions shall be in all respects consistent with this Agreement, the Stock Purchase Agreement and the Plan Sponsorship Agreement; and
- (d) The Specified Restructuring Documents are subject to the SRGL Consent Rights.

## 6 SALIC CLAIMS; DISTRIBUTIONS ON ACCOUNT THEREOF IN WINDING UP PROCEEDINGS

- 6.1 Subject to the completion of the Initial Shares Surrender, SRGL acknowledges and agrees that:
  - (a) SRGL is unconditionally liable, without defense, counterclaim, offset or setoff of any kind to SALIC on account of the SALIC/SRGL Revolver Claims in an amount not less than US\$77,505,389, plus additional interest, fees and other charges that may have accrued or may accrue on or after January 1, 2018;
  - (b) as of the Agreement Date it may be indebted to SALIC or other SALIC Group Companies for certain SALIC Group Services Claims; provided, <u>however</u>, that the Parties agree that any such SALIC Group Services Claims existing as of the Agreement Date do not exceed US\$100,000 in the aggregate; and
  - (c) SALIC, in connection with the Winding Up Proceedings (but not otherwise), shall receive, on account of the SALIC Claims, equal and ratable treatment and distributions with other unsecured claims against SRGL, if any, excluding only any such unsecured claims that are part of a preferred class of claims under applicable Bermuda or Cayman Islands Law.

**6.2** For the avoidance of doubt, this Agreement is without prejudice to the right of SRGL, in accordance with applicable Bermuda and Cayman Islands law governing the Winding Up Proceedings, to reserve such amounts, if any, that SRGL or the Joint Liquidators reasonably determine in good faith to be necessary to fund the reasonable fees and expenses of SRGL, the Joint Liquidators, and the JL Advisors in connection with the Winding Up Proceedings and any Chapter 15 proceedings in respect thereof and any other amounts that SRGL or the Joint Liquidators are permitted to retain under Cayman Islands and Bermuda law, including the Companies (Winding Up) Rules 1982 or the Companies Act 1981; and Cayman Islands law, including the Companies Law (2016 Revision), the Companies Winding Up Rules 2008 (as amended) or Cayman Islands common law.

#### 7 SRGL TruPS CLAIMS AND HOLDINGS; DISTRIBUTIONS ON ACCOUNT THEREOF IN CHAPTER 11 CASES

- 7.1 Subject to the completion of the Initial Shares Surrender, the SALIC Parties acknowledge and agree that as of the Agreement Date: (a) SRGL holds undivided beneficial interests having an aggregate liquidated amount of approximately US\$20,000,000, plus deferred interest as of December 31, 2017 in the amount of US\$5,025,241.00, plus additional interest, fees and other charges that may accrue, in SHST II; (b) SRGL holds undivided beneficial interests having an aggregate liquidated amount of approximately US\$10,000,000, plus deferred interest as of December 31, 2017 in the amount of US\$2,506,585, plus additional interest, fees and other charges that may accrue, in GPIC; (c) SRGL holds undivided beneficial interests having an aggregate liquidated amount of approximately US\$13,000,000, plus deferred interest as of December 31, 2017, in the amount of US\$3,175,458, plus additional interest, fees and other charges that may accrue, in SHST III; (d) SALIC is unconditionally liable under the TruPS Parent Guarantees issued in connection with the SRGL TruPS Holdings in an amount not less than the aggregate of the amounts set forth in clauses (a), (b) and (c) above, without defense, counterclaim, offset or setoff of any kind, which liability is an unsecured, nonpriority Claim against SALIC; (e) SHI is unconditionally liable under the TruPS Sponsor Guarantees issued in connection with the SRGL TruPS Holdings in an amount not less than the aggregate of the amounts set forth in clauses (a), (b) and (c) above, without defense, counterclaim, offset or setoff of any kind, which liability is an unsecured, nonpriority Claim against SHI. The foregoing stipulated amounts are the aggregate amounts that may be asserted on account of the SRGL TruPS Claims, whether by SRGL on account of its SRGL TruPS Holdings or by any trustee or other fiduciary under any TruPS Document acting on behalf or for the benefit of the SRGL TruPS Holdings.
- 7.2 Subject to the completion of the Initial Shares Surrender, the SALIC Parties acknowledge and agree that the treatment afforded the SRGL TruPS Claims under the Chapter 11 Plan shall be consistent with that set forth in the Plan Term Sheet

and not materially disparate from the treatment to be afforded to Other TruPS Claims under the Chapter 11 Plan.

### 8 [RESERVED]

### 9 LIMITATIONS

Notwithstanding any other provision of this Agreement, nothing in this Agreement or the Restructuring Documents shall require any Party or the Joint Liquidators to (whether by action or omission) breach, or procure the breach of (in each case provided that such breach cannot be avoided or removed by taking commercially reasonable steps):

- 9.1 any Applicable Law;
- **9.2** without limiting the scope of Section 9.1 hereof, any obligations of the Joint Liquidators or their respective financial or legal advisors pursuant to, in the case of the Joint Liquidators appointed by the Bermuda Court, the Companies Act 1981 and Companies (Winding Up) Rules 1982 of Bermuda and any Orders of the Bermuda Court and in the case of the Joint Liquidators appointed by the Cayman Islands Court (as applicable), the Companies Law (2016 Revision) and the Companies Winding up Rules 2008 (as amended) of the Cayman Islands and any Orders of the Cayman Islands Court;
- **9.3** any Order, judgment or direction of any relevant court or Governmental Authority; or
- **9.4** any fiduciary duties owed by a Party (or such Party's officers, directors or managers) or Joint Liquidator under Applicable Law.

For the avoidance of doubt, a Party's failure to perform its obligations under this Agreement due to any of the limitations set forth in Sections 9.1 through 9.4 above (the "Limitations") may constitute a material breach of this Agreement entitling the other Parties to terminate this Agreement pursuant to Article 10 hereof; provided, however, that any such material breach shall not entitle other Parties to this Agreement to any damages or other relief under this Agreement.

#### **10 TERMINATION**

#### **10.1 Mutual Voluntary Termination**

This Agreement may be terminated with immediate effect by a signed writing executed by each of the Parties, subject, as to the SALIC Parties, to the prior written consent of the Purchaser.

#### **10.2** Unilateral Termination by Any of the Parties

Any Party may terminate this Agreement without further approval by any Court or other Governmental Authority effective five (5) Business Days after such Party delivers written notice in accordance with Article 13 of this Agreement (such notice, a "**Termination Notice**"), with a copy delivered simultaneously to the Purchaser, of one or more of the following events:

- (a) except through the action or inaction (where such Party has a duty to act) of the Party issuing the Termination Notice, the Bermuda Court or Cayman Islands Court denies on a final basis to make a Winding Up Order for SRGL;
- (b) except through the action or inaction (where such Party has a duty to act) of the Party issuing the Termination Notice, either Winding Up Proceeding is dismissed with prejudice to re-filing a winding up proceeding for SRGL;
- (c) except through the action or inaction (where such Party has a duty to act) of the Party issuing the Termination Notice, the Cayman Islands Court denies on a final basis to grant the Cayman Court Power;
- (d) the board of directors or managers (or comparable governing body) or court-appointed fiduciary (including, in the case of SRGL, the Joint Liquidators), as applicable, of the Party has determined in good faith based upon advice of his, her or its counsel that continued performance under this Agreement would be inconsistent with the Limitations set forth in Article 9 of this Agreement; provided, however, that the SALIC Parties shall not deliver a Termination Notice on this basis without first providing the Purchaser with at least three (3) Business Days' prior written notice of their intent to do so;
- (e) another Party is in material default of such Party's obligations under this Agreement, including but not limited to a material breach of this Agreement resulting from failure to perform due to the Limitations, and, to the extent such default is capable of cure, such default has not been cured within 10 Business Days of the non-breaching Party's provision of written notice of such default in accordance with Article 13 of this Agreement to the defaulting Party; <u>provided</u>, <u>however</u>, that the SALIC Parties shall not deliver a Termination Notice on this basis without the prior written consent of the Purchaser;
- (f) the RIA Order or Confirmation Order ceases to be in full force and effect, is reversed, stayed, vacated or subjected to a stay pending appeal

or is amended, supplemented or otherwise modified in any manner that materially and adversely alters or impairs the rights or obligations under this Agreement of the Party delivering such notice and the RIA Order or Confirmation Order, as applicable, has not been reinstated to its former full force and effectiveness within thirty (30) days after the occurrence of such event; or

(g) the SALIC Parties or the Plan Sponsor terminates the Plan Sponsorship Agreement or the Stock Purchase Agreement pursuant to its terms; <u>provided</u>, <u>however</u>, that termination of the Plan Sponsorship Agreement and/or the Stock Purchase Agreement shall not constitute a basis to deliver a Termination Notice if a Person or Persons other than the Purchaser have been designated the Winning Bidder or Winning Bidders following the Auction and such Winning Bidder or Winning Bidders have executed transaction documents in connection with their Winning Bid or Winning Bids that in the aggregate are on terms and conditions not materially less favorable to the Party that would be delivering the Termination Notice than the terms and conditions contained in the Stock Purchase Agreement and Plan Sponsorship Agreement.

### **10.3 Unilateral Termination by SRGL**

This Agreement may be terminated by SRGL effective, except as otherwise provided below, five (5) Business Days after SRGL's delivery of a Termination Notice to the SALIC Parties, with a copy delivered simultaneously to the Purchaser, each in accordance with Article 13 of this Agreement, if:

- (a) the Bankruptcy Court enters an Order denying the RIA/PSA Motion with prejudice to any future requests by the SALIC Parties in the Chapter 11 Cases to assume this Agreement or otherwise approve its material terms, and in such event the termination of this Agreement shall be automatically effective, without need for SRGL to deliver a Termination Notice, ten (10) Business Days after entry of such Order;
- (b) the closing under the SPA or the Effective Date of the Chapter 11 Plan has not occurred by March 31, 2019, and in such event the termination of this Agreement shall be automatically effective without need for SRGL to deliver a Termination Notice;
- (c) the Bankruptcy Court or any court of competent jurisdiction enters an Order granting relief inconsistent with this Agreement that eliminates or materially and adversely reduces anticipated distributions under the Chapter 11 Plan on account of SRGL TruPS Claims but does not comparably eliminate or materially reduce anticipated distributions under the Chapter 11 Plan on account of the Other TruPS Claims that are

similarly situated; or

(d) the Bankruptcy Court enters or makes an Order (i) disallowing all or any substantial part of the SRGL TruPS Claims (including pursuant to section 502(d) of the Bankruptcy Code), (ii) recharacterizing as equity all or any substantial part of the SRGL TruPS Claims, (iii) subordinating all or any substantial part of the SRGL TruPS Claims to similarly situated Other TruPS Claims, (iv) reducing the SRGL TruPS Claims to an amount less than the stipulated amount of the SRGL TruPS Claims set forth in Section 7.1 of this Agreement or (v) otherwise subject the SRGL TruPS Claims to treatment that is materially inferior to the treatment that similarly situated Other TruPS Claims are receiving in the Chapter 11 Cases.

#### **10.4 Unilateral Termination by SALIC Parties**

This Agreement may be terminated by any SALIC Party effective five (5) Business Days after such SALIC Party's delivery of a Termination Notice to SRGL and the Joint Liquidators, with a copy to the Purchaser, each in accordance with Article 13 of this Agreement, if:

- (a) the Bermuda Court, the Cayman Islands Court or any court of competent jurisdiction enters an Order granting relief inconsistent with this Agreement that materially and adversely reduces the actual or anticipated distributions on account of SALIC/SRGL Revolver Claim relative to the actual or anticipated distributions to *pari passu* Claims in the Winding Up Proceedings; or
- (b) the Bermuda Court, the Cayman Islands Court or any other court of competent jurisdiction makes or enters an Order (i) disallowing all or a substantial part of the SALIC/SRGL Revolver Claim, (ii) recharacterizing as equity all or any substantial part of the SALIC/SRGL Revolver Claim, (iii) subordinating all or any substantial part of the SALIC/SRGL Revolver Claim, (iii) subordinating all or any substantial part of the SALIC/SRGL Revolver Claim, to Claims that otherwise have similar priority of distribution in the Winding Up Proceedings, (iv) reducing the SALIC/SRGL Revolver Claim to an amount less than the stipulated amount of the SALIC/SRGL Revolver Claim set forth in Section 6.1 of this Agreement, or (v) otherwise subjecting the SALIC/SRGL Revolver Claim to treatment that is materially inferior to the treatment that *pari passu* Claims are receiving in the Winding Up Proceedings.

#### **10.5** Effect of Termination

(a) If this Agreement terminates in accordance with this Article 10 (*Termination*), the Parties shall immediately be released from all of their undertakings and other obligations under this Agreement, including any

acknowledgements, stipulations or other agreements pertaining to SRGL TruPS Claims and the SALIC Claims; <u>provided</u>, <u>however</u>, that such termination:

- (i) shall not limit or prejudice the rights of each Party against any other Party which relate to any Party's unexcused breaches of this Agreement; <u>provided</u>, <u>further</u>, that for the avoidance of doubt, no Party or Joint Liquidator shall have or incur any liability to any other Party or Joint Liquidator under this Agreement for any breach of any provision of this Agreement for which such Party or Joint Liquidator's performance was excused solely or primarily pursuant to Article 9 (*Limitations*) of this Agreement;
- (ii) shall not limit the effectiveness of Article 1 (Interpretation), Section 4.6 (Undertakings by the SALIC Parties), Section 4.4(c) (Undertakings by SALIC Parties) (but as to Section 4.4(c), only if termination is pursuant to Section 10.2(e) on account of a SALIC Party's material default of its obligations under this Agreement), Section Termination), 10.5 (Effect of Article 11 (Acknowledgement), Article (Confidentiality 12 and Announcements), Article 13 (Notices), Article 14 (Enforcement by Third Parties), Article 15 (Joint Liquidators' Liability), Article 16 (Governing Law) and Article 17 (General); and
- (iii) shall be without prejudice to any rights or obligations arising under or in connection with the SRGL TruPS Claims, the SALIC Claims, the SRGL Revolver Facility or SALIC/SRGL Revolver Facility Documents; provided, however, that if such termination takes place following the New SALIC Share Issuance, such termination shall not limit the effectiveness of Article 6 (SALIC Claims; Distributions on Account Thereof in Winding Up Proceedings) and Article 7 (SRGL TruPS Claims and Holdings; Distributions on Account Thereof in Chapter 11 Cases).
- (b) In the event that this Agreement is terminated in accordance with its terms after the Initial Shares Surrender has been completed (but before the New SALIC Shares Issuance has occurred) and such termination was for a reason other than pursuant to Section 10.2(e) on account of a breach of this Agreement by SRGL, then within five (5) Business Days after the termination of this Agreement is effective, (x) to the extent practicable, SALIC shall return to SRGL all Ordinary Shares previously surrendered by SRGL to SALIC as part of the Initial Shares Surrender, or (y) to the extent that the Ordinary Shares are no longer capable of being returned to SRGL, to the extent practicable, SALIC shall cause Ordinary Shares with like properties (including their par value) to be issued to SRGL

equal in number to the Ordinary Shares surrendered by SRGL.

#### **10.6** Automatic Stays

To the fullest extent permitted under Applicable Law, the Parties agree that the delivery of a Termination Notice under this Agreement shall not be a violation of any applicable stay that automatically arises in connection with the Winding Up Proceedings or the Chapter 11 Cases (including the automatic stay of section 362 of the Bankruptcy Code) and, to the fullest extent permitted under Applicable Law, the Parties waive the applicability of any such stay to the giving of such Termination Notice.

#### 11 ACKNOWLEDGEMENT

Notwithstanding any other provision herein, this Agreement is not and shall not be deemed to be an offer with respect to any securities or solicitation of votes for the acceptance of a plan of reorganization for purposes of sections 1125 and 1126 of the Bankruptcy Code or otherwise. Any such offer or solicitation will be made only in compliance with all applicable securities laws and provisions of the Bankruptcy Code. The SALIC Parties will not solicit acceptances of the Chapter 11 Plan or any other plan of reorganization in any manner inconsistent with the Bankruptcy Code or applicable bankruptcy law and rules and orders of the Bankruptcy Court.

### 12 CONFIDENTIALITY AND ANNOUNCEMENTS

- **12.1** Subject to Section 12.2, each Party agrees to keep confidential and not disclose any information, including with respect to the Purchaser, that reasonably ought to be considered confidential received by any other Party in connection with this Agreement, the Restructuring Documents and the Restructuring.
- **12.2** Any Party may disclose any information that it is otherwise required to be kept confidential under this Article 12 (*Confidentiality and Announcements*):
  - (a) to the extent that it is already in possession of such information or such information is generally known to the public (in each case, not as a result of a breach of any duty of confidentiality);
  - (b) to its professional advisers, directors, employees and officers; <u>provided</u>, <u>however</u>, that such directors, employees and officers shall have been instructed to treat the information as commercially sensitive, confidential and privileged in accordance with the provisions of this Agreement;
  - (c) to its auditors;
  - (d) to the Plan Sponsor or any other prospective purchaser of some or all of

the stock or assets of the Debtors; <u>provided</u>, <u>however</u>, that the Plan Sponsor or such other prospective purchaser(s) shall execute a form of confidentiality agreement consistent with the terms of this Article 12 (*Confidentiality and Announcements*) as a precondition to accessing such information;

- (e) to Governmental Authorities or other Persons or entities to the extent that disclosure is required by Applicable Law;
- (f) to other Persons or entities to the extent disclosure is required or requested by a relevant regulatory body, tax authority, governmental authority or securities exchange;
- (g) with the prior written approval of the Party that provided the information; and
- (h) to the Joint Liquidators, who may disclose information that it is otherwise required to be kept confidential under this Article 12 (Confidentiality and Announcements) to the extent such disclosure is made in the exercise of the duties of the Joint Liquidators or to the extent such disclosure is required to enable the Joint Liquidators to properly carry out the duties of their office.
- 12.3 No announcement, statement, circulation, or other publicity in connection with this Agreement or its subject matter (unless otherwise permitted by this Agreement) shall be made by or on behalf of the Parties, without the prior written approval of the Parties (such approval not to be unreasonably withheld or delayed).

### **13 NOTICES**

- **13.1** Any confirmation or notice given under this Agreement must be in writing in the English language and may be given in person or by hand, post, courier or email.
- **13.2** The contact details (names, mailing addresses (courier addresses, if different), contact person and title thereof, and email addresses) of the Parties, the Purchaser and the Joint Liquidators for all notices under this Agreement are as follows:

To SRGL:

Prior to the appointment of the Joint Official Liquidators:

Post Office Box HM 2939 Hamilton HM MX, Bermuda Telephone: (441) 298-4375 Facsimile:(441) 295-7576Email:Gregg.Klingenberg@scottishre.comAttention:Gregg Klingenberg

After the appointment of the Joint Official Liquidators:

At the contact information for the Joint Liquidators set forth below.

To the Joint Liquidators:

Eleanor Fisher Kalo (Cayman) Limited PO Box 776 38 Market Street Suite 4208, Canella Court Camana Bay Grand Cayman, KY1-9006 Telephone: (345) 814-4035 Email: efisher@kaloadvisors.com

John C. McKenna Finance & Risk Services Ltd. P.O. Box HM 321 Hamilton HM BX Bermuda Telephone: (441) 292-5526 Email: john.mckenna@frsl.bm

with copies (which shall not constitute notice to the Joint Liquidators for the purposes of this Article 13) to:

Robin J. Mayor Conyers Dill & Pearman Clarendon House 2 Church Street, Hamilton HM 11 Bermuda Telephone: (441) 299-4929 Email: <u>robin.mayor@conyersdill.com</u>

Aristos Galatopoulos Maples and Calder PO Box 309, Ugland House South Church Street KY1-1104 George Town Grand Cayman Cayman Islands Telephone: (345) 814-5241 Email: <u>aristos.galatopoulos@maplesandcalder.com</u>

k

Ned S. Schodek Shearman & Sterling LLP 599 Lexington Avenue New York, NY 10022 Telephone: (212) 848-7052 Email: <u>ned.schodek@shearman.com</u>

To SALIC or SHI:

Post Office Box HM 2939Hamilton HM MX, BermudaTelephone:(441) 298-4375Facsimile:(441) 295-7576Email:Gregg.Klingenberg@scottishre.comAttention:Gregg Klingenberg

with copies (which shall not constitute notice to SALIC for the purposes of this Article 13) to:

Hogan Lovells US LLP 875 Third Avenue New York, New York 10022 Telephone: (212) 918-5560 Facsimile: (212) 918-3100 Email: peter.ivanick@hoganlovells.com Attention: Peter Ivanick, Esq.

Morris, Nichols, Arsht & Tunnell LLP 1201 North Market Street, 16th Floor P.O. Box 1347 Wilmington, Delaware 19899-1347 Telephone: (302) 351-9229 Facsimile: (302) 425-4663 Email: eschwartz@MNAT.com gwerkheiser@MNAT.com Attention: Eric D. Schwartz, Esq. Gregory W. Werkheiser, Esq.

To Purchaser:

HSCM Bermuda Fund Ltd. c/o Hudson Structured Capital Management Ltd. One Dock Street, Suite 404 Stamford, Connecticut 06902 Telephone: (203) 975-4859 Email:ajay.mehra@hscm.comAttention:Ajay Mehra, General Counsel

with a copy (which shall not constitute notice to Purchaser for the purposes of this Article 13 to:

Sidley Austin LLP					
787 Seventh	Avenue				
New York, N	ew York 10019				
Telephone:	(212) 839-7365				
Facsimile:	(212) 839-5599				
Email:	dmanfredi@sidley.com				
	lattanasio@sidley.com				
Attention:	Dennis M. Manfredi, Esq.				
	Lee S. Attanasio, Esq.				

- 13.3 Any Party hereto and any Joint Liquidator may modify its contact details by providing written notice thereof to the other Parties and the Joint Liquidators in accordance with this Article 12, which such changes deemed to be effective three (3) Business Days after the date such notice is provided.
- 13.4 Any notice under this Agreement will be deemed to be given as follows:
  - (a) if in person, at the time of delivery;
  - (b) if by inland post, three Business Days after being deposited in the post, postage prepaid in a correctly addressed envelope;
  - (c) if by international priority courier delivery, three days after delivery to such courier; or
  - (d) if by email or fax, when received in legible form.
- 13.5 For the purpose of this Agreement, an email notice will be treated as being in writing.
- **13.6** Any notice given to SRGL hereunder shall also be given to the Joint Liquidators, whether or not the Joint Liquidators are full powers liquidators at the time of the giving of such notice.

#### **14 ENFORCEMENT BY THIRD PARTIES**

There shall be no third party beneficiaries to this Agreement, except that:

14.1 The Joint Liquidators shall be express third party beneficiaries to this Agreement and entitled, as if they were a party to this Agreement, to enforce and enjoy the benefit of all limitations, exclusions, undertakings and covenants in their favor and in favor of SRGL contained in this Agreement, from which the Joint Liquidators will continue to benefit notwithstanding their discharge from office as Joint Liquidators of SRGL;

- 14.2 Each of the Joint Liquidators' firm, its members, partners, directors, officers, employees, agents, advisers and representatives shall be entitled to rely on, enforce and enjoy the benefit of Article 15 (*Joint Liquidators' Liability*) as if they were a party to this Agreement;
- 14.3 any person who from time to time is appointed as a liquidator of SRGL (including pursuant to an order of the Bermuda Court and/or the Cayman Islands Court) shall be entitled to rely on, enforce and enjoy the benefit of the rights granted to the Joint Liquidators under this Agreement and any such person and his or her firm, its members, partners, directors, officers, employees, agents, advisers and representatives shall be entitled to rely on, enforce and enjoy the benefit of Article 15 (*Joint Liquidators' Liability*) as if they were a party to this Agreement; and
- 14.4 The Plan Sponsor or Purchaser, as applicable, is an intended third-party beneficiary of and may enforce Article 3 (Undertakings by SRGL), Article 4 (Undertakings by SALIC Parties), Article 5 (Restructuring Steps), any provision of Article 10 (Termination) requiring prior notice to or prior written consent of the Purchaser or Plan Sponsor, Article 12 (Confidentiality and Announcements), and this Article 14 (Enforcement by Third Parties).

### 15 JOINT LIQUIDATORS' LIABILITY

Any Restructuring Documents to be executed by the Joint Liquidators (unless expressly otherwise stated therein) or other actions to be taken by the Joint Liquidators in furtherance of the Restructuring Transactions will be done by the Joint Liquidators solely in their capacity as Court-appointed agents of SRGL, acting without personal liability. The Joint Liquidators are not Parties to this Agreement and are third-party beneficiaries to this Agreement as set forth in Article 14 above, and neither the Joint Liquidators, nor any subsequent liquidator (whether appointed pursuant to an order of the Bermuda Court, the Cayman Islands Court or otherwise), nor any of their firm, partners, employees, advisers, representatives or agents shall incur any personal liability whatsoever under this Agreement (or any agreements or arrangements entered into pursuant to or referred to in this Agreement), howsoever arising.

#### 16 GOVERNING LAW

- **16.1** This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware.
- 16.2 The Parties irrevocably agree that prior to the Petition Date only the Bermuda

Court or the Cayman Islands Court shall have jurisdiction to hear and determine any suit, action or proceeding and/or to settle any dispute which may arise out of or in connection with or in any way relate to this Agreement and which involves the rights, claims, duties or liability of SRGL or either Joint Liquidator. Each of the Parties irrevocably waives any objection which it might now or hereafter have to the Cayman Islands Court and the Bermuda Court being nominated as the forum to hear and determine any such suit, action or proceeding and/or any dispute.

- 16.3 The Parties irrevocably agree that, following the Petition Date and except as otherwise provided in Section 16.4, only the Bankruptcy Court shall have jurisdiction to hear and determine any suit, action or proceeding and/or to settle any dispute which may arise out of or in connection with or in any way related to this Agreement. Except as otherwise provided in Section 16.4, each of the Parties irrevocably waives any objection which it might now or hereafter have to the Bankruptcy Court being nominated as the forum to hear and determine any such suit, action or proceeding and/or any dispute.
- Sections 16.2 and 16.3 shall in no way restrict or otherwise affect the supervisory 16.4 jurisdiction of the Bankruptcy Court, the Bermuda Court or the Cayman Islands Court in relation to the respective bankruptcy or liquidation proceedings before those Governmental Authorities and each of those Governmental Authorities shall, among other things, retain exclusive jurisdiction to interpret their own orders. It is expressly understood and agreed by the Parties that the Joint Liquidators are not Parties to this Agreement and do not consent to and shall not be subject to the jurisdiction of the Bankruptcy Court and that only the Bermuda Court or the Cayman Islands Court shall have jurisdiction over any suits, actions, or proceedings naming the Joint Liquidators as parties and/or to settle any dispute which may arise out of or in connection with or in any way related to: (a) the exercise of SRGL of its rights under Section 10.2(a) or 10.2(c) to terminate this Agreement; (b) SRGL's failure to perform its obligations under this Agreement pursuant to Article 9; and (c) any interpretation of Section 12.2(h), Section 14.1, Section 14.2, Section 14.3 and Article 15 (Joint Liquidators' Liability). Each of the Parties irrevocably waives any objection which it might now or hereafter have to the Cayman Islands Court or the Bermuda Court being nominated as the forum to hear and determine any such suit, action or proceeding and/or any such dispute. Further, it is expressly understood and agreed by the Parties that the Share Surrender Documents and the New SALIC Shares Issuance Documents, shall be governed by the laws of the Cayman Islands and subject to the exclusive jurisdiction of the Cayman Islands Court.
- 16.5 The Bermuda Court, and if Cayman Islands Joint Official Liquidators are appointed by the Cayman Islands Court or an order winding up SRGL is made in the Cayman Islands, the Cayman Islands Court, shall have exclusive jurisdiction of any and all matters relating to the liquidation of SRGL and the powers and actions of the Joint Liquidators in those respective jurisdictions.

#### 17 GENERAL

- 17.1 This Agreement may not be modified, amended or supplemented except in a writing executed by each of the Parties and the written consent of the Joint Liquidators.
- **17.2** This Agreement is intended to bind and inure to the benefit of the Parties and their respective successors, assigns and transferees.
- 17.3 Failure by the Parties to require performance of any term or condition of this Agreement shall not prevent the subsequent enforcement of such term or condition nor shall such failure be deemed to be a waiver of any subsequent breach of this Agreement, or any right or remedy granted by this Agreement or by the general law in respect of such breach.
- 17.4 The terms of this Agreement represent the entire agreement between the Parties relating to the subject matter of this Agreement and this Agreement supersedes any previous arrangement between the Parties in relation to the matters dealt with in this Agreement.
- 17.5 This Agreement may be executed in hard copy, by original fax or by pdf copy in any number of counterparts, and by each of the Parties on separate counterparts, each of which so executed and delivered will be an original, but all counterparts will together constitute one and the same Agreement.

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

#### **SCOTTISH RE GROUP LIMITED**

By Name: Grogg Klingenberg

Title: Chief Executive Officer

per esta

SCOTTISH ANNUITY & LIFE INSURANCE COMPANY (CAYMAN) LTD.

By: Name: Gregg Klingenberg Title: Chief Executive Officer.

SCOTTISH HOLDINGS, INC.

By: Thomas & Kelly

Name: Thomas J. Keller Title: EVP, Chief Financial Officer

# <u>Exhibit A</u>

Plan Sponsorship Agreement

[Intentionally Omitted]

# <u>Exhibit B</u>

**Stock Purchase Agreement** 

[Intentionally Omitted]

# <u>EXHIBIT B</u>

BLACKLINE

#### IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re

SCOTTISH HOLDINGS, INC., et al.,

Debtors.<sup>1</sup>

Chapter 11

Case No. 18-10160 (LSS)

Jointly Administered

### **SECOND THIRD** AMENDED JOINT CHAPTER 11 PLAN OF REORGANIZATION OF SCOTTISH HOLDINGS, INC., AND SCOTTISH ANNUITY & LIFE INSURANCE COMPANY (CAYMAN) LTD.

#### HOGAN LOVELLS US LLP

Peter A. Ivanick Lynn W. Holbert John D. Beck 875 Third Avenue New York, NY 10022

Telephone: (212) 918-3000 Facsimile: (212) 918-3100 peter.ivanick@hoganlovells.com lynn.holbert@hoganlovells.com john.beck@hoganlovells.com

#### **MORRIS, NICHOLS, ARSHT & TUNNELL LLP**

Eric D. Schwartz (No. 3134) Gregory W. Werkheiser (No. 3553) Matthew B. Harvey (No. 5186) Paige N. Topper (No. 6470) 1201 N. Market St., 16th Floor P.O. Box 1347 Wilmington, DE 19899-1347 Telephone: (302) 658-9200 Facsimile: (302) 658-3989 eschwartz@mnat.com gwerkheiser@mnat.com ptopper@mnat.com

#### Counsel for Debtors and Debtors in Possession

Dated: June 28, August 9, 2018 Wilmington, Delaware

<sup>&</sup>lt;sup>1</sup>The Debtors, along with the last four digits of their federal tax identification numbers, are as follows: Scottish Holdings, Inc. (4408) and Scottish Annuity & Life Insurance Company (Cayman) Ltd. (3285). The Debtors' mailing address for purposes of these Chapter 11 Cases is 14120 Ballantyne Corporate Place, Suite 300, Charlotte, NC 28277.

# TABLE OF CONTENTS

I.	SUM	MARY	OF THE PLAN	1
II.	DEFI	NITIO	NS AND INTERPRETATION	2
III.	CLA	SSIFIC	ATION OF CLAIMS AND INTERESTS	2
	3.1.	Intro	duction	2
	3.2.	Uncl	assified Claims	3
		(a)	Administrative Claims	3
		(b)	Priority Tax Claims	3
	3.3.	Class	sification of Claims and Interests	3
IV.	TREA	ATMEN	NT OF CLAIMS AND INTERESTS	3
	4.1.	Uncl	assified Claims	3
		(a)	Administrative Claims	3
		(b)	Priority Tax Claims	4
		(c)	Other Provisions Governing Administrative Claims	
		(d)	Indenture Trustee Fees	5
	4.2.	Unin	npaired Classes of Claims and Interests	
		(a)	Class 1 – Secured Claims	7
		(b)	Class 2 – Priority Non-Tax Claims	<u></u> 7 <u>8</u>
		(c)	Class 3 – Intercompany Claims	
		(d)	Class 10 – SALIC Existing Equity Interests	
	4.3.	Impa	aired Voting Classes of Claims	9
		(a)	Class 4 – SHI TruPS Claims	
		(b)	Class 5 – SHI General Unsecured Claims	
		(c)	Class 6 – SALIC TruPS Claims & SFL Claims	11

# Page

		(d)	Class 7 – SALIC General Unsecured Claims	16
	4.4.	Impa	ired Non-Voting Classes of Claims and Interests	
		(a)	Class 8 – Subordinated Claims	
		(b)	Class 9 – SHI Existing Equity Interests	17
V.	ACC	EPTAN	ICE REQUIREMENTS	17
	5.1.	Impa	ired Classes of Claims Entitled to Vote	
	5.2.	Acce	ptance by an Impaired Class	
	5.3.	Presu	amed Acceptance by Unimpaired Classes	18
	5.4.		amed Rejection by Impaired Classes Not Receiving Any ibution under the Plan	18
	5.5.	Conf	irmation Pursuant to Section 1129(b) of the Bankruptcy Code	18
	5.6.	Elim	ination of Vacant Classes	18
	5.7.	Presumed Acceptance by Voting Classes in Which No Votes Are Cast		
	5.8.	Cons	olidation of Classes	19
	5.9.	Separ	rate Classes of Secured Claims	19
VI.	MEANS FOR IMPLEMENTATION OF THE PLAN			
	6.1.	Plan	Transactions	
		(a)	Stock Purchase Agreement Closing	19
		(b)	Funding, Allocation and Use of Plan Funding Payment; Allocation and Use of Debtors' Unrestricted Cash	19
		(c)	Funding of Recapitalization Funding Payment	20
		(d)	Cancellation of SHI Existing Equity Interests; Issuance of New SHI Equity	<del>21<u>20</u></del>

# Case 18-10160-LSS Doc 437-2 Filed 08/09/18 Page 5 of 105

# TABLE OF CONTENTS (Continued)

	(e)	Final Share Surrender	21	
	(f)	New Equity Issuance and Distribution	21	
	(g)	Distribution of Offered New Equity	22	
6.2.	Vestin	ng of Estate Property	22	
6.3.	The D	Distribution Trust	23	
	(a)	Execution of Distribution Trust Agreement	23	
	(b)	Purpose of the Distribution Trust	23	
	(c)	Distribution Trust Assets and Other Property	23	
	(d)	The Administration of the Distribution Trust and Authority and Powers of the Distribution Trustee	23	
	(e)	Mutual Cooperation	24	
	(f)	Establishment and Funding of Distribution Trust Reserves	24	
	(g)	Establishment and Funding of the Indenture Trustee Fee Reserve. 25Intentionally	Omitted.	25
	(h)	Cash Investments	<u>2625</u>	
	(i)	Distributions to Holders of Claims and Distribution Trust Beneficiaries	2625	
	(j)	Federal Income Tax Treatment of Distribution Trust	<del>27<u>26</u></del>	
	(k)	Tax Reporting	<u>2827</u>	
	(1)	Dissolution	<del>29<u>28</u></del>	
6.4.	The R	eorganized Debtors	<del>30<u>29</u></del>	
	(a)	Continued Corporate Existence	<u>3029</u>	

# <u>Page</u>

		(c) Corporate Action	<u>3130</u>
		(d) Effectuating Documents; Further Transactions	<u>3130</u>
	6.5.	Retained Causes of Action	<u>3130</u>
	6.6.	The Closing	<u>3231</u>
	6.7.	Cancellation of Agreements, Securities and Other Documents Relating to TruPS Transactions & SFL Note	<u>-3231</u>
	6.8.	Comprehensive Settlement of Claims and Controversies	<u>33332</u>
VII.	CLAI	MS RESOLUTION & DISTRIBUTIONS	33
	7.1.	Right to Object to Claims	33
	7.2.	Deadline for Objecting to Claims	34
	7.3.	Deadline for Responding to Claim Objections	
	7.4.	Right to Request Estimation of Claims	34
	7.5.	Distribution Procedures Regarding Allowed Claims	
		(a) In General	34
		(b) Distributions on Allowed Claims Only	
		(c) Method of Distributions	35
		(d) Undeliverable Distributions	36
		(e) Unclaimed Distributions	36
		(f) Taxes; Withholding	
		(g) Special Distribution Provisions for TruPS Claims	37
		(h) Additional Procedures Regarding Distributions from the Distribution Trust	38
		(i) Allocation of Distributions between Principal and Interest	38

|--|

	7.6.	Imple	ementation of New Equity and Cash Elections	38
		(a)	Making of New Equity or Cash Election by New Equity Eligible Beneficial Holders	<del>39<u>38</u></del>
		(b)	Implementation of New Equity or Cash Election for SFL Note Claim	39
	7.7.	Claim	ns Paid or Payable by Third Parties	<u>4039</u>
		(a)	Claims Paid by Third Parties	40 <u>39</u>
		(b)	Claims Payable by Third Parties	40
	7.8.	Foreig	gn Currency Exchange Rate	40
	7.9.	Setof	f	40
	7.10.		linimis Distributions	
	7.11.		ional Shares	
	7.12.	No In	iterest	41
VIII.	EXEC	CUTOR	Y CONTRACTS AND UNEXPIRED LEASES	41
	8.1.	Assur	mption of Executory Contracts	41
	8.2.	Rejec	tion of Executory Contracts	
	8.3.		mination of Assumption Disputes and Deemed Consent	
	8.4.	Effec	t of Assumption of Contracts and Leases	43
	8.5.		ent of Cure Amounts	44 <u>43</u>
		(a)	No Admission of Liability	44
		(b)	Reservation of Rights	
		(c)	Rejection Claim Bar Date	
		(d)	Continuing Obligations Owed to the Debtors	4 <u>5</u> 44
		· · /		

# Page

		(e) Postpetition Contracts	45
IX.		DITIONS PRECEDENT TO CONFIRMATION OF THE PLAN OCCURRENCE OF THE EFFECTIVE DATE	45
	9.1.	Conditions to Confirmation	45
	9.2.	Conditions to Effectiveness	45
X.		LEMENT, DISCHARGE, RELEASE, INJUNCTION AND	
	10.1.	Compromise and Settlement of Claims, Interests, and Controversies	46
	10.2.		
	10.3.	Releases by Holders of Claims and Interests	<u>4847</u>
	10.4.	Discharge and Discharge Injunction	
		(a) Discharge of Claims	
		(b) Discharge Injunction	<u>5049</u>
	10.5.	Exculpation	50
	10.6.	Post-Effective Date Indemnification	<u>5150</u>
XI.	RETE	ENTION OF JURISDICTION	<u>5251</u>
	11.1.	Retention of Jurisdiction	<u>5251</u>
	11.2.	Limitation on Jurisdiction	<u>5453</u>
XII.		ELLANEOUS PROVISIONS	54
	12.1.	Legally Binding Effect	
	12.2.	Exemption from Transfer Taxes	<u>5554</u>
	12.3.	Securities Exemption	55

12.4.	Defects, Omissions and Amendments of the Plan	55
12.5.	Due Authorization by Creditors	<del>56<u>55</u></del>
12.6.	Filing of Additional Documentation	56
12.7.	Dissolution of the Official Committee	56
12.8.	Governing Law	56
12.9.	Successors and Assigns	<del>57<u>56</u></del>
12.10.	Transfer of Claims	<del>57<u>56</u></del>
12.11.	Notices	57
12.12.	U.S. Trustee Fees and Reports	59
12.13.	Implementation	60
12.14.	No Admissions	60
12.15.	SRGL Consent Rights Reserved	60
12.16.	Substantial Consummation	60
12.17.	Final Decree	60
#### Case 18-10160-LSS Doc 437-2 Filed 08/09/18 Page 10 of 105

Scottish Holdings, Inc. ("<u>SHI</u>") and Scottish Annuity & Life Insurance Company (Cayman) Ltd. ("<u>SALIC</u>"), debtors and debtors in possession (the "<u>Debtors</u>") in these Chapter 11 Cases, jointly propose this Plan. Reference is made to the Disclosure Statement in support of the Plan for a discussion of the Debtors' history, business, property and results of operations, and for a summary of the Plan and certain related matters.

All Creditors and other parties in interest are encouraged to read the Plan and the Disclosure Statement in their entirety, as well as the Stock Purchase Agreement, the Restructuring Implementation Agreement and the RIA Order, the terms of which are incorporated into the Plan by reference and form integral parts of the Plan. No materials, other than the Disclosure Statement and any exhibits and schedules thereto or referenced therein, have been approved by the Debtors or the Bankruptcy Court for use in soliciting acceptances or rejections of the Plan.

#### I. SUMMARY OF THE PLAN

An overview of the Plan is set forth in the Disclosure Statement. Generally, the Plan provides for, among other things, the following: (1) the reorganization and recapitalization of the Debtors and certain of their non-debtor Affiliates through a new money contribution of \$12,500,000 by the Purchaser in the form of the Recapitalization Funding Payment; (2) the funding of distributions to the Debtors' Creditors through an additional new money contribution of \$21,500,000 by the Purchaser in the form of the Plan Funding Payment subject to reduction by the amount of the TruPS Returned Cash; (3) in exchange for the foregoing payments and other consideration, the issuance or assignment to the Purchaser of one hundred percent (100%) of the New Equity, subject to downward adjustment to no less than seventy percent (70%), to the extent that eligible unsecured creditors elect to receive their pro rata share of up to thirty percent (30%) of the New Equity, in lieu of a cash distribution under the Plan; (4) the assumption by the Reorganized Debtors of the SALIC Net Worth Maintenance Agreements and all or substantially all reinsurance treaties for which SALIC acts as reinsurer or retrocessionaire; (5) the creation of the Distribution Trust (a) for payment of all Secured Claims, Administrative Claims, and Priority Claims to the extent Allowed and not paid or otherwise satisfied prior to the Effective Date, and (b) for the benefit of Holders of SHI TruPS Claims, SHI General Unsecured Claims, SALIC TruPS Claims and SALIC General Unsecured Claims, all to the extent Allowed; and (6) the funding of the Distribution Trust with the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves.

Pursuant to sections 105(a), 363 and 1123 of the Bankruptcy Code and Bankruptcy Rule 9019, and in consideration for the distributions Distributions and other benefits provided pursuant to the Plan, the provisions of the Plan shall constitute a good faith compromise of all Claims, Interests and controversies relating to the contractual, legal, and subordination rights that a Holder of a Claim or Interest may have with respect to any Allowed Claim or Interest, or any distribution Distribution to be made on account of such Allowed Claim or Interest, as well as the allocation of the Plan Funding Payment among the Debtors' Estates and Creditors. In addition, the Plan contains certain, release, injunction and exculpation provisions all as set forth in Article X of the Plan.

## II. DEFINITIONS AND INTERPRETATION

2.1 All capitalized terms used but not defined elsewhere in the Plan have the meanings assigned to them in the Glossary of Defined Terms attached as **Exhibit A** to the Plan. Any capitalized term used and not otherwise defined by the Plan has the meaning ascribed to that term in the Bankruptcy Code and/or Bankruptcy Rules.

2.2 For purposes of the Plan, any reference in the Plan to an existing document or exhibit Filed or to be Filed means that document or exhibit as it may have been or may be amended, supplemented, or otherwise modified.

2.3 The words "herein," "hereof" and "hereunder" and other words of similar import refer to the Plan as a whole and not to any particular section, subsection or clause contained in the Plan, unless the context requires otherwise. Whenever from the context it appears appropriate, each term stated in either the singular or the plural includes the singular and the plural, and pronouns stated in the masculine, feminine or neuter gender include the masculine, feminine and the neuter.

2.4 Captions and headings to articles, sections and exhibits are inserted for convenience of reference only and are not intended to be part of or to affect the interpretation of the Plan.

2.5 The rules of construction set forth in section 102 of the Bankruptcy Code shall apply.

2.6 In computing any period of time prescribed or allowed by the Plan, the provisions of Bankruptcy Rule 9006(a) shall apply.

# III. CLASSIFICATION OF CLAIMS AND INTERESTS

## 3.1. Introduction

All Claims and Interests, except Administrative Claims and Priority Tax Claims, are placed in the Classes set forth below. In accordance with section 1123(a)(1) of the Bankruptcy Code, Administrative Claims and Priority Tax Claims have not been classified, and the treatment of such unclassified Claims is set forth below in <u>Section 4.1</u> of the Plan.

A Claim or Interest is placed in a particular Class only to the extent that the Claim or Interest falls within the description of that Class, and is classified in other Classes to the extent that any portion of the Claim or Interest falls within the description of such other Classes. A Claim is also placed in a particular Class for the purpose of receiving Distributions pursuant to the Plan only to the extent that such Claim is an Allowed Claim in that Class and such Claim has not been paid, released, or otherwise settled prior to the Effective Date.

## **3.2.** Unclassified Claims

- (a) Administrative Claims
- (b) **Priority Tax Claims**

## **3.3.** Classification of Claims and Interests

The following table designates the Classes of Claims against and Interests in the Debtors and specifies which Classes are: (a) Impaired and Unimpaired under the Plan; (b) entitled to vote to accept or reject the Plan in accordance with section 1126 of the Bankruptcy Code; and (c) deemed to accept or reject the Plan:

Class	Type of Claim or Interest	Impairment	<b>Entitled to Vote</b>
Class 1	Secured Claims	Unimpaired	No (deemed to accept)
Class 2	Priority Non-Tax Claims	Unimpaired	No (deemed to accept)
Class 3	Intercompany Claims	Unimpaired	No (deemed to accept)
Class 4	SHI TruPS Claims	Impaired	Yes
Class 5	SHI General Unsecured Claims	Impaired	Yes
Class 6	SALIC TruPS Claims & SFL	Impaired	Yes
	Claims		
Class 7	SALIC General Unsecured Claims	Impaired	Yes
Class 8	Subordinated Claims	Impaired	No (deemed to reject)
Class 9	SHI Existing Equity Interests	Impaired	No (deemed to reject)
Class 10	SALIC Existing Equity Interests	Unimpaired	No (deemed to accept)

# IV. TREATMENT OF CLAIMS AND INTERESTS

# 4.1. Unclassified Claims

## (a) Administrative Claims

Except to the extent that an Allowed Administrative Claim has been satisfied prior to the Effective Date, and except as otherwise provided for herein (including <u>Section</u> <u>4.1(c)(ii)</u> with respect to Professional Fee Claims), each Holder of an Allowed Administrative Claim shall be entitled to receive in full, final and complete settlement, release, and discharge of such Claim, either (i) to the extent such Administrative Claim is Allowed as of the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Administrative Claim as a Closing Date Plan Distribution on the Effective Date, or as soon as reasonably practicable thereafter, or (ii) to the extent such Administrative Claim is Allowed after the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Administrative Claim from the Distribution Trust, in an amount equal to the unpaid portion of such claim, at such time as such Administrative Claim is Allowed, or as soon as reasonably practicable thereafter.

#### (b) **Priority Tax Claims**

Except to the extent that an Allowed Priority Tax Claim has been satisfied prior to the Effective Date, each Holder of an Allowed Priority Tax Claim shall be entitled to receive, in full, final and complete settlement, release, and discharge of such Claim, at the election of the Debtors or the Distribution Trustee, one of the following treatments: (i) to the extent such Priority Tax Claim is Allowed as of the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Priority Tax Claim as a Closing Date Plan Distribution on the Effective Date, or as soon as reasonably practicable thereafter; (ii) to the extent such Priority Tax Claim is Allowed after the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Priority Tax Claim is Allowed after the Distribution Trust, in an amount equal to the unpaid portion of such Allowed thereafter; or (iii) or such other treatment or payment from the Distribution Trust as permitted under section 1129(a)(9)(C) of the Bankruptcy Code.

#### (c) Other Provisions Governing Administrative Claims

i. General Provisions. Except as otherwise provided in this Article IV, requests for payment of Administrative Claims must be included within an application (setting forth the amount of, and basis for, such Administrative Claims, together with documentary evidence) and Filed and served on respective counsel for the Debtors, the Reorganized Debtors, the Purchaser, and the Distribution Trustee no later than the applicable Holders of Administrative Claims (including, without Administrative Claims Bar Date. limitation, Holders of any Claims for federal, state or local taxes) that are required to File a request for payment of such Claims and that do not File such requests by the applicable Administrative Claims Bar Date shall be forever barred from asserting such Claims against the Debtors, Reorganized Debtors, the Purchaser, the Distribution Trust or any of their respective property. Requests for payments of Administrative Claims included within a Proof of Claim are of no force and effect, and are deemed disallowed in their entirety as of the Effective Date, and shall be satisfied only to the extent such Administrative Claim is subsequently Filed in a timely fashion as provided by this subsection and subsequently becomes an Allowed Claim.

ii. *Professionals.* All Professionals or other entitiesEntities requesting compensation or reimbursement of expenses pursuant to sections 327, 328, 330, 331, 503(b) and 1103 of the Bankruptcy Code for services rendered before the Effective Date (including, without limitation, any compensation requested by any Professional or any other Entity for making a substantial contribution in the Chapter 11 Cases) shall File an application for final allowance of compensation and reimbursement of expenses no later than thirty (30) days after the Effective Date and simultaneously serve such application on counsel for the following entities: the Reorganized Debtors, the Purchaser, the Official Committee, the Distribution Trustee, and the U.S. Trustee.

Objections, if any, to a Professional's application for compensation or reimbursement of expenses must be <u>filedFiled</u> no later than twenty-one (21) days after the date the application is <u>filedFiled</u>, and simultaneously served on the applicant (and its counsel, if any)

#### Case 18-10160-LSS Doc 437-2 Filed 08/09/18 Page 14 of 105

and counsel for the following <u>entitiesEntities</u>: the Reorganized Debtors, the Purchaser, the Official Committee, the Distribution Trustee, and the U.S. Trustee. If no objections are received, the Bankruptcy Court may enter a final order approving the applications and authorizing final Allowance and payment of compensation and reimbursement of expenses without a hearing. If any objection cannot be resolved consensually, the Bankruptcy Court will hold a hearing on the affected application(s).

The amount of compensation and reimbursement of expenses Allowed by the Bankruptcy Court (less (i) amounts previously received by the Professional in respect of interim compensation and (ii) any unapplied retainer or advance held by the Professional) shall be paid by the Distribution Trustee from the Professional Fee Reserve.

Any professional fees and reimbursements or expenses incurred by the Distribution Trust subsequent to the Effective Date may be paid in accordance with the Distribution Trust Agreement. Any professional fees and reimbursements or expenses incurred by the Reorganized Debtors subsequent to the Effective Date may be paid without further order of, or application to, the Bankruptcy Court.

#### (d) Indenture Trustee Fees

i. Indenture Trustee Fees Incurred through the Confirmation Date. On the Effective Date, the Debtors shall pay all undisputed Indenture Trustee Fees of the TruPS-Indenture Trustees incurred through the Confirmation Date as set forth in section 6.1(b)(1). For a TruPS Indenture Trustee to be eligible to have its Indenture Trustee Fees incurred through the Confirmation Date paid on the Effective Date or as soon as reasonably practicable after the Effective Date, on or before the thirtieth (30th) day after the Confirmation Date, such TruPS-Indenture Trustee must serve Effective Date. On the DT Initial Distribution Date, prior to making the Distributions under Section 4.3(c) of Cash or Offered New Equity (as applicable) on account of each Allowed TruPS Claim, subject to the review of the IT Fee Notice Parties with invoices setting forth in reasonable detail (but subject to appropriate redactions to preserve confidentialityor any applicable privileges or protections) the Indenture Trustee Fees for which such TruPS-Indenture Trustee seeks payment. If none of the IT Fee Notice Parties has served the applicable TruPS Indenture Trustee with a written objection to the TruPS Indenture Trustee Fees set forth in such invoices within ten (10) Business Days after the date of service of such invoices, then the subject Indenture Trustee Fees shall be paid to the applicable TruPS Indenture Trustee on or assoon as reasonably practicable after the Effective Date as set forth in Section 6.1(b)(1), without the need for application to, or approval by, any court. Each Indenture Trustee will not assert its-Charging Lien to the extent that it receives payment of its Indenture Trustee Fees.as set forth in Section 4.1(d)(iii), the Distribution Trustee shall, in connection with such TruPS Indenture Trustee's Charging Lien, first (a) pay to each TruPS Indenture Trustee in Cash in full the undisputed Indenture Trustee Fees incurred by such TruPS Indenture Trustee in connection with the Chapter 11 Cases through the Effective Date, and then (b) reserve for each TruPS Indenture Trustee in Cash in full any disputed Indenture Trustee Fees incurred by such TruPS Indenture Trustee through the Effective Date. The payment to each TruPS Indenture Trustee shall, as set forth more fully in Section 4.1(d)(ii), come from the DT Initial Distribution made to Holders of

Allowed TruPS Claims related to such trustee, such that:

ii. <u>Indenture Trustee Fees Incurred During the Period from the</u> Confirmation Date through the Effective Date.

A. Prior to the Effective Date, the Debtors, in consultation with the Purchaser, the Official Committee and each of the TruPS Indenture Trustees, will make a good faith estimate of the Indenture Trustee Fees incurred and expected to be incurred during the period from the Confirmation Date through the Effective Date. On the Effective Date, the Indenture Trustee Fee Reserve shall be funded with such estimated amount in accordance with section 6.3(g) of the Plan. With respect to the SHST I TruPS Indenture Trustee, the payment shall come from the DT Initial Distribution made on account of the Allowed SHST I TruPS Claims;

For a TruPS Indenture Trustee to be eligible to have B. its Indenture Trustee Fees incurred during the period commencing from the Confirmation Date through the Effective Date paid from the Indenture Trustee Fee Reserve (and if such Indenture-Trustee Fee Reserve proves to be inadequate, the Available Plan Distribution Funding Amount), on or before the thirtieth (30th) day after the Effective Date, such TruPS Indenture Trustee must serve the Distribution Trustee and the Purchaser with invoices setting forth in reasonable detail (but subject to appropriate redactions to preserve confidentiality or any applicable privileges or protections) the Indenture Trustee Fees for which such TruPS Indenture Trustee seeks payment. If the Distribution Trustee or the Purchaser has not served the applicable TruPS Indenture Trustee with a written objection to the TruPS Indenture Trustee Fees set forth in such invoices within ten (10) Business Days after the date of service of such invoices, then the subject Indenture Trustee Fees shall be paid from the Indenture Trustee Fee Reserve (or the Available Plan Distribution Funding Amount if the Indenture Trustee Fee Reserve is inadequate) to the applicable TruPS Indenture Trustee within ten (10) Business Days after the expiration of such objection period, without the need for application to, or approval by, any court. With respect to the SHST II TruPS Indenture Trustee, the payment shall come from the DT Initial Distribution made on account of the Allowed SHST II TruPS Claims;

<u>C.</u> <u>With respect to the GPIC TruPS Indenture Trustee,</u> the payment shall come from the DT Initial Distribution made on account of the Allowed GPIC <u>TruPS Claims;</u>

<u>D.</u> <u>With respect to the SHST III TruPS Indenture</u> <u>Trustee, the payment shall come from the DT Initial Distribution made on account of the</u> <u>Allowed SHST III TruPS Claims; and</u>

<u>E.</u> <u>With respect to the SFLST I TruPS Indenture</u> <u>Trustee, the payment shall come from the DT Initial Distribution made on account of the</u> <u>Allowed SFLST I TruPS Claims.</u>

ii. <u>The DT Initial Distribution for each Allowed TruPS Claim shall</u>

each be used: (1) first, to satisfy the undisputed Indenture Trustee Fees for the respective TruPS Debenture in Cash; (2) second, to fund in Cash the Post-Effective Date Indenture Trustee Fee Reserve on account of disputed Indenture Trustee Fees for the respective TruPS Debenture incurred before the Effective Date; (3) third, to fund \$50,000 to thePost-Effective Date Indenture Trustee Fee Reserve for the TruPS Indenture Trustee Fees that may be incurred post-Effective Date with respect to the SHST I TruPS, the SHST III TruPS and the SFLST I TruPS, which amount shall be an aggregate amount for all such post-Effective Date Indenture Trustee Fees incurred before the Effective Date, and (4) fourth, to distribute the remaining DT Initial Distribution to be made to Allowed TruPS Claims in Cash or Offered New Equity based upon the election of the respective Holders of Allowed TruPS Claims; provided, that, the amount of Offered New Equity to be distributed to Holders of Allowed TruPS Claims that make the New Equity Election will be adjusted downward to account for such Allowed TruPS Claims' respective share of Indenture Trustee Fees payable and reserved.

iii. Indenture Trustee Fees Incurred Following the Effective Date. Prior to the Effective Date, the Debtors, in consultation with the Purchaser, the Official Committee and each of the TruPS Indenture Trustees, will make a good faith estimate of the Indenture Trustee Fees expected to be incurred following the Effective Date and the Indenture Trustee Fee Reserve shall be funded with such estimated amount in accordance with section 6.3(g) of the Plan. NoNotwithstanding the foregoing, the IT Fee Notice Parties shall have the right to review such Indenture Trustee Fees as set forth in Section 4.1(d)(iv). The Charging Lien of each Indenture Trustee will be satisfied to the extent that such trustee receives payment of its Indenture Trustee Fees; provided, however, that nothing herein will prevent the a TruPS Indenture Trustee from asserting its respective Charging Lien for unpaid Indenture Trustee Fees owed to it, including the SFL I TruPS Indenture Trustee asserting its Charging Lien against any distributions it receives from SFL. Other than as expressly set forth herein, no TruPS Indenture Trustee shall be eligible to receive payment from the Distribution Trust or SRGL or to maintain a Charging Lien, for Indenture Trustee Fees incurred after the Effective Date for fees or expenses relating to the SHST II TruPS, the SHST II TruPS Documents, the GPIC TruPS, the GPIC TruPS Documents or anyservices related to Distributions made on account of Claims arising from the SHST II TruPS, the SHST II Debentures, the GPIC TruPS, the GPIC Debentures or any other TruPS Document related to the foregoing TruPS transactions. on account of Allowed TruPS Claims. For the avoidance of doubt, the Distribution Trustee shall be under no obligation to satisfy or reserve any amount on account of post-Effective Date Indenture Trustee Fees that may be incurred by the TruPS Indenture Trustees for the SHST II Debentures or the GPIC Debentures.

iv. Indenture Trustee Fee Cap. The aggregate amount of Indenture Trustee Fees recoverable from the Debtors and the Distribution Trust by the TruPS Indenture Trustees shall not exceed the Indenture Trustee Fee Cap. The Indenture Trustee Fee Cap may be increased upon the consent of the Debtors, the Official Committee and the Purchaser at any time prior to the Effective Date, and upon the consent of the Distribution Trustee and the Purchaser on or after the Effective Date. <u>Review of Indenture Trustee Fees and Disputes Regarding Indenture</u> <u>Trustee Fees.</u> <u>A.</u> (x) Five (5) days after the conclusion of the Confirmation Hearing and (y) five (5) days after the Effective Date, each TruPS Indenture Trustee shall deliver by email to the IT Fee Notice Parties and their counsel invoices setting forth in reasonable detail (but subject to appropriate redactions to preserve confidentiality or any applicable privileges or protections) the Indenture Trustee Fees incurred by such TruPS Indenture Trustee and its counsel and agents. If an IT Fee Notice Party has not served the applicable TruPS Indenture Trustee with a written objection to the TruPS Indenture Trustee Fees set forth in such invoices within ten (10) Business Days after the date of service of such invoices, then the subject Indenture Trustee Fees shall be Allowed and paid in accordance with section 4.1(d)(i) without the need for application to, or approval by, any court.

В. v. Disputes Regarding Indenture Trustee Fees. If any IT Fee Notice Party disputes any requested Indenture Trustee Fees, such party shall notify the applicable TruPS Indenture Trustee and the Distribution Trustee, and the other IT Fee Notice Parties, and, upon such notification, the applicable TruPS Indenture Trustee may (a) assert its Charging Lien to pay the disputed portion of the Indenture Trustee Fees and/or (b) submit such dispute for resolution to the Bankruptcy Court. If the dispute is not resolved in the TruPS Indenture Trustee's favor, any amounts for which the TruPS Indenture Trustee asserted itscharging lien on account of such disputed Indenture Trustee Fees must be returned. Notwithstanding the pendency of an objection to a portion of a TruPS Indenture Trustee's Indenture Trustee Fees, the Debtors or Distribution Trust, as applicable, shall pay any undisputed portion of Indenture Trustee. Nothing herein shall be deemed to impair, waive, discharge, or negatively affect any Charging Lien for any fees, costs and expenses not paid by the Debtors or the Distribution Trustee and otherwise claimed by a TruPS Indenture Trustee pursuant to the procedures set forth in this Section 4.1(d) of the Plan; provided, however, that no TruPS-Indenture Trustee shall be eligible to receive payment from the Distribution Trust or maintain a Charging Lien for Indenture Trustee Fees incurred after the Effective Date for services related to-Distributions to SRGL on account of its holdings of SHST II TruPS, GPIC TruPS or any corresponding SRGL TruPS Claims, on an expedited basis if the IT Fee Notice Party and the applicable TruPS Indenture Trustee are unable to resolve such dispute consensually within five (5) days of receipt of a written objection. The Distribution Trustee will reserve the disputed Indenture Trustee Fees in the Post-Effective Date Indenture Trustee Fee Reserve pending resolution of the dispute. If there are excess funds in the Post-Effective Date Indenture Trustee Fee Reserve after resolution of the dispute, such funds will be distributed to the applicable holders of Allowed TruPS Claim.

#### 4.2. Unimpaired Classes of Claims and Interests

## (a) Class 1 – Secured Claims

i. *Classification*. Class 1 consists of all Secured Claims, to the extent such Claims have not already been satisfied during the Chapter 11 Cases.

ii. *Treatment.* Unless a Holder of an Allowed Secured Claim agrees to lesser treatment, on the Effective Date, or as soon as reasonably practicable thereafter, each

Holder of an Allowed Secured Claim shall receive one of the following treatments on account of such Allowed Secured Claim, at the option of the Debtors or the Distribution Trustee, as applicable: (a) reinstatement of the Allowed Secured Claim as against any collateral or proceeds thereof held by the Distribution Trust; (b) with the consent of the Purchaser, reinstatement of the Allowed Secured Claim as against any collateral or proceeds thereof held by the Reorganized Debtors; (c) in full and final satisfaction, compromise, settlement, release, and discharge of and in exchange for such Allowed Secured Claim, Cash equal to the full Allowed amount of such Secured Claim, with such Cash to be paid (i) as a Closing Date Plan Distribution to the extent that such Secured Claim is Allowed as of the Effective Date or (ii) from the assets of the Distribution Trust to the extent that such Secured Claim is allowed Allowed after the Effective Date; or (d) with the consent of the Purchaser as to any asset that is not a Distribution Trust Asset, delivery of the collateral securing any such Claim and payment of any interest required under section 506(b) of the Bankruptcy Code.

iii. *Voting.* Class 1 is Unimpaired and the Holders of Claims in Class 1 will be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code. Therefore, the Holders of Claims in Class 1 will not be entitled to vote to accept or reject the Plan.

# (b) Class 2 – Priority Non-Tax Claims

i. *Classification*. Class 2 consists of all Priority Non-Tax Claims, to the extent such Claims have not already been satisfied during the Chapter 11 Cases.

ii. *Treatment.* Unless a Holder of an Allowed Priority Non-Tax Claim agrees to lesser treatment, on the Effective Date, or as soon as reasonably practicable thereafter, each Holder of an Allowed Priority Non-Tax Claim shall receive in full, final and complete satisfaction, settlement, release, and discharge of such Allowed Priority Non-Tax Claim, either: (i) to the extent such Priority Non-Tax Claim is Allowed as of the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Priority Non-Tax Claim as a Closing Date Plan Distribution on the Effective Date, or as soon as reasonably practicable thereafter, or (ii) to the extent such Priority Non-Tax Claim is Allowed after the Effective Date, payment in full in Cash of the unpaid portion of such Allowed Priority Non-Tax Claim from the Distribution Trust at such time as such Priority Non-Tax Claim is Allowed, or as soon as reasonably practicable thereafter.

iii. *Voting.* Class 2 is Unimpaired and the Holders of Claims in Class 2 will be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code. Therefore, the Holders of Claims in Class 2 will not be entitled to vote to accept or reject the Plan.

# (c) Class 3 – Intercompany Claims

i. *Classification*. Class 3 consists of all Intercompany Claims, to the extent such Claims have not already been satisfied during the Chapter 11 Cases.

## Case 18-10160-LSS Doc 437-2 Filed 08/09/18 Page 19 of 105

ii. *Treatment.* Intercompany Claims shall be paid, adjusted, continued, settled, reinstated, discharged, eliminated, or otherwise managed, in each case to the extent determined to be appropriate by the applicable Debtor(s) or Reorganized Debtor(s) and certain of their non-debtor Affiliates with the consent of the Purchaser. For the avoidance of doubt, Intercompany Claims shall not receive a Distribution and shall not otherwise be entitled to any of the assets of the Distribution Trust.

iii. *Voting.* Class 3 is Unimpaired and the Holders of Claims in Class 3 will be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code. Therefore, the Holders of Claims in Class 3 will not be entitled to vote to accept or reject the Plan.

## (d) Class 10 – SALIC Existing Equity Interests

Interests.

i. Classification. Class 10 consists of all SALIC Existing Equity

ii. *Treatment.* SALIC Existing Equity Interests are Unimpaired by the Plan and will be treated in accordance with the Stock Purchase Agreement, the New SALIC Shares Issuance Documents, the Share Surrender Documents, and the Restructuring Implementation Agreement, as provided in <u>Section 6.1</u> of the Plan.

iii. *Voting.* Class 10 is Unimpaired and the Holders of Interests in Class 10 will be conclusively deemed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code. Therefore, the Holders of Interests in Class 10 will not be entitled to vote to accept or reject the Plan.

# 4.3. Impaired Voting Classes of Claims

# (a) Class 4 – SHI TruPS Claims

i. *Classification*. Class 4 consists of all SHI TruPS Claims.

ii. *Allowance*. The SHI TruPS Claims shall be allowed as follows (inclusive of all principal amount issued pursuant to the applicable TruPS Indentures and related documents and accrued but unpaid interest as of the Petition Date at the applicable rates specified in the applicable TruPS Indentures and related documents, other than Indenture Trustee Fees), and shall not be subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual or otherwise) except as provided herein, counterclaim, cross-claim, defense or disallowance under Applicable Law:

TruPS Debenture Issuance	TruPS Indenture Trustee	Holder of Allowed SHI TruPS Claim	Principal	Interest through Petition Date	Total Allowed SHI TruPS Claim
SHST I	U.S. Bank,	U.S. Bank, as			

TruPS Debenture Issuance	TruPS Indenture Trustee	Holder of Allowed SHI TruPS Claim	Principal	Interest through Petition Date	Total Allowed SHI TruPS Claim
TruPS Debentures	as Indenture Trustee	Indenture Trustee, on behalf of Beneficial Holders	\$18,042,000.00	\$4,805,863.87	\$22,847,863.87
SHST II TruPS Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, exclusively for SRGL	\$20,619,000.00	\$5,528,239.88	\$26,147,239.88
GPIC TruPS Debentures	BNYM, as Indenture Trustee	BNYM, as Indenture Trustee, exclusively for SRGL	\$10,310,000.00	\$2,561,006.29	\$12,873,506.29
SHST III TruPS Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, on behalf of Beneficial Holders	\$32,990,000.00	\$8,310,979.84	\$41,300,979.84
TOTAL			\$81,961,000.00	\$21,206,089.88	\$103,169,589.88

iii. Treatment.

A. With respect to Eligible SHI TruPS Claims, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all Eligible SHI TruPS Claims, each Beneficial Holder of TruPS on account of its Allocated Portion of SHI TruPS Claim will receive (to be distributed in accordance with the procedures set forth in Sections 6.1(f) & (g), 6.3(i), 7.5 and 7.6 of this Plan the following Distributions:

Beneficial Holder, the following:

(1) If such Beneficial Holder is a New Equity Eligible

(a) Either (x) if the Beneficial Holder makes the New Equity Election, then the Beneficial Holder's TruPS Claims Equity Distribution Amount; or (y) if the Beneficial Holder makes the Cash Election, the Beneficial Holder's TruPS/GUC Claims Cash Distribution Amount; and (b) The Beneficial Holder's applicable percentage of the Distribution Trust Asset Proceeds, calculated as the amount of such New Equity Eligible Beneficial Holder's Allocated Portion of the Allowed Eligible SHI TruPS Claim divided by the TruPS/GUC Claims Aggregate Amount.

(2) If such Beneficial Holder is SRGL, the following:

(a) SRGL's TruPS/GUC Claims Cash

Distribution Amount; and

(b) SRGL's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of SRGL's Allocated Portion of the Allowed Eligible SHI TruPS Claim divided by the TruPS/GUC Claims Aggregate Amount.

B. With respect to SRGL Exclusively Held SHI TruPS Claims, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all SRGL Exclusively Held SHI TruPS Claims, SRGL shall receive (to be distributed in accordance with the procedures set forth in Sections 6.1(f) & (g), 6.3(i), 7.5 and 7.6 of this Plan):

Amount; and

(1) SRGL's TruPS/GUC Claims Cash Distribution

(2) SRGL's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SRGL Exclusively Held SHI TruPS Claims divided by TruPS/GUC Claims Aggregate Amount.

iv. *Voting.* Class 4 is Impaired, and the Holders of Allowed Class 4 Claims will be entitled to vote to accept or reject the Plan.

# (b) Class 5 – SHI General Unsecured Claims

i. Classification. Class 5 consists of all SHI General Unsecured

Claims.

ii. *Treatment*.

A. On or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all SHI General Unsecured Claims, each Holder of an Allowed SHI General Unsecured Claim shall receive the following Distributions:

(1) The Holder's TruPS/GUC Claims Cash Distribution

Amount; and

(2) The Holder's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed Amount of the Holder's SHI General Unsecured Claims divided by the TruPS/GUC Claims Aggregate Amount.

iii. *Voting.* Class 5 is Impaired, and the Holders of Allowed Class 5 Claims will be entitled to vote to accept or reject the Plan.

## (c) Class 6 – SALIC TruPS Claims & SFL Claims

the SFL Note Claim.

Classification. Class 6 consists of all SALIC TruPS Claims and

## ii. *Allowance*.

i.

A. <u>SALIC TruPS Claims</u>: The SALIC TruPS Claims shall be allowed as follows (inclusive of all principal amount issued pursuant to the applicable TruPS Indentures and related documents and accrued but unpaid interest as of the Petition Date at the applicable rates specified in the applicable TruPS Indentures and related documents, other than Indenture Trustee Fees), and shall not be subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual or otherwise) except as provided herein, counterclaim, cross-claim, defense or disallowance under Applicable Law:\_

TruPS Debenture Issuance	TruPS Indenture Trustee	Holder of Allowed SALIC TruPS Claim	Principal	Interest through Petition Date	Total Allowed SALIC TruPS Claim
SHST I Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, on behalf of Beneficial Holders	\$18,042,000.00	\$4,805,863.87	\$22,847,863.87
SHST II Debentures	U.S. Bank, as Indenture Trustee	U.S. Bank, as Indenture Trustee, exclusively for SRGL	\$20,619,000.00	\$5,528,239.88	\$26,147,239.88
GPIC Debentures	BNYM, as Indenture Trustee	BNYM, as Indenture Trustee, exclusively for SRGL	\$10,310,000.00	\$2,561,006.29	\$12,873,506.29
SHST III Debentures	U.S. Bank, as	U.S. Bank, as Indenture	\$32,990,000.00	\$8,310,979.84	\$41,300,979.84

TruPS Debenture Issuance	TruPS Indenture Trustee	Holder of Allowed SALIC TruPS Claim	Principal	Interest through Petition Date	Total Allowed SALIC TruPS Claim
	Indenture Trustee	Trustee, on behalf of Beneficial Holders			
SFLST I Debentures	WTC, as Indenture Trustee	Wilmington Trust CompanyWTC, as Indenture Trustee, on behalf of Beneficial Holders	\$51,547,000.00	\$11,989,041.32	\$63,536,041.32
TOTAL			\$133,508,000.00	\$33,195,131.20	\$166,705,631.20

#### B. <u>SFL Note Claim</u>:

(1) If the SFLST I TruPS CDO Facility Holders each have voted their respective Allocated Portions of SALIC TruPS Claims to accept the Plan, then (a) the SFL Note Claim Allowance Conditions (as defined below) will be deemed satisfied, such that the Holder of the SFL Note Claim will be deemed (i) to have voted the entire SFL Note Claim to accept the Plan and (ii) to not have opted out of the "Releases by Holders of Claims and Interests" set forth in <u>Section 10.3</u> of the Plan, and (b) the SFL Note Claim shall be deemed Allowed as a Class 6 Claim in the amount of \$63,536,014.32, and shall not be subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual or otherwise) except as provided herein, counterclaim, cross-claim, defense or disallowance under Applicable Law.

(2) If one or more of the SFLST I TruPS CDO Facility Holders has not voted its Allocated Portion of SALIC TruPS Claims to accept the Plan, then whether or not the SFL Note Claim will be Allowed pursuant to this Plan as a Class 6 Claim in the amount of \$63,536,014.32 will depend on whether the SFL Note Claim Allowance Conditions are satisfied or waived as follows:

(a) If the Holder of the SFL Note Claim (1) votes the SFL Note Claim to accept the Plan, (2) does not object to confirmation of the Plan, and (3) does not opt out of the "Releases by Holders of Claims and Interests" set forth in <u>Section 10.3</u> of the Plan (together, the "<u>SFL Note Claim Allowance Conditions</u>"), then upon the occurrence of the Effective Date, the SFL Note Claim shall be deemed Allowed as a Class 6 Claim in the amount of \$63,536,014.32, and shall not be subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual

or otherwise) except as provided herein, counterclaim, cross-claim, defense or disallowance under Applicable Law, and shall be entitled, at such Holder's option to make (i) the New Equity Election, (ii) the Cash Election, or (iii) elect to allocate its Claim between the New Equity Election or the Cash Election, as set forth in Section 4.3(c)(iii) below.

(b) If the Holder of the SFL Note Claim fails to satisfy one or more of the SFL Note Claim Allowance Conditions, then, unless otherwise agreed in a writing signed by an authorized representative of the Debtors (or, if on or after the Effective Date, the Distribution Trustee) and consented to by the Purchaser (which consent shall not be unreasonably withheld) or adjudicated by a Final Order of the Bankruptcy Court, the SFL Note Claim shall (a) remain fully subject to objection, challenge, reduction, offset, avoidance, setoff, recharacterization, impairment, subordination (whether equitable, contractual or otherwise), counterclaim, cross-claim, defense or disallowance under Applicable Law and treated as a Disputed Claim and (b) be deemed to have made the Cash Election and will be reserved for in Cash as a Disputed Class 6 Claim. For the avoidance of doubt, after the Effective Date, the Distribution Trustee shall have the sole right and authority, but not the obligation, to object to, litigate, and settle the amount, priority or extent of the SFL Note Claim and to make a Cash Distribution thereon to the extent Allowed.

#### iii. Treatment:

A. With respect to Eligible SALIC TruPS Claims, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all Eligible SALIC TruPS Claims, each Beneficial Holder of TruPS on account of its Allocated Portion of SALIC TruPS Claim arising from or relating to the TruPS issuance for which it is a Beneficial Holder shall receive (to be distributed in accordance with the procedures set forth in <u>Sections</u> <u>6.1(f) & (g), 6.3(i), 7.5 and 7.6</u> of this Plan) the following Distributions:

(1) If such Beneficial Holder is a New Equity Eligible

Beneficial Holder, the following:

(a) Either (x) if the Beneficial Holder makes the New Equity Election, then the Beneficial Holder's TruPS Claims Equity Distribution Amount; or (y) if the Beneficial Holder makes the Cash Election, the Beneficial Holder's TruPS/GUC Claims Cash Distribution Amount; and

(b) The Beneficial Holder's applicable percentage of the Distribution Trust Asset Proceeds, calculated as the amount of such New Equity Eligible Beneficial Holder's Allocated Portion of the Allowed Eligible SALIC TruPS Claim divided by the TruPS/GUC Claims Aggregate Amount.

(2) If such Beneficial Holder is SRGL, the following:

(a) SRGL's TruPS/GUC Claims Cash

Distribution Amount; and

(b) SRGL's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of SRGL's Allocated Portion of the Allowed Eligible SALIC TruPS Claim divided by the TruPS/GUC Claims Aggregate Amount.

B. With respect to SRGL Exclusively Held SALIC TruPS Claims, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Date(s) (as applicable), in full and final satisfaction of and in exchange for all SRGL Exclusively Held SALIC TruPS Claims, SRGL shall receive (to be distributed in accordance with the procedures set forth in Sections 6.1(f) & (g), 6.3(i), 7.5 and 7.6 of this Plan) the following Distributions:

Amount; and

(1) SRGL's TruPS/GUC Claims Cash Distribution

(2) SRGL's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SRGL Exclusively Held SALIC TruPS Claims divided by the TruPS/GUC Claims Aggregate Amount.

C. With respect to the SFL Note Claim, on or as soon as reasonably practicable after the DT Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all SFL Claims, the Holder of the SFL Note Claim shall receive (to be distributed in accordance with the procedures set forth in Sections 6.1(f) & (g), 6.3(i), 7.5 and 7.6 of this Plan) the following:

(1) If the SFL Note Claim Allowance Conditions are deemed satisfied in accordance with Section 4.3(c)(ii)(B)(1) of this Plan because the SFLST I TruPS CDO Facility Holders each have voted their respective Allocated Portions of SALIC TruPS Claims to accept the Plan, then the Distributions to be received on account of the Allowed SFL Note Claim shall consist of the following:

(a) Subject to the SFL Receiver's right to make the SFL Bankruptcy Administration Senior Cash Election, the Holder of the SFL Note Claim will be deemed to have made the New Equity Election for the remaining portion of the Allowed amount of the SFL Note Claim (*i.e.*, the Allowed SFL Note Claim net of the portion for which the SFL Receiver has made the SFL Bankruptcy Administration Senior Cash Election) that corresponds to the percentage of SFLST I TruPS to have made the New Equity Election (as determined by the Ballots received from the SFLST I TruPS CDO Facility Holders). Accordingly, the Holder of the Allowed SFL Note Claim will be entitled to receive the TruPS Claims Equity Distribution Amount calculated based on-that the portion of the Allowed SFL Note Claim for which it has deemed to have made the New Equity Election as described immediately above. Any remaining portion of the Holder's Allowed SFL Note Claim shall be treated as if the Cash Election hashed been made and entitled to receive the TruPS/GUC Claims Cash Distribution Amount calculated based on such remaining portion of the Allowed SFL Note Claims. (b) The Holder of the Allowed SFL Note Claim's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SFL Note Claim divided by the TruPS/GUC Claims Aggregate Amount.

(2) If one or more of the SFLST I TruPS CDO Facility Holders has not voted its Allocated Portion of SALIC TruPS Claims to accept the Plan, but the Holder of the SFL Note Claim has otherwise satisfied the SFL Note Claim Allowance Conditions in accordance with Section  $4.3(c)(ii)(B)(\frac{12}{a})$  of this Plan, then the Distributions to be received on account of the Allowed SFL Note Claim shall consist of the following:

(a) Either (x) if the Holder of the Allowed SFL Note Claim makes the New Equity Election, such Holder's TruPS Claims Equity Distribution Amount; or (y) if the Holder of the Allowed SFL Note Claim makes the Cash Election, such Holder's TruPS/GUC Claims Cash Distribution Amount<del>, and</del>

(b) The Holder of the Allowed SFL Note Claim's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SFL Note Claim divided by the TruPS/GUC Claims Aggregate Amount.

(3) If the SFL Note Claim Allowance Conditions are not satisfied or deemed satisfied in accordance with this Plan, then subject to and upon the Allowance of the SFL Note Claim post-Effective Date as a Class 6 Claim, then the Distributions to be received on account of the Allowed SFL Note Claim shall consist of the following:

(a) The Holder of the Allowed SFL Note Claim's TruPS/GUC Claims Cash Distribution Amount<del>.; and</del>

(b) The Holder of the Allowed SFL Note Claim's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed SFL Note Claim divided by the TruPS/GUC Claims Aggregate Amount.

D. <u>Voting</u>. Class 6 is Impaired, and the Holders of Allowed Class 6 Claims will be entitled to vote to accept or reject the Plan; *provided, however,* that with respect to the Allowed SFL Note Claim, if the SFLST I TruPS CDO Facility Holders each have voted their respective Allocated Portions of SALIC TruPS Claims to accept the Plan as contemplated by Section 4.3(c)(ii)(B)(1) of this Plan, then the Holder of the SFL Note Claim shall be deemed to have voted the SFL Note Claim to accept the Plan.

# (d) Class 7 – SALIC General Unsecured Claims

i. *Classification*. Class 7 consists of all SALIC General Unsecured

Claims.

- ii. *Treatment*.
  - A. On or as soon as reasonably practicable after the DT

Initial Distribution Date and the DT Subsequent Distribution Dates (as applicable), in full and final satisfaction of and in exchange for all Allowed SALIC General Unsecured Claims, each Holder of an Allowed SALIC General Unsecured Claim shall receive the following Distributions:

(1) The Holder's TruPS/GUC Claims Cash Distribution

Amount; and

Interests.

(2) The Holder's applicable percentage of the Distribution Trust Assets Proceeds, calculated as the amount of the Allowed Amount of the Holder's SALIC General Unsecured Claims divided by the TruPS/GUC Claims Aggregate Amount.

iii. *Voting.* Class 7 is Impaired, and the Holders of Allowed Class 7 Claims will be entitled to vote to accept or reject the Plan.

## 4.4. Impaired Non-Voting Classes of Claims and Interests

## (a) Class 8 – Subordinated Claims

i. *Classification.* Class 8 consists of all Subordinated Claims.

ii. *Treatment.* Holders of Subordinated Claims will not receive or retain any property on account of such Claims. On the Effective Date, Subordinated Claims shall be deemed automatically cancelled, released, and extinguished without further action by any Debtor, any Reorganized Debtor or the Distribution Trustee, and the obligations of the Debtors thereunder shall be forever discharged.

iii. *Voting.* Class 8 is Impaired, and each Holder of a Subordinated Claim will be conclusively deemed to have rejected the Plan pursuant to section 1126(g) of the Bankruptcy Code. Therefore, Holders of Subordinated Claims shall not be entitled to vote on the Plan.

## (b) Class 9 – SHI Existing Equity Interests

i. Classification. Class 9 consists of all SHI Existing Equity

ii. *Treatment.* All SHI Existing Equity Interests will be cancelled and reissued at the direction of the Purchaser as described in <u>Section 6.1(d)</u> of the Plan.

iii. Voting. Class 9 is Impaired, and each Holder of an SHI Existing

Equity Interest will be conclusively deemed to have rejected the Plan pursuant to section 1126(g) of the Bankruptcy Code. Therefore, Holders of SHI Existing Equity Interests shall not be entitled to vote on the Plan.

# V. ACCEPTANCE REQUIREMENTS

# 5.1. Impaired Classes of Claims Entitled to Vote

Holders of Allowed Claims in each Impaired Class that will receive a Distribution are entitled to vote as a Class to accept or reject the Plan. Accordingly, only the votes of Holders of Claims in Classes 4, 5, 6 and 7 shall be solicited with respect to the Plan. A Holder of a Disputed Claim which has not been temporarily allowed for purposes of voting on the Plan may vote only such Disputed Claim in an amount equal to the portion, if any, of such Claim shown as fixed, liquidated, and undisputed in the Schedules.

# 5.2. Acceptance by an Impaired Class

In accordance with section 1126(c) of the Bankruptcy Code, and except as provided in section 1126(e) of the Bankruptcy Code, an Impaired Class shall have accepted the Plan if the Plan is accepted by the Holders of at least two-thirds (2/3) in dollar amount and more than one-half (1/2) in number of the Allowed Claims in such Class that have timely and properly voted to accept or reject the Plan.

# 5.3. Presumed Acceptance by Unimpaired Classes

Classes 1, 2, 3, and 10 are Unimpaired under the Plan. Under section 1126(f) of the Bankruptcy Code, Holders of such Unimpaired Claims and Interests are conclusively presumed to have accepted the Plan, and the votes of Holders of such Claims and Interests shall not be solicited.

# 5.4. Presumed Rejection by Impaired Classes Not Receiving Any Distribution under the Plan

Classes 8 and 9 are Impaired under the Plan, and Holders of Claims and Interests in such Classes will not receive or retain any property under the Plan on account of such Claims or Interests. Under section 1126(fg) of the Bankruptcy Code, Holders of such Claims and Interests are conclusively presumed to have rejected the Plan, and the votes of Holders of such Claims and Interests shall not be solicited.

# 5.5. Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code

To the extent that any Impaired Class rejects the Plan or is deemed to have rejected the Plan, the Debtors shall request Confirmation of the Plan under section 1129(b) of the Bankruptcy Code. The Debtors, subject to the terms of the Plan, the Stock Purchase Agreement, and the Restructuring Implementation Agreement, reserve the right to alter, amend, modify, revoke, or withdraw the Plan, the Plan Supplement, or any schedule or exhibit, including to amend or modify it to satisfy the requirements of section 1129(b) of the Bankruptcy Code, if necessary.

In connection with requesting Confirmation of the Plan under section 1129(b) of the Bankruptcy Code, the Debtors, with the consent of the Purchaser, reserve the right to seek Confirmation of the Plan on a "per plan" basis (as opposed to a "per debtor" basis) consistent with *In re Matter of Transwest Resort Properties, Inc.*, 881 F.3d 724 (9th Cir. 2018).

#### 5.6. Elimination of Vacant Classes

Any Class that, as of the date of commencement of the Confirmation Hearing, does not contain any Allowed Claim or Interest, or any Claim temporarily allowed under Bankruptcy Rule 3018, shall be deemed eliminated from the Plan for purposes of voting to accept or reject the Plan, and disregarded for purposes of determining whether the Plan satisfies section 1129(a)(8) of the Bankruptcy Code with respect to such Class.

# 5.7. Presumed Acceptance by Voting Classes in Which No Votes Are Cast

If a Class contains Claims eligible to vote and no Holder of a Claim eligible to vote in such Class votes to accept or reject the Plan, the Plan shall be presumed accepted by the Holders of such Claims in such Class.

## 5.8. Consolidation of Classes

To the extent permitted under section 1122 of the Bankruptcy Code, and subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement, the Debtors reserve the right to consolidate one or more Classes of Claims, including for purposes of sections 1126, 1129(a)(8), 1129(a)(10) or 1129(b) of the Bankruptcy Code.

## 5.9. Separate Classes of Secured Claims

Although all Secured Claims have been placed in one Class for purposes of nomenclature within the Plan, each Secured Claim, to the extent secured by a Lien on Collateral different from the Collateral collateral securing another Secured Claim, shall be treated as being in a separate sub-Class for the purposes of receiving Distributions.

# VI. MEANS FOR IMPLEMENTATION OF THE PLAN

## 6.1. Plan Transactions

## (a) Stock Purchase Agreement Closing

On or prior to the Effective Date, and as a condition to the Effective Date, each of the actions, transactions, and deliveries described in the Stock Purchase Agreement shall occur and shall have occurred, including, without limitation, the actions, transactions, and deliveries described in section 2.4 of the Stock Purchase Agreement.

## (b) Funding, Allocation and Use of Plan Funding Payment; Allocation and Use of Debtors' Unrestricted Cash

On the Effective Date, the Plan Funding Payment shall be funded by the Purchaser in accordance with the terms of the Stock Purchase Agreement and this Plan and allocated and used as follows (such waterfall, the "<u>Allocation/Use Priorities</u>"):

- (1) <u>First</u>, to fund-(a) all Closing Date Plan Distributions, (b) the Indenture Trustee Fees payable as set forth in section 4.1(d)(i), and (c) the Indenture Trustee Fee Reserve on account of the Pre-Effective Date Indenture Trustee Fee Estimate, to the extent such amounts in (a) (c) are not fully funded from the unrestricted Cash of SALIC and SHI; provided, however, that such amounts shall be funded on or before the Effective Date from the unrestricted Cash of SALIC and SHI to the extent of such unrestricted Cash;
- (2) <u>Second</u>, to fund the Professional Fee Reserve;
- (3) <u>Third</u>, to fund the Trust Administration Reserve <u>up to the Trust</u> <u>Administration Reserve Cap</u>;
- (4) <u>Fourth</u>, to fund the Disputed Claims Reserve to be maintained by the Distribution Trust;
- (5) <u>Fifth</u>, to fund the Indenture Trustee Fee Reserve on account of the Post-Effective Date Indenture Trustee Fee Estimate; (6) <u>Sixth</u>, to fund Distributions to the Holders of Allowed SHI General Unsecured Claims, Allowed SALIC General Unsecured Claims, Allowed SALIC General Unsecured Claims, Allowed SHI TruPS Claims (that make the Cash Election), Allowed SALIC TruPS Claims (that make the Cash Election), and Allowed SFL Note Claim (if the Cash Election is made or deemed to have been made) with such Distributions to be made on the DT Initial Distribution Date or DT Subsequent Distribution Dates, as applicable.

The Plan Funding Payment shall be adjusted downward by the TruPS Returned Cash on account of any amount of the Available Plan Funding Distribution Amount allocable to Allowed TruPS Claims that elect to receive the New Equity. For the avoidance of doubt, the Purchaser shall not fund the TruPS Returned Cash shall be returned by the Distribution Trust to the Hildene Master Funds or their designee in such proportion as they direct on the DT Initial Distribution Date. Any portion of the Plan Funding Payment that is subsequently released from the Disputed Claims Reserve, the Professional Fee Reserve, or the Trustee Administration Reserve, or the

Indenture Trustee Fee Reserve on account of the Post-Effective Date Indenture Trustee Fee Estimate, shall be released by the Distribution Trustee pro rata to (x) the PurchaserHildene Master Funds or their designee on account of the TruPS Returned Cash and (y) Holders of Allowed SHI General Unsecured Claims, Allowed SALIC General Unsecured Claims, Allowed SHI TruPS Claims (that make the Cash Election), Allowed SALIC TruPS Claims (that make the Cash Election), and Allowed SFL Note Claim (if the Cash Election is made or deemed to have been made). Any portion of the Indenture Trustee Fee Reserve on account of the Pre-Effective Date Indenture Trustee Fee Estimate that has not been paid to satisfy Indenture Trustee Fees shall be returned to the Reorganized Debtors to the extent that such amount was funded with unrestricted cash of SALIC and SHI. Except as stated in Section 6.1(b)(1) of the Plan, the unrestricted Cash of SALIC and SHI shall remain with the Reorganized Debtors.

## (c) Funding of Recapitalization Funding Payment

On the Effective Date, the Recapitalization Funding Payment shall be funded to Reorganized SALIC by the Purchaser in accordance with the terms of the Stock Purchase Agreement and this Plan. The Recapitalization Funding Payment shall not be used to make Distributions.

# (d) Cancellation of SHI Existing Equity Interests; Issuance of New SHI Equity

On the Effective Date, all SHI Existing Equity Interests shall be cancelled and New SHI Equity shall be issued to the Purchaser or to another entity at the direction of the Purchaser in its sole discretion. Unless the Purchaser determines otherwise in its sole discretion, the New SHI Equity shall be deemed immediately contributed by the Purchaser to Reorganized SALIC.

# (e) Final Share Surrender

On the Effective Date and immediately following the New SALIC Equity issuance to Purchaser, in accordance with the terms and conditions of the Restructuring Implementation Agreement, the Stock Purchase Agreement and this Plan, SRGL shall complete the Final Share Surrender (as defined in the Restructuring Implementation Agreement). For the avoidance of doubt, as a result of the Plan, SRGL as the holder of the SALIC Existing Equity Interests shall not receive or retain any property under the Plan on account of such SALIC Existing Equity Interests.

# (f) New Equity Issuance and Distribution

On the Effective Date, without further act or action under Applicable Law (other than as required by Applicable Law of the Cayman Islands with respect to SRGL and SALIC and provided for in the Restructuring Implementation Agreement and the RIA Order), in accordance with the terms and conditions of the Stock Purchase Agreement, the Restructuring Implementation Agreement, the RIA Order and this Plan, the New Equity shall be issued and distributed by Reorganized SALIC or New Holdco, as applicable as reflected in the

organizational chart included in the Plan Supplement. Such New Equity shall be issued and distributed free and clear of all Liens, Claims and other Interests, except as expressly provided in this Plan.

On or before the deadline established by the Disclosure Statement Order for the filing of the Plan Supplement, the <u>PurchaserDebtors</u> shall File a notice stating whether the New Equity will be issued by Reorganized SALIC or New Holdco, which notice may be Filed as part of the Plan Supplement<u>in the form of an organizational chart</u>. Any recipient or subsequent holder of shares of New Equity shall be required to enter into the Stockholders Agreement, whether such recipient or holder acquires such shares as of the Effective Date or subsequent thereto. The New Corporate Governance Documents (including the Stockholders Agreement) will include certain restrictions on transfers of the New Equity, which shall be reasonably acceptable to the Purchaser in consultation with the Official Committee and the Debtors, and disclosed in the Plan Supplement.

The New Equity when issued or distributed as provided in the Plan, will be duly authorized, validly issued and, if applicable, fully paid and nonassessable. Each Distribution and issuance of such New Equity shall be governed by the terms and conditions set forth in the Plan applicable to such Distribution or issuance and by the terms and conditions of the instruments evidencing or relating to such Distribution or issuance, which terms and conditions shall bind each Person receiving such Distribution or issuance.

The Debtors, the Purchaser, the Indenture Trustees, the Official Committee, SRGL, the Voting Agent, and each of their respective Representatives have, and upon Confirmation of the Plan shall be deemed to have, participated in good faith and in compliance with the applicable provisions of the Bankruptcy Code and Applicable Law with regard to the distributionDistribution of the New Equity under the Plan, and therefore are not, and on account of such distributionsDistributions will not be, liable at any time for the violation of any Applicable Law, rule, or regulation governing the solicitation of acceptances or rejections of the Plan or such distributionsDistributions made pursuant to the Plan. Upon entry of the Confirmation Order, all provisions of the Plan addressing distributionDistribution of the New Equity shall be deemed necessary and proper.

## (g) Distribution of Offered New Equity

On the <u>EffectiveDT Initial Distribution</u> Date as soon as practicable thereafter, the Offered New Equity shall be distributed to all New Equity Eligible Holders that make the New Equity Election.

On the Effective Date, all New Equity, other than the Offered New Equity that is distributed to New Equity Eligible Holders that <del>do</del>-make the New Equity Election, shall be distributed to the <u>PurchaserHildene Master Funds or their designee in such proportion as they</u> <u>direct</u>.

Any shares of Offered New Equity that, as of the <u>EffectiveDT Initial Distribution</u> Date, have not already been distributed to (or earmarked for distribution to) New Equity Eligible Holders or the Holder of the Allowed SFL Note Claim, shall be distributed to the <u>PurchaserHildene Master Funds or their designee in such proportion as they direct</u>.

## 6.2. Vesting of Estate Property

On the Effective Date, all property of the Debtors and their Estates shall vest automatically in the Reorganized Debtors or the Distribution Trust as described in this section of the Plan.

On the Effective Date, except as otherwise expressly provided in the Confirmation Order, the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves shall automatically vest in the Distribution Trust free and clear of all Claims, Liens and Interests (other than the Purchaser and Reorganized Debtors' respective reversionary interests in the Distribution Trust Reserves).

Except for the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves, or as otherwise expressly provided in the Confirmation Order, pursuant to sections 1123(b)(3) and 1141(b)–(c) of the Bankruptcy Code, on the Effective Date, all of the property and assets of each Debtor shall automatically vest in the respective Reorganized Debtor, free and clear of all Claims, Liens and Interests. For the avoidance of doubt, such property and assets of each Debtor that will automatically vest in the Reorganized Debtors includes, but is not limited to, any SALIC Group Services Claim incurred on or after the Petition Date. The Reorganized Debtors may operate their business and may use, acquire, and dispose of such property free of any restrictions of the Bankruptcy Code, the Bankruptcy Rules, and the Bankruptcy Court. As of the Effective Date, all such property of the Reorganized Debtors shall be free and clear of all Claims, Liens and Interests, except as specifically provided in the Confirmation Order, and the Reorganized Debtors shall receive the benefit of any and all discharges and injunctions under the Plan.

# 6.3. The Distribution Trust

# (a) Execution of Distribution Trust Agreement

On or prior to the Effective Date, the Debtors shall execute the Distribution Trust Agreement, and shall take all other necessary steps to establish the Distribution Trust, which shall be for the payment of Allowed Administrative Claims, Allowed Priority Claims, and Allowed Secured Claims not satisfied by Closing Date Plan Distributions, and for the benefit of the Distribution Trust Beneficiaries. In the event of any conflict between the terms of this Section 6.3(a) and the terms of the Distribution Trust Agreement as such conflict relates to the establishment of the Distribution Trust, the terms of this Section 6.3(a) shall govern. The Distribution Trust Agreement may provide powers, duties and authorities in addition to those explicitly stated herein, but only to the extent that such powers, duties, and authorities do not affect the status of the Distribution Trust as a "liquidating trust" for United States federal income tax purposes. The Distribution Trust Agreement shall be Filed with the Plan Supplement.

## (b) **Purpose of the Distribution Trust**

The Distribution Trust shall be established for the sole purpose of liquidating and distributing the assets of the Debtors contributed to such Distribution Trust in accordance with Treas. Reg. § 301.7701-4(d), with no objective to continue or engage in the conduct of a trade or business.

## (c) Distribution Trust Assets and Other Property

The Distribution Trust shall consist of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves. On the Effective Date, all of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves shall transfer to and be vested in the Distribution Trust. All attorney-client privilege associated with the Retained Causes of Action remains with and vests in the Reorganized Debtors.

# (d) The Administration of the Distribution Trust and Authority and Powers of the Distribution Trustee

The Distribution Trust shall be administered by the Distribution Trustee pursuant to the Distribution Trust Agreement. The initial Distribution Trustee shall be a Person selected by the Official Committee and reasonably acceptable to the Debtors and the Purchaser. The identity of the initial Distribution Trustee will be disclosed in the Plan Supplement, and any successor Distribution Trustee shall be appointed in the manner set forth in the Distribution Trust Agreement. In the event of any inconsistency between the Plan and the Distribution Trust Agreement as such conflict relates to anything other than the establishment of the Distribution Trust, the Plan shall control. All compensation for the Distribution Trustee and other costs of administration for the Distribution Trust Assets in accordance with this Plan and the Distribution Trust Agreement. The Distribution Trust estable he are presentative of each Debtor's Estate in accordance with section 1123(b)(3)(B) of the Bankruptcy Code for the purposes of the DT Post-Closing Rights.

# (e) Mutual Cooperation

As the Reorganized Debtors or the Distribution Trustee may reasonably request, each shall use commercially reasonable efforts to cooperate with the other with respect to the implementation of the Plan (including, without limitation, the resolution of Disputed Claims, the determination of taxes and the preparation and filing of tax returns), with all reasonable out-of-pocket expenses incurred by the (i) Reorganized Debtors in connection with a request by the Distribution Trustee made pursuant to this paragraph being borne by the Distribution Trustee and (ii) Distribution Trustee in connection with a request by the Reorganized Debtors made pursuant to this paragraph being borne by the Reorganized Debtors; *provided, however*, that neither party shall be required to (i) provide information, records or employees or other personnel under circumstances which the providing party believes in its sole reasonable determination may waive privilege, confidentiality or a similar protection or expose it to material liability to any person or may prejudice any legal interest of the providing party, or (ii) take any action that in the providing party's reasonable determination unreasonably interferes with its business. For the avoidance of doubt, nothing herein is intended to limit the DT Post-Closing Rights of the Distribution Trust and the Distribution Trustee.

## (f) Establishment and Funding of Distribution Trust Reserves

On the Effective Date, the following Distribution Trust Reserves shall be established and funded from the Plan Funding Payment, each in accordance with the Allocation/Use Priorities:

i. Disputed Claims Reserve. The Disputed Claims Reserve shall be established and funded with Cash (including an amount for the SFL Note Claim if the SFL Note Claim Allowance Conditions are not met as of the Effective Date) in an amount sufficient to cover pro rata distributions to each Disputed Claim that, as of the Effective Date, is neither an Allowed Claim nor a Disallowed Claim, and includes, without limitation, a Claim that is the subject of a timely objection or request for estimation with the Bankruptcy Court, which has not been withdrawn, settled or overruled by a Final Order; *provided, however*, that if the Disputed Claim is an Administrative Claim (other than a Professional Fee Claim), Priority Claim or Secured Claim, an amount sufficient to cover payment in full of the Face Amount of such Disputed Claim shall be funded to the Disputed Claims Reserve; provided further, that if any dispute arises regarding any increase or reduction of the Disputed Claims Reserve, the Distribution Trustee shall consult with the Purchaser and shall obtain approval of the Bankruptcy Court, which shall have jurisdiction and power to set the amount of the reserve applying the principals of section 502(c) of the Bankruptcy Code to estimate any Claim.

ii. *Professional Fee Reserve.* The Professional Fee Reserve shall be established and funded in an amount that the Debtors estimate in good faith, after consultation with the relevant Professionals, the Purchaser, and the Official Committee, to be necessary to pay in full all amounts then owing or that may later become owing to such Professionals for professional fees and expenses incurred through the Effective Date. For the avoidance of doubt, the estimated amount initially funded to the Professional Fee Reserve is not intended as and shall not be deemed to be a cap on the funds available to pay or satisfy Allowed Administrative Claims of Professionals for compensation or reimbursement of expenses. Following the Effective Date, the Distribution Trustee shall have the discretion to increase the Professional Fee Reserve as the Distribution Trustee deems necessary or appropriate to pay or satisfy Allowed Administrative Claims of Professionals for compensation or reimbursement of expenses.

For the avoidance of doubt, the KBW Reserved Funds will not be part of the Professional Fee Reserve. Subject to the Bankruptcy Court's entry of an Order approving its Contingent Fee (as defined in the KBW Retention Order), the KBW Reserved Funds shall be distributed directly to KBW. If a Final Order is entered denying KBW's final application for allowance and payment of compensation and reimbursement of expenses or reducing the amount otherwise payable to KBW to such a degree that KBW is not entitled to the KBW Reserved Funds, then the KBW Reserved Funds shall be disbursed to Reorganized SALIC.

Trust Administration Reserve. The Trust Administration Reserve iii. shall be established and funded\_in an amount, mutually agreed by the Debtors, the Official Committee, and the Purchaser, estimated in good faith to be necessary to cover the costs of administration of the Distribution Trust and subject in all respects to the Trust Administration Reserve Cap. The Trust Administration Reserve shall be used to administer the Distribution Trust, including to (a) fund the reasonable fees and expenses of the Distribution Trustee and any employees, attorneys, accountants, financial advisors, consultants, other professional persons or independent contractors that the Distribution Trustee may engage to assist him, her or it in the discharge of the Distribution Trustee's duties under the Plan and the Distribution Trust Agreement, including, without limitation, fees and expenses related to prosecution and resolution of Causes of Action and objections to Claims; (b) fund premium payments for an errors and omissions insurance policy for the benefit of the Distribution Trust, the Distribution Trustee and the Distribution Trustee's agents and representatives, (c) meet contingent liabilities and to maintain the value of the Distribution Trust Assets during liquidation, (d) pay other reasonably incurred or anticipated expenses (including, without limitation, any taxes imposed on or payable by the Distribution Trust or in respect of the Distribution Trust Assets, including with respect to such assets as are allocable to Disputed Claims), and (e) satisfy other liabilities incurred or anticipated by such Distribution Trust in accordance with the Plan or Distribution Trust Agreement.

#### (g) Establishment and Funding of the Indenture Trustee Fee Reserve. Intentionally Omitted.

i. The Indenture Trustee Fee Reserve shall have two accounts for each of (a) the Pre-Effective Date Indenture Trustee Fee Estimate and (b) the Post-Effective Date Indenture Trustee Fee Estimate. The Indenture Trustee Fee Reserve shall be established and funded, in the following manner:

#### Case 18-10160-LSS Doc 437-2 Filed 08/09/18 Page 37 of 105

A. For the Pre-Effective Date Indenture Trustee Fee Estimate, first, from the unrestricted Cash available to SALIC and SHI, and to the extent not fully funded from the unrestricted Cash of SALIC and SHI, then from the Plan Funding Payment, in an amount that the Debtors estimate in good faith, after consultation with the Purchaser and the relevant TruPS Indenture Trustees, to be necessary to pay in full, but subject to the relevant Indenture Trustee Fee Caps, and

B. For the Post-Effective Date Indenture Trustee Fees, from the Plan Funding Payment. For the avoidance of doubt, the Distribution Trustee shall be underno obligation to reserve any amount in the Indenture Trustee Fees Reserve on account of post-Effective Date Indenture Trustee Fees that may be incurred by the TruPS Indenture Trustees for the SHST II Debentures or the GPIC Debentures.

ii. The Indenture Trustee Fee Reserve shall be held by the Distribution Trust and administered by the Distribution Trustee, but shall not constitute a Distribution Trust Reserve.

iii. Any remaining funds in the Indenture Trustee Fee Reserve after payment and satisfaction of all Indenture Trustee Fees, shall be released in accordance with Section 6.1(b) of the Plan.

#### (h) Cash Investments

The Distribution Trustee may invest Cash (including any earnings thereon or proceeds therefrom); *provided, however*, that such investments are investments permitted to be made by a "liquidating trust" within the meaning of Treas. Reg. § 301.7701-4(d), as reflected therein, or under applicable IRS guidelines, rulings or other controlling authorities.

#### (i) Distributions to Holders of Claims and Distribution Trust Beneficiaries

The Distribution Trustee shall be responsible for making all Distributions to Holders of Allowed Claims required to be made on or after the Effective Date pursuant to the Plan; *provided*, that the Reorganized Debtors or the Disbursing Agent (as applicable) shall make the Distributions to Holders of Allowed Claims on the Effective Date on behalf of the Distribution Trustee. The Distribution Trustee will make all Distributions to Holders of Allowed Claims, or in the case of Holders of Allowed Claims in Classes 4 and 6, the Beneficial Holders thereof, as required by this Plan at: (i) the address of any such Holder on the books and records of the Debtors, the Distribution Trustee or their agents; or (ii) at the address in any written notice of address change delivered to the Debtors, the Distribution Trustee or the Distribution Trustee or the Disbursing Agent, including any addresses included on any transfer of Claim filed pursuant to Bankruptcy Rule 3001 or as otherwise provided in the Distribution Trust Agreement. The procedures for Distributions to be made to the holder of the SFL Note Claim shall be set forth in the SFL Distribution Agreement.

i. DT Initial Distribution

As soon as reasonably practicable after (i) funding of all Distribution Trust Reserves, (ii) the Indenture Trustee Reserve (as applicable) and (ii) payment in full (or reserving for payment in full) of all Allowed Administrative Claims, Allowed Priority Claims, and Allowed Secured Claims, the Distribution Trustee shall distribute to the Holders of Allowed Claims in Classes 4, 5, 6 and 7, or in the case of Holders of Allowed Claims in Classes 4 and 6, to the Beneficial Holders thereof, on account of their Allowed Claims their pro rata share of the Available Plan Distribution Funding Amount and Distribution Trust Asset Proceeds, as applicable to Allowed Claims.

#### ii. DT Subsequent Distribution

After completion of the DT Initial Distribution, the Distribution Trustee shall make the DT Subsequent Distribution(s) in a reasonably timely manner after Distribution Trust Assets Proceeds become available, and in accordance with such procedures as may be provided in the Distribution Trust Agreement. Such DT Subsequent Distributions shall be made no less frequently than every twelve (12) months; *provided, however*, that the Distribution Trustee shall not be required to make a Distribution pursuant to this <u>Section 6.3(h)(ii)</u> of the Plan if the Distribution Trustee determines that the expense associated with making the Distribution would likely utilize a substantial portion of the amount to be distributed, thus making the Distribution impracticable.

#### (j) Federal Income Tax Treatment of Distribution Trust

Subject to definitive guidance from the IRS or a court of competent jurisdiction to the contrary (including the receipt of an adverse determination by the IRS upon audit if not contested by such Distribution Trustee), for all U.S. federal income tax purposes, all parties (including, without limitation, the Debtors, the Distribution Trustee and Distribution Trust Beneficiaries) shall treat the transfer of Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves to the Distribution Trust as (1) a transfer of property (subject to any and all Allowed Administrative Claims, Allowed Priority Claims, and Allowed Secured Claims to the extent not satisfied by the Debtors on or prior to the Effective Date, that are payable by the Distribution Trust pursuant to the Plan), followed by (2) the transfer by such beneficiaries to the Distribution Trust of Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves. Accordingly, except in the event of contrary definitive guidance, Distribution Trust Beneficiaries shall be treated for U.S. federal income tax purposes as the grantors and owners of their respective share of Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves (other than that which is allocable to Disputed Claims). The foregoing treatment shall also apply, to the extent permitted by Applicable Law, for state and local income tax purposes. For the avoidance of doubt, the term "party" as herein used shall not include the United States or any agency or department thereof, or any officer or employee thereof acting in such capacity. The Distribution Trustee shall not take any action inconsistent with the purposes of the Distribution Trust and the qualification of the Distribution Trust as a "liquidating trust" for U.S. federal income tax purposes.

#### (k) Tax Reporting

i. The Distribution Trustee shall file tax returns for the Distribution Trust treating such Distribution Trust as a grantor trust pursuant to Treas. Reg. § 1.671-4(a) and in accordance with this <u>Section 6.3</u>. The Distribution Trustee also shall annually send or otherwise provide to each Holder of the Distribution Trust Interest a separate statement regarding the receipts and expenditures of the Distribution Trust as relevant for U.S. federal income tax purposes.

ii. Allocations of Distribution Trust taxable income among Distribution Trust Beneficiaries (other than taxable income allocable to any assets allocable to, or retained on account of, Disputed Claims, if such income is otherwise taxable at the Distribution Trust) shall be determined by reference to the manner in which an amount of Cash representing such taxable income would be distributed (were such Cash permitted to be distributed at such time) if, immediately prior to such deemed Distribution, the Distribution Trust had distributed all its assets (valued at their tax book value, other than, if applicable, assets allocable to Disputed Claims) to the Holders of Distribution Trust Interests, adjusted for prior taxable income and loss and taking into account all prior and concurrent Distributions from the Distribution Trust. Similarly, taxable loss of the Distribution Trust shall be allocated by reference to the manner in which an economic loss would be borne immediately after a hypothetical liquidating distribution of the remaining Distribution Trust Assets, Available Plan Distribution Funding Amount and Distribution Trust Reserves. The tax book value of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves for purpose purposes of this paragraph shall equal their fair market value on the date such assets are transferred to the Distribution Trust, adjusted in accordance with tax accounting principles prescribed by the IRC, the applicable Treasury Regulations, and other applicable administrative and judicial authorities and pronouncements.

iii. As soon as reasonably practicable after the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves are transferred to the Distribution Trust, the Distribution Trustee shall make a good faith valuation of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves. Such valuation shall be made available from time to time to all parties to the Distribution Trust (including, without limitation, the Debtors (or, as the case may be, the Reorganized Debtors), and the Distribution Trust Beneficiaries), to the extent relevant to such parties for tax purposes, and shall be used consistently by such parties for all U.S. federal income tax purposes.

iv. Subject to definitive guidance from the IRS or a court of competent jurisdiction to the contrary (including the receipt by the Distribution Trustee of a private letter ruling if the Distribution Trustee so requests one, or the receipt of an adverse determination by the IRS upon audit if not contested by such Distribution Trustee), the Distribution Trustee (i) shall treat any Distribution Trust Reserves allocable to Disputed Claims as a "disputed ownership fund" governed by Treas. Reg. § 1.468B-9 (and make any necessary elections with respect thereto) and (ii) to the extent permitted by Applicable Law, shall report

consistently for state and local income tax purposes. All parties (including the Distribution Trustee, the Debtors and Distribution Trust Beneficiaries) shall report for U.S. federal, state and local income tax purposes consistently with the foregoing.

v. The Distribution Trustee shall be responsible for payment, out of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves, of any taxes imposed on the Distribution Trust or its assets (including with respect to assets allocable to Disputed Claims).

vi. The Distribution Trustee may request an expedited determination of taxes of the Distribution Trust, including any reserve for Disputed Claims, or of the Debtors as to whom the Distribution Trust was established, under section 505(b) of the Bankruptcy Code for all tax returns filed for, or on behalf of, such Distribution Trust or the Debtors for all taxable periods through the dissolution of such Distribution Trust.

## (l) Dissolution

i. The Distribution Trustee and Distribution Trust shall be discharged or dissolved, as the case may be, at such time as (i) all of the Distribution Trust Assets, the Available Plan Distribution Funding Amount and the Distribution Trust Reserves have been expended or distributed pursuant to the Plan and the Distribution Trust Agreement, (ii) the Distribution Trustee determines, in its sole discretion, that the administration of any remaining Distribution Trust Assets, Available Plan Distribution Funding Amount or Distribution Trust Reserves is not likely to yield sufficient additional Distribution Trust proceeds to justify further pursuit, or (iii) all Distributions required to be made by the Distribution Trustee under the Plan and the Distribution Trust Agreement have been made; provided, however, that in no event shall the Distribution Trust be dissolved later than three (3) years from the creation of such Distribution Trust pursuant to this Section 6.3 of the Plan, unless the Bankruptcy Court, upon motion within the six-month period prior to the third (3rd) anniversary (or within the six-month period prior to the end of an extension period), determines that a fixed period extension (not to exceed three (3) years, together with any prior extensions, without a favorable private letter ruling from the IRS or an opinion of counsel in form and substance satisfactory to the Distribution Trustee that any further extension would not adversely affect the status of the trust as the Distribution Trust for U.S. federal income tax purposes) is necessary to facilitate or complete the recovery and liquidation of the Distribution Trust Assets.

ii. If at any time the Distribution Trustee determines, in reliance upon such professionals as the Distribution Trustee may retain, that the expense of administering the Distribution Trust so as to make a final Distribution to Distribution Trust Beneficiaries is likely to exceed the value of the assets remaining in such Distribution Trust, such Distribution Trustee may apply to the Bankruptcy Court for authority to (i) reserve any amount necessary to dissolve such Distribution Trust, (ii) donate any balance to a charitable organization (A) described in section 501(c)(3) of the IRC, (B) exempt from U.S. federal income tax under section 501(a) of the IRC, (C) not a "private foundation", as defined in section 509(a) of the IRC, and (D) that is unrelated to the Debtors, such Distribution Trust, and any insider of such Distribution Trustee, and (iii) dissolve such Distribution Trust.

## 6.4. The Reorganized Debtors

## (a) Continued Corporate Existence

Except as otherwise provided in the Plan, each of the Debtors, as Reorganized Debtors, shall continue to exist after the Effective Date as a corporate entity, with all of the powers of a corporation or limited company, as the case may be, under Applicable Law in the jurisdiction in which such Debtor is incorporated or organized and pursuant to the New Corporate Governance Documents. After the Effective Date, the Reorganized Debtors may operate their business and use, acquire, and dispose of property without the supervision of the Bankruptcy Court, free of any restrictions of the Bankruptcy Code or Bankruptcy Rules.

The New Corporate Governance Documents shall be consistent with section 1123(a)(6) of the Bankruptcy Code, to the extent applicable, and in form and substance acceptable to the Purchaser.

## (b) Directors and Officers of the Reorganized Debtors

The officers and the members of each board of directors of each of the Reorganized Debtors shall be selected and appointed in the sole discretion of the Purchaser. To the extent required by section 1129(a)(5) of the Bankruptcy Code, the identity of such officers and members shall be disclosed prior to the Confirmation Hearing.

Except to the extent that a member of the board of directors of a Debtor continues to serve as a director of such Debtor following the Effective Date, the members of the board of directors of each Debtor prior to the Effective Date, in their capacities as such, shall have no continuing obligations to the Reorganized Debtors on or after the Effective Date and each such member will be deemed to have resigned or shall otherwise cease to be a director of the applicable Debtor on the Effective Date without any further action required on the part of any such Debtor or member. Commencing on the Effective Date, each of the directors of each of the Reorganized Debtors shall serve pursuant to the terms of the applicable organizational documents of such Reorganized Debtor and may be replaced or removed in accordance with such organizational documents.

Except as otherwise provided in the Stock Purchase Agreement with respect to the Employment Agreements (as defined therein), subject to the discretion of the Reorganized Debtors' boards of directors, the Reorganized Debtors shall enter into new employment agreements with key executives on a case-by-case basis.

# (c) Corporate Action

On the Effective Date, the adoption and filing of the New Corporate Governance Documents, the appointment of officers of the Reorganized Debtors, and all actions contemplated by the Plan will be authorized and approved in all respects pursuant to the Plan. On the Effective Date, pursuant to section 1142(b) of the Bankruptcy Code and section 303 of the Delaware General Corporation Law (to the extent applicable) and any comparable provision of other Applicable Law, the appropriate officers or directors of each Reorganized Debtor shall be authorized and directed to issue, execute, and deliver the agreements, documents, securities, and instruments contemplated by the Plan with like effect as if exercised and taken by unanimous action of the directors and stockholders of each Debtor.

## (d) Effectuating Documents; Further Transactions

On the Effective Date or as soon as reasonably practicable thereafter, the Reorganized Debtors and/or the Purchaser may take all actions as may be necessary or appropriate to effect any transaction described in, approved by, or necessary or appropriate to effectuate the Plan, including: (1) the execution and delivery of appropriate agreements or other documents of merger, consolidation, restructuring, conversion, disposition, transfer, dissolution or liquidation to the extent consistent with the terms of the Plan and the Plan Documents; (2) the execution and delivery of appropriate instruments of transfer, assignment, assumption or delegation of any asset, property, right, liability, debt or obligation on terms consistent with the terms of the Plan and Plan Documents or having other terms to which the Debtors, the Reorganized Debtors, the Purchaser, and other applicable parties may agree; (3) the filing of appropriate certificates or articles of incorporation, reincorporation, merger, consolidation, conversion or dissolution pursuant to applicable state law; and (4) all other actions that the Purchaser and any other applicable Entities determine to be necessary or appropriate, including making filings or recordings that may be required by Applicable Law.

## 6.5. Retained Causes of Action

Except to the extent any Claim against an Entity is expressly waived, relinquished, exculpated, released, compromised or settled in the Plan or by a Final Order or is a Distribution Trust Asset, all Causes of Action of the Debtors (the "Retained Causes of Action") shall, in accordance with section 1123(b) of the Bankruptcy Code, vest in and be retained by the Reorganized Debtors, including any Causes of Action against the Specified SALIC Officers and Directors. The applicable Reorganized Debtors (with respect to the Retained Causes of Action and any Causes of Action arising after the Petition Date), in accordance with section 1123(b) of the Bankruptcy Code, shall retain and may enforce, sue on, settle, or compromise (or decline to do any of the foregoing) all Claims, rights, Causes of Action, suits, and proceedings, whether in law or in equity, whether known or unknown, that they each may respectively hold against any Person without the approval of the Bankruptcy Court and the Reorganized Debtors' rights to commence, prosecute, or settle such Causes of Action shall be preserved notwithstanding the occurrence of the Effective Date. For the avoidance of doubt, all Causes of Action against the SRGL Equity Holders shall not be Retained Causes of Action, and shall be transferred to and vest in the Distribution Trust. On the Effective Date, all Causes of Action that are Distribution Trust Assets shall, in accordance with Section 1123(b) of the Bankruptcy Code, vest in the Distribution Trust, and the Distribution Trust may enforce, sue on, settle, or compromise (or decline to do any of the foregoing) all such Claims, rights, Causes of Action, suits, and proceedings, whether in law or in equity, known or unknown, without approval of the

Bankruptcy Court, and the Distribution Trust's rights to commence, prosecute, or settle such Causes of Action shall be preserved notwithstanding the occurrence of the Effective Date.

#### 6.6. The Closing

The Closing as defined in the Stock Purchase Agreement shall be subject to the conditions in the Stock Purchase Agreement, including, without limitation, the conditions to closing set out in Article VII of the Stock Purchase Agreement and the actions and deliveries set out in section 2.4 of the Stock Purchase Agreement, unless waived in accordance with the Stock Purchase Agreement. The Closing shall occur simultaneously with the Effective Date of the Plan.

#### 6.7. Cancellation of Agreements, Securities and Other Documents Relating to TruPS Transactions & SFL Note

Except for purposes of evidencing a right to a Distribution under the Plan or otherwise as provided in the Plan, the Confirmation Order or the Distribution Trust Agreement, on the Effective Date, the TruPS Indentures, the TruPS Debentures, the TruPS Declarations, the TruPS Sponsor Guarantees, the TruPS Parent Guarantees, all other TruPS Documents, the SFL Note and all corresponding documents issued in connection with such documents shall be deemed automatically cancelled, terminated and of no further force or effect, without further act or action under any applicable agreement, law, regulation, order, or rule and the obligations of the Debtors, the TruPS Indenture Trustees, and the TruPS Institutional Trustees, as applicable, thereunder shall be deemed terminated; provided, however, that TruPS Indentures and TruPS Declarations shall continue in effect only as follows: (1) for the TruPS Indenture Trustees and the TruPS Institutional Trustees to discharge any responsibilities they have under the Plan, the Confirmation Order, the Distribution Trust Agreement, the TruPS Indentures, the TruPS Parent Guarantees, the TruPS Sponsor Guarantees and the TruPS Declarations in connection with Distributions to be made to the Holders of the TruPS Debentures, the Beneficial Holders of TruPS and SRGL in accordance with the terms of the Plan (including Sections 4.3(a), and 4.3(c) thereof), the Confirmation Order, the Distribution Trust Agreement, the TruPS Indentures, the TruPS Declarations and, as to SRGL only, the Restructuring Implementation Agreement and RIA Order, it being understood that the TruPS Indentures, the TruPS Parent Guarantees, the TruPS Sponsor Guarantees and the TruPS Declarations shall continue in effect only so long as is necessary to permit such Distributions to be made; (2) subject to Section 4.1(d)(iii) hereof, to allow each TruPS Indenture Trustee and any predecessor trustee under any of the TruPS Indentures to exercise its Charging Lien for the payment of its fees and expenses incurred post-Closing (if any) and for indemnification as provided in the applicable TruPS Indentures solely to the extent requested by the Debtors, the Reorganized Debtors or the Distribution Trustee; (3) to preserve any rights of the TruPS Indenture Trustees pursuant to any direction provided by Holders of the TruPS pursuant to the terms of the TruPS documents; and (4) solely with respect to the SFLST I TruPS Indenture, the SFLST I TruPS Debentures, the SFLST I Trust Declaration, the SFLST I TruPS Sponsor Guarantee, and any other SFLST I TruPS Documents (except the SFLST I TruPS Parent Guarantee), the foregoing SFLST I TruPS Documents shall not be deemed cancelled, terminated or of no force or effect as against SFL.

For the avoidance of doubt, nothing in this <u>Section 6.7</u> is intended to or shall extinguish or impair any liability or obligation of SFL under any SFLST I TruPS Document.

The TruPS Indenture Trustees, in their capacity as such, shall have no liability in connection with Distributions made in accordance with the terms of the Plan and shall have no obligation to take any action in connection with such Distributions or otherwise; provided, however, that for so long as the SFLST I TruPS Indenture, the SFLST I TruPS Debentures, the SFLST I Trust Declaration, the SFLST I TruPS Sponsor Guarantee, and any other SFLST I TruPS Documents (except the SFLST I TruPS Parent Guarantee) remain in effect, the SFLST I TruPS Indenture Trustee shall only have such liability and obligations as are expressly set forth therein. To the extent that, after the Effective Date, (x) the Debtors, the Reorganized Debtors or the Distribution Trustee requests services from the TruPS Indenture Trustees in connection with Distributions or otherwise, or (v) material action by the TruPS Indenture Trustee is reasonably necessitated in response to actions taken by a Holder of an Allowed TruPS Claim, a TruPS Indenture Trustee's reasonable and documented costs and out-of-pocket expenses, including professional fees, for such post-Effective Date services of the TruPS Indenture Trustees shall be paid by the Distribution Trust from the Post-Effective Date Indenture Trustee Fee Reserve of \$50,000 and such invoices for such Indenture Trustee Fees shall be subject to the review procedures set forth in section 4.1(d). Such payment shall be in lieu of the TruPS Indenture Trustee's ability to effectuate its Charging Liens under the TruPS Indentures. The \$50,000 in aggregate to be held in the Post-Effective Date Indenture Trustee Fee Reserve for payment of post-Effective Date Indenture Trustee Fees shall be held by the Distribution Trustee for a period of one (1) year after the Effective Date and any remainder of such funds will be released thereafter from the Post-Effective Date Indenture Trustee Fee Reserve ratably to the Holders of Allowed TruPS Claims. For the avoidance of doubt, the Distribution Trustee shall be under no obligation to satisfy or reserve any amount on account of post-Effective Date Indenture Trustee Fees that may be incurred by the TruPS Indenture Trustees for the SHST II Debentures or the GPIC Debentures.

## 6.8. Comprehensive Settlement of Claims and Controversies

Pursuant to sections 363 and 1123 of the Bankruptcy Code and Bankruptcy Rule 9019, and in consideration for the distributions and other benefits provided pursuant to the Plan, the provisions of the Plan shall constitute a good faith compromise of all Claims, Interests and controversies relating to the contractual, legal, and subordination rights that a Holder of a Claim may have with respect to any Allowed Claim, or any distribution to be made on account of such Allowed Claim, as well as the allocation of the Plan Funding Payment among the Debtors' Estates and Creditors. The entry of the Confirmation Order shall constitute the Bankruptcy Court's approval of the compromise or settlement of all such Claims, Interests, and controversies, as well as a finding by the Court that such compromise or settlement is in the best interests of the Debtors, their Estates, and Holders of Claims and Interests and is fair, equitable, and reasonable. In accordance with the provisions of the Plan, pursuant to Bankruptcy Rule 9019, without any further notice to or action, order, or approval of the Court, after the Effective Date, the Reorganized Debtors or the Distribution Trustee, as applicable, may compromise and settle

Claims against, and Interests in, the Debtors and their Estates and Causes of Action against other entities.

# VII. CLAIMS RESOLUTION & DISTRIBUTIONS

## 7.1. Right to Object to Claims

The Distribution Trustee shall have the authority, but not the obligation, to object to, litigate, and settle, the amount, priority or the extent of any Administrative Claim, Secured Claim, Priority Claim, SHI General Unsecured Claim, SALIC General Unsecured Claim, any SFL Claims (if not settled on or before the Effective Date) or Subordinated Claim (including, with respect to any other of the foregoing, to argue that such Claim constitutes a Subordinated Claim). Notwithstanding anything to the contrary herein, subject to the terms and conditions set forth in the Distribution Trust Agreement, and notwithstanding any requirements that may be imposed pursuant to Bankruptcy Rule 9019, except insofar as a Claim is Allowed under the Plan on and after the Effective Date, the Distribution Trustee shall have the authority, but not the obligation, to: (1) file, withdraw or litigate to judgment objections to and requests for estimation of Claims; (2) settle or compromise any Disputed Claim without any further notice to or action, order or approval by the Bankruptcy Court; and (3) administer and adjust the Claims register to reflect any such settlements or compromises without any further notice to or action, order or approval by the Bankruptcy Court. The Distribution Trustee shall succeed to any pending objections to Claims filed by the Debtors prior to the Effective Date, and, at the Distribution Trustee's election, any other pending objections to Claims filed by any other party, and shall have and retain any and all rights and defenses the Debtors had immediately prior to the Effective Date with respect to any Disputed Claim, including pursuant to the DT Post-Closing Rights. The Reorganized Debtors shall provide commercially reasonable assistance and cooperation to the Distribution Trustee in connection with the Distribution Trustee's prosecution of objections to Claims, including, without limitation, access to the books and records of the Debtors or the Reorganized Debtors (as the case may be) and other information reasonably requested by the Distribution Trustee to enable the Distribution Trustee to perform its obligations under the Distribution Trust Agreement, including pursuant to the DT Post-Closing Rights.

# 7.2. Deadline for Objecting to Claims

Objections to Claims must be Filed with the Bankruptcy Court, and a copy of the objection must be served on the subject Creditor, before the expiration of the Claims Objection Deadline (unless such period is further extended by subsequent orders of the Bankruptcy Court); otherwise such Claims shall be deemed Allowed in accordance with section 502 of the Bankruptcy Code. The objection shall notify the Creditor of the deadline for responding to such objection.

# 7.3. Deadline for Responding to Claim Objections

Within twenty-one (21) days after service of an objection, or such other date as is indicated on such objection or the accompanying notice thereof, the Creditor whose Claim was
objected to must File a written response to the objection with the Bankruptcy Court and serve a copy on the Distribution Trustee. Failure to <u>fileFile</u> a written response within such time period shall constitute a waiver and release of that portion of the subject Claim that was subject to the objection, and shall constitute cause for the Bankruptcy Court to enter a default judgment against the non-responding Creditor or grant the relief requested in the Claim objection.

## 7.4. Right to Request Estimation of Claims

Pursuant to section 502(c) of the Bankruptcy Code, the Debtors, the Reorganized Debtors, and the Distribution Trustee may request estimation or liquidation of any Disputed Claim that is contingent or unliquidated or any Disputed Claim arising from a right to an equitable remedy or breach of performance.

## 7.5. Distribution Procedures Regarding Allowed Claims

## (a) In General

Other than Closing Date Plan Distributions and initial distributionsDistributions of New Equity to New Equity Eligible Holders that make the New Equity Election, the Distribution Trustee shall make all Distributions required to be made under the Plan, including Distributions from the Distribution Trust. Each Creditor or Holder receiving any Distribution from the Distribution Trust shall be deemed to have ratified and become bound by the terms and conditions of the Distribution Trust Agreement.

## (b) Distributions on Allowed Claims Only

Distributions from the Distribution Trust shall be made only to the Holders of Allowed Claims or in the case of Holders of Allowed Claims in Classes 4 and 6, the Beneficial Holders thereof. Until a Disputed Claim becomes an Allowed Claim, the Holder of that Disputed Claim shall not receive a Distribution. Other than as specifically set forth in section 6.1(b)(i), Allowed Claims shall not be entitled to distributions from any source other than the Plan Funding Payment or the Distribution Trust.

## (c) Method of Distributions

i. Use of Distribution Agent. The Reorganized Debtors with respect to Closing Date Plan Distributions and initial Distributions of Offered New Equity and the Distribution Trustee with respect to Distributions from the Distribution Trust shall have the authority, in their respective sole discretion, to enter into agreements with a third-party Distribution Agent to facilitate the Distributions required hereunder. For the avoidance of doubt, Prime Clerk, LLC, which was previously engaged to serve as the Debtors' Voting Agent, is an acceptable choice to serve as the Distribution Agent and shall be engaged as the shared Distribution Agent in the event that the Reorganized Debtors and Distribution Trustee both want to use the services of a Distribution Agent and cannot agree to an alternate choice. The Distribution Trustee shall be authorized, but not directed, to pay to any third-party Distribution Agent all reasonable and documented fees and expenses of such Distribution Agent without the need for any approvals, authorizations, actions, or consents. The Distribution Agent shall be authorized, but not directed, to submit detailed invoices to the Debtors or the Reorganized Debtors, as applicable, for all fees and expenses for which the Distribution Agent seeks reimbursement and the Distribution Trustee shall pay those amounts from the Trust Administration Reserve that it, in its sole discretion, deems reasonable, and shall object in writing to those fees and expenses, if any, that the Distribution Trustee deems to be unreasonable. In the event that the Distribution Agent's invoice, the Distribution Trustee and such Distribution Agent shall endeavor, in good faith, to reach mutual agreement on the amount of the appropriate payment of such disputed fees or expenses. In the event that the Distribution Agent are unable to resolve any differences regarding disputed fees or expenses, either party shall be authorized to move to have such dispute heard by the Bankruptcy Court.

ii. *Cash Distributions.* Except as otherwise specified in the Plan or the Distribution Trust Agreement (with respect to Distributions other than Closing Date Plan Distributions), any Distribution of Cash made by the Reorganized Debtors as a Closing Date Plan Distribution or made by the Distribution Trustee from the Available Plan Funding Distribution Amount shall be made by mailing such Distribution to the Creditor or Holder, as applicable, at the address listed in any Proof of Claim filedFiled by such Entity or at such other address as such Entity shall have specified for payment purposes in a written notice received by the Reorganized Debtors or Distribution Trustee, as applicable, at least twenty-one (21) days before a Distribution Date. If a Creditor or Holder has not filedFiled a Proof of Claim or sent the Distribution Trustee a written notice of payment address, then the Distribution(s) for such Entity will be mailed to the address identified in the Schedules. Notwithstanding the foregoing, any Cash may be distributed by wire, check, or such other method as the Distribution Trustee or Reorganized Debtors, as applicable, deem appropriate under the circumstances. AnAny Cash Distribution to be made to SRGL under the Plan shall be made by wire transfer unless otherwise agreed by SRGL.

iii. Offered New Equity Distributions. On or as soon as practicable after the Effective Date, the Reorganized Debtors (or the New HoldCo (as applicable)) shall distribute to the New Equity Eligible Holders that have made the New Equity Election a Distribution of each such holder's TruPS Claims Equity Distribution Amount. For the avoidance of doubt, such Distribution shall not be certificated, and will rather be in book-entry form.

iv. *Tax Information Required for Distributions*. Before receiving any Distributions, all Creditors and Holders, at the request of the Reorganized Debtors or Distribution Trustee, as applicable, must provide written notification of their respective Federal Tax Identification Numbers or Social Security Numbers to the requesting Entity; otherwise, the Reorganized Debtors or Distribution Trustee, as applicable, may suspend Distributions to any Creditors or Holders who have not provided their Federal Tax Identification Numbers or Social Security Numbers.

v. Notwithstanding the foregoing, the Debtors may cause there to be

established a register on which there shall be recorded the Beneficial Holders of the Claims in Class 4 and Class 6, and the New Equity Election made by any of them. The Reorganized Debtors or the Distribution Trustee, as applicable, shall make Distributions of TruPS/GUC Claims Cash Distribution Amount, the TruPS Claims Equity Distribution Amounts and/or the Distribution Trust Asset Proceeds, as the case may be, to such Beneficial Holders, or their transferees, as recorded on such register in accordance with such transfer procedures as the Debtors may establish. The provisions of Section 7.5(c)(iv) shall similarly apply to Distributions to such Beneficial Holders or their transferees

#### (d) Undeliverable Distributions

If a Distribution is returned as undeliverable, the Distribution Trustee shall use reasonable efforts to determine such Creditor's or Holder's then current address. If the Distribution Trustee cannot determine, or is not notified of, a Creditor's or Holder's then current address within six (6) months after the Effective Date, the Distribution reserved for such Creditor or Holder shall be deemed an unclaimed Distribution, and <u>Section 7.5(e)</u> of the Plan shall be applicable thereto.

#### (e) Unclaimed Distributions

If the current address for a Creditor or Holder entitled to a Distribution (whether in the form of Cash or Offered New Equity) under the Plan has not been determined within six (6) months after the Effective Date or such Entity has otherwise not been located, or if such Entity has not submitted a valid Federal Tax Identification Number or Social Security Number to the Distribution Trustee within six (6) months after the Effective Date, then such Creditor or Holder, as applicable, (i) shall no longer be a Creditor or Holder and (ii) shall be deemed to have released such Claim and Interest, if any. If such <u>Unclaimedunclaimed</u> Distribution consists of Cash, then the Cash shall remain property of the Distribution Trust and be used or distributed in accordance with the terms of this Plan and the Distribution Trust Agreement. If such <u>Unclaimedunclaimed</u> Distribution consists of New Equity, then such New Equity shall not be issued to the forfeiting Claim Holder and shall instead be issued to the <u>PurchaserHildene Master</u> <u>Funds</u>.

#### (f) Taxes; Withholding

In connection with the Plan, any party issuing any instrument or making any Distribution described in the Plan shall comply with all applicable withholding and reporting requirements imposed by any federal, state, or local taxing authority, and all Distributions and all related agreements shall be subject to any such withholding or reporting requirements. Notwithstanding the foregoing, each Holder of an Allowed Claim or any other Person that receives a Distribution shall have responsibility for any taxes imposed by any Governmental Unit, including, without limitation, income, withholding, and other taxes, on account of such Distribution. Any party issuing any instrument or making any Distribution has the right, but not the obligation, to not make a Distribution until such Holder has made arrangements satisfactory to such issuing or disbursing party for payment of any such tax obligations. The Distribution Trustee, in the exercise of its sole discretion and judgment, may enter into agreements with taxing or other authorities for the payment of such amounts that may be withheld in accordance with the provisions of this section. Any party entitled to receive any property as an issuance or distribution under the Plan shall, upon request, by the Debtors or Distribution Trustee, as applicable, provide an appropriate Form W-9 or (if the payee is a foreign Person, as applicable) Form W-8. If such request is made and such party fails to comply before the date that is 180 days after the request is made, the amount of such Distribution shall irrevocably revert to the Distribution Trust, and any Claim in respect of such Distribution shall be discharged and forever barred from assertion against the Debtors, the Reorganized Debtors, the Distribution Trust and their respective property.

#### (g) Special Distribution Provisions for TruPS Claims

#### i. Service of TruPS Indenture Trustees; In General.

Except with respect to SRGL Exclusively Held TruPS Claims, Distributions on account of the TruPS Claims shall be made by the Distribution Trustee to (i)directly to each beneficial holder of Allowed TruPS Claims in accordance with the Distribution Trust Agreement, rather than to the applicable TruPS Indenture Trustee or (ii) with the prior written consent of such TruPS Indenture Trustee, through the facilities of a Securities Depository or Custodian, by means of book entry exchange through the facilities of a Securities Depository or Custodian in accordance with the customary practices of such Securities Depository or Custodian, as applicable. If a Distribution is made to the TruPS Indenture Trustee, the TruPS Indenture Trustee, in its capacity as a disbursing agent, shall administer the Distributions in accordance with the terms of this Plan, the relevant TruPS Indenture, the relevant TruPS Declaration and any other applicable TruPS Documents.Security Depository or Custodian. In the event that a beneficial holder of Allowed TruPS Claims is unknown to the Debtors, the Reorganized Debtors or the Distribution Trustee (as applicable), then any Distribution for such unknown beneficial holder shall be made to the applicable Security Depository or Custodian.

#### ii. Distributions Relating to SRGL Exclusively Held TruPS

Notwithstanding anything to the contrary in this Plan, Distributions on account of the SRGL Exclusively Held TruPS Claims shall be made by the Distribution Trustee directly to SRGL.

#### iii. Distributions Arising From SALIC Claims Against SRGL

Upon any distribution from the SRGL estate to the Distribution Trust in respect of the SALIC Claims, the Joint Liquidators shall apply such provisioning or withholding (if any) as they deem appropriate in accordance with their duties under Cayman Islands law.

All distributions from the SRGL estate to the Distribution Trust in respect of the SALIC Claims shall be distributed pro rata to all Holders of Allowed Claims in Classes 4, 5, 6 and 7, except for SRGL.

Notwithstanding anything contained herein, (i) any funds received by SRGL from the Distribution Trust and any distributions by SRGL on account of such funds shall be governed by, and construed in accordance with, the laws of the Cayman Islands and (ii) any funds received by the Debtors or the Distribution Trust from SRGL and any <u>distributionDistribution</u> from the Distribution Trust or the Debtors on account of such funds shall be subject to the terms of the Plan and the jurisdiction of the Bankruptcy Court.

## (h) Additional Procedures Regarding Distributions from the Distribution Trust

Additional procedures regarding Distributions from the Distribution Trust to Holders of Allowed Claims shall be governed by the Distribution Trust Agreement.

# (i) Allocation of Distributions between Principal and Interest

Except as otherwise provided in the Plan, to the extent that any Claim entitled to a Distribution under the Plan is comprised of indebtedness and accrued but unpaid interest thereon, such Distribution shall be allocated first to the principal amount (as determined for U.S. federal income tax purposes) of the Claim, and then to accrued but unpaid interest.

## 7.6. Implementation of New Equity and Cash Elections

The Plan affords New Equity Eligible Holders the opportunity to make the New Equity Election or the Cash Election and with respect to the Holder of the SFL Note Claim, deems a New Equity Election and/or Cash Election to have been made under certain circumstances.

## (a) Making of New Equity or Cash Election by New Equity Eligible Beneficial Holders

Each New Equity Eligible Beneficial Holder must elect to take its entire Distribution (other than for its applicable percentage of the Distribution Trust <u>AssetsAsset</u> Proceeds) either exclusively in the form of Offered New Equity (by making the New Equity Election) or exclusively in the form of Cash (by making the Cash Election). For the avoidance of doubt, a New Equity Eligible Beneficial Holder may not split its Distribution between New Equity and Cash.

A New Equity Election will be recognized as valid only if the electing New Equity Eligible Holder checks the box for the "New Equity Election" on its Ballot and such Ballot is otherwise properly completed and timely received by the Voting Agent in accordance with the requirements of the Disclosure Statement Order. Any New Equity <u>electionElection</u> that is not properly made as set forth herein shall be disregarded and such New Equity Eligible Holder shall be deemed to have made the Cash Election. Except as otherwise agreed in writing by (a) if prior to the Effective Date, by the Debtors, the Official Committee and the Purchaser or (b) if on or after the Effective Date, the Distribution Trustee and the Reorganized Debtors, all

New Equity Elections and Cash Elections (including any deemed Cash Elections) will be final and irrevocable after the Voting Deadline.

## (b) Implementation of New Equity or Cash Election for SFL Note Claim

Notwithstanding the foregoing, with respect to Distributions to be made under the Plan on account of the Allowed SFL Note Claim:

i. If the SFL Note Claim Allowance Conditions are deemed satisfied in accordance with  $\underline{\text{Section 4.3(c)(ii)(B)(1)}}$  of this Plan because the SFLST I TruPS CDO Facility Holders each have voted their respective Allocated Portions of SALIC TruPS Claims to accept the Plan, then the Holder of the Allowed SFL Note Claim shall be deemed to have made the New Equity Election and Cash Election in the manner described in  $\underline{\text{Section 4.3(c)(iii)(C)(1)(a)}}$  of this Plan, subject to the SFL Receiver's right to make the SFL Bankruptcy Administration Senior Cash Election.

ii. If one or more of the SFLST I TruPS CDO Facility Holders has not voted its Allocated Portion of SALIC TruPS Claims to accept the Plan, but the Holder of the SFL Note Claim has otherwise satisfied the SFL Note Claim Allowance Conditions in accordance with Section 4.3(c)(ii)(B)(42)(a) of this Plan, then the Holder of the Allowed SFL Note Claim shall be entitled to: (a) apply its Allowed SFL Note Claim amount entirely to the New Equity Election; (b) apply its Allowed SFL Note Claim amount entirely to the Cash Election; or (c) allocate its Allowed SFL Note Claim Amount between the New Equity Election and the Cash Election.

iii. If the SFL Note Claim Allowance Conditions are not satisfied or deemed satisfied in accordance with this Plan, then the Holder of the SFL Note Claim shall be deemed to have made the Cash Election with respect to the entire SFL Note Claim, subject to the SFL Note Claim becoming an Allowed Class 6 Claim.

## 7.7. Claims Paid or Payable by Third Parties

## (a) Claims Paid by Third Parties

To the extent a Holder has received a Distribution on account of a Claim and also receives payment from a party that is not a Debtor or the Distribution Trustee on account of such Claim, such Holder shall, within 30 calendar days of receipt thereof, repay and/or return the Distribution to the Distribution Trustee to the extent the recipient-Holder's total recovery on account of such Claim from the third party and under this Plan exceeds the amount of the Holder's Allowed Claim as of the date of any such distributionDistribution under this Plan.

Any such Claim shall be expunded from the official claims register without a claims objection having to be filed and without any further notice to or action, order, or approval of the Bankruptcy Court, to the extent that the Holder receives payment in full on account of such Claim; *provided, however,* that to the extent the non-Debtor party making the payment is

subrogated to such Holder's Claim, the non-Debtor party shall have a 30-calendar-day grace period to notify the Distribution Trustee of such subrogation rights and, if they are valid and enforceable, the expungement will be reversed to the extent of such subrogation rights.

## (b) Claims Payable by Third Parties

To the extent that one or more of the Debtors' insurers agrees to satisfy a Claim (if and to the extent adjudicated by a court of competent jurisdiction), then immediately upon such insurers' agreement, such Claim may be expunged on the official claims register (in each case to the extent of any agreed-upon satisfaction) by the Clerk of Court or Distribution Trustee, as applicable, without a claims objection having to be filed and without any further notice to or action, order, or approval of the Bankruptcy Court.

## 7.8. Foreign Currency Exchange Rate

As of the Effective Date, any Claim asserted in a currency other than U.S. dollars shall be automatically deemed converted to the equivalent U.S. dollar value using the exchange rate on the Petition Date, as quoted at 4:00 p.m., mid-range spot rate of exchange for the applicable currency as published in The Wall Street Journal, Eastern Edition, on the day after the Petition Date.

## 7.9. Setoff

Except as otherwise provided in this Plan, the Restructuring Implementation Agreement, the RIA Order or another Final Order of the Bankruptcy Court, (a) nothing contained in this Plan shall constitute a waiver or release by the Debtors, the Reorganized Debtors, the Distribution Trustee or the Distribution Trust of any right of setoff or recoupment that any of the foregoing Entities may have against any Entity, and (b) to the extent permitted by Applicable Law, the Distribution Trustee or Reorganized Debtors, as applicable, may setoff or recoup (but shall not be required to do so) against any Claim (and any Interest) and the payments or other Distributions to be made under the Plan in respect of such Claim (or Interest), claims of any nature whatsoever that the Debtors may have against the Holder of such Claim or Interest.

### 7.10. De Minimis Distributions

If any interim Distribution under the Plan to the <u>holderHolder</u> of an Allowed Claim or Interest would be less than \$100.00 or a fractional number of Offered New Equity, the Distribution Trustee or Reorganized Debtors, as applicable, may withhold such Distribution until the next Subsequent Distribution Date or the date of a final Distribution, as applicable, is made to such Holder. If any final Distribution under the Plan to the <u>holderHolder</u> of an Allowed Claim or Interest would be less than \$25.00 or a fractional number of Offered New Equity, then such Distribution may be canceled in its entirety. Any unclaimed Distributions pursuant to this <u>Section 7.9</u> shall be treated as an <u>Unclaimedunclaimed</u> Distribution under <u>Section 7.5(e)</u> of the Plan.

#### 7.11. Fractional Shares

No fractional shares or number of the Offered New Equity shall be issued or distributed under the Plan. The actual Distribution of shares or number of the Offered New Equity shall be rounded to the next higher or lower whole number as follows: (i) fractions less than one-half ( $\frac{1}{2}$ ) shall be rounded to the next lower whole number and (ii) fractions equal to or greater than one-half ( $\frac{1}{2}$ ) shall be rounded to the next higher whole number. The total amount of shares or number of Offered New Equity to be distributed hereunder shall be adjusted as necessary to account for such rounding. No consideration shall be provided in lieu of fractional shares or numbers that are rounded down.

#### 7.12. No Interest

Other than as provided by section 506(b) of the Bankruptcy Code or as specifically provided for in the Plan, the Confirmation Order, or another Final Order of the Bankruptcy Court, postpetition interest shall not accrue or be paid on Claims and no Holder of a Claim shall be entitled to interest accruing on or after the Petition Date on any Claim or right. Additionally, and without limiting the foregoing, interest shall not accrue or be paid on any Claim or Disputed Claim with respect to the period from and after the Effective Date; provided, however, that nothing in this Section 7.117.12 shall limit any rights of any governmental unit (as defined in section 101(27) of the Bankruptcy Code) to interest under sections 503, 506(b), 1129(a)(9)(A), or 1129(a)(9)(C) of the Bankruptcy Code or as otherwise provided for under Applicable Law.

### VIII. EXECUTORY CONTRACTS AND UNEXPIRED LEASES

#### 8.1. Assumption of Executory Contracts

Unless otherwise specified herein, as of, and subject to, the occurrence of the Effective Date and the payment of any applicable Cure Amounts, all Executory Contracts and Unexpired Leases to which either Debtor is a party, and which have not expired by their own terms on or prior to the Confirmation Date shall be deemed assumed except for any executory contract or unexpired lease Executory Contract or Unexpired Lease that (a) previously has been assumed or rejected pursuant to a Final Order of the Bankruptcy Court, (b) is the subject of a separate motion for assumption or rejection under section 365 of the Bankruptcy Code filed by a Debtor before the Confirmation Date, (c) is designated as a contract or lease to be rejected on the Rejection Schedule, or (d) is the subject of a pending Assumption Dispute. Subject to the occurrence of the Effective Date, entry of the Confirmation Order by the Bankruptcy Court shall constitute approval of the assumptions and rejections provided for in the Plan pursuant to sections 365(a) and 1123 of the Bankruptcy Code. Each Executory Contract and Unexpired Lease assumed pursuant to the Plan shall vest in, and be fully enforceable by, the Reorganized Debtors in accordance with its terms, except as modified by the provisions of the Plan, and any order of the Bankruptcy Court authorizing and providing for its assumption or applicable lawApplicable Law.

For the avoidance of doubt, unless otherwise expressly provided in the Plan, the Plan Supplement or the Confirmation Order, all SRUS Retrocession Agreements, Third-Party Reinsurance Agreements, Trust Agreements, SALIC Net Worth Maintenance Agreements, and any and all other reinsurance treaties and trust agreements shall be deemed assumed by SALIC.

#### 8.2. Rejection of Executory Contracts

On the Effective Date, all Executory Contracts identified on the Rejection Schedule shall be deemed rejected. The initial Rejection Schedule shall be filed with, and as a part of, the Plan Supplement, and, except as otherwise provided in the Stock Purchase Agreement, may be amended by the Purchaser at any time prior to the Effective Date. Entry of the Confirmation Order shall constitute, subject to the occurrence of the Effective Date, approval of the rejection of such Executory Contracts under sections 365 and 1123 of the Bankruptcy Code.

### 8.3. Determination of Assumption Disputes and Deemed Consent

(a) On or before the date that is at least twenty-one (21) calendar days prior to the Confirmation Objection Deadline (as defined in the Disclosure Statement Order), the Debtors shall fileFile and serve on parties to Executory Contracts and Unexpired Leases to be assumed a notice reflecting the Debtors' intention to assume the contract or lease in connection with the Plan and setting forth the proposed Cure Amount (if any). Such notice shall not prejudice any of the defenses and counterclaims the Debtors, Reorganized Debtors or Distribution Trust may have with respect to any such Executory Contract or Unexpired Lease. If the counterparty believes any different Cure Amount is due in connection with the assumption or otherwise objects to the assumption, it must assert such Cure Amount or any other Assumption Dispute by filingFiling an objection with the Bankruptcy Court and serving such objection on counsel for the notice parties identified in Section 12.11 hereof so as to be received by such parties by no later than twenty-one (21) calendar days after the date of service of such notice.

(b) Cure Amount disputes shall be resolved by the Debtors or Reorganized Debtors and the applicable counterparty in the ordinary course, and the agreed-upon amounts shall be paid by the Debtors or Reorganized Debtors in the ordinary course, with any such payments of Cure Amounts to be funded in accordance with Section 2.4(d) of the Stock Purchase Agreement. If there is an Assumption Dispute pertaining to assumption of an Executory Contract or Unexpired Lease, such dispute shall be heard by the Bankruptcy Court prior to such assumption being effective; *provided, however*, the Debtors or the Reorganized Debtors, as applicable, may settle any dispute regarding the Cure Amount or the Assumption Dispute without any further notice to any party or any action, order, or approval of the Bankruptcy Court. To the extent the Assumption Dispute is resolved or determined by a Final Order unfavorable to the Debtors or Reorganized Debtors, as applicable, shall have fourteen (14) calendar days following entry of such Final Order to fileFile a motion to reject such contract or lease.

Any counterparty to an Executory Contract or Unexpired Lease that fails (c) to object timely to the notice of the proposed assumption of such contract or lease within twenty-one (21) calendar days after the date of service thereof shall be deemed to have assented to (i) the proposed Cure Amount, and (ii) the assumption of the applicable contract or lease notwithstanding any provision thereof that purports to (A) prohibit, restrict, or condition the transfer or assignment of such contract or lease, (B) terminate or modify, or permit the termination or modification of, a contract or lease as a result of any direct or indirect transfer or assignment of the rights of a Debtor under such contract or lease or a change, if any, in the ownership or control of a Debtor, (C) increase, accelerate, or otherwise alter any obligations or liabilities of a Debtor under such Executory Contract or Unexpired Lease, or (D) create or impose a Lien upon any property or asset of a Debtor or a Reorganized Debtor. Each such provision shall be deemed to not apply to the assumption of such Executory Contract or Unexpired Lease pursuant to the Plan and counterparties to assumed Executory Contracts or Unexpired Leases that fail to object to the proposed assumption in accordance with the terms set forth in this Section 8.3(c) shall forever be barred and enjoined from objecting to the validity of such assumption or to the Cure Amount, and from taking any other action prohibited by the foregoing on account of transactions contemplated by the Plan.

#### 8.4. Effect of Assumption of Contracts and Leases

Subject to resolution of any Assumption Dispute, all Cure Amounts shall be satisfied by the Purchaser (subject to the terms of Section 2.4(d) of the Stock Purchase Agreement) or the Reorganized Debtors, as the case may be, upon assumption of the applicable **contracts**<u>Executory Contracts</u> and <u>unexpired leases</u><u>Unexpired Leases</u> in the ordinary course. Assumption of any Executory Contract or Unexpired Lease pursuant to the Plan, or otherwise, subject to satisfaction of the Cure Amount, shall result in the full release and satisfaction of any Claims or defaults, whether monetary or nonmonetary, including defaults of provisions restricting the change in control or ownership interest composition or other bankruptcy-related defaults, arising under any assumed Executory Contract or Unexpired Lease at any time before the effective date of the assumption. On the Effective Date (or with respect to any assumption of an Executory Contract or Unexpired Lease that becomes effective), any proofs of Claim filedFiled with respect to an Executory Contract or Unexpired Lease that has been assumed shall be deemed withdrawn without further notice to, action, order, or approval of the Bankruptcy Court or any other Entity.

#### 8.5. Payment of Cure Amounts

Within thirty (30) days after the Effective Date, the Reorganized Debtors shall pay, in Cash (or as otherwise agreed or ordered by the Bankruptcy Court), all Cure Amounts related to Executory Contracts and Unexpired Leases that are assumed pursuant to this <u>Article VIII of the Plan</u>, other than Cure Amounts that are subject to an Assumption Dispute; *provided*, that the Reorganized Debtors shall pay all Cure Amounts that are subject to an Assumption <u>ObjectionDispute</u> on the Effective Date within twenty-one (21) days after entry of an order by the Bankruptcy Court resolving the Assumption Dispute or approving an agreement between the

parties concerning the Cure Amount. For the avoidance of doubt, funding of Cure Amounts shall be subject to sections 2.2(b), 2.3(f) and 2.4(e) of the Stock Purchase Agreement; in particular, the amount contributed by the Purchaser for payment of the Cure Amounts shall not exceed \$100,000 and the Recapitalization Funding Payment shall be used by the Reorganized Debtors to pay any amounts in respect of the Cure Amounts in excess of \$100,000.

## (a) No Admission of Liability

Neither the inclusion nor exclusion of any Executory Contract by the Debtors on the Assumption Schedule, nor anything contained in the Plan, shall constitute an admission by the Debtors that any such contract or lease is in fact an Executory Contract or that the Debtors, the Reorganized Debtors, the Purchaser or Distribution Trust has any liability thereunder.

## (b) Reservation of Rights

Nothing in the Plan shall waive, excuse, limit, diminish, or otherwise alter any of the defenses, claims, causes of action, or other rights of the Debtors, the Reorganized Debtors or Distribution Trust under any executory or non-executory contract or any unexpired or expired lease, nor shall any provision of the Plan increase, augment, or add to any of the duties, obligations, responsibilities, or liabilities of the Debtors or the Reorganized Debtors under any such contract or lease.

## (c) Rejection Claim Bar Date

Each Claim resulting from the rejection of an Executory Contract pursuant to <u>Section 8.2</u> of the Plan shall be Filed with the Bankruptcy Court no later than the Rejection Claim Bar Date. Any Claim resulting from the rejection of an Executory Contract not Filed by the applicable deadline shall be discharged and forever barred, and shall not be entitled to any Distributions under the Plan. The Distribution Trustee shall have the right to object to any rejection damage Claim. All rejection damage Claims shall be treated in Class 5 or 7, respectively, and shall be paid out of the Distribution Trust.

## (d) Continuing Obligations Owed to the Debtors

Any continuing obligations of third parties to the Debtors under insurance policies, contracts, or leases that have otherwise ceased to be executory or have otherwise expired on or prior to the Effective Date, including, without limitation, continuing obligations to pay insured claims, to defend against and process claims, to refund premiums or overpayments, to provide indemnification, contribution or reimbursement, to grant rights of first refusal, to maintain confidentiality, or to honor releases, will continue and will be binding on such third parties, notwithstanding any provision to the contrary herein, unless otherwise specifically terminated by the Debtors or the Reorganized Debtors, or by order of the Bankruptcy Court.

To the extent any insurance policy under which the insurer has a continuing obligation to pay the Debtors or a third party on behalf of the Debtors is held by the Bankruptcy Court to be an Executory Contract, such insurance policy will be treated as though it is an Executory Contract that is assumed by the Reorganized Debtors pursuant to section 365 of the Bankruptcy Code and <u>Sections 8.1</u> and <u>8.3</u> of the Plan.

## (e) **Postpetition Contracts**

The Debtors will not be required to assume or reject any contract or lease entered into by the Debtors after the Petition Date. Any such contract or lease will continue in effect in accordance with its terms after the Effective Date, unless the Reorganized Debtors have obtained a Final Order of the Bankruptcy Court approving rejection of such contract or lease. Contracts or leases entered into after the Petition Date will be performed by the Reorganized Debtors in the ordinary course of their business.

## IX. CONDITIONS PRECEDENT TO CONFIRMATION OF THE PLAN AND OCCURRENCE OF THE EFFECTIVE DATE

## 9.1. Conditions to Confirmation

The Confirmation Order will not be effective unless the final version of the Plan, Plan Supplement, and any other documents, or schedules thereto, including the <u>filedFiled</u> Confirmation Order, shall have been <u>filedFiled</u> in form and substance acceptable to the Purchaser in its reasonable discretion, and the Restructuring Implementation Agreement shall not have been terminated and shall be in full force and effect.

## 9.2. Conditions to Effectiveness

Unless the following conditions (except with respect to the Distribution Trust Agreement and the Restructuring Implementation Agreement) are waived by the Purchaser, the Plan will not be effective unless: (a) the conditions to Confirmation above have either been satisfied, or (except with respect to the Restructuring Implementation Agreement) waived by the Purchaser; (b) the Confirmation Order has been entered by the Bankruptcy Court, is not subject to appeal, and no stay or injunction is in effect with respect thereto; (c) the Closing shall have occurred or shall occur simultaneously with the Effective Date; (d) the PurchaserHildene Master Funds shall acquire the New Equity (subject to the New Equity Election), directly or indirectly, free and clear of all Liens, Claims, and Interests and in accordance with the Plan; (e) the Distribution Trust Agreement shall have been executed by all parties thereto; (f) the Restructuring Implementation Agreement shall not have been terminated and shall be in full force and effect; and (g) the Purchaser shall have demonstrated to the reasonable satisfaction of the Debtors and the Official Committee that all actions have occurred or will occur on or before the Effective Date necessary to fund the Plan Funding Payment to the Distribution Trust and the Recapitalization Funding Payment to Reorganized SALIC, each as provided in the Plan and the Stock Purchase Agreement; and (h) all governmental, judicial, and third party approvals and consents that are required in connection with the transactions contemplated by the Plan, including regulatory approvals of the transaction in Delaware by the Delaware DOI in accordance with Chapter 101 of Title 29 and Chapter 50 of Title 18 of the Delaware Code Annotated, the

Cayman Islands, Bermuda and Ireland, shall have been obtained, not subject to unfulfilled conditions, and shall be in full force and effect.

### X. SETTLEMENT, DISCHARGE, RELEASE, INJUNCTION AND RELATED PROVISIONS

## 10.1. Compromise and Settlement of Claims, Interests, and Controversies

As discussed in detail in the Disclosure Statement and as otherwise provided herein, pursuant to sections 105(a), 363 and 1123 of the Bankruptcy Code and Bankruptcy Rule 9019 and in consideration for the distributions and other benefits provided pursuant to the Plan, which distributions and other benefits shall be irrevocable and not subject to challenge upon the Effective Date, the provisions of the Plan, and the distributions and other benefits provided hereunder, shall constitute a good-faith compromise and settlement of all Claims, Interests, and controversies relating to the contractual, legal, and subordination rights that a holderHolder of a Claim or Interest may have with respect to any Allowed Claim or Interest, or any distribution to be made on account of such Allowed Claim or Interest. Without limiting the foregoing, the Plan incorporates and is predicated upon the good-faith compromise and settlement of (i) any disputes regarding the appropriate allocation of general and administrative costs across the Debtors' assets, (ii) any disputes regarding the allocation of the Plan Funding Payment and any other value received by the Debtors under the Stock Purchase Agreement between the Debtors' Estates, and (iii) any disputes regarding whether and, if so, to what extent the Debtors' assets and liabilities should be pooled for voting, distribution and other purposes into a single, substantively consolidated estate.

The Plan shall be deemed a motion to approve the good-faith compromise and settlement of all such Claims, Interests, and controversies pursuant to Bankruptcy Rule 9019, and the entry of the Confirmation Order shall constitute the <u>Bankruptcy</u> Court's approval of the compromise or settlement of all such Claims, Interests, and controversies, as well as a finding by the <u>Bankruptcy</u> Court that such compromise or settlement is in the best interests of the Debtors, their Estates, and <u>holdersHolders</u> of Claims and Interests and is fair, equitable, and reasonable. In accordance with the provisions of the Plan, pursuant to Bankruptcy Rule 9019, without any further notice to or action, order, or approval of the <u>Bankruptcy</u> Court, after the Effective Date, the Distribution Trustee may compromise and settle Claims against, and Interests in, the Debtors and their Estates and Causes of Action against other Entities.

#### **10.2.** Releases by the Debtors in Favor of Third Parties

As of the Effective Date, for good and valuable consideration, the adequacy of which is hereby confirmed, the Debtors, the Reorganized Debtors, and any Person or Entity seeking to exercise the rights of the Estates, including, without limitation the Distribution Trust, the Distribution Trustee, any successor to the Debtors or any Estate representative appointed or selected pursuant to section 1123(b)(3) of the Bankruptcy Code, hereby forever release, waive and discharge, and shall be deemed to forever release,

waive, and discharge each of the Released Parties from any and all Claims, obligations, suits, judgments, damages, demands, debts, rights, causes of action, and liabilities whatsoever, whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, at equity, or otherwise, that are based in whole or in part on any act, omission, transaction, event or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the conduct of the Debtors' business, the Chapter 11 Cases, the Disclosure Statement, the Plan, or other documents implementing the Plan, provided, however, that nothing in this Section 10.2 of the Plan shall be deemed to release, or otherwise to prohibit the Reorganized Debtors or the Distribution Trustee from asserting and enforcing, any Claims, obligations, suits, judgments, demands, debts, rights, causes of action, or liabilities any of them may hold related to, or arising out of, the Plan, the DT Post-Closing Rights, the SALIC Claims, the Admitted SALIC/SRGL Revolver Claim, the Retained Causes of Action (solely as to the Reorganized Debtors), Causes of Action that are Distribution Trust Assets (solely as to the Distribution Trustee), the Stock Purchase Agreement, the Restructuring Implementation Agreement, the Distribution Trust Agreement, and the other documents implementing the Plan, provided, further, that nothing in this Section 10.2 of the Plan; (i) shall be deemed to release, or otherwise to prohibit the Reorganized Debtors or the Distribution Trustee, or anyone claiming through them from enforcing any confidentiality or non-disclosure agreement or any Claim, right or cause of action related thereto, (ii) shall be deemed to release, impair, or otherwise affect any parties' rights or interests under any Executory Contract or Unexpired Lease that is assumed by the Reorganized Debtors, and all such rights and interests shall be unaffected by the Plan and this Section 10.2 (subject, however, to the effects of Section 8.3(a), (c), and (h) of the Plan); (iii) shall be deemed to release any Intercompany Claims; (iv) shall be deemed to release any Causes of Action specifically identified in this Plan as Distribution Trust Assets; (v) shall be deemed to release any Claims or liabilities arising out of or relating to any act or omission of a Released Party that constitutes willful misconduct, actual fraud, or gross negligence as determined by a Final Order; or (vi) shall be deemed to release any Person's rights under the Plan.

Entry of the Confirmation Order shall constitute the <u>Bankruptcy</u> Court's approval, pursuant to Bankruptcy Rule 9019, of the Releases by the Debtors described in this Section 10.2 which includes by reference each of the related provisions and definitions contained herein, and, further, shall constitute the <u>Bankruptcy</u> Court's finding that such releases are: (1) in exchange for the good and valuable consideration provided by the Released Parties; (2) a good faith settlement and compromise of the Claims released by such releases; (3) in the best interests of the Debtors and all <u>holdersHolders</u> of Claims and Interests; (4) fair, equitable and reasonable; (5) given and made after due notice and opportunity for hearing; and (6) a bar to any of the Debtors, the Estates or the Distribution Trust asserting any Claim or cause of action released pursuant to such releases.

#### **10.3.** Releases by Holders of Claims and Interests

To the fullest extent permitted by law and except as otherwise specifically provided in the Plan, for good and valuable consideration, including the contributions of the Released Parties in facilitating the expeditious reorganization of the Debtors and the implementation of the restructuring contemplated by the Plan, effective as of the Effective Date, (i) Holders of Claims who vote to accept the Plan, (ii) Holders of Claims who are Unimpaired under the Plan, (iii) Holders of Claims entitled to vote on the Plan who do not submit a **<u>ballot Ballot</u>** and do not timely object to or opt-out of the releases, and (iv) each of the Purchaser, the Joint Liquidators, SRGL, and for each of the foregoing, their respective Representatives (other than with respect to SRGL<sub>5</sub> or the SRGL Equity Holders), hereby release, waive and discharge, and shall be deemed to forever release, waive, and discharge the Released Parties of any and all Claims, obligations, suits, judgments, damages, demands, debts, rights, causes of action, and liabilities whatsoever in connection with or related to the Debtors, whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, at equity, or otherwise, including those that any of the Debtors, the Reorganized Debtors, the Estates, or their Affiliates would have been legally entitled to assert in their own right (whether individually or collectively) or on behalf of the Holder of any Claim or Interest or other Entity based on or relating to, or in any manner arising from, in whole or in part, the Debtors, the conduct of the Debtors' business, the Chapter 11 Cases, the Disclosure Statement, the Plan or the documents implementing the Plan, and the negotiation, formulation, or preparation thereof, the solicitation of votes with respect to the Plan, the subject matter of, or the transactions or events giving rise to, any Claim or Interest that is treated in the Plan, the business or contractual arrangements between any of the Debtors and any Released Party, the restructuring of Claims and Interests prior to or in the Chapter 11 Cases, or upon any other act or omission, transaction, agreement, event or other occurrence taking place on or before the Effective Date, other than Claims or liabilities arising out of or relating to any act or omission of a Released Party that constitutes willful misconduct, actual fraud, or gross negligence as determined by a Final Order. Notwithstanding the foregoing, nothing in this Section 10.3, shall cause (i) a release, waiver or discharge of SFL from any Claim against it under the SFLST I TruPS Documents or (ii) SRGL to release, waive or discharge any Claim it may have against any of SRGL's Representatives (except for the Debtors and the Debtors' Representatives).

Each Person or Entity providing releases under the Plan, including the Debtors, the Estates, and the Holders of Claims (regardless of whether such Holder is a Released Party), shall be deemed to have granted the releases set forth above notwithstanding that such Person or Entity may hereafter discover facts in addition to, or different from, those which it now knows or believes to be true, and without regard to the subsequent discovery or existence of such different or additional facts, and such Person or Entity expressly waives any and all rights that it may have under any statute or common law principle which would limit the effect of such releases to those Claims or causes of action actually known or suspected to exist at the time of execution of such release.

Entry of the Confirmation Order shall constitute the Court's approval, pursuant to Bankruptcy Rule 9019, of the Releases by Holders of Claims and Interests described in this Section 10.3, which includes by reference each of the related provisions and definitions contained herein, and, further, shall constitute the Court's finding that such releases are: (1) in exchange for the good and valuable consideration provided by the Released Parties; (2) a good faith settlement and compromise of the Claims released by such releases; (3) in the best interests of the Debtors and all Holders of Claims and Interests; (4) fair, equitable and reasonable; (5) given and made after due notice and opportunity for hearing; and (6) a bar to any of the Releasing Parties asserting any elaimClaim or cause of action released pursuant to such releases.

#### 10.4. Discharge and Discharge Injunction

(a) Discharge of Claims

On and after the Effective Date: (a) the rights afforded in the Plan and the treatment of all Claims and Interests shall be in exchange for and in complete satisfaction, discharge, and release of all Claims and Interests of any nature whatsoever, including any interest accrued on such Claims from and after the Petition Date, against the Debtors, the Reorganized Debtors or any of their assets, property, or estateEstates; (b) the Plan shall bind all Holders of Claims and Interests, notwithstanding whether any such Holders failed to vote to accept or reject the Plan or voted to reject the Plan; (c) all Claims and Interests shall be satisfied, discharged, and released, and the Debtors' and Reorganized Debtors' liability with respect thereto shall be extinguished completely, including any liability of the kind specified under Bankruptcy Code section 502(g); and (d) all entities shall be precluded from asserting against the Debtors, the Reorganized Debtors, the Estates, the Distribution Trust, the Distribution Trustee their successors and assigns, and their assets and properties any other Claims or Interests based upon any documents, instruments, or any act or omission, transaction, or other activity of any kind or nature that occurred prior to the Effective Date, provided, however, that the foregoing discharge shall not apply to the ability of Holders of Allowed Claims to recover from the Distribution Trust on account of such Allowed Claims and/or Interests, all in accordance with the terms of the Plan and **Distribution Trust Agreement.** 

(b) Discharge Injunction

Except as provided in the Plan, to the fullest extent permitted by law, or the Confirmation Order, as of the Effective Date, all Persons that have held, currently hold, may hold, or allege that they hold, a Claim, Interest, or other debt or liability that is satisfied, released and discharged pursuant to the terms of the Plan are permanently enjoined from taking any of the following actions against the Debtors, the Reorganized Debtors, the Distribution Trust, the Distribution Trustee and their respective subsidiaries or their property on account of any such discharged Claims, debts, liabilities or rights: (i) commencing or continuing, in any manner or in any place, any action or other proceeding; (ii) enforcing, attaching, collecting, or recovering in any manner any judgment, award,

decree, or order; (iii) creating, perfecting, or enforcing any Lien or encumbrance; (iv) asserting a setoff, right of subrogation, or recoupment of any kind against any debt, liability, or obligation due to the Debtors or the Reorganized Debtors; or (v) commencing or continuing any action or other proceeding of any kind, in each such case in any manner, in any place, or against any Person that does not comply with or is inconsistent with the provisions of the Plan.

Except as provided in the Plan or the Confirmation Order, as of the Effective Date, all Persons that have held, currently hold, or may hold, a Claim, Interest, obligation, suit, judgment, damage, demand, debt, right, cause of action, or liability that is released pursuant to Article X of the Plan are permanently enjoined from taking any of the following actions on account of such released Claims, Interests, obligations, suits, judgments, damages, demands, debts, rights, causes of action, or liabilities or rights: (i) commencing or continuing, in any manner or in any place, any action or other proceeding; (ii) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree, or order; (iii) creating, perfecting, or enforcing any Lien or encumbrance; (iv) asserting a setoff against any debt, liability, or obligation due to any Released Party; or (v) commencing or continuing any action, in any manner, in any place, or against any Person or Entity that does not comply with or is inconsistent with the provisions of the Plan.

Without limiting the effect of the foregoing provisions of this <u>Section 10.4</u> of the Plan upon any Person or Entity, by accepting <u>distributionsDistributions</u> pursuant to the Plan, each Holder of an Allowed Claim shall be deemed to have specifically consented to the injunctions set forth in this <u>Section 10.4</u> of the Plan.

#### 10.5. Exculpation

To the fullest extent permitted by law and except as otherwise specifically provided in the Plan, each of the Exculpated Parties will not have or incur any liability for any act or omission in connection with, or arising out of, the formulation, negotiation, preparation, dissemination, implementation or pursuit of approval of the Plan, the Disclosure Statement, the Restructuring Implementation Agreement, the Stock Purchase Agreement, the Plan Supplement or any documents, instruments or agreements implementing or related to the foregoing, or the solicitation of votes for or Confirmation of the Plan, or the consummation of the Plan, the Restructuring Implementation Agreement, the Stock Purchase Agreement, the Plan Supplement, or the transactions contemplated, implemented and effectuated thereby or the administration of the Plan or the property to be distributed under the Plan, or any other act or omission during the administration of the Debtors' Estates or in contemplation of the Chapter 11 Cases, except for willful misconduct, actual fraud or gross negligence as determined by a Final Order, and in all respects, will be entitled to rely upon the advice of counsel with respect to their duties and responsibilities under the Plan.

#### 10.6. Post-Effective Date Indemnification

Indemnification Obligations of the Debtors that are owed to Indemnified D&O Parties will be deemed to be, and will be treated as though they are, Executory Contracts that are assumed by the Reorganized Debtors pursuant to section 365 of the Bankruptcy Code, and such Indemnification Obligations shall not be discharged or otherwise impaired by Confirmation of the Plan.

From and after the Closing Date, to the extent permitted by Applicable Law, the certificate of incorporation, certificate of formation, bylaws or limited liability company operating agreement (or similar organizational documents) of each SALIC Group Company shall continue to contain provisions no less favorable with respect to indemnification, advancement of expenses and exculpation of each Indemnified D&O Party than are set forth in the organizational documents of the SALIC Group Companies as of the date hereof, which provisions shall not be amended, repealed or otherwise modified for a period of six (6) years after the Closing Date in any manner that would adversely affect the rights thereunder of any such individual.

For a period of six (6) years from and after the Closing Date, to the extent that the Indemnified D&O Parties are not otherwise covered as insureds under an existing policy of directors' and officers' liability insurance in accordance with the requirements set forth in Section 5.8(b) of the Stock Purchase Agreement, Purchaser shall cause the SALIC Group Companies to maintain in effect policies of directors' and officers' liability insurance comparable to those maintained by the SALIC Group Companies or SRGL with respect to matters existing or occurring at or prior to the Closing Date; provided, that Purchaser or the SALIC Group Companies may substitute therefor policies of at least the same coverage containing terms and conditions that are not less advantageous than the existing policies (including with respect to the period covered); provided, further, that in lieu of maintaining the current policies of directors' and officers' liability insurance, Purchaser may (or may cause the SALIC Group Companies to) purchase "tail" coverage or otherwise replace such policies with coverage with a scope, policy limits and retained coverage not less favorable than the scope, policy limits and retained coverage currently provided. Notwithstanding the foregoing, at Purchaser's direction and in satisfaction of Purchaser's obligations under Section 5.8(b) of the Stock Purchase Agreement, SALIC shall obtain such "tail" coverage in respect of SALIC's existing policy of directors' and officers' liability insurance identified in Section 3.15 of the SALIC Disclosure Schedules (Policy No. ELU154535-18) on terms acceptable to Purchaser, to be effective as of the Closing Date, provided that the cost of such coverage shall be funded from unrestricted Cash of SALIC and SHI.

#### XI. RETENTION OF JURISDICTION

#### 11.1. Retention of Jurisdiction

Pursuant to sections 105(a) and 1142 of the Bankruptcy Code, and notwithstanding entry of the Confirmation Order and occurrence of the Effective Date, except as otherwise ordered by the Bankruptcy Court, or contemplated by the RIA Order or Restructuring

Implementation Agreement, the Bankruptcy Court will retain jurisdiction over all matters arising out of, and related to, the Chapter 11 Cases and the Plan to the fullest extent permitted by law, including, among other things, jurisdiction to:

(a) allow, disallow, determine, liquidate, classify, estimate, or establish the priority, secured, unsecured, or subordinated status of any Claim or Interest not otherwise Allowed under the Plan (other than personal injury or wrongful death Claims, unless agreed by the Holder), including the resolution of any request for payment of any Administrative Claim and the resolution of any objections to the Allowance or priority of Claims;

(b) hear and determine all applications for compensation and reimbursement of expenses of Professionals under the Plan or under sections 327, 328, 330, 331, 503(b), 1103, and 1129(a)(4) of the Bankruptcy Code; provided, however, that from and after the Effective Date, the payment of the fees and expenses of the professionals of the Reorganized Debtors or the Distribution Trust shall be made in the ordinary course of business and shall not be subject to the approval of the Bankruptcy Court;

(c) hear and determine all matters with respect to the assumption or rejection of any Executory Contract or Unexpired Lease to which a Debtor is a party or with respect to which a Debtor may be liable, including, if necessary, the nature or amount of any required Cure or the liquidation or Allowance of any Claims arising therefrom;

(d) effectuate performance of and payments under the provisions of the Plan and enforce remedies upon any default under the Plan;

(e) hear and determine any and all adversary proceedings, motions, applications, and contested or litigated matters arising out of, under, or related to, the Chapter 11 Cases or the Plan;

(f) enter such orders as may be necessary or appropriate to execute, implement, or consummate the provisions of the Plan and all contracts, instruments, releases, and other agreements or documents created in connection with the Plan, the Disclosure Statement or the Confirmation Order;

(g) hear and determine disputes arising in connection with the interpretation, implementation, consummation, or enforcement of the Plan, including disputes arising under agreements, documents, or instruments executed in connection with the Plan;

(h) consider any modifications of the Plan, cure any defect or omission, or reconcile any inconsistency in any order of the Bankruptcy Court, including, without limitation, the Confirmation Order;

(i) issue injunctions, enter and implement other orders, or take such other actions as may be necessary or appropriate to restrain interference by any Person or Entity with the implementation, consummation, or enforcement of the Plan or the Confirmation Order;

(j) enter and implement such orders as may be necessary or appropriate if the Confirmation Order is for any reason reversed, stayed, revoked, modified, or vacated;

(k) hear and determine any matters arising in connection with or relating to the Plan, the Plan Supplement, the schedules to the Plan, the Disclosure Statement, the Confirmation Order, or any contract, instrument, release, or other agreement or document created in connection with the Plan, the Plan Supplement, the Schedules to the Plan, the Disclosure Statement, or the Confirmation Order;

(1) hear and determine any matters arising in connection with or relating to the Distribution Trust, the interpretation, implementation or operation of the Distribution Trust Agreement or the consummation of the transactions contemplated thereby;

(m) enforce, interpret, and determine any disputes arising in connection with any stipulations, orders, judgments, injunctions, releases, exculpations, indemnifications, and rulings entered in connection with the Chapter 11 Cases (whether or not the Chapter 11 Cases have been closed);

(n) except as otherwise limited in the Plan, recover all assets of the Debtors and property of the Estates, wherever located;

(o) hear and determine matters concerning state, local, and federal taxes in accordance with sections 346, 505, and 1146 of the Bankruptcy Code;

(p) hear and determine all disputes involving the existence, nature, or scope of the Debtors' discharge;

(q) hear and determine such other matters as may be provided in the Confirmation Order or as may be authorized under, or not inconsistent with, provisions of the Bankruptcy Code;

(r) hear and determine matters relating to the Stock Purchase Agreement and the Restructuring Implementation Agreement, to the extent provided for in such documents; and

(s) enter a final decree closing the Chapter 11 Cases.

If the Bankruptcy Court abstains from exercising, or declines to exercise, jurisdiction or is otherwise without jurisdiction over any matter arising in, arising under, or related to the Chapter 11 Cases, including the matters set forth in <u>Section 11.1</u> of the Plan, the provisions of Article XI of the Plan will have no effect upon and will not control, prohibit, or limit the exercise of jurisdiction by any other court having jurisdiction with respect to such matter.

## 11.2. Limitation on Jurisdiction

In no event shall the provisions of the Plan be deemed to confer in the Bankruptcy Court jurisdiction greater than that established by the provisions of 28 U.S.C. §§ 157 and 1334, as well as the applicable circumstances that continue jurisdiction for defense and enforcement of the Plan and Plan Documents. For the avoidance of doubt, however, such jurisdiction shall be deemed, by the entry of the Confirmation Order, to:

(a) Permit entry of a final judgment by the Bankruptcy Court in any core proceeding referenced in 28 U.S.C. § 157(b) and to hear and resolve such proceedings in accordance with 28 U.S.C. § 157(c) and any and all related proceedings, including, without limitation, (i) all proceedings concerning disputes with, or Causes of Action or Claims against, any Entity that the Distribution Trust, the Debtors or the Reorganized Debtors or their successors or assigns, may have, and (ii) any and all Causes of Action or other Claims against any Entity for harm to or with respect to (x) any property of the Estates, or (y) any property of the Estate liened or transferred by the Debtors to any other Entity;

(b) Include jurisdiction over the recovery of any property of the Estates (or property transferred by the Debtors with Bankruptcy Court approval) from any Entity wrongly asserting ownership, possession or control of the same, whether pursuant to sections 542, 543, 549, 550 of the Bankruptcy Code or otherwise, as well as to punish any violation of the automatic stay under section 362 of the Bankruptcy Code or any other legal rights of the Debtors under or related to the Bankruptcy Code; and

(c) Permit the taking of any default judgment against any Entity that has submitted itself to the jurisdiction of the Bankruptcy Court.

## XII. MISCELLANEOUS PROVISIONS

### 12.1. Legally Binding Effect

The provisions of the Plan shall bind all Creditors and Interest Holders, whether or not they accept the Plan and wherever located. On and after the Effective Date, all Holders of Claims and Interests shall be precluded and enjoined from asserting any Claim against or Interest in the Debtors or their assets or properties based on any transaction or other activity of any kind that occurred prior to the Effective Date except as may be expressly provided for by the Plan.

### **12.2.** Exemption from Transfer Taxes

Pursuant to section 1146 of the Bankruptcy Code and the Plan, any of the following acts or any similar act otherwise contemplated in the Plan will not be subject to any stamp tax, transfer tax, filing or recording tax, or other similar tax: (a) the issuance, transfer or exchange of notes, debt instruments and equity securities under or in connection with the Plan; (b) the creation, assignment, recordation or perfection of any lien, pledge, other security interest or other instruments of transfer; (c) the making or assignment of any lease; (d) the creation, execution and delivery of any agreements or other documents creating or evidencing the

formation of the Reorganized Debtors or the issuance or ownership of any interest in the Reorganized Debtors; or (e) the making or delivery of any deed or other instrument of transfer under the Plan in connection with the vesting of the Debtors' assets in the Reorganized Debtors or the Distribution Trust or Distribution Trustee pursuant to or in connection with the Plan, including, without limitation, merger agreements, stock purchase agreement, agreements of consolidation, restructuring, disposition, liquidation or dissolution, and transfers of tangible property.

#### 12.3. Securities Exemption

Any rights issued under, pursuant to or in effecting the Plan, including, without limitation, the New Equity, the New SHI Equity and any beneficial interests in the Distribution Trust, and the offering and issuance thereof by any party, including without limitation the Debtors, the Estates, or New Holdco (if applicable), shall be exempt from Section 5 of the Securities Act of 1933, if applicable, and from any state or federal securities laws requiring registration for offer or sale of a security or registration or licensing of an issuer of, underwriter of, or broker or dealer in, a security, and shall otherwise enjoy all exemptions available for Distributions of securities under a plan of reorganization in accordance with all Applicable Law, including without limitation section 1145 of the Bankruptcy Code. If the issuance of the New Equity and the New SHI Equity does not qualify for an exemption under section 1145 of the Bankruptcy Code, the New Equity and the New SHI Equity shall be issued in a manner that qualifies for any other available exemption from registration, whether as a private placement under Section 4(a)(2) of the Securities Act and/or the safe harbor provisions promulgated thereunder, Regulation D of 1993, or otherwise.

#### 12.4. Defects, Omissions and Amendments of the Plan

The Plan may be amended, modified, or supplemented by the Debtors, subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement, in the manner provided for by section 1127 of the Bankruptcy Code or as otherwise permitted by law, without additional disclosure pursuant to section 1125 of the Bankruptcy Code, except as otherwise ordered by the Bankruptcy Court. In addition, after the Confirmation Date, so long as such action does not materially and adversely affect the treatment of Holders of Allowed Claims pursuant to the Plan, the Debtors, subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement, may remedy any defect or omission or reconcile any inconsistencies in the Plan or the Confirmation Order with respect to such matters as may be necessary to carry out the purposes or effects of the Plan, and any Holder of a Claim that has accepted the Plan shall be deemed to have accepted the Plan as amended, modified, or supplemented. Subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement, prior to the Effective Date, the Debtors may make appropriate technical adjustments and modifications to the Plan without further order or approval of the Bankruptcy Court; provided, that such technical adjustments and modifications do not adversely affect in a material way the treatment of Holders of Claims under the Plan.

## 12.5. Due Authorization by Creditors

Each and every Creditor who elects to participate in the Distributions provided for herein warrants that the Creditor is authorized to accept in consideration of its Claim against the Debtors the Distributions provided for in the Plan, and that there are no outstanding commitments, agreements, or understandings, express or implied, that may or can in any way defeat or modify the rights conveyed or obligations undertaken by the Creditor under the Plan.

### 12.6. Filing of Additional Documentation

No later than seven (7) calendar days prior to the Voting Deadline, subject to the terms of the Stock Purchase Agreement and the Restructuring Implementation Agreement and subject to approval in form and substance by the Purchaser, the Debtors may file with the Bankruptcy Court such Plan Supplement, agreements and other documents as may be reasonably necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan or any Plan Document, which shall also constitute "Plan Documents."

## 12.7. Dissolution of the Official Committee

On the Effective Date, the Official Committee shall dissolve and all members, employees, or agents thereof shall be released and discharged from all rights and duties arising from or related to the Chapter 11 Cases, provided, however, that (a) the Official Committee and its respective Professionals shall be retained with respect to applications Filed or to be Filed by Professionals pursuant to sections 330 and 331 of the Bankruptcy Code and (b) the Distribution Trust shall be deemed the successor of the Official Committee with respect to any motions seeking to enforce the Plan and the transactions contemplated hereunder or the Confirmation Order and any pending appeals and related proceedings.

### 12.8. Governing Law

Except to the extent the Bankruptcy Code or the Bankruptcy Rules are applicable, the rights and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with the laws of the State of Delaware, without giving effect to the principles of conflicts of law thereof.

### 12.9. Successors and Assigns

The rights, benefits and obligations of any Entity named or referred to in the Plan or any Plan Document shall be binding on, and shall inure to the benefit of, any heir, executor, administrator, successor or assign of such Entity.

### **12.10.** Transfer of Claims

Any transfer of a Claim shall be in accordance with Bankruptcy Rule 3001(e) and the terms of this <u>Section 12.10</u>. Notice of any such transfer shall be forwarded to the Debtors by registered or certified mail, as set forth in <u>Section 12.11</u> hereof. Both the transferee and transferor

shall execute any notice, and the signatures of the parties shall be acknowledged before a notary public. The notice must clearly describe the interest in the Claim to be transferred. No transfer of a partial Claim shall be allowed. All transfers must be of one hundred percent (100%) of the transferor's interest in the Claim.

#### **12.11.** Notices

All notices, requests, and demands required or permitted to be provided to the Debtors, the Purchaser, the Reorganized Debtors, the Official Committee, or the Distribution Trust under the Plan shall be in writing and shall be deemed to have been duly given or made (a) when actually delivered (i) by certified mail, return receipt requested, (ii) by hand delivery or (iii) by mail, postage prepaid or, (b) in the case of notice by facsimile transmission, when received and confirmed, addressed (in all instances, with a simultaneous copy by electronic mail, which shall not independently constitute notice) as follows:

(a) If to the Debtors, at:

Scottish Holdings, Inc. Scottish Annuity & Life Insurance Company (Cayman) Ltd. 14120 Ballantyne Corporate Place, Suite 300 Charlotte, NC 28277 Facsimile: (704) 752-7736 Attn: Gregg Klingenberg, Chief Executive Officer Gregg.Klingenberg@scottishre.com

with copies to:

Hogan Lovells US LLP 875 Third Avenue New York, NY 10022 Facsimile: (212) 918-3100 Attn: Peter Ivanick, Esq. Lynn W. Holbert, Esq. John D. Beck, Esq. Email: peter.ivanick@hoganlovells.com lynn.holbert@hoganlovells.com

john.beck@hoganlovells.com

-and-

Morris, Nichols, Arsht & Tunnell LLP 1201 N. Market St., 16th Floor PO Box 1347 Wilmington, DE 19899-1347 Facsimile: (302) 658-3989 Attn: Eric D. Schwartz, Esq. Gregory W. Werkheiser, Esq. Matthew B. Harvey, Esq. Email: eschwartz@mnat.com gwerkheiser@mnat.com mharvey@mnat.com

-and-

Mayer Brown LLP 1221 Avenue of the Americas New York, New York 10020 Telephone: (212) 506-2227 Facsimile: (212) 262-1910 Email: fmonaco@mayerbrown.com Attn: Stephen G. Rooney, Esq. Francis R. Monaco, Esq. Email: fmonaco@mayerbrown.com srooney@mayerbrown.com

(b) If to the Purchaser or the Reorganized Debtors, at:

with a copy to:

Kramer Levin Naftalis & Frankel LLP 1177 Avenue of the Americas	
New York, New York 10036	
(212) 715-9100	
(212) 715-8000	
Attention: Stephen Zide, Esq.	
Ernest S. Wechsler, Esq.	
<u>Anupama Yerramalli, Esq.</u>	
<u>Seth R. Merl, Esq.</u>	
szide@kramerlevin.com	
ewechsler@kramerlevin.com	
ayerramalli@kramerlevin.com	
smerl@kramerlevin.com	
-Stephen Zide, Esq.	
-Ernest S. Wechsler, Esq.	
-Anupama Yerramalli, Esq.	
-Seth R. Merl, Esq.	

(c) If to the Official Committee, at:

Pepper Hamilton LLP Hercules Plaza, Suite 5100 1313 Market Street P.O. Box 1709 Wilmington, DE 19899-1709 (Courier Deliveries: 19801-1151) Facsimile: (302) 421-8390 Attn: David M. Fournier, Esq. H. Peter Haveles Jr., Esq. John Henry Schanne II, Esq. Email: fournierd@pepperlaw.com havelesp@pepperlaw.com schannej@pepperlaw.com

(d) If to the Distribution Trustee, at the contact information to be supplied in the notice of the occurrence of the Effective Date.

#### 12.12. U.S. Trustee Fees and Reports

The Debtors will pay<u>All</u> pre-confirmation fees <u>owed todue and payable pursuant</u> to section 1930 of Title 28 of the U.S. Trustee on or before Code prior to the Effective Date of the Plan. After confirmationshall be paid by the Debtors. On and after the Effective Date, the Distribution Trustee will file with the court and serve on the U.S. Trustee quarterly financial reports in a format prescribed by the U.S. Trustee, and the Distribution Trustee will pay from

theshall cause the Distribution Trust to pay any and all such fees when due and payable from the Trust Administration Reserve, and shall file with the Bankruptcy Court quarterly reports in a form reasonably acceptable to the United States Trustee. The Distribution Trust post-confirmationshall remain obligated to pay quarterly fees to the U.S. Trustee until a final decree is entered or the case is converted or dismissed as provided in 28 U.S.C. § 1930(a)(6).Office of the United States Trustee for each of the Chapter 11 Cases until such Chapter 11 Case has been closed, dismissed, or converted to a case under Chapter 7 of the Bankruptcy Code.

#### 12.13. Implementation

The Debtors, the Reorganized Debtors, the Purchaser, and the Distribution Trustee shall be authorized to perform all reasonable, necessary and authorized acts to consummate the terms and conditions of the Plan and the Plan Documents.

#### 12.14. No Admissions

Notwithstanding anything herein to the contrary, nothing contained in the Plan shall be deemed an admission by the Debtors with respect to any matter set forth herein, including, without limitation, liability on any Claim or Interest or the propriety of the classification of any Claim or Interest.

### 12.15. SRGL Consent Rights Reserved

For the avoidance of doubt, all SRGL Consent Rights relating to Specified Restructuring Documents are reserved in their entirety. Nothing herein shall affect SRGL's rights under the Restructuring Implementation Agreement, all of which are incorporated herein by reference, including in respect of the Restructuring Documents and the Specified Restructuring Documents. Without limiting the generality of the foregoing, (a) the Restructuring Documents shall, unless otherwise expressly indicated in the Restructuring Implementation Agreement, be consistent in all material respects with the Restructuring Implementation Agreement, and (b) the Specified Restructuring Documents shall be consistent in all material respects with the Restructuring Implementation Agreement, nothing anything to the contrary in the Restructuring Implementation Agreement, nothing set forth in the Restructuring Implementation Agreement or this Plan shall operate as a waiver or release of (i) the Admitted SALIC/SRGL Revolver Claim; (ii) the SALIC Claims; or (iii) any Causes of Action against the SRGL Equity Holders.

#### 12.16. Substantial Consummation

The Plan shall be deemed substantially consummated on the Effective Date.

## 12.17. Final Decree

On full consummation and performance of the Plan and Plan Documents, the Distribution Trustee may request the Bankruptcy Court to enter a final decree closing the Chapter 11 Cases and such other orders that may be necessary and appropriate.

Scottish Holdings, Inc. Scottish Annuity & Life Insurance Company (Cayman) Ltd.

<u>Gregg Klingenberg</u>

Gregg Klingenberg Chief Executive Officer

## <u>Exhibit A</u>

**Glossary of Defined Terms** 

#### EXHIBIT A Glossary of Defined Terms

1.1 "<u>Administrative Claim</u>" means a Claim for any costs or expenses of administration of the Estates under sections 503(b), 507(b) or 1114(e)(2) of the Bankruptcy Code, including, without limitation, for: (a) any actual and necessary costs and expenses incurred after the Petition Date of preserving the Estates and operating the businesses of the Debtors; (b) any payment to be made under the Plan to cure a default on an assumed Executory Contract or assumed Unexpired Lease; (c) any postpetition cost, indebtedness or contractual obligation duly and validly incurred or assumed by the Debtors in the ordinary course of its business or by order of the Bankruptcy Court; (d) any Allowed Claims that are entitled to be treated as Administrative Claims pursuant to a Final Order under section 546(c)(2)(A) of the Bankruptcy Code; (e) any Allowed Claims of Professionals in the Chapter 11 Cases; and (f) any fees and charges assessed against the EstateEstates under chapter 123 of title 28 of the United States Code, 28 U.S.C. §§ 1911–1930.

1.2 "<u>Administrative Claims Bar Date</u>" means, (i) for any Administrative Claim (other than a Professional Fee Claim) incurred prior to the Confirmation Date, the date that is thirty (30) days after the Confirmation Date or such earlier deadline applicable to such Administrative Claim as established by order of the Bankruptcy Court entered before the Confirmation Date ; or (ii) for any Administrative Claim (other than a Professional Fee Claim) incurred between the Confirmation Date and the Effective Date, the date that is thirty (30) days after the Effective Date or such earlier deadline applicable to such Administrative Claim as established by order of the Bankruptcy Court entered before the Effective Date and the Effective Date.

1.1 "<u>Admitted SALIC/SRGL Revolver Claim</u>" shall have the meaning set forth in paragraph 11 of the RIA Order.

1.2 "<u>Affiliate</u>" means "affiliate" as defined in section 101(2) of the Bankruptcy Code. With respect to any Person that is not a Debtor, the term "Affiliate" shall apply to such <u>personPerson</u> as if the Person were a Debtor.

1.3 "<u>Allocated Portion</u>" means with respect to a particular issuance of TruPS, the portion of the TruPS Claim allocable on a pro rata basis to a Beneficial Holder on account of its beneficial ownership of such TruPS, as determined in accordance with the applicable TruPS Declaration.

1.4 "<u>Allocation/Use Priorities</u>" means the waterfall described in Section 6.1(b) of the Plan.

1.5 "<u>Allowed</u>" or "<u>Allowance</u>" means with respect to any Claim (including any Administrative Claim) or portion thereof (to the extent such Claim is not Disputed or Disallowed) or any Interest (a) any Claim or Interest, proof of which (i) was timely Filed with the Bankruptcy Court or its duly appointed claims agent, (ii) was deemed timely filed pursuant to section 1111(a) of the Bankruptcy Code, or (iii) by the Bar Date Order or other Final Order, was not required to be Filed; (b) any Claim or Interest that has been, or hereafter is, listed in the Schedules as liquidated in an amount other than zero or unknown and not Disputed or Contingent

(or as to which the applicable Proof of Claim has been withdrawn or Disallowed); and (c) any Claim or Interest which has been allowed (whether in whole or in part) by the Plan, the Restructuring Implementation Agreement or the RIA Order or other Final Order (but only to the extent so allowed), and, in (a) and (b) above, as to which no objection to the allowance thereof, or action to subordinate, avoid, classify, reclassify, expunge, estimate or otherwise limit recovery with respect thereto, has been Filed within the applicable period of limitation fixed by the Plan, the Bankruptcy Code, the Bankruptcy Rules or a Final Order; (d) any Claim or Interest allowed under or pursuant to the terms of the Plan; (e) any Claim arising from the recovery of property under sections 550 or 553 of the Bankruptcy Code which has been allowed in accordance with section 502(h) of the Bankruptcy Code; (f) a Claim relating to a rejected Executory Contract or Unexpired Lease that either (i) is not a Disputed Claim or (ii) has been allowed by a Final Order, in either case only if a Proof of Claim has been Filed by the applicable Bar Date or has otherwise been deemed timely Filed under applicable law Applicable Law; or (g) which is a Professional Claim for which a fee award amount has been approved by order of the Bankruptcy Court; provided, however, that Claims or Interests allowed solely for the purpose of voting to accept or reject the Plan pursuant to an order of the Bankruptcy Court shall not be considered "Allowed" hereunder. For the avoidance of doubt, a Proof of Claim Filed after the Bar Date or a request for payment of an Administrative Claim Filed after the Administrative Claims Bar Date, as applicable, shall not be Allowed for any purposes whatsoever absent entry of a Final Order allowing such late-filed Claim.

1.6 "<u>Applicable Law</u>" means any foreign, domestic, state, federal, national, international, multinational, regional or local law, statute, ordinance, rule, regulation, writ, directive, treaty, order, judgment, decree, injunction or other legally binding obligation imposed by or on behalf of a Governmental Unit.

1.7 "<u>Assumption Dispute</u>" <u>Assumption Dispute</u> means a pending objection relating to the assumption of an Executory Contract or Unexpired Lease pursuant to section 365 of the Bankruptcy Code, including, but not limited to, as to the Cure Amount.

1.8 "Available Plan Distribution Funding Amount" means all of the Plan Funding Payment *less* the amounts set forth in Sections 6.1(b)(1) through (54) of the Plan.

1.9 "<u>Ballantyne</u>" means Ballantyne Re II plc.

1.10 "<u>Ballot</u>" has the meaning set forth in paragraph D of the Disclosure Statement Order.

1.11 "<u>Bankruptcy Code</u>" means the Bankruptcy Reform Act of 1978, as codified in title 11 of the United States Code, 11 U.S.C. §§ 101-1532, as in effect on the Petition Date, together with all amendments and modifications thereto that were subsequently made applicable to the Chapter 11 Cases.

1.12 "<u>Bankruptcy Court</u>" means the United States Bankruptcy Court for the District of Delaware or such other court as may have jurisdiction over the Chapter 11 Cases.

1.13 "<u>Bankruptcy Rules</u>" means, when referenced generally, (i) the Federal Rules of Bankruptcy Procedure and the Official Bankruptcy Forms, as amended and promulgated under section 2075 of title 28 of the United States Code, (ii) the applicable Federal Rules of Civil Procedure, as amended and promulgated under section 2072 of title 28 of the United States Code, (iii) the applicable Local Rules of Bankruptcy Practice and Procedure for the United States Bankruptcy Court for the District of Delaware, and (iv) any standing orders governing practice and procedure issued by the Bankruptcy Court, each as in effect on the Petition Date, together with all amendments and modifications thereto that were subsequently made applicable to the Chapter 11 Cases or proceedings therein, as the case may be; <u>provided</u>, <u>however</u>, when a specific Bankruptcy Rule is referenced (e.g., Bankruptcy Rule 9019), such reference shall be to such Rule under the Federal Rules of Bankruptcy Procedure.

1.14 "<u>Bar Date</u>" means, for any Claim, the date set by the Bankruptcy Court by the Bar Date Order as the last day for Filing a Proof of Claim on account of such Claim against the Debtors in the Chapter 11 Cases.

1.15 "<u>Bar Date Order</u>" means the Order (A) Establishing Bar Dates For Filing Proofs Of Claim, (B) Approving The Form And Manner For Filing Proofs Of Claim, (C) Approving Notice Thereof, And (D) Granting Related Relief entered by the Bankruptcy Court on March 28, 2018, at Docket No. 189.

1.16 "<u>Beneficial Holder</u>" means, with respect to any TruPS, the person or entity having "beneficial ownership" of such TruPS (as determined pursuant to Rule 13d-3 of the Securities Exchange Act of 1934). For the avoidance of doubt, the term Beneficial Holder includes SRGL with respect to its TruPS holdings.

1.17 "<u>BNYM</u>" means The Bank of New York Mellon Trust Company, N.A., as trustee.

1.18 "<u>Business Day</u>" means any day, excluding Saturdays, Sundays, or "legal holidays" (as defined in Bankruptcy Rule 9006(a)), on which commercial banks are open for business in Wilmington, Delaware.

1.19 "<u>Cash</u>" or "<u>\$</u>" means legal tender of the United States of America or the equivalent thereof, including bank deposits, checks and cash equivalents.

1.20 "<u>Cash Election</u>" means an election made by a New Equity Eligible Holder to take its Distribution from the Plan Funding Payment under the Plan in the form of Cash, which shall be made by <u>not</u> checking the "<u>Cash DistributionNew Equity Election</u>" box on its timely completed and submitted Ballot. In the event that a New Equity Eligible Holder fails to make an election or fails to submit a Ballot, then such New Equity Eligible Holder shall be deemed to have made the Cash Election.

1.21 "<u>Causes of Action</u>" means any and all Claims, causes of action, controversies, obligations, suits, judgments, damages, demands, debts, rights, preference actions, fraudulent conveyance actions and other claims or causes of action under sections 510, 544, 545, 546, 547, 548, 549, 550 and 553 of the Bankruptcy Code and other similar state law claims and causes of action, Liens, indemnities, guaranties, suits, liabilities, judgments, accounts, defenses, offsets,

powers, privileges, licenses and franchises of any kind or character whatsoever, whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, suspected or unsuspected, disputed or undisputed, secured or unsecured, assertable directly or derivatively, whether arising before, on, or after the Petition Date, in contract or in tort, arising in law, equity or pursuant to any other theory of law. For the avoidance of doubt, Causes of Action also includes: (a) any right of setoff, counterclaim or recoupment and any claim on contracts or for breaches of duties imposed by law or in equity; (b) the right to object to Claims or Interests; (c) any claim pursuant to sections 362 or chapter 5 of the Bankruptcy Code; (d) any claim or defense including fraud, mistake, duress and usury, and any other defenses set forth in section 558 of the Bankruptcy Code; and (e) any state law fraudulent transfer claim.

1.22 "<u>Cayman Islands Court</u>" means the Grand Court of the Cayman Islands, Financial Services Division.

1.23 "<u>Cayman Islands Winding Up Proceedings</u>" means winding up proceedings of SRGL in the Cayman Islands Court.

1.24 "Cerberus" means Cerberus Capital Management, L.P., and SRGL Acquisition.

1.25 "<u>Certificate</u>" means any instrument evidencing a Claim or Interest.

1.26 "<u>Chapter 11 Cases</u>" mean the jointly administered chapter 11 cases of Scottish Holdings, Inc., and Scottish Annuity & Life Insurance Company (Cayman) Ltd., Case No. 18-10160 (LSS) in the United States Bankruptcy Court for the District of Delaware.

1.27 "<u>Charging Lien</u>" means any Lien or other priority in payment to which a TruPS Indenture Trustee is entitled under the terms of a TruPS Indenture to assert against Distributions to be made to Holders of Claims under such TruPS Indenture.

1.28 "<u>Claim</u>" has <u>the same</u> meaning as "claim" under section 101(5) of the Bankruptcy Code-section.

1.29 "<u>Claims Objection Deadline</u>" means the date that is one hundred eighty (180) days after the Effective Date or such later date as may be extended by order of the Bankruptcy Court.

1.30 "<u>Class</u>" means a category of Holders of Claims or Interests pursuant to section 1122(a) of the Bankruptcy Code, as described in Articles III and IV of the Plan.

1.31 "<u>Closing</u>" has the meaning as defined in the Stock Purchase Agreement.

1.32 "<u>Closing Date</u>" has the meaning as defined in the Stock Purchase Agreement.

1.33 "<u>Closing Date Plan Distributions</u>" means Distributions, and the Cash to be distributed on account thereof, to the Holders of Secured Claims, Administrative Claims, Priority Claims, and Priority Non-Tax Claims all to the extent Allowed as of the Effective Date, <u>provided</u> that in no event shall the aggregate of Closing Date Plan Distributions exceed the Plan Funding Payment.

1.34 "<u>Confirmation</u>" means the entry of the Confirmation Order on the docket of the Chapter 11 Cases, subject to all conditions specified having been (a) satisfied, or (b) waived.

1.35 "<u>Confirmation Date</u>" means the date upon which the Bankruptcy Court enters the Confirmation Order on the docket of the Chapter 11 Cases, within the meaning of Bankruptcy Rules 5003 and 9021.

1.36 "<u>Confirmation Hearing</u>" means the hearing held by the Bankruptcy Court pursuant to section 1128 of the Bankruptcy Code to consider Confirmation of the Plan, as such hearing may be adjourned or continued from time to time.

1.37 "<u>Confirmation Order</u>" means the order of the Bankruptcy Court confirming the Plan pursuant to, among others, section 1129 of the Bankruptcy Code, which shall be in form and substance reasonably acceptable to the Purchaser.

1.38 "Contract Objection" has the meaning set forth in <u>Section 8.3(b)</u> of the Plan.

1.39 "<u>Contract Objection Deadline</u>" has the meaning set forth in <u>Section 8.3(b)</u> of the Plan.

1.40 "Creditor" means any Holder of a Claim.

1.41 "<u>Cure</u>" means the Distribution of Cash, or such other property as may be agreed upon by the parties or ordered by the Bankruptcy Court, with respect to the assumption or assumption and assignment of an Executory Contract or Unexpired Lease, pursuant to Bankruptcy Code section 365(b), in an amount equal to all unpaid monetary obligations, without interest, or such other amount as may be agreed upon by the parties, under such Executory Contract or Unexpired Lease, to the extent such obligations are enforceable under the Bankruptcy Code and applicable bankruptcy law and as determined pursuant to Section 8.3 of the Plan.

1.42 "<u>Cure Amount</u>" means, for any Executory Contract or Unexpired Lease, the amount of the Cure asserted by the Debtors or the counterparty, as applicable.

1.43 "<u>Debtors</u>" means Scottish Holdings, Inc., and Scottish Annuity & Life Insurance Company (Cayman) Ltd., the debtors and debtors in possession in these Chapter 11 Cases.

1.44 "<u>Delaware DOI</u>" means the Delaware Department of Insurance.

1.45 "<u>Disallowed</u>" means, with respect to any Claim or Interest or portion thereof, any Claim against or Interest in a Debtor which is not Allowed and: (i) has been disallowed, in whole or part, by a Final Order; (ii) has been withdrawn by agreement of the Holder thereof and the applicable Debtor(s), in whole or in part; (iii) has been withdrawn, in whole or in part, by the Holder thereof; (iv) if listed in the Schedules as zero or as Disputed, contingent or unliquidated and in respect of which a Proof of Claim or a Proof of Interest, as applicable, has not been timely Filed or deemed timely Filed pursuant to the Plan, the Bankruptcy Code or any Final Order or other applicable lawApplicable Law; (v) has been reclassified, expunged, subordinated or estimated to the extent that such reclassification, expungement, subordination or estimation results in a reduction in the Filed amount of any Proof of Claim or Proof of Interest; (vi) is

evidenced by a Proof of Claim or a Proof of Interest which has been Filed, or which has been deemed to be Filed under applicable lawApplicable Law or order of the Bankruptcy Court or which is required to be Filed by order of the Bankruptcy Court but as to which such Proof of Claim or Proof of Interest was not timely or properly Filed; (vii) is unenforceable to the extent provided in section 502(b) of the Bankruptcy Code; (viii) where the holderHolder of a Claim is a Person or Entity from which property is recoverable under sections 542, 543, 550, or 553 of the Bankruptcy Code or that is a transferee of a transfer avoidable under sections 522(f), 522(h), 544, 545, 547, 548, 549, or 724(a) of the Bankruptcy Code, unless such Person, Entity or transferee has paid the amount, or turned over any such asset or property, for which such Person, Entity or transferee is liable under section 522(i), 542, 543, 550, or 553 of the Bankruptcy Code; or (ix) is for reimbursement or contribution that is contingent as of the time of allowance or disallowance of such claimClaim. In each case a Disallowed Claim is disallowed only to the extent of disallowance, withdrawal, reclassification, expungement, subordination or estimation.

1.46 "<u>Disallowed Claim</u>" means a Claim, or any portion thereof, that is Disallowed.

1.47 "<u>Disclosure Statement</u>" means the disclosure statement for the Plan, as amended, supplemented or modified from time to time, describing the Plan, that is prepared and distributed in accordance with, among others, sections 1125, 1126(b) and 1145 of the Bankruptcy Code, Bankruptcy Rule 3018 and other applicable law<u>Applicable Law</u>.

1.48 "<u>Disclosure Statement Order</u>" means the order of the Bankruptcy Court approving the Disclosure Statement and solicitation procedures with respect to the Plan<del>, which can be found at Docket No. 380.</del>

"Disputed" means, with respect to any Claim, including Priority Claims and 1.49 Administrative Claims, that has not been Allowed, (a) if no Proof of Claim has been Filed by the applicable Bar Date or has otherwise been deemed timely Filed under applicable law Applicable Law: (i) a Claim that is listed on the Debtor's Schedules as other than disputed, contingent or unliquidated, but as to which the Debtors or Reorganized Debtors or, prior to the Effective Date, any other party in interest, has Filed an objection by the Claims Objection Deadline, and such objection has not been withdrawn or denied by a Final Order; or (ii) a Claim that is listed on the Debtors' Schedules as disputed, contingent or unliquidated; or (b) if a Proof of Claim or request for payment of an Administrative Claim has been Filed by the applicable Bar Date or has otherwise been deemed timely Filed under applicable law Applicable Law: (i) a Claim for which no corresponding Claim is listed on the Debtor's Schedules; (ii) a Claim for which a corresponding Claim is listed on the Debtor's Schedules as other than disputed, contingent or unliquidated, but the nature or amount of the Claim as asserted in the Proof of Claim varies from the nature and amount of such Claim as it is listed on the Schedules; (iii) a Claim for which a corresponding Claim is listed on the Debtor's Schedules as disputed, contingent or unliquidated; (iv) a Claim for which an objection has been Filed by the Debtors or Reorganized Debtors or, prior to the Effective Date, any other party in interest, by the Claims Objection Deadline, and such objection has not been withdrawn or denied by a Final Order; or (v) a tort claim.

1.50 "<u>Disputed Claim Amount</u>" means (a) if a liquidated amount is set forth in the Proof of Claim relating to a Disputed Claim that has not been Allowed, (i) the liquidated amount set forth in the Proof of Claim relating to the Disputed Claim; (ii) an amount agreed to by the

Debtors or the Reorganized Debtors, as applicable, and the Holder of such Disputed Claim; or (iii) if a request for estimation is Filed by any party, the amount at which such Disputed Claim is estimated by the Bankruptcy Court; (b) if no liquidated amount is set forth in the Proof of Claim relating to a Disputed Claim, (i) an amount agreed to by the Debtors or the Reorganized Debtors, as applicable, and the Holder of such Disputed Claim or (ii) the amount estimated by the Bankruptcy Court with respect to such Disputed Claim; or (c) zero, if the Disputed Claim was listed on the Schedules as unliquidated, contingent or disputed and no Proof of Claim was Filed, or deemed to have been Filed, by the applicable Bar Date and the Claim has not been resolved by written agreement of the parties or an order of the Bankruptcy Court.

1.51 "Disputed Claims Reserve" means, Cash set aside by the Distribution Trustee in the amount required pursuant to Section 6.3(f)(i) of the Plan for each Disputed Claim that, as of the Effective Date, is neither an Allowed Claim nor a Disallowed Claim, and includes, without limitation, a Claim that is the subject of a timely objection or request for estimation with the Bankruptcy Court, which has not been withdrawn, settled or overruled by a Final Order; *provided, however,* if any dispute arises regarding any increase or reduction of the Disputed Claims Reserve, the Distribution Trustee shall consult with the Purchaser and the Reorganized Debtors, and obtain approval of the Bankruptcy Court, which shall have jurisdiction and power to set the amount of the reserve applying the principals of section 502(c) of the Bankruptcy Code to estimate any claim.

1.52 "<u>Distribution</u>" means any distribution pursuant to the Plan to the Holders of Allowed Claims or Interests.

1.53 "<u>Distribution Date</u>" means (a) with respect to Distributions made as Closing Date Plan Distributions, the Effective Date, (b) with respect to a DT Initial Distribution, the DT Initial Distribution Date, and (c) with respect to a DT Subsequent Distribution, the DT Subsequent Distribution Date.

1.54 "<u>Distribution Trust</u>" means the trust to be formed on or prior to the Effective Date in accordance with the provisions of <u>Section 6.3</u> of the Plan and the Distribution Trust Agreement for the payment of Allowed Claims and for the benefit of the Distribution Trust Beneficiaries as set forth herein.

1.55 "<u>Distribution Trust Agreement</u>" means the trust agreement that establishes the Distribution Trust and governs the powers, duties, and responsibilities of the Distribution Trustee. The Distribution Trust Agreement shall be part of the Plan Supplement, and shall be in form and substance reasonably acceptable to the Official Committee, the Debtors and the Purchaser.

1.56 "Distribution Trust Assets" means: (a) the Admitted SALIC/SRGL Revolver Claim; (b) all SALIC Group Services Claims incurred prior to the Petition Date; (c) any distributions or rights to distribution from SRGL or its estate that are made in connection with the Cayman Islands Winding Up Proceeding at any time on account of (i) the Admitted SALIC/SRGL Revolver Claim and/or (ii) any SALIC Group Services Claims incurred prior to the Petition Date; (d) all DT Post-Closing Rights; (e) the SFL Shares; and (f) any and all Causes of Action against the SRGL Equity Holders. For the avoidance of doubt, any SALIC Group
Services Claims incurred on or after the Petition Date are not Distribution Trust Assets and, instead, shall re-vest in the Reorganized Debtors.

1.57 "<u>Distribution Trust Asset Proceeds</u>" means the proceeds of the Distribution Trust Assets, net of expenses incurred in preserving and/or monetizing such Distribution Trust Assets.

1.58 "<u>Distribution Trust Beneficiaries</u>" means the Holders of Allowed Claims in Classes 4, 5, 6, and 7, whether Allowed on or after the Effective Date.

1.59 "<u>Distribution Trust Reserves</u>" means the Disputed Claims Reserve, the Professional Fee Reserve, and the Trust Administration Reserve.

1.60 "<u>Distribution Trustee</u>" the Person appointed under the Plan and Distribution Trust Agreement to administer the Distribution Trust, which Person shall be selected by the Official Committee and reasonably acceptable to the Debtors and the Purchaser; *provided, however*, that the Distribution Trustee shall not take any action inconsistent with the purposes of the Distribution Trust and the qualification of the Distribution Trust as a "liquidating trust" for U.S. federal income tax purposes.

1.61 "<u>DT Initial Distribution</u>" means the Distribution set forth in section  $6.3(\underline{i})(\underline{i})$  of the <u>Plan</u>.

1.62 "<u>DT Initial Distribution Date</u>" means, for Distributions of New Equity, the Effective Date, and for Distributions of Cash, the date that is as soon as reasonably practicable after the Administrative Claims Bar Date, <u>but in no event later than thirty (30) days after the Administrative Claims Bar Date</u>, when Distributions from the Distribution Trustof New Equity and Cash (as applicable) shall commence to Holders of Allowed Claims.

1.63 "<u>DT Subsequent Distribution</u>" means after the completion of the DT Initial Distribution, the distributionsDistributions to the Holders of Allowed Claims in Classes 4, 5, 6 and 7 on account of their Allowed Claims from Distribution Trust Asset Proceeds; *provided*, *however*, as set forth in the Plan, that the Distribution Trustee shall not be required to make a Distribution pursuant to this Section 6.3(i) of the Plan if the Distribution Trustee determines that the expense associated with making the Distribution would likely utilize a substantial portion of the amount to be distributed, thus making the Distribution impracticable.

1.64 "<u>DT Subsequent Distribution Date</u>" means any later date that a Distribution not made on the DT Initial Distribution Date is made.

1.65 "<u>DT Post-Closing Rights</u>" has <u>the</u> meaning as defined in the Stock Purchase Agreement.

1.66 "Effective Date" means any Business Day following the date on which all conditions to consummation set forth in Section 9.1 of the Plan have been satisfied or, if capable of being duly and expressly waived, as provided in Section 9.2 of the Plan, any conditions to the occurrence of consummation set forth in the Plan have been satisfied or waived.

1.67 "<u>Eligible SALIC TruPS Claims</u>" means all TruPS <u>ClaimClaims</u> against SALIC arising from or relating to the SHST I TruPS, the SHST III TruPS or the SFLST I TruPS.

1.68 "<u>Eligible SHI TruPS Claims</u>" means all TruPS <u>ClaimClaims</u> against SHI arising from or relating to the SHST I TruPS or the SHST III TruPS.

1.69 "<u>Entity</u>" means a Person, estate, trust, Governmental Unit, and U.S. Trustee, within the meaning of Bankruptcy Code section 101(15).

1.70 "<u>Estates</u>" means the estates of the Debtors in the Chapter 11 Cases created pursuant to section 541 of the Bankruptcy Code.

1.71 "<u>Exculpated Parties</u>" means (a) the Debtors, (b) the Reorganized Debtors, (c) the Official Committee (including the current and former Official Committee members in their capacities as such); (d) the Voting Agent; and (e) for each of the foregoing, their respective Representatives, each to the extent that they held such office or capacity during the pendency of the Chapter 11 Cases; *provided, however*, that, the SRGL Equity Holders shall not be an Exculpated Party or a Representative of an Exculpated Party.

1.72 "<u>Executory Contract</u>" means any contract or Unexpired Lease to which the Debtor is a party that is subject to assumption or rejection under section 365 of the Bankruptcy Code.

1.73 "Face Amount" means the outstanding amount of any Certificate or Claim.

1.74 "<u>File</u>," "<u>Filed</u>" or "<u>Filing</u>," means, respectively, file, filed or filing with the Bankruptcy Court in the Chapter 11 Cases; <u>provided</u>, <u>however</u>, that with respect to Proofs of Claim and Proofs of Interest only, "Filed" shall mean delivered and received in the manner provided by the Bar Date Order or as otherwise established by order of the Bankruptcy Court.

1.75 "<u>Final Order</u>" means an order, ruling, judgment, the operation or effect of a judgment or other decree issued and entered by the Bankruptcy Court or by any state or other federal court or other court of competent jurisdiction which has not been reversed, vacated, stayed, modified or amended and as to which (i) the time to appeal or petition for review, rehearing, certiorari, reargument or retrial has expired and as to which no appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal or petition for review, rehearing, certiorari, reargument or retrial has been finally decided and no further appeal and provided, however, that the possibility that a motion under Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules, may be filed with respect to such order or judgment shall not cause such order or judgment not to be a Final Order.

1.76 "<u>General Unsecured Claim</u>" means any unsecured Claim that is not an Administrative Claim, Priority Claim, Priority Tax Claim, Professional Fee Claim, Secured Claim, a claim for U.S. Trustee Fees, Intercompany Claim, or Subordinated Claim.

1.77 "<u>General Unsecured Creditor</u>" means any Holder of a General Unsecured Claim.

## Case 18-10160-LSS Doc 437-2 Filed 08/09/18 Page 84 of 105

1.78 <u>"Governmental Unit</u>" has the meaning of such term under Bankruptcy Code section 101(27).

1.79 "<u>GPIC TruPS</u>" has the meaning ascribed to such term in the definition of "TruPS".

1.80 "<u>GPIC TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

<u>1.81</u> <u>"Hildene Master Funds" means Hildene Opportunities Master Fund, Ltd. and Hildene Opportunities Master Fund II, Ltd.</u>

<u>1.82</u> <u>1.81</u>-"<u>Holder</u>" means the legal or beneficial holder of a Claim or Interest (and, when used in conjunction with a Class or type of Claim or Interest, means a Holder of a Claim or Interest in such Class or of such type). Additionally, in reference to the Distribution Trust, the term "Holder" means the legal or beneficial holder of an Allowed Claim.

<u>1.83</u> <u>1.82</u> "<u>Impaired</u>" means, when used with reference to a Claim or Interest, a Claim or Interest that is impaired within the meaning of section 1124 of the Bankruptcy Code.

<u>1.84</u> <u>1.83</u> "<u>Impaired Class</u>" means a Class of Claims or Interests that are Impaired.

<u>1.85</u> <u>1.84</u>-"<u>Indemnified D&O Parties</u>" means each present and former director and officer of each of the SALIC Group Companies, as set forth in <u>Section 5.8(a) of the Stock</u> <u>Purchase Agreement.</u>

<u>1.86</u> <u>1.85</u>-"<u>Indemnification Obligation</u>" means any Claim against or obligation of the Debtors to indemnify, reimburse, advance expenses or provide contribution to or with respect to any present or former officers, directors or employees pursuant to by-laws, articles of incorporation, agreements, contracts, common law or otherwise as may be in existence immediately prior to the Effective Date.

1.86 "Indenture Trustee Fee Reserve" means the reserve to be established at the 1.87 Distribution Trust and funded on or before the Effective Date in an amount that the Debtorsestimate in good faith, after consultation with the Purchaser, the Official Committee, and the relevant TruPS Indenture Trustees, to be necessary to pay in full, but subject to the relevant Indenture Trustee Fee Caps, (a) any amounts payable to the TruPS Indenture Trustees for Indenture Trustee Fees incurred prior to the Effective Date but not paid to the relevant Indenture-Trustees as Closing Date Plan Distributions and (b) any amounts reasonably estimated to beincurred after the Effective Date for TruPS Indenture Trustees for the SHST I Debentures, the SHST III Debentures and the SFLST I Debentures. For the avoidance of doubt, the Distribution Trustee shall be under no obligation to reserve any amount in the Indenture Trustee Fees Reserveon account of post-Effective Date Indenture Trustee Fees that may be incurred by the TruPS-Indenture Trustees for the SHST II Debentures or the GPIC Debentures. 1.87-"Indenture Trustee Fees" means the reasonable and documented fees and out-of-pocket expenses of the TruPS Indenture Trustees that are recoverable by such TruPS Indenture Trustees in accordance with the terms of their respective TruPS Indentures. For the avoidance of doubt, the Indenture-Trustee Fees are subject in all respects to the applicable Indenture Trustee Fee Cap.

1.88 <u>"Indenture Trustee Fee Cap</u>" means the cap on the aggregate amount of Indenture Trustee Fees incurred by Wilmington Trust Company, U.S. Bank, and Bank of New York Melonin connection with their respective roles as TruPS Indenture Trustees, which shall be allocated inspecific amounts among the three TruPS Indenture Trustees and shall be set forth in the Plan-Supplement.

<u>1.88</u> <u>1.89</u> "<u>Insider</u>" has the meaning set forth in section 101(31) of the Bankruptcy Code.

<u>1.89</u> <u>1.90</u> "<u>Intercompany Claims</u>" means all Claims between or among any Debtors and any other SALIC Group Company. For the avoidance of doubt, Claims by or against SFL and SRGL are not Intercompany Claims for <u>purposepurposes</u> of the Plan.

<u>1.90</u> <u>1.91</u> "<u>Interest</u>" means the legal, equitable, contractual interests, equity interests or ownership interests, or other rights of any Person in the Debtors including all capital stock, stock certificates, common stock, preferred stock, partnership interests, limited liability company or membership interests, rights, treasury stock, options, warrants, contingent warrants, convertible or exchangeable securities, investment securities, subscriptions or other agreements and contractual rights to acquire or obtain such an interest or share in the Debtors, partnership interests in the Debtors' stock appreciation rights, conversion rights, repurchase rights, redemption rights, dividend rights, preemptive rights, subscription rights and liquidation preferences, puts, calls, awards or commitments of any character whatsoever relating to any such equity, common stock, preferred stock, ownership interests or other shares of capital stock of the Debtors or obligating the Debtors to issue, transfer or sell any shares of capital stock whether or not certificated, transferable, voting or denominated "stock" or a similar security.

<u>1.91</u> <u>1.92</u> "<u>IT Fee Notice Parties</u>" means (a) for any notice to be provided or service to be made as of a date prior to the Effective Date, (i) the Debtors, (ii) the Official Committee, (iii) the Purchaser, and (iv) the U.S. Trustee, and (b) for any notice to be provided or service to be made as of a date on or after the Effective Date, (i) the Reorganized Debtors, (ii) the Distribution Trustee, (iii) the Purchaser, and (iv) the U.S. Trustee.

<u>1.92</u> <u>1.93</u> "Joint Liquidators" has the meaning set forth in the Restructuring Implementation Agreement.

1.94—"<u>KBW</u>" means Keefe, Bruyette & Woods, Inc., the Debtors' investment banker.

1.95—"<u>KBW Reserved Funds</u>" means, as contemplated by Paragraph 3 of the KBW Retention Order, Cash set aside in the amount of \$300,000 by the Debtors in the client trust account of their bankruptcy counsel on account of Monthly Service Fees otherwise due KBW for March, April and May, 2018.

1.96—"<u>KBW Retention Order</u>" means that certain Order under Sections 327(a), 328(a) and 1107(b) of the Bankruptcy Code, Bankruptcy Rules 2014 and 2016, and Local Rules 2014-1and 2016-2(h) Authorizing Retention and Employment of Keefe, Bruyette & Woods, Inc. as Debtors' Investment Banker <u>nunc pro tunc</u> to the Petition Date, entered in the Chapter 11 Cases on March 12, 2018 (D.I. 155). <u>1.93</u> <u>1.97</u> "<u>Lien</u>" means, with respect to any asset or property (or the rents, revenues, income, profits or proceeds therefrom), and in each case, whether the same is consensual or nonconsensual or arises by contract, operation of law, legal process or otherwise: (a) any and all mortgages or hypothecation to secure payment of a debt or performance of an obligation, liens, pledges, attachments, charges, leases evidencing a capitalizable lease obligation, conditional sale or other title retention agreement, or other security interest or encumbrance or other legally cognizable security devices of any kind in respect of any asset or property, or upon the rents, revenues, income, profits or proceeds therefrom; or (b) any arrangement, express or implied, under which any asset or property is transferred, sequestered or otherwise identified for the purpose of subjecting or making available the same for the payment of debt or performance of any other obligation in priority to the payment of general unsecured<u>General Unsecured</u> Creditors.

<u>1.94</u> <u>1.98</u> "<u>Mass Mutual</u>" means Massachusetts Mutual Life Insurance Company, MassMutual Capital Partners LLC, Benton Street Partners I, L.P., Benton Street Partner II, L.P., and Benton Street Partners III, L.P.

<u>1.95</u> <u>1.99</u> "<u>New Corporate Governance Documents</u>" means the corporate governance documents for the Reorganized Debtors and New Holdco (if applicable), including charters, bylaws, memoranda and article of association, operating agreements, or other organization or formation documents, as applicable, including, but not limited to, the Stockholders Agreement. For the avoidance of doubt, all New Corporate Governance Documents shall be in form and substance acceptable to the Purchaser in the Purchaser's sole discretion (other than that such New Corporate Governance Documents must not be materially inconsistent with the terms of the Plan).

<u>1.96</u> <u>1.100</u> "<u>New Equity</u>" means (i) if the Purchaser elects for SALIC to issue New SALIC Equity pursuant to the Plan, the New SALIC Equity; or (ii) if the Purchaser elects in accordance with Section 6.1(f) of this Plan to form New Holdco and for New Holdco to issue its voting capital stock the voting capital stock of or membership interests, the voting capital stock of or membership interests in New Holdco, as applicable in New Holdco</u>.

<u>1.97</u> <u>1.101</u> "<u>New Equity Election</u>" means an election made by a New Equity Eligible Holder to take its Distribution in the form of Offered New Equity, which shall be made by checking the "New Equity <u>DistributionElection</u>" box on its timely completed and submitted Ballot. In the event that a New Equity Eligible Holder fails to make a New Equity Election or fails to submit a Ballot, then such New Equity Eligible Holder shall be deemed to have made the Cash Election.

<u>1.98</u> <u>1.102</u> "<u>New Equity Eligible Beneficial Holder</u>" means a Beneficial Holder of TruPS other than SRGL.

<u>1.99</u> <u>1.103</u> "<u>New Equity Eligible Holders</u>" means (i) all New Equity Eligible Beneficial Holders and (ii) the SFL Receiver with respect to the SFL Note Claim, the extent that the SFL Note Claim Allowance Conditions have been satisfied.

<u>1.100</u> <u>1.104</u> "<u>New Holdco</u>" means a newly organized Entity, which, if organized by the Purchaser, shall have 100% of the New SALIC Equity as its primary asset and shall engage in no business other than to act as a holding company for such New SALIC Equity.

<u>1.101</u> <u>1.105</u> "<u>New SALIC Equity</u>" means 19,999,999 Ordinary Shares of SALIC to be issued under the terms of the Plan, the Restructuring Implementation Agreement, the Share Surrender Documents, the New SALIC Shares Issuance Documents, and the Stock Purchase Agreement.

<u>1.102</u> 1.106 "New SHI Equity" means 100% of the new common stock of SHI to be issued to, or at the direction of, the Purchaser.

<u>1.103</u> <u>1.107</u> "<u>New SALIC Shares Issuance Documents</u>" means each of the documents, directions and resolutions reasonably required of SALIC to effectuate the New SALIC Shares Issuance (as such term is defined in the Restructuring Implementation Agreement). The New SALIC Shares Issuance Documents shall be in form and substance reasonably satisfactory to the Purchaser, and otherwise subject to the SRGL Consent Rights.

<u>1.104</u> <u>1.108</u> "<u>Offered New Equity</u>" refers to the thirty percent (30%) of <u>the</u> New Equity to be offered to New Equity Eligible Holders pursuant to the Plan.

<u>1.105</u> <u>1.109</u> "<u>Official Committee</u>" means the Official Committee of Unsecured Creditors appointed by the U.S. Trustee in the Chapter 11 Cases pursuant to section 1102(a) of the Bankruptcy Code, as the membership of such committee is constituted and reconstituted from time to time.

<u>1.106</u> <u>1.110</u> "<u>Ordinary Shares</u>" shall have the meaning ascribed to such term in the Stock Purchase Agreement.

<u>1.107</u> <u>1.111</u> "<u>Orkney Re II</u>" means Orkney Re II plc.

1.108 <u>1.112</u> "Person" means and includes a natural person, individual, partnership, corporation (as defined in section 101(a) of the Bankruptcy Code), or organization including, without limitation, corporations, limited partnerships, limited liability companies, general partnerships, joint ventures, joint stock companies, trusts, land trusts, business trusts, unincorporated organizations or associations, Official Committee, or ad hoc committee, or other organizations, irrespective of whether they are legal entities, governmental bodies (or any agency, instrumentality or political subdivision thereof), or any other form of legal entities; provided, however, the term "Person" does not include Governmental Units, except that a Governmental Unit that (a) acquires an asset from a Person (i) as a result of the operation of a loan guarantee agreement or (ii) as receiver or liquidating agent of a Person; (b) is a guarantor of a pension benefit payable by or on behalf of a Debtor or an Affiliate of a Debtor; or (c) is the legal or beneficial owner of an asset of (i) an employee pension benefit plan that is a governmental plan, as defined in section 414(d) of the Internal Revenue Code of 1986 or (ii) an eligible deferred compensation plan, as defined in section 457(b) of the Internal Revenue Code of 1986, shall be considered for purposes of section 1102 of the Bankruptcy Code to be a Person with respect to such asset or such benefit.

<u>1.109</u> <u>1.113</u>-"<u>Petition Date</u>" means January 28, 2018, the date on which each Debtor Filed its petition for relief commencing the Chapter 11 Cases.

<u>1.110</u> <u>1.114</u> "<u>Plan</u>" means the <u>Third Amended</u> Joint Chapter 11 Plan of Reorganization of Scottish Holdings, Inc., and Scottish Annuity & Life Insurance Company (Cayman) Ltd., as it may be altered, amended, modified or supplemented from time to time including in accordance with its terms, any Plan Supplement, the Bankruptcy Code or the Bankruptcy Rules.

<u>1.111</u> <u>1.115</u>-"<u>Plan Documents</u>" means, collectively, those material documents executed or to be executed in order to consummate the transactions contemplated under the Plan, including without limitation the Plan Supplement, which shall be Filed with the Bankruptcy Court in advance of the Confirmation Hearing, which shall be in form and substance reasonably acceptable to the Purchaser.

<u>1.112</u> <u>1.116</u> "<u>Plan Funding Payment</u>" means an amount equal to twenty-one million five hundred thousand dollars (\$21,500,000), subject to downward adjustment in an amount equal to the TruPS Returned Cash, which shall be funded by Purchaser to the <u>Reorganized SALIC which</u> <u>shall deliver such amount to the</u> Distribution Trust (subject to the payments and reserves provided for in Sections 6.1(b)(1) through (5) of the Plan on the Effective Date as provided in the <u>SPAStock Purchase Agreement</u> and as a condition to occurrence of the Effective Date].

1.117—"<u>Plan Sponsorship Agreement</u>" means that certain Plan Sponsorship Agreement among the Stalking Horse and the Debtors, dated as of January 28, 2018, and approved by order of the Bankruptcy Court entered on February 27, 2018, at Docket No. 115.

**1.113 1.118** "Plan Supplement" means the supplement to the Plan to be Filed hereafter to supplement or clarify aspects of the Plan, and which shall include, among others: (a) the following documents, each of which shall be in form and substance acceptable to the Purchaser, (i) the Rejection Schedule, (ii) the New Corporate Governance Documents, (iii) the identity of any individual proposed to serve, after Confirmation of the Plan, as a director, officer, or voting trustee of the Reorganized Debtors, (iv) the identity of any insider that will be employed or retained by the Reorganized Debtors and the nature of any compensation for such insider, (v) a notice stating whether the New Equity will be issued by Reorganized SALIC or New Holdco, and (vi) the Stockholders Agreement, and (vii) the Indenture Trustee Fee Cap; and (b) the Distribution Trust Agreement, which shall be in form and substance reasonably acceptable to the Debtors, the Official Committee and the Purchaser; and (c) the identity of the Distribution Trustee's proposed compensation if not already set forth in the Distribution Trust Agreement.

1.119 <u>"Pre-Effective Date Indenture Trustee Fee Estimate</u>" means any amounts reasonably estimated to be incurred after the Confirmation Date for the TruPS Indenture Trustees for Indenture Trustee Fees incurred prior to the Effective Date but not paid to the relevant Indenture Trustees on the Effective Date.

1.120—"<u>Post-Effective Date Indenture Trustee Fee Estimate</u>" means any amountsreasonably estimated to be incurred after the Effective Date for TruPS Indenture Trustees for the SHST I Debentures, the SHST III Debentures and the SFLST I Debentures ("Post-Effective Date-Indenture Trustee Fees").

<u>1.114</u> <u>1.121</u> "<u>Priority Claims</u>" means any and all Priority Tax Claims and Priority Non-Tax Claims.

<u>1.115</u> <u>1.122</u> "<u>Priority Non-Tax Claim</u>" means any and all Allowed Claims accorded priority in right of payment under section 507(a) of the Bankruptcy Code, other than a Priority Tax Claim.

<u>1.116</u> <u>1.123</u> "<u>Priority Tax Claim</u>" means any and all Claims of a Governmental Unit of the kind specified in section 507(a)(8) of the Bankruptcy Code.

<u>1.117</u> "Post-Effective Date Indenture Trustee Fee Reserve" means the reserve to be established by the Distribution Trust in the following amounts: (a) the amount of any disputed Indenture Trustee Fees incurred through the Effective Date and (b) \$50,000 in aggregate for Indenture Trustee Fees incurred by the TruPS Indenture Trustees post-Effective Date with respect the SHST I TruPS, the SHST III TruPS, and the SFLST I TruPS as provided for in Section 4.1(d)(ii) of the Plan. For the avoidance of doubt, the Distribution Trustee Fee Reserve on account of post-Effective Date Indenture Trustee Fees that may be incurred by the TruPS Indenture Trustees for the SHST II Debentures or the GPIC Debentures.

<u>1.118</u> <u>1.124</u> "<u>Professional</u>" means any professional employed in these Chapter 11 Cases pursuant to Bankruptcy Code sections 327, 328, or 1103, or for which compensation and reimbursement has been Allowed by the Bankruptcy Court pursuant to section 503(b)(4) of the Bankruptcy Code.

<u>1.119</u> <u>1.125</u> "<u>Professional Fee Claim</u>" means a Claim of a Professional for compensation for services rendered or reimbursement of costs, expenses, or other charges incurred after the Petition Date and on or before the Effective Date.

<u>1.120</u> <u>1.126</u> "<u>Professional Fee Reserve</u>" means a professional fee reserve to be maintained by the Distribution Trust in an amount mutually agreed by the Debtors, the Official Committee, and the Purchaser, and estimated in good faith, after consultation with the relevant Professionals, to be necessary to pay in full all amounts then owing or that may later become owing to such Professionals for professional fees and expenses incurred through the Effective Date.

<u>1.121</u> <u>1.127</u> "<u>Proof of Claim</u>" means a proof of claim Filed with the Bankruptcy Court or its duly appointed claims agent in connection with the Chapter 11 Cases pursuant to section 501 of the Bankruptcy Code.

<u>1.122</u> <u>1.128</u> "<u>Purchaser</u>" means Hildene Re Holdings, LLC, and its permitted designees, successors and assigns.

<u>1.123</u> <u>1.129</u> "<u>Recapitalization Funding Payment</u>" means an amount equal to twelve million five hundred thousand dollars (\$12,500,000), which shall be utilized by Reorganized

SALIC (x) to recapitalize the SALIC Group Companies after Closing, as such amount is reduced on a dollar-for-dollar basis by any Cure Amounts paid and (y) used by the Debtors to payCure Amounts on behalf of the Purchaser in excess of \$100,000 in accordance with Sectionsection 2.2(b) of the Stock Purchase Agreement.

<u>1.124</u> <u>1.130-"Reinstate," "Reinstated</u>" or "<u>Reinstatement</u>" means with respect to Claims and Interests, that the Claim or Interest shall be rendered unimpaired in accordance with section 1124 of the Bankruptcy Code.

<u>1.125</u> <u>1.131</u> "<u>Rejection Schedule</u>" means that certain schedule initially to be <u>filedFiled</u> as part of the Plan Supplement and identifying any Executory Contract or Unexpired Lease to be rejected by the Debtors effective as of the Effective Date. Except as otherwise provided in the Stock Purchase Agreement, the Rejection Schedule may be modified at any time prior to the Effective Date, including to add or remove Executory Contracts or Unexpired Leases to be rejected by the Debtors.

<u>1.126</u> <u>1.132</u> <u>"Rejection Claim Bar Date</u>" means the date that is <u>thirty (30twenty-one</u> (21) days after the Effective Date; *provided, however*, to the extent that the Bankruptcy Court grants a motion to reject an Executory Contract or Unexpired Lease <u>filedFiled</u> after the Effective Date as permitted by <u>Section 8.3(b)</u> of the Plan, the Rejection Claim Bar Date for such contract or lease shall be thirty (30) days after the date of entry of the Bankruptcy Court's order granting the post-Effective Date rejection of such contract or lease.

<u>1.127</u> <u>1.133</u> "<u>Released Parties</u>" means: (a) the Debtors; (b) the Reorganized Debtors; (c) the Official Committee (including the current and former Official Committee members in their capacities as such); (d) the Purchaser; (e) the <u>Hildene Master Funds; (f) the Stalking Horse; (fg</u>) the Joint Liquidators; (gh) SRGL; (hi) the Voting Agent; and (ij) for each of the foregoing, all of their respective Representatives; *provided, however,* that "Released Parties" specifically excludes (x) the SRGL Equity Holders, and their respective Representatives, in their capacity as such and (y) the Specified SALIC Officers and Directors.</u>

<u>1.128</u> <u>1.134</u> "<u>Reorganized Debtors</u>" means Reorganized SALIC and Reorganized SHI, and for the avoidance of doubt, includes New Holdco.

<u>1.129</u> <u>1.135</u> "<u>Reorganized SALIC</u>" means SALIC as reorganized upon the Effective Date pursuant to the Plan.

<u>1.130</u> <u>1.136</u>-"<u>Reorganized SHI</u>" means SHI as reorganized upon the Effective Date pursuant to the Plan.

<u>1.131</u> <u>1.137</u> "<u>Representatives</u>" means, with respect to an Entity, all of that Entity's current and former managed and controlled affiliates, subsidiaries, officers, directors, managers, managing members, principals, shareholders, members, partners, employees, agents, advisors, attorneys, professionals, accountants, investment bankers, consultants and other representatives and such persons' respective heirs, executors, estates, servants and nominees, in each case in their capacity as such.

<u>1.132</u> <u>1.138</u> "<u>Restructuring Documents</u>" has the meaning set forth in the Restructuring Implementation Agreement.

<u>1.133</u> <u>1.139</u> "<u>Restructuring Implementation Agreement</u>" means that certain Restructuring Implementation Agreement among the Debtors and SRGL, made January 28, 2018, and approved by the RIA Order, as such Restructuring Implementation Agreement may be amended from time to time and as confirmed by SRGL in accordance with section 6.1(c) of the Stock Purchase Agreement. A copy of the Restructuring Implementation Agreement is annexed to the Plan as <u>Exhibit C-1</u> and the evidence of such confirmation is annexed to the Plan as <u>Exhibit C-2</u>.

<u>1.134</u> <u>1.140</u> "<u>Retained Causes of Action</u>" shall have the meaning set forth in <u>Section 6.5</u> of the Plan.

<u>1.135</u> <u>1.141</u> "<u>RIA Order</u>" means the *Order Authorizing Debtors To Assume Restructuring Implementation Agreement And Related Relief* entered by the Bankruptcy Court on March 19, 2018, at Docket No. 170 (as may be amended).

<u>1.136</u> <u>1.142</u>-"<u>SALIC</u>" means Scottish Annuity & Life Insurance Company (Cayman) Ltd.

<u>1.137</u> <u>1.143</u> "<u>SALIC Existing Equity Interests</u>" means all issued and outstanding Ordinary Shares of SALIC existing prior to the Effective Date.

<u>1.138</u> <u>1.144</u>."<u>SALIC General Unsecured Claim</u>" means any Claim against SALIC that is not an Administrative Claim, Secured Claim, Priority Claim, Intercompany Claim, TruPS Claim, SFL Claim, SFL Note Claim, or Subordinated Claim.

<u>1.139</u> <u>1.145</u> "<u>SALIC Group Companies</u>" means SALIC, SHI, SRD, SRUS and SRLB.

<u>1.140</u> <u>1.146-"SALIC Group Services Claims</u>" has the meaning ascribed to such term in the Restructuring Implementation Agreement.

<u>1.141</u> <u>1.147</u> "<u>SALIC Net Worth Maintenance Agreements</u>" means (i) that certain Net Worth Maintenance Agreement, dated as of January 1, 2002, by and between SRD and SALIC, as amended from time to time, and (ii) that certain Net Worth Maintenance Agreement, restated as of February 1, 2002, by and between SRUS and SALIC, as amended from time to time.\_

<u>1.142</u> <u>1.148-</u>"<u>SALIC TruPS Claim</u>" means, other than any Subordinated Claim, any and all TruPS Claims related to the TruPS Parent Guarantees, but does not include Indenture Trustee Fees.

<u>1.143</u> <u>1.149</u> "<u>SALIC Claims</u>" has the meaning as set forth in the Restructuring Implementation Agreement.

<u>1.144</u> <u>1.150</u> "<u>Schedules</u>" means the schedules of assets and liabilities, schedules of Executory Contracts and Unexpired Leases and statements of financial affairs Filed by the Debtors pursuant to section 521 of the Bankruptcy Code and in substantial accordance with the

Official Bankruptcy Forms, as the same may have been amended, modified or supplemented from time to time.

<u>1.145</u> <u>1.151</u> "<u>Scottish Re</u>" means the Debtors collectively with their non-debtor affiliates.

<u>1.146</u> <u>1.152</u>-"<u>Secured Claim</u>" means a Claim that is secured by a Lien which is not subject to avoidance under the Bankruptcy Code or otherwise invalid under the Bankruptcy Code or applicable state law, on property in which the Estate has an interest, or a Claim that is subject to setoff under section 553 of the Bankruptcy Code; to the extent of the value of the Holder's interest in the Estate's interest in such property or to the extent of the amount subject to setoff, as applicable; as determined by a Final Order pursuant to section 506(a) of the Bankruptcy Code, or in the case of setoff, pursuant to section 553 of the Bankruptcy Code, or in either case as otherwise agreed upon in writing by the Debtors or the Reorganized Debtors and the Holder of such Claim. The amount of any Claim that exceeds the value of the Holder's interest in the Estate's interest in property or the amount subject to setoff shall be treated as an SHI General Unsecured Claim or SALIC General Unsecured Claim, as applicable.

<u>1.147</u> <u>1.153</u> "Security Depository or Custodian" means a securities depository or custodian or any nominee for such a securities depository or custodian, *e.g.*, Depository Trust Company and its nominee Cede & Co.

1.148 "SFL" means Scottish Financial (Luxembourg) S.á r.l.

<u>1.149</u> <u>1.154</u>-"<u>SFL Bankruptcy Administration Senior Cash Election</u>" means election exercisable by the SFL Receiver on or before the Voting Deadline applicable to the SFL Note Claim to designate that a portion of the SFL Note Claim should receive its Distribution in the form of Cash. The SFL Bankruptcy Administration Senior Cash Election is made available to the SFL Receiver to ensure that the SFL Receiver, notwithstanding any inconsistent New Equity Election having been made by one or more of the SFLST I TruPS CDO Facility Holders, will have sufficient Cash available to fulfill the SFL Receiver's duties as the insolvency receiver for SFL in its bankruptcy proceeding in Luxembourg. Once the initial SFL Bankruptcy Administration Senior Cash Election, it may not thereafter be modified, except with the prior written consent of each of the SFLST I TruPS CDO Facility Holders.

1.155—"<u>SFL</u>" means Scottish Financial (Luxembourg) S.á r.l.

<u>1.150</u> <u>1.156</u> "<u>SFL Claims</u>" <u>means</u> any and all Claims and Causes of Action owned or held by SFL or the SFL Receiver on SFL's behalf against SALIC or SHI, including, but not limited, to the SFL Note Claim.

<u>1.151</u> <u>"SFL Distribution Agreement" means that certain agreement entered into between</u> <u>SFL and the Debtors on or prior to the Effective Date setting forth the mechanics for</u> distributions made on account of the SFL Note Claim.

<u>1.152</u> <u>1.157</u> "<u>SFL Note</u>" means that certain Floating Rate Junior Subordinated Deferrable Interest Debenture of Scottish Re (Dublin) Limited, dated December 15, 2004, as

amended, in the original principal amount of fifty-one million five hundred and forty-seven thousand dollars (\$51,547,000) issued by SRD, as obligor, to SFL, as obligee, and subsequently assigned to, and assumed by, SALIC, as obligor, as such note may have been amended from time to time.

<u>1.153</u> <u>1.158</u> "<u>SFL Note Claim</u>" <u>meanmeans</u> any and all Claims and Causes of Action against any of the Debtors arising out of or relating to the SFL Note, including, but not limited to, any amendment to the SFL Note or any payment made or not made on account of the SFL Note.

<u>1.154</u> <u>1.159</u> "SFL Note Claim Allowance Conditions" has the meaning ascribed to such term in Section 4.3(c)(ii)(B) of this the Plan.

<u>1.155</u> <u>1.160</u> "<u>SFL Receiver</u>" means Max Mailliet, as insolvency receiver for SFL in its bankruptcy proceeding in Luxembourg or any successor thereto.

<u>1.156</u> <u>1.161</u> "<u>SFL Shares</u>" means the 47,046 shares of SFL, representing all of the issued and outstanding equity interest in SFL.

<u>1.157</u> <u>1.162-</u>"<u>SFLST I TruPS</u>" has the meaning ascribed to such term in the definition of "TruPS".

<u>1.158</u> <u>1.163</u> "<u>SFLST I TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

<u>1.159</u> <u>1.164</u> <u>"SFLST I TruPS CDO Facility Holders</u>" means the Preferred Term Securities XVI, the Preferred Term Securities XVII <u>(Cusip</u> and the Preferred Term Securities XVIII collateraized debt obligation facilities, each as a Beneficial Holder of SFLST I TruPS, holding through BNYM, as nominee.

<u>1.160</u> <u>1.165</u> "<u>Share Surrender Documents</u>" has the meaning set forth in the Restructuring Implementation Agreement.

<u>1.161</u> <u>1.166</u> "<u>SHI</u>" means Scottish Holdings, Inc.

<u>1.162</u> <u>1.167</u> "<u>SHI Existing Equity Interests</u>" means all issued and outstanding common shares of SHI existing prior to the Effective Date.

<u>1.163</u> <u>1.168</u> "<u>SHI General Unsecured Claim</u>" means any Claim against SHI that is not an Administrative Claim, Secured Claim, Priority Claim, Intercompany Claim, TruPS Claim, SFL Claim, SFL Note Claim, or Subordinated Claim.

<u>1.164</u> <u>1.169</u> "<u>SHI TruPS Claim</u>" means, other than any Subordinated Claim, any and all TruPS Claims related to the SHST I TruPS, the SHST II TruPS, the GPIC TruPS, and the SHST III TruPS but does not include Indenture Trustee Fees.

<u>1.165</u> <u>1.170-"SHST I TruPS</u>" has the meaning ascribed to such term in the definition of "<u>TruPS</u>".

### Case 18-10160-LSS Doc 437-2 Filed 08/09/18 Page 94 of 105

<u>1.166</u> <u>1.171</u> "<u>SHST I TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

<u>1.167</u> <u>1.172</u> "<u>SHST II TruPS</u>" has the meaning ascribed to such term in the definition of "TruPS".

<u>1.168</u> <u>1.173-</u>"<u>SHST II TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

<u>1.169</u> <u>1.174</u> "<u>SHST III TruPS</u>" has the meaning ascribed to such term in the definition of "TruPS".

<u>1.170</u> <u>1.175-</u>"<u>SHST III TruPS Debentures</u>" has the meaning ascribed to such term in the definition of "<u>TruPS Debentures</u>".

<u>1.171</u> <u>1.176</u> "<u>SLD</u>" means Security Life of Denver Insurance Company, a claimant in these Chapter 11 Cases.

<u>1.172</u> <u>1.177</u> "<u>Specified Restructuring Documents</u>" has the meaning set forth in the Restructuring Implementation Agreement.

<u>1.173</u> <u>"Specified SALIC Officers and Directors" means any former officer or director of</u> <u>SALIC who held said position at any point during the period January 1, 2011 through December</u> <u>31, 2013, exclusive of Gregg Klingenberg and Mark Harris (who are each a Released Party).</u>

<u>1.174</u> <u>1.178</u> "<u>SRD</u>" means Scottish Re (Dublin) dac.

<u>1.175</u> <u>1.179</u> "<u>SRGL</u>" means Scottish Re Group Limited.

<u>1.176</u> <u>1.180</u> "<u>SRGL Consent Rights</u>" has the meaning set forth in the Restructuring Implementation Agreement.

<u>1.177</u> <u>1.181</u>-"<u>SRGL Equity Holders</u>" means each Person that (i) holds or at any time held any direct or indirect equity interest of any kind in SRGL, including without limitation Cerberus, MassMutual and their current and former affiliates<u>and</u>, subsidiaries and assigns<u>in</u> <u>each case</u>, holding or having held a direct or indirect equity interest of any kind in SRGL, or (ii) at any time received directly or indirectly any Transfer (or proceeds thereof) from any Debtors pursuant to, or on account of any such direct or <u>indirect</u> equity interest in SRGL, including without limitation any transferee of a kind referenced in <u>Sectionsection</u> 550 of the Bankruptcy Code.

<u>1.178</u> <u>1.182-"SRGL Exclusively Held SALIC TruPS Claims</u>" means all SALIC TruPS Claim arising from or relating to the SHST II TruPS or the GPIC TruPS, of which SRGL is the sole Beneficial Holder.

<u>1.179</u> <u>1.183</u> "<u>SRGL Exclusively Held SHI TruPS Claims</u>" means all SHI TruPS Claim arising from or relating to the SHST II TruPS or the GPIC TruPS, of which SRGL is the sole Beneficial Holder.

<u>1.180</u> <u>1.184</u> "<u>SRGL Exclusively Held TruPS Claims</u>" means all SRGL Exclusively Held SHI TruPS Claims and all SRGL Exclusively Held SALIC TruPS Claims.

<u>1.181</u> <u>1.185</u> "<u>SRGL Revolving Credit Agreement</u>" means that certain Revolving Credit Agreement, dated as of September 20, 2009, by and between SALIC, as lender, and SRGL, as borrower, as amended, novated, supplemented, extended, or restated from time to time.

<u>1.182</u> <u>1.186</u> "<u>SRGL Revolver Facility</u>" means the credit facility provided for under the SRGL Revolving Credit Agreement.

<u>1.183</u> <u>1.187</u> "<u>SRGL Revolver Facility Documents</u>" means the SRGL Revolving Credit Agreement, together with any related agreement, instrument, report or other document executed in connection therewith or otherwise evidencing Claims or other obligations arising thereunder, each as amended, novated, supplemented, extended, or restated from time to time.

<u>1.184</u> <u>1.188</u> "<u>SRGL TruPS Claims</u>" has the meaning set forth in the Restructuring Implementation Agreement.

<u>1.185</u> <u>1.189</u> "<u>SRLB</u>" means Scottish Re Life (Bermuda) Limited.

<u>1.186</u> <u>1.190</u> "<u>SRUS</u>" means Scottish Re (U.S.), Inc.

<u>1.187</u> <u>1.191</u>-"<u>SRUS Retrocession Agreements</u>" means all retrocession agreements to which SALIC and SRUS were party as of the Petition Date, including the following: (a) Coinsurance/Modified Coinsurance Reinsurance Agreement by and between SRUS (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocessionaire) and SRUS (as reinsurer), effective January 1, 2001; (b) Coinsurance Reinsurance Agreement by and between SRUS (as retrocedent) and SALIC (assumed from SRUS (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocessionaire), effective January 1, 2001; (c) Coinsurance Reinsurance Agreement by and between SRUS (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocedent) and SALIC (assumed from SRD under the Portfolio Transfer Agreement) (as retrocessionaire), effective March 31, 2005; and (e) Modco Reinsurance Agreement by and between SRUS (as reinsurer) and SALIC (as retrocessionaire), as of December 31, 2015.

<u>1.188</u> <u>1.192</u> "<u>Stalking Horse</u>" means HSCM Bermuda Fund Ltd.

<u>1.189</u> <u>1.193</u> "Stock Purchase Agreement" means that certain Stock Purchase Agreement by and among SALIC, SHI and the Purchaser, dated as of June 11, 2018, and Filed with the Bankruptcy Court on June 11, 2018, at Docket No. 342, as such Stock Purchase Agreement may be amended from time to time. A copy of the Stock Purchase Agreement is annexed to the Plan as **Exhibit B**.

<u>1.190</u> <u>1.194</u> "<u>Stockholders Agreement</u>" means the agreement among (x) <u>New Holdco</u> and the Reorganized <u>SALIC or New HoldeoDebtors</u>, on the one hand, and (y) each holder of New Equity, on the other hand, whether as of the Effective Date or subsequent thereto, to be set forth in the Plan Supplement. <u>1.191</u> <u>1.195</u> "<u>Subordinated Claims</u>" means any Claims arising under section 510(b) of the Bankruptcy Code or other Claims that are subordinated to general unsecured claims under the Bankruptcy Code; *provided, however,* for the avoidance of doubt, that the SHI TruPS Claims, the SALIC TruPS Claims and the SFL Note Claim shall not be classified or treated as Subordinated Claims for purposes of the Plan.

<u>1.192</u> <u>1.196</u> "Third-Party Reinsurance Agreements" means all reinsurance type agreements to which SALIC and any third-party unaffiliated insurer were party as of the Petition Date, including the following: (a) Coinsurance Agreement, dated as of July 28, 1999, by and between Highmark Life Insurance Co., as ceding company, and SALIC, as reinsurer; (b) Coinsurance Agreement, dated as of October 21, 1999, by and between Highmark Life Insurance Co., as ceding company, and SALIC, as reinsurer; (c) Coinsurance Agreement, dated as of April 1, 2000, by and between Investors Heritage Life Insurance Co., as ceding company, and SALIC, as reinsurer; (d) Coinsurance Agreement, dated as of April 1, 2000, by and between Lincoln Heritage Life Insurance Co., as ceding company, and SALIC, as reinsurer; (e) Modco Agreement, dated as of February 15, 2000, by and between Lincoln National Life Ins. Co., as ceding company, and SALIC, as reinsurer; and (f) Automatic Monthly Renewable Term Reinsurance Agreement, dated as of June 1, 2016, between C.M. Life Insurance Co. and Mass Mutual Life Insurance Co., each a ceding insurer, and SALIC, as reinsurer.

<u>1.193</u> <u>1.197</u> "<u>Transfer</u>" means to sell, resell, reallocate, use, pledge, assign, transfer, hypothecate, participate, donate, or otherwise encumber or dispose of, directly or indirectly (including through derivatives, options, swaps, pledges, forward sales, or other transactions); provided, however, that holding Certificates in an account with a broker-dealer where the brokerdealer holds a security interest or other encumbrance over property in the account generally, which security interest or other encumbrance is released upon transfer of such securities, shall not constitute a "Transfer" under the Plan.

<u>1.194</u> <u>1.198</u> "<u>TruPS</u>" means:

(i) the undivided beneficial interests, having an aggregate liquidation amount of \$17,500,000.00, in Scottish Holdings Statutory Trust I, a Connecticut statutory trust ("SHST <u>I</u>"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of December 4, 2002, among State Street Bank and Trust Company of Connecticut, National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>SHST I TruPS</u>");

(ii) the undivided beneficial interests, having an aggregate liquidation amount of \$20,000,000.00, in Scottish Holdings, Inc. Statutory Trust II, a Connecticut statutory trust ("<u>SHST II</u>"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of October 29, 2003, among U.S. Bank National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>SHST II</u> <u>TruPS</u>");

(iii) the undivided preferred beneficial interests, having an aggregate liquidation

## Case 18-10160-LSS Doc 437-2 Filed 08/09/18 Page 97 of 105

amount of \$10,000,000.00, in GPIC Holdings Inc. Statutory Trust, a Delaware statutory trust ("<u>GPIC</u>"), issued pursuant to that certain Amended and Restated Trust Agreement dated as of November 14, 2003, among JPMorgan Chase Bank as property trustee, Chase Manhattan Bank USA, National Association as Delaware trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>GPIC TruPS</u>");

(iv) the undivided beneficial interests, having an aggregate liquidation amount of \$32,000,000.00, in Scottish Holdings, Inc. Statutory Trust III, a Connecticut statutory trust ("<u>SHST III</u>"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of May 12, 2004, among U.S. Bank National Association, as institutional trustee, the administrators of the issuer named therein, SHI as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>SHST III</u> <u>TruPS</u>"); and

(v) the undivided beneficial interests, having an aggregate liquidation amount of \$50,000,000.00, in SFL Statutory Trust I, a Delaware statutory trust ("<u>SFLST I</u>," and collectively with SHST I, SHST II, GPIC, and SHST III, the "<u>TruPS Trusts</u>"), issued pursuant to that certain Amended and Restated Declaration of Trust, dated as of December 15, 2004, among Wilmington Trust Company, as institutional trustee, the administrators of the issuer named therein, SFL as the sponsor of the issuer, and the holders from time to time of undivided beneficial interests in the assets of the issuer (the "<u>SFLST I TruPS</u>").

<u>1.195</u> <u>1.199</u> "<u>TruPS Claims</u>" means Claims of any Person relating to or arising out of any TruPS, TruPS Junior Subordinated Debentures or TruPS Documents, including any Claims relating to or arising out of any TruPS Documents. For the avoidance of doubt, TruPS Claims shall include all of the Claims set forth in the preceding sentence that could be asserted by one or more of the several parties thereto without duplication. When referring to TruPS Claims arising out of a particular TruPS issuance, the words "TruPS Claims" will be preceded by the name of the applicable TruPS issuance (*e.g.*, "SHST I TruPS Claims").

<u>1.196</u> <u>1.200</u> "<u>TruPS Claims Aggregate Amount</u>" means (i) all Allowed SHI TruPS Claims, (ii) all Allowed SALIC TruPS Claims, and (iii) \$63,536,014.32 on account of the SFL Note Claim.

<u>1.197</u> <u>1.201</u>."<u>TruPS Claims Equity Distribution Amount</u>" means (i) with respect to a Beneficial Holder of TruPS, the amount of the Offered New Equity to be distributed to a Beneficial Holder that elects to receive New Equity under Section 4.3 of the Plan, calculated based on such Beneficial Holder's Allocated Portion of the Allowed TruPS Claim in respect of which it is a Beneficial Holder divided by the TruPS Claims Aggregate Amount; and (ii) with respect to the Holder of the SFL Note Claim, for which the SFL Claims Allowance Conditions have been or deemed satisfied or deemed satisfied, the amount of the Offered New Equity to be distributed on account of the Allowed SFL Note Claim, if the Holder thereof elects or is deemed to have elected to receive New Equity under Sectionsection</u> 4.3 of the Plan, calculated based on the portion of the Allowed SFL Note Claim for which the New Equity Election has been made or has been deemed to have been made divided by the TruPS Claims Aggregate Amount.

## <u>1.198</u> <u>1.202</u> "<u>TruPS Debentures</u>" means:

(i) with respect to the SHST I TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Holdings, Inc., and State Street Bank and Trust Company of Connecticut, National Association, due 2032 (the "<u>SHST I TruPS</u> <u>Debentures</u>");

(ii) with respect to the SHST II TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Holdings, Inc., and U.S. Bank National Association, due 2033 (the "<u>SHST II TruPS Debentures</u>");

(iii) with respect to the GPIC TruPS, that certain Floating Rate Junior Subordinated Note Due 2033 between Scottish Holdings, Inc., and JPMorgan Chase Bank (the "<u>GPIC TruPS Debentures</u>");

(iv) with respect to the SHST III TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Holdings, Inc., and U.S. Bank National Association, due 2034 (the "<u>SHST III TruPS Debentures</u>"); and

(v) with respect to the SFLST I TruPS, that certain Floating Rate Junior Subordinated Deferrable Interest Debenture between Scottish Financial (Luxembourg) S.à r.l. and Wilmington Trust Company, due 2034 (the "<u>SFLST I TruPS Debentures</u>").

<u>1.199</u> <u>1.203</u> "<u>TruPS Declarations</u>" means:

(i) with respect to the SHST I TruPS, that certain Amended and Restated Declaration of Trust by and among State Street Bank and Trust Company of Connecticut, National Association, as Institutional Trustee, Scottish Holdings, Inc., as Sponsor, and Paul Goldean and Oscar R. Scofield, as Administrators, dated as of December 4, 2002 (the "<u>SHST I Trust Declaration</u>").

(ii) with respect to the SHST II TruPS, that certain Amended and Restated Declaration of Trust by and among U.S. Bank National Association, as Institutional Trustee, Scottish Holdings, Inc., as Sponsor, and Paul Goldean and Oscar R. Scofield, as Administrators, dated as of October 29, 2003 (the "SHST II Trust Declaration").

(iii) with respect to the GPIC TruPS, that certain Amended and Restated Declaration of Trust by and among Scottish Holdings, Inc., as Depositor, JPMorgan Chase Bank, as Property Trustee, Chase Manhattan Bank USA, National Association, as Delaware Trustee, and The Administrative Trustees Named [T]Herein, as Administrative Trustees, dated as of November 14, 2003 (the "<u>GPIC Trust Declaration</u>").

(iv) with respect to the SHST III TruPS, that certain Amended and Restated Declaration of Trust by and among U.S. Bank National Association, as Institutional Trustee, Scottish Holdings, Inc., as Sponsor, and Paul Goldean and Oscar R. Scofield, as Administrators, dated as of May 12, 2004 (the "SHST III Trust Declaration").

(v) with respect to the SFLST I TruPS, that certain Amended and Restated

Declaration of Trust by and among Wilmington Trust Company, as Institutional Trustee, Wilmington Trust Company, as Delaware Trustee, Scottish Financial (Luxembourg) S.à r.l., as Sponsor, and Paul Goldean and George Scott, as Administrators, dated as of December 15, 2004 (the "<u>SFLST Trust Declaration</u>").

<u>1.200</u> <u>1.204</u> "<u>TruPS Documents</u>" means collectively all TruPS Indentures, all documents evidencing TruPS Debentures, all TruPS Declarations, all TruPS Sponsor Guarantees, all TruPS Parent Guarantees and all related and ancillary documents and instruments, each as altered, amended, modified or supplemented from time to time and including all exhibits and schedules thereto.

<u>1.201</u> <u>1.205</u> "<u>TruPS/GUC Claims Aggregate Amount</u>" means (i) all Allowed SHI TruPS Claims, (ii) all Allowed SALIC TruPS Claims, (iii) \$63,536,014.32 on account of the SFL Note Claim. (iv) all Allowed SHI General Unsecured Claims, (v) all Allowed SALIC General Unsecured Claims, and (vi) the Disputed Claim Reserve Amount on account of all Disputed General Unsecured Claims.

<u>1.202</u> <u>1.206</u> "<u>TruPS/GUC Claims Cash Distribution Amount</u>" means the pro rata amount of the Available Plan Distribution Funding Amount to be distributed to a Beneficial Holder, Holder of the SFL Note Claim or Holder of a General Unsecured Claim, as applicable, calculated based on: (i) with respect to a Beneficial Holder of TruPS, such Beneficial Holder's Allocated Portion of the Allowed TruPS Claim in respect of which it is a Beneficial Holder, divided by the TruPS/GUC Claims Aggregate Amount; (ii) with respect to the Allowed SFL Note Claim, the portion of such Holder's Allowed SFL Note Claim for which the Cash Election has been made (or deemed made), divided by the TruPS/GUC Claims Aggregate Amount; and (iii) with respect to an Allowed General Unsecured Claim, the Allowed amount of such General Unsecured Claim, the Allowed amount of such General Unsecured Claim, the Allowed amount of such General Unsecured Claim, divided by the TruPS/GUC Claims Aggregate Amount.

<u>1.203</u> <u>1.207</u> "<u>TruPS Indentures</u>" means:

(i) with respect to the SHST I TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2032, between Scottish Holdings, Inc. as Issuer, and State Street Bank and Trust Company of Connecticut, National Association, as Trustee, dated as of December 4, 2002;

(ii) with respect to the SHST II TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2033, between Scottish Holdings, Inc., as Issuer, and U.S. Bank National Association, as Trustee, dated as of October 29, 2003;

(iii) with respect to the GPIC TruPS, that certain Junior Subordinated Indenture, due 2033, between Scottish Holdings, Inc., as Issuer, and JPMorgan Chase Bank, as Trustee, dated as of November 14, 2003;

(iv) with respect to the SHST III TruPS, that certain Indenture for the Floating Rate Junior Subordinated Deferrable Interest Debentures, due 2034, between Scottish Holdings, Inc., as Issuer, and U.S. Bank National Association, as Trustee, dated as of May 12, 2004; and

(v) with respect to the SFLST I TruPS, that certain Indenture for the Floating

Rate Junior Subordinated Deferrable Interest Debentures, due 2034, between Scottish Financial (Luxembourg) S.à r.l., as Issuer, and Wilmington Trust Company, as Trustee, dated as of December 15, 2004.

<u>1.204</u> <u>1.208</u> "<u>TruPS Indenture Trustee</u>" means an Entity serving as an indenture trustee under a TruPS Indenture during the period from the Petition Date through the Effective Date.

<u>1.205</u> <u>1.209</u> "<u>TruPS Institutional Trustee</u>" means an Entity serving as the trustee pursuant to a TruPS Declaration, a TruPS Parent Guarantee or a TruPS Sponsor Guarantee.

<u>1.206</u> <u>1.210</u> "<u>TruPS Parent Guarantees</u>" means:

(i) with respect to the SHST I TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and State Street Bank and Trust Company of Connecticut, National Association, dated as of December 4, 2002;

(ii) with respect to the SHST II TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and U.S. Bank National Association, dated as of October 29, 2003;

(iii) with respect to the GPIC TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and JPMorgan Chase Bank, dated as of November 14, 2003;

(iv) with respect to the SHST III TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and U.S. Bank National Association, dated as of May 12, 2004; and

(v) with respect to the SFLST I TruPS, that certain Parent Guarantee Agreement by and between Scottish Annuity & Life Insurance Company (Cayman) Ltd. and Wilmington Trust Company, dated as of December 15, 2004.

<u>1.207</u> <u>1.211</u> "<u>TruPS Returned Cash</u>" means the amount of the Available Plan Distribution Funding Amount that is not required to be funded by the <u>Purchaser or is returned to</u> Purchaser on account of the Beneficial Holders of TruPS that make the New Equity Election, calculated based on the aggregate Beneficial Holders who made, or are deemed to make, the New Equity Election, divided by the TruPS/GUC Claims Aggregate Amount.

<u>1.208</u> <u>1.212</u> "<u>TruPS Shares</u>" means the Offered New Equity. Any reference to the TruPS Shares in the Stock Purchase Agreement shall mean the Offered New Equity.

<u>1.209</u> <u>1.213</u> "<u>TruPS Sponsor Guarantees</u>" means:

(i) with respect to the SHST I TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and State Street Bank and Trust Company of Connecticut, National Association, dated as of December 4, 2002;

(ii) with respect to the SHST II TruPS, that certain Guarantee Agreement by and

between Scottish Holdings, Inc., and U.S. Bank National Association, dated as of October 29, 2003;

(iii) with respect to the GPIC TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and JPMorgan Chase Bank, dated as of November 14, 2003;

(iv) with respect to SHST III TruPS, that certain Guarantee Agreement by and between Scottish Holdings, Inc., and U.S. Bank National Association, dated as of May 12, 2004; and

(v) with respect to the SFLST I TruPS, that certain Guarantee Agreement by and between Scottish Financial (Luxembourg) S.à r.l and Wilmington Trust Company, dated as of December 15, 2004.

<u>1.210</u> <u>1.214</u> "<u>Trust Administration Reserve</u>" means a reserve to be maintained by the Distribution Trust in an amount, mutually agreed by the Debtors, the Official Committee, and the Purchaser, estimated in good faith to be necessary to cover the costs of administration of the Distribution Trust, as further defined in Section 6.3(f)(iii) of the Plan.

<u>1.211</u> <u>"Trust Administration Reserve Cap" means the cap on the aggregate amount to be</u> funded into the Trust Administration Reserve, which shall be mutually acceptable to the Debtors, the Purchaser and the Official Committee. The Trust Administration Reserve Cap shall be disclosed at or before the commencement of the Confirmation Hearing.

1.212 1.215 "Trust Agreements" means all trust agreements relating to Third-Party Reinsurance Agreements to which SALIC, as grantor, and any ceding company were party as of the Petition Date, including the following: (a) Trust Agreement dated as of July 8, 1999, by and among SALIC, as grantor, Highmark Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (b) Trust Agreement dated as of October 21, 1999, by and among SALIC, as grantor, Highmark Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (c) Trust Agreement dated as of April 1, 2000, by and among SALIC, as grantor, Investors Heritage Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (d) Trust Agreement dated as of April 1, 2000, by and among SALIC, as grantor, Lincoln Heritage Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (e) Trust Agreement dated as of July 12, 2016, by and among SALIC, as grantor, C.M. Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; (f) Trust Agreement dated as of July 12, 2016, by and among SALIC, as grantor, Massachusetts Mutual Life Insurance Company, as beneficiary, and The Bank of New York Mellon, as trustee; and (g) Security Trust Agreement dated as of July 12, 2016, by and among SALIC, as grantor, Massachusetts Mutual Life Insurance Company and C.M. Life Insurance Company, each as beneficiary, and The Bank of New York Mellon, as trustee.

<u>1.213</u> <u>1.216-"Unexpired Lease</u>" means a lease of non-residential real property to which any Debtor is a party that is subject to assumption or rejection under section 365 of the Bankruptcy Code.

## Case 18-10160-LSS Doc 437-2 Filed 08/09/18 Page 102 of 105

<u>1.214</u> <u>1.217</u> "<u>Unimpaired</u>" means, when used with reference to a Claim or Interest, a Claim or Interest that is not impaired within the meaning of section 1124 of the Bankruptcy Code.

<u>1.215</u> <u>1.218</u> "<u>Unimpaired Class</u>" means a Class of Claims that are not impaired within the meaning of section 1124 of the Bankruptcy Code.

<u>1.216</u> <u>1.219</u> "<u>U.S. Bank</u>" means U.S. Bank National Association, as trustee.

<u>1.217</u> <u>1.220-"U.S. Trustee</u>" means the Office of the United States Trustee for the District of Delaware.

<u>1.218</u> <u>1.221</u> "<u>Voting Agent</u>" means Prime Clerk, LLC, the Entity the Bankruptcy Court has authorized to serve as the "Voting Agent" pursuant to the Disclosure Statement Order.

<u>1.219</u> <u>1.222</u> "<u>Voting Deadline</u>" means [August <u>13],13</u>, 2018 at 4:00 p.m. (Eastern Time), which date and time has been established by the Bankruptcy Court pursuant to the Disclosure Statement Order as the deadline by which all Ballots to accept or reject the Plan must be received in order to be counted for purposes of section 1126 of the Bankruptcy Code; *provided, however*, that the Voting Deadline for the SFL Receiver as the Holder of the SFL Note Claim shall be the later of (x) [August <u>13],13</u>, 2018 at 4:00 p.m. (Eastern Time), and (y) two (2) Business Days after the Debtors or the Voting Agent has provided counsel for the SFL Receiver with notice in writing (which may take the form of an email to the SFL Receiver's counsel) of the outcome of voting on the Plan by the SFLST I TruPS CDO Facility Holders.

<u>1.220</u> <u>1.223</u> "<u>WTC</u>" means Wilmington Trust Company, as trustee.

# <u>Exhibit B</u>

Stock Purchase Agreement

[Intentionally Omitted]

# <u>Exhibit C</u>

**Restructuring Implementation Agreement** 

[Intentionally Omitted]

Document comparison by Workshare 9.5 on Thursday, August 9, 2018 5:50:48 PM

Input:		
Document 1 ID	interwovenSite://WIL-DMS/WILM/12008978/5	
Description	#12008978v5 <wilm> - Scottish Second Amended Plan</wilm>	
Document 2 ID	interwovenSite://WIL-DMS/WILM/12105785/2	
Description	#12105785v2 <wilm> - Scottish-Third_Amended_Plan</wilm>	
Rendering set	Standard	

Legend:		
Insertion		
Deletion-		
Moved from		
Moved to		
Style change		
Format change		
Moved deletion		
Inserted cell		
Deleted cell		
Moved cell		
Split/Merged cell		
Padding cell		

Statistics:		
	Count	
Insertions	172	
Deletions	149	
Moved from	7	
Moved to	7	
Style change	0	
Format changed	0	
Total changes	335	